Copyright is owned by the Author of the thesis. Permission is given for a copy to be downloaded by an individual for the purpose of research and private study only. The thesis may not be reproduced elsewhere without the permission of the Author.

### Metagenomic analysis and culture-based methods to examine the prevalence and distribution of antimicrobial resistance on two New Zealand dairy farms

A thesis presented in partial fulfilment of the requirements for the degree

of

Doctor of Philosophy

 $_{\mathrm{in}}$ 

Veterinary Science

at Massey University, Palmerston North, New Zealand

Rose Moyra Collis<br/>2022

## Abstract

Antimicrobial resistance (AMR) is a global threat to human and animal health, with the misuse and overuse of antimicrobials being suggested as the main driver of resistance. In a global context, New Zealand (NZ) is a relatively low user of antimicrobials in animal production. However, antimicrobial usage on NZ dairy farms and its potential for driving the spread of AMR within the dairy farm environment is under-researched.

This research addresses the hypothesis that antimicrobial use on NZ dairy farms influences the prevalence of AMR in dairy farm environments, taking into consideration seasonality and contrasting farm management practices. The aims of this study, focused on two NZ dairy farm environments over an 15 month period, were to (i) determine the prevalence and distribution of AmpC- and extended-spectrum  $\beta$ -lactamase (ESBL)-producing *Escherichia coli*, utilising culture-based methods, and (ii) to determine the abundance and diversity of antimicrobial resistance genes (ARGs), utilising a metagenomic approach, and lastly (iii), to assess the impact of systemic antimicrobial treatment on the bovine faecal microbiome.

Overall, the research presented in this thesis has shown a low sample level prevalence of ESBL-producing E. coli from two NZ dairy farms (faeces 0%, 1.7%; farm dairy effluent (FDE) 0%, 6.7% from Dairy 1 and Dairy 4, respectively) but AmpC-producing E. coli were more frequently isolated across both farms (faeces 3.3%, 8.3%; FDE 38.4%, 6.7% from Dairy 1 and Dairy 4, respectively). AmpC- and ESBL-producing *E. coli* were isolated in spring and summer, during months with varying levels of antimicrobial use. Analysis at the individual animal level showed a decrease in bacterial diversity and richness during systemic antimicrobial treatment and in many cases the microbiome diversity recovered post-treatment when the cow re-entered the milking herd. Compared to overseas data in a similar context, NZ dairy farm environments had a low abundance of ARGs, with the highest abundance detected in soil (0.20 - 0.63 copies of ARG per 16S rRNA gene). However, many of the ARGs identified in soil are not frequently found in human pathogens or acquired genes. FDE had a lower ARG abundance but the ARGs were more diverse (0.03 - 0.37 copies of ARG per 16S rRNA gene). There was no association between the normalised ARG abundance and antimicrobial use or collection date, however the low ARG abundance in the farm samples may have made any associations difficult to detect.

AMR is a burden for human, animal and environmental health and requires a holistic "One Health" approach to address. The outcomes from this research improve our understanding of the current levels of AMR on two NZ dairy farms and identifies areas for future research. Prevention is better than a cure and urgent action is required to slow the development and dissemination of AMR and to improve antimicrobial stewardship in humans and animals.

# List of publications

#### Peer-reviewed:

Collis, R. M., Burgess, S. A., Biggs, P. J., Midwinter, A. C., French, N. P., Toombs-Ruane, L., & Cookson, A. L. (2019). Extended-spectrum beta-lactamase-producing *Enterobacteri*aceae in dairy farm environments: A New Zealand perspective. *Foodborne Pathogens and Disease*, 16(1), 5-22. doi:10.1089/fpd.2018.2524

#### In preparation:

Collis, R. M., Burgess, S. A., Biggs, P.J., Midwinter, A. C., Brightwell, G., & Cookson, A. L. Prevalence and distribution of extended-spectrum  $\beta$ -lactamase and AmpC-producing *Escherichia coli* in two New Zealand dairy farm environments.

Collis, R. M., Burgess, S. A., Biggs, P.J., Midwinter, A. C., Brightwell, G., & Cookson, A. L. Impact of systemic antimicrobial therapy on the faecal microbiome in dairy cows.

Collis, R. M., Burgess, S. A., Biggs, P.J., Midwinter, A. C., Brightwell, G., & Cookson, A. L. A longitudinal study of the resistome on two New Zealand dairy farms.

## List of presentations

#### 2021:

Collis, R. M., Biggs, P.J., Burgess, S. A., Midwinter, A. C., Brightwell, G., & Cookson, A. L. Metagenomic analysis and culture-based methods to examine the prevalence and distribution of antibiotic resistance on two New Zealand dairy farms. One Health Aotearoa Symposium. December 8<sup>th</sup> 2021. (Oral) *Awarded the One Health Aotearoa best student oral presentation.*

#### 2020:

- Collis, R. M., Biggs, P.J., Burgess, S. A., Midwinter, A. C., Brightwell, G., & Cookson, A. L. Metagenomic analysis and culture-based methods to examine the prevalence and distribution of antibiotic resistance on two New Zealand dairy farms. One Health Aotearoa Symposium, Palmerston North, New Zealand. December 8<sup>th</sup> 2020. (Oral)
- Collis, R. M., Biggs, P.J., Burgess, S. A., Midwinter, A. C., Brightwell, G., & Cookson, A. L. Metagenomic analysis and culture-based methods to examine the prevalence and distribution of antibiotic resistance on two New Zealand dairy farms. New Zealand Microbiological Society Conference. November 27<sup>th</sup> 2020. (Oral) Awarded 1<sup>st</sup> place in the student oral competition.
- Collis, R. M., Biggs, P.J., Burgess, S. A., Midwinter, A. C., Brightwell, G., & Cookson, A. L. Metagenomic analysis and culture-based methods to examine the prevalence and distribution of antibiotic resistance on two New Zealand dairy farms. New Zealand Food Safety Science and Research Centre Spotlight Seminar series. New Zealand. 23<sup>rd</sup> October 2020. (Oral)
- Collis, R. M., Biggs, P.J., Burgess, S. A., Midwinter, A. C., Brightwell, G., & Cookson, A. L. Metagenomic analysis and culture-based methods to examine the prevalence and distribution of antibiotic resistance on two New Zealand dairy farms. Presented my research to Hon Prime Minister Jacinda Ardern and Hon Minister Grant Robertson, representing the New Zealand Food Safety Science and Research Centre. Food HQ, Palmerston North, New Zealand. September 17<sup>th</sup> 2020. (Oral)
- Collis, R. M., Biggs, P.J., Burgess, S. A., Midwinter, A. C., Brightwell, G., & Cookson, A. L. Metagenomic analysis and culture-based methods to examine the prevalence and distribution of antibiotic resistance on two New Zealand dairy farms.

Massey University Three Minute Thesis competition. New Zealand. August 2020. (Oral - Three Minute Thesis)

### 2019:

- Collis, R. M., Biggs, P.J., Burgess, S. A., Midwinter, A. C., Brightwell, G., & Cookson, A. L. Metagenomic analysis and culture-based methods to examine the prevalence and distribution of antibiotic resistance on two New Zealand dairy farms. One Health Aotearoa Symposium, Wellington, New Zealand. December 10<sup>th</sup> 2019. (Poster) Awarded One Health Aotearoa poster prize.
- Collis, R. M., Biggs, P.J., Burgess, S. A., Midwinter, A. C., Brightwell, G., & Cookson, A. L. Metagenomic analysis and culture-based methods to examine the prevalence and distribution of antibiotic resistance on two New Zealand dairy farms. New Zealand Microbiological Society Conference, Palmerston North, New Zealand. November 26<sup>th</sup> 2019. (Poster)
- Collis, R. M., Biggs, P.J., Burgess, S. A., Midwinter, A. C., Brightwell, G., & Cookson, A. L. Metagenomic analysis and culture-based methods to examine the prevalence and distribution of antibiotic resistance on two New Zealand dairy farms. Participated in an antimicrobial resistance meeting with a focus on Global One Health. Nanjing Agricultural University, Nanjing, Jiangsu, China. November 2019. (Oral)

#### 2018:

- Collis, R. M., Biggs, P.J., Burgess, S. A., Midwinter, A. C., Brightwell, G., & Cookson, A. L. Metagenomic analysis and culture-based methods to examine the prevalence and distribution of antibiotic resistance on two New Zealand dairy farms. I presented my research to the Primary Production Select Committee, representing the New Zealand Food Safety Science and Research Centre. Parliament, Wellington, New Zealand. September 20<sup>th</sup> 2018. (Oral)
- Collis, R. M., Biggs, P.J., Burgess, S. A., Midwinter, A. C., Brightwell, G., & Cookson, A. L. Metagenomic analysis and culture-based methods to examine the prevalence and distribution of antibiotic resistance on two New Zealand dairy farms. New Zealand Food Safety Science and Research Centre conference. July 2<sup>nd</sup> 2018. (Oral Three Minute Thesis)

## Acknowledgements

Firstly, a huge thank you to my four supervisors for their invaluable help, support and encouragement throughout my PhD journey: A/Prof. Adrian Cookson, Prof. Patrick Biggs, Dr Anne Midwinter and Dr Sara Burgess. I am fortunate to have supervisors who are not only excellent scientists and mentors, but are also kind and supportive individuals.

Adrian, thank you for all of the help, support and guidance throughout my PhD. Your enthusiasm, knowledge and ability to think about the 'bigger picture' really helped me to grow as a scientist. You were always happy to answer my many questions and read countless drafts of my work - thank you! Patrick, thank you for helping me to develop my coding skills and for all of your guidance and support analysing the large metagenomic dataset. This was a huge task and I am happy with the final outcome and the skills I have learnt along the way. Thank you for your attention to the small details and helping me to make my plots look as aesthetically pleasing as possible! Sara, thank you for all of help and support, especially for your help in the lab and with the MinION sequencing and with Toastmasters to improve my presentation skills. Anne, thank you for all of your support and guidance in the lab, for helping me to always consider the relevance of my research and for always being there to support me through all aspects of my PhD journey.

Thank you to the many people who have given advice and helped me with various aspects of my PhD project: Dr Jinxin Liu and Prof. David Mills (University of California, Davis) for their guidance on the shotgun metagenomic sequencing and code development and Dan Sun and Alan McCulloch (AgResearch Ltd.) for all of their help using the AgResearch High Performance Computing Servers to analyse my large dataset! Thank you to the Food Integrity AgResearch Ltd. Team Leader Dr Gale Brightwell for her help and support. Thank you to Xiaoxiao Lin (Massey University Genome Service) for his help and advice on the many sequencing components of my project, to Dr Ahmed Fayaz (Massey University) for the database development and maintenance and to Prof. Scott McDougall (Anexa FVC) for his study design advice.

A huge thank you to the Massey University Dairy 1 and Dairy 4 farm staff, especially Fiona Sharland and the farm managers Jolanda Amoore, Hamish Doohan and Joshua Mitchell. Thank you for all of your help with the sample collection and on farm components of my research. Thank to my colleagues who also helped with the sample collection on farm: Amila Nawarathna, Alexis Risson, Eden Esteves and Karina Sakamoto.

I have really enjoyed being a part of the Food System Integrity team at AgResearch Ltd. and the <sup>m</sup>EpiLab at Massey University. Thank you to all of my colleagues who I asked questions or bounced ideas off and for the many laughs and chats in the hallway or over a cup of coffee. A special thank you to my fellow PhD students that I shared an office with:

Amila Nawarathna, Alexis Risson, Soundarya Karamcheti, Eden Esteves and Zack Zhang. And thank you to all of the post-graduate students from the <sup>m</sup>EpiLab, Olivia Angelin-Bonnet for her friendship and R knowledge and my car-pooling friends Ellie Bradley and Holly Gray. No words can express how grateful I am for your friendship, memories and support over the last three years.

I would like to acknowledge the financial support of the New Zealand Food Safety Science and Research Centre, the AgResearch Ltd. Food Integrity SSIF programme, the Massey University School of Veterinary Science post-graduate research fund and the Eric Ojala Trust post-graduate scholarship in agricultural policy research for funding components of this project. I am grateful for the support of One Health Aotearoa and the New Zealand Microbiological Society through the oral presentation and poster prizes I was awarded.

I would like to thank the following Trusts for the post-graduate scholarships I have received throughout my studies: C. Alma Baker Trust post-graduate scholarship, Sydney Campbell Memorial Trust post-graduate scholarship, Catherine Baxter post-graduate scholarship, Freemasons post-graduate scholarship and the Tararua Provincial Rural Women NZ tertiary resource/book grant.

Lastly, thank you to my amazing family for their support throughout my PhD. Thank you to my parents, Tracey and Mike, who have always encouraged my passion for science and learning from a young age and have inspired me to work hard and to always do my best. Thank you to my three sisters Deana, Grace and Lucy and my brother-in-law Ash, for supporting me over the last three years and always making me smile. Thank you to my mother- and father-in-law, Viv and Graeme, for their support and to Viv for cooking us dinner every Thursday. And of course, thank you to my kind and supportive husband, Louis. You have always supported me throughout my studies and encouraged me to follow my dreams. Your love and support throughout the good and bad times has helped get me through – thank you!

# Contents

A	bstra	act	ii
Li	st of	f publications	iii
$\mathbf{Li}$	st of	f presentations	iv
A	cknov	wledgements	vi
A	bbre	viations	xv
1	Intr	roduction	1
	1.1	General background	1
	1.2	Antimicrobial resistance on New Zealand dairy farms	2
	1.3	Thesis aims	3
	1.4	Thesis structure	4
<b>2</b>	Lite	erature review	6
	2.1	Introduction	6
	2.2	Antimicrobial resistance mechanisms	7
	2.3	Bacterial pathogens of concern	8
		2.3.1 Mastitis-associated pathogens	8
		2.3.2 Enterobacteriaceae	9
	2.4	$\beta$ -lactamases	10
		2.4.1 ESBL enzymes	12
		2.4.2 AmpC $\beta$ -lactamase enzymes	13
		2.4.2.1 Chromosomal $ampC$ genes $\ldots$ $\ldots$ $\ldots$ $\ldots$	13
		2.4.2.2 Plasmid-mediated AmpC-producing $E. \ coli$	14
		2.4.2.3 Extended-spectrum AmpC $\beta$ -lactamases	15
	2.5	Methods for detecting AmpC- and ESBL-producing	
		Enterobacteriaceae	15
		2.5.1 ESBL-producing <i>Enterobacteriaceae</i>	16
		2.5.2 AmpC-producing <i>Enterobacteriaceae</i>	17
	2.6	Prevalence of AmpC- and ESBL-E in the dairy farm	
		environment	18
		2.6.1 Prevalence of AmpC- and ESBL-E in dairy cattle, calves and farm	
		dairy effluent	18
		2.6.2 Prevalence of AmpC- and ESBL-E in milk	28
	2.7	Shotgun metagenomics for the detection of ARGs	31
	2.8	AMU on dairy farms	33
	2.9	Other farm management practices which may influence AMR on dairy farms	36

	2.10	"One	Health" approach	42
	2.11	Trans	mission of antimicrobial resistant bacteria and ARGs in the dairy farm	
		enviro	nment	42
	2.12	Concl	usion	46
3	Pre	valenc	e and distribution of extended-spectrum $eta$ -lactamase and AmpO	<b>]-</b>
	pro	ducing	g Escherichia coli in two New Zealand dairy farm environments	48
	3.1	Introd	luction	48
	3.2	Metho	ds	49
		3.2.1	Study population and sample collection	49
		3.2.2	Sample processing	50
		3.2.3	Microbiological methods	52
		3.2.4	Antimicrobial susceptibility tests	53
		3.2.5	Molecular characterisation	54
		3.2.6	DNA extraction, library preparation, and whole genome	
			sequencing	58
		3.2.7	Bioinformatic analysis	59
		3.2.8	Statistical tests	60
	3.3	Result	ts	60
		3.3.1	Antimicrobial resistance profiles	60
		3.3.2	Molecular characterisation of AmpC- and	
			ESBL-producing <i>E. coli</i>	61
		3.3.3	Population structure and comparative genomics	62
		3.3.4	Antimicrobial resistance genes	66
		3.3.5	Virulence factors and <i>E. coli</i> pathotypes	68
		3.3.6	Plasmid characteristics	70
		3.3.7	Prevalence of AmpC- and ESBL-producing <i>E. coli</i> from farm envi-	
			ronmental samples	76
		3.3.8	Estimated antimicrobial use on dairy farms between October 2018	
			and December 2019	77
	3.4	Discus	ssion	78
	3.5	Concl	usion	90
4	Imp	oact of	systemic antimicrobial therapy on the faecal microbiome in	
	dair	y cow	S	91
	4.1	Introd	luction	91
	4.2	Metho	$ds \dots \dots$	92
		4.2.1	Study population and sample collection	92
		4.2.2	Sample processing	94
		4.2.3	16S rRNA V3-V4 sequencing	94
			4.2.3.1 DNA extraction and sequencing	94
			4.2.3.2 Bioinformatics	95
		424	Statistical tests	96

		4.2.5	Microbiological methods and molecular characterisation $\ldots$ 96
		4.2.6	Whole genome sequencing
			4.2.6.1 DNA extraction
			4.2.6.2 Genome analysis
	4.3	Result	s
		4.3.1	Microbiome diversity
		4.3.2	Case studies
			4.3.2.1 Procaine penicillin G treated cows
			4.3.2.2 Penethamate hydriodide treated cows $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots 103$
			4.3.2.3 Marbofloxacin/penethamate hydriodide treated cow $\ldots$ . 105
			4.3.2.4 Ceftiofur treated cows
		4.3.3	AMR in action
			4.3.3.1 Whole genome sequencing $\ldots \ldots \ldots$
	4.4	Discus	sion $\ldots \ldots \ldots$
	4.5	Conclu	usion
_		•	
5	A lo	ongituc	linal study of the resistome on two New Zealand dairy farms 122
	5.1	Introd	uction
	5.2	Metho	$ds \dots \dots$
		5.2.1	Study population, sample collection and processing
		5.2.2	DNA extraction
		5.2.3	Shotgun metagenomic sequencing
		5.2.4	Bioinformatic analysis
		5.2.5	Resistance gene analysis
	- 0	5.2.6	Statistical analysis and data visualisation
	5.3	Result	s
		5.3.1	Microbial community composition
		5.3.2	Resistome analysis
		5.3.3	Bacterial host range harbouring contigs containing acquired ARGs 138
		5.3.4	Examination of sequencing depth for the detection of antimicrobial,
			heavy metal and biocide resistance genes
	F 4	5.3.5 D'	Sequencing controls
	5.4 5.7	Discus	sion
	5.5	Conclu	ISION
6	Gen	eral di	iscussion 156
	6.1	Overvi	ew
	6.2	Findin	gs, implications and general discussion
		6.2.1	Antimicrobial resistance in the New Zealand dairy farm
			environment
		6.2.2	Selection of antimicrobial resistant bacteria
	6.3	Potent	ial limitations and future work
		631	Farm choice 164

6.	.3.2	AMR risk factors	65
6.	.3.3	Sample variables	66
6.	.3.4	Sequencing methodologies	66
6.	.3.5	Computational limitations	67
6.	.3.6	Analysis of faecal samples enriched in media containing specific an-	
		tibiotics $\ldots \ldots \ldots$	67
6.	.3.7	Waste milk	68
6.	.3.8	Environmental vectors	68
6.4 C	Conclu	$1 sion \ldots \ldots$	70
7 Bibliography			72
8 Append	dices	21	18

# List of Figures

1.1	General outline of the PhD research
2.1	Potential transmission pathways of ESBL-producing <i>Enterobacteriaceae</i> and antimicrobial resistant bacteria in dairy farm environments
3.1	Workflow of sample screening, antimicrobial susceptibility testing and ES- BL/AmpC confirmation tests 55
3.2	Resistance profiles of <i>E. coli</i> (n=52) isolated across 14 farm samples $\dots$ 61
0.0	analysis of AmpC and ESBL-producing <i>E. coli</i>
3.4	Hierarchical cluster tree constructed using Jaccard distances for the presence         or absence data from 37 virulence genes         69
3.5	Annotation of the partial copy of the $bla_{\text{TEM-105}}$ gene and the IS26 transposase in the pDF0059.2e_3 plasmid
3.6	Representative IncI1 complete circular plasmid pDF0049.2e_1 and IncFI- A/IncQ1 plasmid pDF0059.2e 3
3.7	<b>A:</b> Collection dates of samples positive for AmpC and/or ESBL-producing $E_{i}$ coli <b>B:</b> Antimicrobial use per month 78
4.1	Timeline of faecal sampling
4.2	Diversity between treated and control (untreated) cows for <b>A</b> : Shannon diversity and <b>B</b> : Chaol diversity
4.3	Principal Component Analysis for the faecal microbiome in treated animals
4.4	Procaine penicillin G case study
$4.5 \\ 4.6$	Penethamate hydriodide case study $\dots \dots \dots$
4.7	Marbofloxacin/penethamate hydriodide case study: Abundance of the top
4.8	S0 ASVS    106      Ceftiofur case study    108
5.1	Workflow for bioinformatic analysis of shotgun metagenomic sequencing reads127
$5.2 \\ 5.3$	NMDS of microbiome profiles at the phylum level of farm samples 130 Normalised antimicrobial resistance gene abundance in farm dairy effluent,
54	faeces, soil and milk samples
U.T	faeces, soil and milk samples on Dairy 1 and Dairy 4
5.5	Antimicrobial, heavy metal and biocide resistance gene abundance across farm dairy effluent, faeces, milk and soil samples on Dairy 1

5.6	Antimicrobial, heavy metal and biocide resistance gene abundance across
	farm dairy effluent, faeces, milk and soil samples on Dairy 4 $\ .$
5.7	Normalised heavy metal and biocide resistance gene abundance in farm dairy
	effluent, faeces, soil and milk samples
5.8	Relative abundance of heavy metal and biocide resistance genes in farm
	dairy effluent, faeces, soil and milk samples on Dairy 1 and Dairy 4 $\ldots$ . 137
5.9	Antimicrobial, heavy metal and biocide resistance gene abundance identified
	in the waste milk sample
5.10	The predicted bacterial phyla of 200 contigs harbouring acquired antimicro-
	bial resistance genes
5.11	The predicted bacterial family of 58 contigs harbouring acquired antimicro-
	bial resistance genes
5.12	Rarefaction analysis of antimicrobial, heavy metal and biocide resistance
	genes detected in sequencing reads
6.1	Summary of sample collection, methodology and outputs to address the
	research objectives in this study

# List of Tables

2.1	Characteristics of $\beta$ -lactamase enzymes
2.2	Plasmid AmpC $\beta$ -lactamase variant groups $\ldots \ldots \ldots$
2.3	Prevalence of AmpC- and ESBL-E in dairy cattle, calves and farm dairy
	effluent
2.4	Prevalence of AmpC- and ESBL-producing <i>Enterobacteriaceae</i> in milk 29
2.5	Factors which may influence AMR in agricultural environments
3.1	Sample collection methods
3.2	Sample processing methods
3.3	Discs for antimicrobial susceptibility tests
3.4	PCR primer sequences and resulting amplicon lengths
3.5	Genome composition of <i>E. coli</i> sequenced in this study
3.6	Resistance profiles of AmpC- and ESBL-producing <i>E. coli</i>
3.7	Characteristics of plasmids from <i>E. coli</i>
3.8	Core genome single nucleotide polymorphism analysis of IncI1 plasmids car-
	rying the $bla_{\rm CMY-2}$ gene $\ldots$ 76
3.9	Number of positive AmpC- and ESBL-producing <i>E. coli</i> samples and isolates 76
3.10	Amount (mg/PCU) of the antibiotic classes used on Dairy 1 and Dairy 4 $$ 77
4.1	Isolates identified from faecal enrichments of ceftiofur treated cows $(n=3)$
	and associated control cows (n=3) $\ldots \ldots \ldots$
4.2	Antimicrobial susceptibility testing and molecular characterisation of the $E$ .
	coli isolates
4.3	Genome composition of three <i>E. coli</i> isolates
4.4	Core genome single nucleotide polymorphism analysis of three AmpC-producing
	$E. \ coli \ \ldots \ $
5.1	Taxonomic classification of contigs co-harbouring two acquired resistance
	genes
5.2	The expected and actual proportion of reads taxonomically classified in the
	DNA mock community sequencing control
5.3	Genomic DNA content and the proportion of reads taxonomically classified
	in the microbial community log distribution standard
5.4	The most abundant contaminants in the PBS and blank reagent controls 146 $$

# Abbreviations

AA	Amino acid		
AMR	Antimicrobial resistance		
AMU	Antimicrobial usage		
ARG	Antimicrobial resistance gene		
ASTs	Antimicrobial susceptibility tests		
ASV	Amplicon sequence variant		
BPW	Buffered peptone water		
CAZ	Ceftazidime		
CLSI	Clinical and Laboratory Standards Institute		
CTX	Cefotaxime		
DCT	Dry cow therapy		
ESAC	Extended-spectrum AmpC $\beta$ -lactamases		
ESBL	Extended-spectrum $\beta$ -lactamase enzymes		
ESBL-E	Extended-spectrum $\beta$ -lactamase-producing Enterobacteriaceae		
EUCAST	European Committee on Antimicrobial Susceptibility Testing		
FDE	Farm dairy effluent		
HGT	Horizontal gene transfer		
MALDI TOF MS	Matrix-assisted laser desorption/ionization-time of flight		
MALDI-IOF MS	mass spectrometry		
MAGs	Metagenome-assembled genomes		
MC	MacConkey agar		
MIC	Minimum inhibitory concentration		
MLST	Multi-locus sequence type		
NMDS	Non-metric multidimensional scaling		
NZ	New Zealand		
NZVA	New Zealand Veterinary Association		
pAmpC	Plasmid-mediated AmpC $\beta$ -lactamase		
PBS	Phosphate buffered saline		
PCA	Principal component analysis		
PMA	Propidium monoazide		
PCR	Polymerase chain reaction		
PCU	Population correction unit		
rRNA	Ribosomal ribonucleic acid		
SCC	Somatic cell count		
SNP	Single nucleotide polymorphism		
ST	Sequence type		
WGS	Whole genome sequencing		
WHO	World Health Organization		

### Chapter 1

## Introduction

### 1.1 General background

Antimicrobial resistance (AMR) is growing global public and animal health concern. The misuse and over-use of antimicrobials in both human and animal health have been suggested as the main drivers of AMR [1]. However, the development and dissemination of AMR is complex [2] and other factors such as heavy metal use may co-select for resistance [3]. Of particular concern are antimicrobial resistance genes (ARGs) which can be encoded on mobile genetic elements and shared to bacteria via horizontal gene transfer (HGT) [4]. The number of antimicrobial resistant infections in humans has been increasing and at current rates, it is estimated that by 2050, 10 million human deaths per year will be attributable to resistant infections [5].

Antimicrobial resistant bacteria and ARGs have been detected globally in agricultural environments including dairy and beef cattle and calves [6, 7, 8, 9, 10, 11, 12, 13]. Antimicrobial usage (AMU) in food-producing animals has become an established consumer concern [14]. A survey of 1,000 participants in the USA found that 90.7% of respondents were concerned that AMU on dairy farms posed a threat to human health [15]. NZ is a comparatively low user of antimicrobials in animal production [16, 17], however given the prominence of the NZ dairy industry in the local and export markets, it is important to consider the role of the NZ dairy farm environment in the development and transmission of AMR. NZ dairy farm systems are largely pasture-based [18] and have a lower prevalence of some cattle diseases such as coliform mastitis [19] and compared with more intensive farming systems such as poultry and swine, dairy farming is less reliant on antimicrobials in NZ [20]. The majority of antibiotics used on NZ dairy farms are for the treatment and prevention of mastitis [21].

Traditionally, AMR surveillance and research has focused on specific target organisms and genes using culture-based methods and molecular techniques. With the development and increased availability of next-generation sequencing methods, the focus of AMR prevalence studies has shifted to also include whole genome sequencing (WGS) which allows a comparison of the genomic epidemiology of isolates and a high-resolution analysis of ARGs. Shotgun metagenomic sequencing enables a broader analysis of ARGs present in a sample. If the sequencing depth is sufficient, the genomic context and bacterial host identification of ARGs can also be inferred. A combination of targeted culture-based analysis and nextgeneration sequencing methods, coupled with shotgun metagenomic sequencing, allows a deep and holistic understanding of AMR in the NZ dairy farm environment.

### 1.2 Antimicrobial resistance on New Zealand dairy farms

Studies researching AMR in NZ dairy farm environments have predominantly focussed on mastitis-associated pathogens such as Staphylococcus aureus, Streptococcus dysgalactiae and Streptococcus uberis [22, 23, 24] as well on the AmpC- and extended-spectrum  $\beta$ -lactamase (ESBL)-producing *Enterobacteriaceae* (ESBL-E) [25, 26]. Of the limited studies conducted in NZ dairy farm environments, the prevalence of ESBL- and plasmidmediated AmpC-producing E. coli was low in dairy cattle faeces [25, 26] and they were not isolated from calf faecal enrichments (n=40) [27]. Farms which were positive for chromosomal mediated AmpC-hyperproducing E. coli (7 of 26 farms) were associated with a higher use of injectable amoxicillin antibiotics [26]. AMU and farm management practices likely drive the prevalence of AmpC- and ESBL-E in agricultural environments and associated food products. Most studies investigating the prevalence of AmpC- and ESBL-E on dairy farms have been conducted in the northern hemisphere or are cross-sectional studies [25, 26, 28]. Therefore, additional studies investigating the prevalence of AmpCand ESBL-E in NZ dairy farm environments are required, particularly those using a longitudinal study design. Such studies would provide useful insight into any effects farm management practices and seasonal variation could have on AMR in agricultural environments. Few longitudinal studies globally, and to the best of my knowledge none in NZ, have been undertaken using culture-independent metagenomic approaches to investigate the abundance and diversity of ARGs in dairy cattle and the dairy farm environment. Such studies would be beneficial to examine the transmission pathways and reservoirs of AMR by providing a high level of detection of ARGs in these environments, and this would aid AMR surveillance. Further research is required to understand the role of dairy farms in the development and transmission of AMR in NZ, with a particular focus on pathogens of concern such as AmpC- and ESBL-E as well as a broader approach to investigate the abundance of ARGs in the wider dairy farm environment.

### 1.3 Thesis aims

This research, using a longitudinal approach, addresses the hypothesis that antimicrobial use on NZ dairy farms influences the prevalence of AMR in dairy farm environments and takes into consideration seasonality and contrasting farm management practices. This research utilises a combination of phenotypic, molecular and next-generation sequencing techniques to study AMR in NZ dairy farm environments.

The specific research objectives of this study were:

- 1. To investigate the sample level prevalence of AmpC- and ESBL-producing *Escherichia* coli from two NZ dairy farm environments.
- 2. To characterise AmpC- and ESBL-producing *E. coli* isolated from the two dairy farm environments and compare the genomic epidemiology (with a particular focus on ARGs) of a subset of strains.
- 3. Assess the impact of systemic antimicrobial treatment on the bovine faecal microbiome composition and diversity, as well as the microbiome recovery post antimicrobial treatment.
- 4. To identify the abundance and diversity of antimicrobial, heavy metal and biocide resistance genes from farm environmental samples.
- 5. To infer the bacterial host of acquired ARGs identified in draft bacterial contigs assembled from farm environmental samples and infer the ARG risk.

Therefore, a three-fold approach was used to achieve these key research objectives in this study:

- Targeting specific antibiotic resistant *E. coli* using culture-based methods (Chapter 3).
- 2. Investigating the impact of systemic antimicrobial therapy on the bovine faecal microbiome and the emergence of antimicrobial resistant bacteria at the individual animal level (Chapter 4).
- 3. An extensive approach utilising shotgun metagenomic sequencing to estimate the abundance and diversity of antimicrobial, heavy metal and biocide resistance genes in farm environmental samples collected over a 15 month period and to identify the bacterial host and genomic context of acquired ARGs (Chapter 5).

### 1.4 Thesis structure

Chapter 1 provides a general background and information on AMR in the NZ dairy farm environment and outlines the structure of this thesis. The literature review in **Chapter** 2 highlights and critically evaluates research relevant to this PhD project. Sections of the literature review were previously published in Foodborne Pathogens and Disease in 2018 [28] and have been updated in this thesis, as well as including additional sections relevant to the research. In Chapter 3, the isolation of AmpC- and ESBL-producing E. coli from farm environmental samples collected during the longitudinal study is presented along with the molecular characterisation of isolates, antimicrobial resistance profiles and the genomic epidemiology of a subset of strains (Figure 1.1). Chapter 4 presents the research of the microbiome study investigating the impact of systemic antimicrobial therapy on the bovine faecal microbiome. Chapter 5 presents the results from the longitudinal study looking at antimicrobial, heavy metal and biocide resistance genes in two NZ dairy farm environments and inferring the bacterial host of acquired ARGs. Chapters 3 - 5 are presented in publication style, with a separate introduction, methods and discussion section per chapter. As a result, there will be some overlap of concepts as each chapter is intended to be understood as an individual piece of work. The final thesis chapter, Chapter 6, provides a holistic general discussion of the overall thesis themes and highlights research gaps and recommends future work to study AMR in NZ.

This thesis improves the understanding of the prevalence and genomic epidemiology of AmpC- and ESBL-producing *E. coli* from NZ dairy farms, whilst taking into consideration fluctuations in AMU and changes in seasonality and farm management practices. This research investigated the impact of a range of systemic antimicrobial treatments on the bovine faecal microbiome and demonstrated an overall reduction within sample taxonomic diversity during treatment and that perturbations in the bovine faecal microbiome generally recovered post-treatment. To the best of my knowledge, this research provides the first comprehensive understanding of the on-farm abundance and diversity of ARGs and improves our overall understanding AMR in NZ dairy farm environments.



Figure 1.1: General outline of the PhD research showing the two on-farm studies and how they relate to the research objectives

### Chapter 2

## Literature review

This chapter is partially based on the publication by Collis *et al.*, [28], as described previously.

### 2.1 Introduction

Antibiotics are a subclass of antimicrobials, along with antifungals, antivirals and antiparasitics [29]. Antibiotics are molecules which can either inhibit (bacteriostatic, prevents growth) or kill (bactericidal) bacteria [30, 31]. Antimicrobial resistance (AMR) occurs when bacteria acquire, or inherently have, the ability to circumvent a drug's mechanism of action [32]. AMR is a global public and animal health concern, primarily due to infections caused by antimicrobial resistant bacteria being difficult to treat and often associated with increased disease severity [33]. AMR has been described as a "slow-burning" major public health concern [34, 35]. If left unchecked and at current rates it is estimated that by 2050, 10 million human deaths per year will be attributable to AMR globally [5]. A recent study estimated the AMR burden across 204 countries and territories and found that globally in 2019, 4.95 million deaths and 1.27 millions deaths were associated with or attributable to bacterial AMR, respectively [36]. Of the deaths attributable to AMR in 2019, E. coli was the bacterial pathogen associated with the most deaths [36]. Among the bacterial pathogens, resistance to fluoroquinolones and  $\beta$ -lactam antibiotics (carbapenems, cephalosporins and penicillins) accounted for more than 70% of human deaths attributable to AMR [36].

The misuse and over-use of antimicrobials in both human and animal health has been suggested as the main drivers of resistance [1, 2], however other factors such as heavy metal, biocide or disinfectant use may co-select for AMR [3, 37, 38]. The emergence, persistence and spread of AMR is often more intricate than a single driver [2] and antimicrobial resistance genes (ARGs) evolved in natural environments prior to the use of antibiotics in clinical settings [39]. ARGs in human pathogens are hypothesised to originate from environmental organisms [40] and the misuse and over-use of antimicrobials in human and animal health has exacerbated the selection pressures for resistant phenotypes.

In New Zealand (NZ) animal production, prophylactic use of antimicrobials is only permitted with a veterinary prescription, and the use of antimicrobials for growth promotion is prohibited [41]. A target has been set by the NZ Veterinary Association (NZVA), that by 2030, NZ will no longer rely on antimicrobials for the maintenance of animal health and wellness [42]. Reports analysing antimicrobial sales data from 30 countries identified NZ as the third lowest user of antimicrobials in animal production [16, 17], using approximately 12.9 times less than in human medicine [16]. Compared to poultry and swine, dairy farming is less dependent on antimicrobials, and in comparison to international standards, a study found NZ dairy herds (n=477) were generally low users of antimicrobials, however some high-use herds were identified [43]. On NZ dairy farms, the majority of antibiotics are used for the treatment and prevention of mastitis [21], such as for dry cow therapy, which is the prophylactic use of antibiotics to reduce further mastitis events. The NZ dairy production system is largely pasture-based, which may contribute to the low antimicrobial use in this industry.

The development and transmission of antimicrobial resistant bacteria between humans, animals and the environment is multi-factorial and complex, and several potential transmission pathways of resistant bacteria between humans, the dairy farm environment and vice versa have been proposed [28]. The extensive use of antimicrobials in food-producing animals globally has become a consumer concern [14]. Compared to other farming systems, little is known about the contribution of dairy farming as a source of antimicrobial resistant bacteria and ARGs, however epidemiological relationships between genetically similar antimicrobial resistant bacteria found in both humans and livestock have been proposed [44, 45, 46].

### 2.2 Antimicrobial resistance mechanisms

Key mechanisms of AMR include (i) target site modification or protection, (ii) efflux pumps, (iii) degradation or modification of the antibiotic, (iv) reduced permeability and (v) expression changes [30, 32]. AMR can either be intrinsic, acquired or adaptive. Intrinsic resistance is the inherent characteristic of a resistant phenotype in a given bacterial species [40]. For example, Gram-negative bacteria are intrinsically resistant to glycopeptides due to their inability to permeate the outer membrane of Gram-negative bacteria [32]. In contrast, acquired resistance occurs when a previously susceptible bacteria acquires resistance towards an antibiotic through either mutation or acquisition of new genetic material via horizontal gene transfer (HGT) [32]. For example, acquisition of the bla<sub>CTX-M</sub> genes in E. coli. HGT can occur through either conjugation, transduction or transformation [2]. Plasmids play a key role in the transfer of ARGs [47] and they may harbour multiple resistance genes, resulting in multi-drug resistant phenotype which is of particular concern [48, 49, 50]. Mobile genetic elements such as transposons, integrons and insertion sequences are involved in the transfer and acquisition of ARGs on plasmids [32]. Thirdly, adaptive resistance occurs in response to environmental conditions and is not the result of genetic changes [32].

Natural environments are a key source of antibiotics and thus are locations where resistance has evolved [51], however some ARGs may have functions other than resistance in the natural environment. For example multi-drug efflux pumps are also involved in the detoxification of intracellular metabolites, cell homeostasis and bacterial virulence within plant and animal hosts [52]. Therefore, to understand AMR risk, particularly in sequencebased studies, it is essential to determine the bacterial host of an ARG (e.g. a human pathogen) and the genetic context (e.g. intrinsic or acquired resistance) [53].

### 2.3 Bacterial pathogens of concern

The World Health Organisation (WHO) has defined a priority list for 20 bacterial pathogens of concern to human health, and categorised them as critical, high and medium, according to ten criteria relating to health and resistance [54]. This list includes several pathogens of concern which are also associated with dairy cattle including ESBL-E, fluoroquinolone-resistant *Salmonella* spp., fluoroquinolone resistant *Campylobacter* spp., methicillin-resistant and vancomycin intermediate and resistant *Staphylococcus aureus* [55, 56, 57].

### 2.3.1 Mastitis-associated pathogens

Numerous bacterial pathogens may cause bovine mastitis and can be categorised into two groups according to their main reservoir: contagious or environmental pathogens [58]. Contagious mastitis-causing pathogens generally include *S. aureus*, *Streptococcus agalactiae* and *Streptococcus dysgalactiae* [58], although *Strep. dysgalactiae* can also be considered as an environmental pathogen [59]. Environmental pathogens also include members of the *Enterobacteriaceae* family where *Escherichia coli*, *Klebsiella pneumoniae* and *Klebsiella oxytoca* are of particular importance, as are *Streptococcus uberis*, coagulase negative staphylococci [60, 61, 58], *Serratia* spp., *Enterococcus* spp. [59] and *Pseudomonas aeruginosa* [62]. Coliforms are not a common cause of mastitis in NZ [19].

The prevalence and diversity of mastitis-associated pathogens varies between studies, geographic regions, can be herd-specific [59, 63, 64, 65, 66, 67] and may be due to different methods used for the detection of these pathogens. Variations may also occur in case definitions of clinical mastitis as well as national and local policies regarding AMU in livestock animals and farming practices. The wide range of bacterial pathogens that may trigger bovine mastitis highlights the difficulty in identifying the causative organism(s) and incorrect treatment (either antibiotics to which the aetiological agent is resistant to or using third-generation cephalosporins when penicillin would suffice) could potentially contribute to the development and spread of AMR in dairy farm environments. The use of on-farm bacterial culture and subsequent selective antimicrobial therapy was associated with a decrease in AMU compared to blanket antimicrobial treatment for clinical mastitis cases in a study on NZ dairy farms (n=7) [68]. In addition, a network meta-analysis suggested that critically important antimicrobials are not required to treat nonsevere clinical cases of mastitis caused by *E. coli* or *Klebsiella* spp. [69] and that the treatment efficacy of clinical mastitis with both critically important and not critically important antimicrobials was comparable for the most prevalent mastitis-associated pathogens including *S. aureus*, *E. coli*, *Klebsiella* spp., non-*aureus* staphylococci and non-*agalactiae* streptococci [69]. NZ veterinary clinical pathology laboratories had reported an increase in  $\beta$ -lactam resistance among *Strep. uberis* bovine clinical isolates and a recent study found that this resistance was mediated by mutations in penicillin binding proteins [24]. A study of *S. aureus* (n=57) isolated from bovine mastitis cases in NZ found that AMU patterns were not associated with ARGs in the *S. aureus* isolated across the dairy farms studied (n=17), although some genes were over-represented among herds using ampicillin/cloxacillin dry cow therapy, however these genes were not all associated with AMR [23].

### 2.3.2 Enterobacteriaceae

Enterobacteriaceae are part of the normal microflora in the intestinal tract of many mammals including humans and dairy cows and can survive in variable environmental conditions. E. coli, which belong to the Enterobacteriaceae family, are a diverse species which are naturally found as commensals in the intestinal tract of healthy humans and animals, however a range of E. coli pathotypes such as Shiga toxin-producing E. coli (STEC) and enteropathogenic E. coli (EPEC) can also cause severe human disease [70]. E. coli are classified serologically according to O (O-specific polysaccharide component of lipopolysaccharide), H (flagellar) and K (capsular) antigens [71, 72]. Currently, 188 O-serogroups and 53 H antigens are included in the E. coli serotyping scheme [73]. Specific combinations of certain O and H antigens are defined as a serotype [72]. E. coli can also be classified using multi-locus sequence typing (MLST) [74] and by phylogroups [75, 76], which is useful to help understand the phylogenetic relationship between isolates and for epidemiological studies. E. coli can be divided into eight E. coli phylogroups (A, B1, B2, C, D, E, F and G) [76, 77, 78] that can be identified using PCR or by *in silico* analysis of whole genome sequencing (WGS) data [76, 78, 77]. It has been proposed that certain E. coli phylogroups may be more abundant in different niches [79, 80], although such associations have been debated and may be complicated by the genetic fluidity of E. coli and their ability to adapt to different ecological niches. In addition, E. coli can be characterised by pathotypes such as STEC and EPEC according to the symptoms they cause and pathotype-specific virulence factors [72, 81]. These pathotypes can be difficult to distinguish due to the lack of definitive defining features between pathotypes [81] which is further complicated by the potential for acquisition of virulence factors via HGT.

Enterobacteriaceae can develop resistance to a range of antimicrobials including colistin [46], tetracycline [82] and carbapenems [83]. E. coli isolated from dairy cattle and calves in NZ were resistant to  $\beta$ -lactams including third-generation cephalosporins, tetracycline,

streptomycin and a small number to ciprofloxacin [25, 27]. However one of the most widespread mechanisms of AMR in *Enterobacteriaceae* is the production of AmpC and ESBL  $\beta$ -lactamase enzymes. A study assessing the global burden of AMR in human populations found that 59,900 and 50,100 AMR attributable deaths in 2019 were caused by thirdgeneration cephalosporin resistant *E. coli* and *K. pneumoniae*, respectively [36]. AmpC-, ESBL-E and mastitis-associated pathogens are both significant bacterial pathogens from dairy farm environments when considering human and animal health and the development and transmission of AMR. AmpC- and ESBL-E are discussed in sections 2.4.1 and 2.4.2, with a particular focus on *E. coli*.

### 2.4 $\beta$ -lactamases

 $\beta$ -lactamase enzymes are found in both Gram-negative and Gram-positive bacteria [84] and they existed prior to the widespread use of  $\beta$ -lactam antibiotics [85]. The production of  $\beta$ -lactamase enzymes encoded by the *bla* genes is a common mechanism of resistance to  $\beta$ -lactams in Gram-negative bacteria [86], which is concerning as the  $\beta$ -lactams are commonly used as first-line therapy to treat infections caused by Gram-negative bacteria [87].  $\beta$ -lactamase enzymes inactivate  $\beta$ -lactam antibiotics by hydrolysis [88], specifically by cleaving the  $\beta$ -lactam amide bond [89].  $\beta$ -lactam enzymes have been classified using either amino acid (AA) sequence similarity in the Ambler classification system (A to D [84]) or using functional classification (substrate spectrum and inhibitor profiles) in the Bush-Jacoby-Medeiros scheme (1 to 4 with subgroups [90, 86]). The characteristics of  $\beta$ -lactamase enzymes, including the substrate spectrum, is shown in Table 2.1. A large number of variants within the  $\beta$ -lactamase families have been described [85].

Table 2.1: Characteristic	s of $\beta$ -lactamase	enzymes			
Enzyme	Ambler classification	Bush- Jacoby- Medeiros group	Examples	Substrate spectrum	Inhibitors
Broad spectrum penicillinases	Class A	2b	TEM-1, TEM-2, SHV- 1	Penicillins, narrow-spectrum cephalosporins	Clavulanic acid
ESBLs	Class A	2be	CTX-M, TEM and SHV (not parent type enzymes)	Penicillins, monobactams, cephalosporins (1st- and 3rd-gen.)	Clavulanic acid, tazobactam, sulbactam
KPC carbapenemases	Class A	2f	KPC	Penicillins, cephalosporins, carbapenems	Clavulanic acid (weak), tazobactam, boronic acid
Metallo- $\beta$ -lactamases	Class B	3	IMP, VIM, NDM	Penicillins, cephalosporins, cephamycins, carbapenems	Metal chelators
AmpC $\beta$ -lactamases	Class C	1	CMY, FOX, ACT MOX, ACC, DHA	Penicillins, monobactams, cephalosporins (1st, 2nd and 3rd-gen.), cephamycins	Cloxacillin, boronic acid
Oxacillinases	Class D	2d	OXA-1	Penicillins, narrow-spectrum cephalosporins, increased hydrolysis of cloxacillin or oxacillin	NaCl
Oxacillinases	Class D	2df	OXA-48	Penicillins, cloxacillin or oxacillin, cephalosporins, carbapenems <sup>b</sup>	NaCl
<ul> <li><sup>a</sup> Information in the tab</li> <li>92, 86, 93].</li> <li><sup>b</sup> Substrate spectrum var</li> </ul>	le is adapted fro ies. Some variar	m Pitout (20 its also hydr	112), Rubin and Pitout ( olyse expanded-spectrum	2014), Bush and Jacoby (2010), Jacoby and oxyimino-cephalosporins [94].	d Munoz-Price (2005) [91,

### 2.4.1 ESBL enzymes

Extended-spectrum  $\beta$ -lactamase (ESBL) enzymes belong to the Amber Class A and Bush-Jacoby-Medeiros group 2be [86]. ESBLs confer resistance to a broad range of  $\beta$ -lactam antibiotics, including penicillins, monobactams and first- and third-generation cephalosporins (Table 2.1) [92, 95] and have variable levels of resistance to fourth-generation cephalosporins [96]. Third- and fourth-generation cephalosporins have been identified as 'critically important' antimicrobials for use in human medicine as they are the antibiotics of last resort to treat serious human infections and pathogens causing these infections could be acquired via non-human sources [97, 98]. Resistance to third-generation cephalosporins in human clinical infections caused by *E. coli* and *K. pneumoniae* enhances the reliance on the remaining antimicrobials available to treat such infections (e.g. carbapenems) [99].

The most common ESBL variants include the sulfhydryl reagent variable (SHV), Temoneira (TEM) (excluding the parent type) enzymes and the CTX-M variants [91]. Less frequently identified ESBL variants include the SFO, PER, BES, BEL, TLA, GES and VEB types [100]. The TEM- and SHV-type enzymes are derived from the narrow spectrum parenttype enzymes (TEM-1, TEM-2, TEM-13 and SHV-1) by AA substitutions, however these changes can result in an extended hydrolytic spectrum [86, 92, 96]. CTX-M type ESBLs were named based on their increased activity against cefotaxime compared to ceftazidime [101], however some CTX-M types such as CTX-M-15 confer higher levels of resistance to ceftazidime [96]. CTX-M type ESBLs likely emerged from chromosomal  $\beta$ -lactamase genes from *Kluyvera* spp. which were mobilised onto plasmids [102]. Various genetic elements, particularly IS Ecp1 and IS CR1, have been associated with the mobilisation of the  $bla_{CTX-M}$ genes onto plasmids, likely on multiple occasions [96, 101, 102]. Over 172 CTX-M types have been identified [103] and are classified into five main groups (CTX-M-1, CTX-M-2, CTX-M-8, CTX-M-9 and CTX-M-25) according to their AA sequence [96]. Each cluster was named after the first member of each group that was identified [103]. Changes in the AA sequence can give rise to CTX-M variants within each group [104, 105]. AA changes in the parent OXA-type enzymes has given rise to some variants with an extended-spectrum of hydrolysis and are thus classified as ESBLs [106]. The dissemination of ESBLs has largely been driven by the spread of plasmids harbouring ARGs between related bacteria [107] and by the spread of epidemic clones harbouring ESBLs such as sequence type 131 (ST131) in humans [108, 109].

The predominant ESBL variants differ geographically and can exhibit temporal shifts [110, 111, 112], however CTX-M-14 and CTX-M-15 (which belong to CTX-M-9 and CTX-M-1 groups, respectively) are the predominant genotypes in most geographic regions [110]. In livestock and animal sectors, CTX-M-1, CTX-M-14 and CTX-M-15 ESBL types have frequently been detected [113, 114, 115, 116, 117]. ESBLs have been identified in numerous *Enterobacteriaceae*, such as *K. pneumoniae*, *E. coli*, *Enterobacter* spp., *Serratia* spp., *Proteus* spp., *Providencia* spp., *Salmonella* spp. and *Morganella morganii* [95], but they are most frequently detected in *E. coli* isolates [33], which is of high concern to public health.

ESBL-E can cause urinary tract and bloodstream infections [33] and are associated with both hospital-acquired [118] and community-acquired infections [119]. Recently there has been an increase in the number of community-acquired infections associated with ESBL-E [120]. In NZ, the national period-prevalence rate of ESBL-E (from a clinical sample) was 11.1 people per 100,000 population in 2016, according to a one month survey [121]. The majority of ESBL-E identified from clinical samples during the study were *E. coli* (74.1%, 386 of 521) and the infections were mostly categorised as community-related (60.8%)[121].

ESBLs may be located on the bacterial chromosome but are often encoded on plasmids [102], the latter of which is of particular concern as such plasmids can be transferred via HGT and these plasmids may harbour additional ARGs, resulting in a multi-drug resistant phenotype [48]. Many ESBL-E are resistant to multiple classes of antibiotics [122, 123, 124] and of particular concern are those that are also resistant to carbapenems. The WHO bacterial priority list used this resistance class as a marker for multi-resistant and pan-resistant (resistance to agents in all antimicrobial classes) bacteria [54] and carbapenems are often the 'last line of defence' for treating serious infections [98]. Another 'critically important' antimicrobial is colistin which is a polymyxin antibiotic [125] that is often used to treat infections caused by multi-drug resistant Gram-negative bacteria. Worryingly, three ESBL-producing *E. coli* isolated from dairy farms in China also harboured the plasmid-mediated colistin resistance gene, mcr-1 [126].

ESBL-E may cause disease in dairy cattle and have been isolated from milk from cows with mastitis in France [127], Turkey [128], Greece [124], Japan [129] and China [55]. Although less frequent, ESBL-E abundance may be underestimated if sampling of apparently healthy dairy cattle with subclinical levels of mastitis does not occur [55]. In NZ, coliform mastitis (caused by members of the *Enterobacteriaceae* family) is rare [19, 130, 68]. The prevalence of ESBL-producing *E. coli* was low in recent cross-sectional surveys of dairy farms in NZ [25, 26] and was not detected from recto-anal mucosal swabs from calves fed waste milk in NZ [27].

### 2.4.2 AmpC $\beta$ -lactamase enzymes

Class C AmpC  $\beta$ -lactamase enzymes utilise serine for  $\beta$ -lactam hydrolysis and belong to the Bush group 1  $\beta$ -lactamases [86]. These enzymes are located in the bacterial periplasm [131] and confer resistance to penicillins, monobactams, first-, second- and third-generation cephalosporins and cephamycins (Table 2.1).

### 2.4.2.1 Chromosomal ampC genes

Enterobacteriaceae with chromosomally-encoded AmpC  $\beta$ -lactamases include Enterobacter spp., Escherichia albertii, Escherichia fergusonii, E. coli, Hafnia alvei and Serratia spp. [131]. The chromosomal ampC gene (synonym for the  $bla_{\rm EC}$  gene [132]) is absent in K.

pneumoniae, K. oxytoca and Salmonella spp. [131, 133]. Inducible  $\beta$ -lactamases, which are usually produced at low levels but are inducible when exposed to certain  $\beta$ -lactam antibiotics, are found in Enterobacter spp., M. morganii, Citrobacter freundii, Serratia marcescens and Providencia spp. [134]. However, the chromosomal ampC gene in E. coli is non-inducible as E. coli lacks the transcriptional regulator ampR gene [135] and it is instead regulated by promoter and attenuator mechanisms [136] and therefore is weakly expressed [91]. Mutations in the promoter and attenuator regions can result in overexpression of the ampC gene, increased production of the AmpC  $\beta$ -lactamase and thus resistance [137]. Common positions for mutations in the promoter region of the ampCgene include -42, -18, -1, +58, +70 and +81 [138, 139]. Mutations which are important in strengthening the ampC promoter include creating a consensus -35 box (transversions at either position -32  $[T \rightarrow A]$  or -42  $[C \rightarrow T]$  and base pair insertions that result in a promoter which more closely resembles the E. coli consensus sigma promoter sequence (insertions which increase the distance between the -35 and -10 boxes from 16 bp to either 17 or 18 bp) [140, 139]. Mutations can also arise in the attenuator region and are thought to destabilise the mRNA hairpin structure, leading to increased transcription of the ampCgene [139]. No association was identified between specific mutations found in the ampCpromoter region and the level of AmpC  $\beta$ -lactamase production [138, 140]. Other factors such as porin expression and the rate of substrate delivery to the enzyme can also influence the spectrum of resistance [131].

### 2.4.2.2 Plasmid-mediated AmpC-producing E. coli

Plasmid-borne AmpC  $\beta$ -lactamase genes are derived from bacteria which have a chromosomal ampC that has been mobilised onto plasmids (Table 2.2) [133]. According to AA sequence similarity, the plasmid AmpC  $\beta$ -lactamase variants are classified into five main groups (Table 2.2) [131]. In addition to the AmpC  $\beta$ -lactamase variant groups listed in Table 2.2, a novel plasmid-mediated AmpC  $\beta$ -lactamase CFE-1 was detected in Japan in a clinical E. coli isolate [141] and more recently in 2018, CFE-2 was characterised from a clinical isolate of C. freundii [142]. The nomenclature for AmpC  $\beta$ -lactamase enzymes is inconsistent, with some variants named according to their preferred substrate, acronyms for Amber Class C and even abbreviations of the location where the variant was first identified [133, 143]. The majority of the pAmpC genes are constitutively expressed [143], with the exception of DHA-1 and ACT-1 which are inducible by  $\beta$ -lactam antibiotics [144, 145]. Surprisingly, the ACC-1 AmpC  $\beta$ -lactamase variant is susceptible to cephamycins, including cefoxitin [146] which is often used to screen for AmpC-producing Enterobacteriaceae. This complicates the isolation of *Enterobacteriaceae* producing this AmpC  $\beta$ -lactamase variant and may result in its true prevalence being under-reported. However, CMY-2 is the most prevalent plasmid-mediated AmpC  $\beta$ -lactamase globally [131].

Group	$\begin{array}{c} \mathbf{AmpC} \ \beta \text{-lactamase} \\ \mathbf{types} \end{array}$	Likely origin of the $ampC$ gene	GC content (%)	Ref.
CIT	CMY-2 to CMY-7, CMY-12 to CMY-18, CMY-21 to CMY-23, LAT-1 to LAT-3, BIL-1	Citrobacter freundii	52	[147]
EBC	ACT-1, MIR-1	$Enterobacter { m spp.}$	55	[148, 149]
DHA	DHA-1, DHA-2	Morganella morganii	52	[144, 150]
ACC	ACC-1, ACC-2	Hafnia alvei	46	[146]
MOX/FOX	MOX-1, MOX-2, CMY-1, CMY-8 to CMY-11, CMY-19, FOX-1 to FOX-5	Aeromonas spp.	63	[151, 152]

Table 2.2: Plasmid AmpC  $\beta$ -lactamase variant groups

<sup>a</sup> Information in the table adapted from Jacoby (2009) [131].

### 2.4.2.3 Extended-spectrum AmpC $\beta$ -lactamases

Extended-spectrum AmpC (ESAC)  $\beta$ -lactamases can arise from insertions, deletions and substitutions of AA in the AmpC  $\beta$ -lactamase, specifically in the 10 or 11 helix, R2 loop or  $\Omega$  loop [153]. This can result in more flexibility in the enzymes active site and thus a broadened hydrolysis spectrum, including against fourth-generation cephalosporins such as cefepime [153, 154]. ESAC  $\beta$ -lactamases can arise from both chromosomal-mediated and plasmid-mediated *ampC* genes. Worryingly, some ESAC  $\beta$ -lactamases have an increased efficiency towards the carbapenem imipenem [153], which is a critically important antimicrobial in human medicine [155]. ESAC-producing *E. coli* have been isolated from human clinical samples [132] as well as from bovine milk and faeces in Brazil [154], *E. coli* isolated from faecally contaminated environments on UK dairy farms [156] and from healthy and diseased cattle in France [157]. ESAC  $\beta$ -lactamases have also been reported in other *Enterobacteriaceae* including *Enterobacter* spp. [158, 159] and *S. marcescens* [160].

### 2.5 Methods for detecting AmpC- and ESBL-producing Enterobacteriaceae

Traditionally, bacteria have been classified as 'susceptible' or 'resistant' according to the likelihood of therapeutic treatment success or failure respectively, with a given antibiotic [40]. Minimum inhibitory concentration (MIC) breakpoints and disc diffusion tests with zone diameter (mm) breakpoints have been defined using the pharmacodynamics and pharmacokinetics of the antibiotic, information regarding the wild-type distribution of MICs as well as clinical outcomes of infections that have been treated with the specific antibiotics [161]. In addition, isolates with zone diameters which fall in-between 'susceptible' or 'resistant' are classified as 'intermediate', where the therapeutic outcome is unclear due to an intermediate level of antimicrobial activity. Recently, the European Union Committee

on Antimicrobial Susceptibility Testing (EUCAST) guidelines reclassified this category as 'susceptible, increased exposure', to reflect the uncertainty and that changes to various factors such as the dosing regimen or antibiotic concentration at the infection site may result in an increased likelihood of therapeutic success [162]. Such breakpoints should be reviewed after the antibiotic has been used in clinical settings to take into consideration resistance mechanisms which may have developed [161]. Published guidelines and standards for antimicrobial susceptibility testing (AST) interpretation of breakpoints and cutoffs have been provided by the Clinical and Laboratory Standards Institute (CLSI) [163] and EUCAST [164, 165]. More recently, epidemiological cutoff (ECOFF) values have been proposed which can be used to determine whether a strain is categorised as fully susceptible (wild-type) or less susceptible (non wild-type) within a specific bacterial species [166]. The use of ECOFF values removes the biases associated with using clinical breakpoints according to the likelihood of treatment failure or success, although current ECOFF estimations rely on databases which are likely to be biased towards bacteria of importance for human health [167]. Normalised resistance interpretation (NRI) methods can be used to define ECOFF values within a specific bacterial population [168], thus removing these biases. These methods are particularly useful for non-human associated or environmental bacterial populations. For example, these methods have been used to assess AMR in E. coli and Salmonella spp. from wild ungulates in Portugal [169]. The use of NRI defined ECOFFs may be limited when the number of wild-type (fully susceptible) isolates for a given population is low [168].

### 2.5.1 ESBL-producing Enterobacteriaceae

Published guidelines are available from CLSI and EUCAST for the isolation and detection of ESBL-E [163, 164, 170]. Detection of ESBL-E is based on resistance to oxyminocephalosporins in a screening test, followed by subsequent phenotypic confirmation [170] which has been shown to be highly sensitive to detect ESBL-E [171]. Cefpodoxime is hydrolysed by TEM, SHV and CTX-M type ESBLs and is the most sensitive individual cephalosporin indicator [96, 170]. However, screening with only ceppodoxime may lead to a high number of false-positives as mechanisms other than ESBL production can result in cefpodoxime resistance, such as the production of AmpC  $\beta$ -lactamases. Ceftazidime is an effective substrate for TEM and SHV ESBLs [96]. The hydrolytic activity of ESBL variants towards some  $\beta$ -lactams may differ [172], therefore it is recommended to use a combination of indicator third-generation cephalosporins, such as cefotaxime (or ceftriaxone) and ceftazidime for screening [170, 171], which is more specific than testing cefpodoxime alone. According to EUCAST and CLSI guidelines, a screening breakpoint of >1 mg/L is recommended for cefotaxime, ceftriaxone, ceftazidime and cefpodoxime for ESBL-E [170, 163]. ESBL production can be confirmed using either the double-disc synergy, ESBL gradient or the broth microdilution tests [170]. Double-disc synergy tests are commonly used for ESBL confirmation and they aim to demonstrate a synergistic effect between a third-generation cephalosporin and an ESBL inhibitor [172]. For example, double-disc synergy tests that compare the zone diameters for cefotaxime and ceftazidime with/without clavulanic acid is recommended for phenotypic confirmation of *Enterobacteriaceae* with no chromosomal de-repressed or inducible AmpC (e.g. *E. coli*, *Klebsiella* spp.). For *Enterobacteriaceae* with inducible AmpC (e.g. *Enterobacter* spp., *Serratia* spp.) it is recommended to use cefepime, which is stable to AmpC hydrolysis [170]. The latter method is also recommended for indeterminate results using the cefotaxime/ceftazidime and clavulanic acid confirmation method [170]. Phenotypic confirmation tests are essential to reduce false positive results. AmpC  $\beta$ -lactamases can complicate ESBL confirmation tests as they are not inhibited by ESBL inhibitors and are resistant to third-generation cephalosporins (Table 2.1) [143].

Detection of ESBL genes using either PCR [173, 7] or WGS is useful for both AMR surveillance and epidemiological studies. As next-generation sequencing is becoming cheaper and more widely available, prevalence studies are more frequently utilising these methods [25, 26, 174] which allows for a higher resolution analysis of  $\beta$ -lactam resistance genes as well as additional ARGs and enables a comparison of the genomic epidemiology of ESBL-E. WGS may be particularly useful when phenotypic confirmation methods are inconclusive [175].

### 2.5.2 AmpC-producing Enterobacteriaceae

Similarly, a two-step method can be used for the screening and confirmation of AmpCproducing *Enterobacteriaceae*. Resistance to cefoxitin can be used to differentiate AmpC production from other  $\beta$ -lactamases [131, 157], therefore reduced susceptibility to cefotaxime and/or ceftazidime and cefoxitin can be used in the screening process, followed by confirmation with a synergy test containing either cloxacillin or boronic acid [170]. A three-disc comparison assay can also be used for AmpC  $\beta$ -lactamase confirmation in which the zone diameters are compared between discs containing (i) cefpodoxime and an AmpC inducer, (ii) cefpodoxime, AmpC inducer and an ESBL inhibitor and (iii) cefpodoxime, an AmpC inducer, an ESBL inhibitor and AmpC inhibitors (D69C AmpC disc test, Mast Group Ltd., Liverpool, United Kingdom). Plasmid-mediated ACC-1 AmpC  $\beta$ -lactamases are susceptible to cephanycins however [146] and therefore will not be included in any subsequent analyses if reduced susceptibility to cefoxitin is a requirement for AmpC confirmation. After AmpC confirmation, it is important for AMR surveillance to distinguish between plasmid-mediated and chromosomal AmpC, which cannot be distinguished using phenotypic methods [176]. Instead, in E. coli they can be differentiated using either a multiplex PCR for the pAmpC genes [177, 178], mutations in the ampC  $\beta$ -lactamase gene promoter region can be amplified using PCR and subsequent Sanger sequencing of the PCR product [138] or the resistance mechanism can be identified through WGS. Mechanisms other than AmpC  $\beta$ -lactamase production can result in resistance to cefoxitin in Enterobacteriaceae, such as porin loss [179], that may lead to false positive results after the screening test and highlights the importance of confirmatory tests. Isolates showing false positive results should be further analysed to identify the resistance mechanism for cefoxitin and third-generation cephalosporins, which is important for AMR surveillance. Reduced susceptibility to cefoxitin and cefepime in *E. coli* may be indicative of ESAC  $\beta$ -lactamase production [163, 154] or isolates co-producing plasmid-mediated AmpC and ESBL  $\beta$ -lactamases [153]. ESBL confirmation tests and identification of pAmpC genes should be carried out for *E. coli* with suspected ESAC  $\beta$ -lactamase production. Once these resistance mechanisms are excluded, the *ampC* gene should be sequenced in these isolates and examined to confirm ESAC production [154].

New diagnostic methods to detect AmpC- and ESBL-E have been developed, such as the matrix-assisted laser desorption/ionization-time of flight (MALDI-TOF) mass spectrometry (MS) based direct-on-target microdroplet growth assay which relies on the synergistic effect of indicator cephalosporins and inhibitors [175] and a similar method has also been developed for the detection of carbapenemases in *Enterobacterales* [180]. These methods provide fast detection of AmpC- and ESBL-E which enables prompt patient treatment, however further validation and standardisation of such methods is required before they can be routinely used [175].

### 2.6 Prevalence of AmpC- and ESBL-E in the dairy farm environment

# 2.6.1 Prevalence of AmpC- and ESBL-E in dairy cattle, calves and farm dairy effluent

Many studies have investigated the prevalence and incidence of AmpC- and ESBL-E in dairy cattle, calves and effluent and of those, the prevalence and incidence varies widely (Table 2.3). Study comparisons are often confounded due to the study-specific sampling and detection methods used as well as the different sample types, which often results in data from study populations that cannot be extrapolated [7]. The culture-based selection methods and enrichment protocols may differ in several studies (Table 2.3) resulting in varying screening sensitivities. For example, many studies use non-selective enrichment of faecal samples or effluent followed by plating on selective agar (Table 2.3), whereas some studies do not use a pre-enrichment step [181] and others use indicator antibiotics in the pre-enrichment step followed by plating on selective agar [7, 182, 183, 184]. In addition, numerous types of selective and chromogenic agar plates are available for the isolation of AmpC- and ESBL-E (Table 2.3). A study comparing AmpC- and/or ESBL-producing E. coli isolated from pig caeca samples on either MacConkey (MC) agar supplemented with cefotaxime (1 mg/L) or Brilliance<sup>TM</sup> ESBL agar found that in 2013 and 2017 the recovery of isolates with a multi-drug resistant phenotype was higher on *Brilliance*<sup>TM</sup> ESBL agar and the proportions of multi-drug resistant isolates from both agars was similar in 2015 [185]. This finding may be partially due to *Enterobacteriaceae* producing either carbapenemases, ESBLs or AmpC  $\beta$ -lactamases, being able to grow on MC agar with cefotaxime,
whereas  $Brilliance^{TM}$  ESBL agar selects for ESBLs only and the genes encoding ESBLs are often associated with additional ARGs, which can result in a multi-drug resistance phenotype [50, 48, 49]. ESBL-producing *E. coli* are also commonly found as contaminants in both Bolton broth and modified charcoal cefoperazone deoxycholate agar used for the isolation of *Campylobacter* spp. [186, 187], which both contain the third-generation cephalosporin cefoperazone and other antimicrobial agents, and may be brought about by the presence of additional ARGs co-harboured by ESBL-E. Therefore, multiple selective agars should be used in parallel for AMR surveillance studies. Such variations in the culture-based detection methods can complicate comparisons between prevalence studies and lead to estimates that differ from the actual prevalence. Farm selection criteria and smaller sample sizes may also influence the prevalence estimates.

In some prevalence studies AmpC- and ESBL-E are not differentiated [188, 189, 190] which is an important step in AMR surveillance and epidemiological studies. Ideally, more than one isolate per sample should be analysed, as distinct ESBL-producing *E. coli* can be isolated from the same sample [126]. In some studies only one presumptive isolate was analysed per plate [182, 183, 191] which could potentially reduce the diversity observed in prevalence studies if further characterisation of the isolates is not completed. A combination of both culture-based isolation methods and WGS are advantageous for surveillance of AmpC- and ESBL-producing *E. coli* [185]. WGS, particularly both short- and long-read sequencing, enables the detection and genomic context of a range of ARGs, high-resolution analysis of plasmids and can be used to help identify novel ARGs in conjunction with culture-based methods [185]. Generally the phenotypic and genotypic data are concordant, however for a small number of antimicrobial classes, the presence of a gene does not always correlate with a resistant phenotype [185] which highlights the importance of using both methods in parallel.

In NZ, few studies have investigated the prevalence of antimicrobial resistant bacteria, including AmpC- and ESBL-E, in animal-derived food products, dairy cattle and their environment. One regional cross-sectional study of dairy farms in NZ found a low prevalence of ESBL-producing E. coli (1 of 116; 0.9%) in pooled faecal samples [25]. No E. coli with pAmpC genes were isolated but 7.89% (9 of 114) of faecal samples were positive for putative AmpC hyperproducers with mutations in the promoter region of the ampCgene [25]. Similarly, a nationwide cross-sectional study (n=26 farms) in NZ did not detect any ESBL-producing E. coli [26] but chromosomal AmpC-producing E. coli were isolated from 14% (11 of 78) of pooled faecal enrichments originating from seven farms [26]. The lower prevalence of AmpC- and ESBL-E on NZ dairy farms compared to overseas studies may be due to the comparatively low use of antimicrobials in NZ animal production systems [16, 17] or the generally less intensive pasture-based farming system. A crosssectional study of ruminant farming systems in Spain found beef cattle herds and sheep flocks, which have a semi-intensive pasture-based farming system, had significantly lower herd level prevalence of cefotaxime-resistant E. coli compared to dairy cattle, which are generally housed inside pens [9]. The prevalence of ESBL-E has been demonstrated to be

lower on organic dairy farms than on conventional dairy farms within the same country [7, 192]. The detection of ESBL-E on organic farms which have reduced AMU compared to conventional farms suggests that factors other than AMU may also contribute to the development of antimicrobial resistant bacteria [192].

The prevalence of antimicrobial resistant bacteria shed in faeces is typically highest in young calves compared to adult dairy cattle [193]. The age distribution of antimicrobial resistant E. coli from calves and dairy cattle in the USA was examined and tetracycline resistant E. coli were more prevalent in calves compared to lactating cows [193]. Removing oxytetracycline from the diet of calves did not reduce the proportion of tetracycline resistant E. coli shed in calf faces [193], which suggests that antimicrobial exposure may not be required for the persistence and shedding of antimicrobial resistant E. coli in calf faeces. E. *coli* isolated from calves in the study population were frequently resistant to streptomycin, sulfadiazine and tetracycline (SSuT) [193]. To understand the factors driving the high prevalence of this resistant phenotype in young calves, experimental oral inoculation studies were performed in young calves and heifers (12 to 14 months) [193]. The shedding of SSuT resistant E. coli compared to susceptible strains was significantly higher in neonatal calves, whereas no difference was observed for heifers. In neonatal calves, the SSuT resistant E. coli out-competed susceptible E. coli, whereas in heifers no competitive advantage in the resistant isolates was observed [193]. These findings suggest that the SSuT resistant E. coli have a competitive advantage in the calf gut, which may be partially related to their diet as the prevalence of SSuT resistant E. coli was highest when the calves were fed a milk diet [193].

A higher prevalence of AmpC- and/or ESBL-E has been identified in calves compared to adult dairy cattle [10, 194, 195]. A study analysing recto-anal mucosal swabs from waste milk fed dairy calves (n=40) in NZ did not identify any ESBL-producing *E. coli*, however 25% of enrichments (10 of 40) were putative AmpC hyperproducing *E. coli* [27]. Additionally, an NZ study between 2009 - 2010 found no ESBL-producing *E. coli* (0 of 300) from bobby calf carcasses (4 to 10 days old) [196]. However, the detection of antimicrobial resistant bacteria from veal calf faecal samples and bobby veal in published studies overseas (Table 2.3) suggest that carcass contamination and transfer to the food chain may represent a public health concern if good hygienic slaughter practices are not adhered to. This food safety concern may be enhanced if non-optimal transport, cooking or storage conditions occurs.

Calf faeces may also contain a number of ARGs. A shotgun metagenomic study comparing the faecal resistome in preweaned calves compared to lactating dairy cows across 17 commercial farms in the USA found that the abundance of ARGs was significantly higher in preweaned calves (0.43 - 2.9 ARGs per copy of 16S rRNA gene) compared to lactating dairy cattle (0.11 - 0.6 ARGs per copy of 16S rRNA gene) [13]. Antimicrobial resistant bacteria shed in calf faeces may contaminate the environment, especially in calf pens, and this may be a potential transmission route for antimicrobial resistant bacteria and ARGs between calves and within the farm environment. A study utilising shotgun metagenomic sequencing found that the faecal resistome in young dairy calves significantly changed over time and the total abundance of ARGs decreased significantly with increasing age [11]. The researchers found that changes in the diet and the age of calves contributed to the gut microbiome composition, with the faecal microbiota diversity increasing over time as the calf diets became more diverse, which subsequently influenced the faecal resistome [11]. To determine whether colostrum is a source of ARGs to young calves, the resistome between calf faeces and colostrum was compared. The faecal resistome structure of early-life calves and colostrum [11]. Therefore early-life colostrum feeding may provide a mechanism of transmission where ARGs are seeded in the developing calf gut.

Ref.	[25]	[26]	[27]	8	
Study population	Regional (Manawatū) cross- sectional study of 15 randomly selected dairy farms.	Nationwide cross-sectional study of 26 dairy farms.	Four farms from the Canterbury region which used waste milk were randomly selected from a larger cross-sectional study of dairy farms (n=102) across six regions in NZ [197].	Randomly selected conventional dairy herds (n=196).	
Prevalence <sup>c</sup>	0.9% samples positive for ESBL- producing <i>E. coli</i> (1 of 116) and 7.9% (9 of 114 samples; samples from three farms) were positive for putative chromosomal AmpC hy- perproducers.	14% of pooled faecal enrichments (11 of 78) from seven farms were positive for putative chro- mosomal AmpC hyperproducers. No ESBL- or plasmid-mediated AmpC-producing $E. coli$ were de- tected.	25% of calves (10 of 40) were pos- itive for AmpC-producing $E. \ coli$ (putative AmpC hyperproducers).	59.6% (109 of 183) of dairy herds had at least one sample positive for AmpC and/or ESBL-producing $E.$ coli	
Detection method <sup>b</sup>	Enriched in BPW and plated on MC agar, MC + CTX, MC + CAZ and CHROMagar <sup>TM</sup> ESBL.	Enriched in BPW and plated on MC agar, MC + CTX, MC + CAZ and CHROMagar <sup>TM</sup> ESBL.	Swabs enriched with modified tryptone soy broth (mTSB) and enrichments plated on MC agar, MC + CTX, MC + CAZ and CHROMagar <sup>TM</sup> ESBL.	The faeces were plated on MC agar No. 3 with 1mg/L CTX and Luria-Bertani (LB) broth with 1mg/L CTX.	
Country	ZN	ZN	ZN	Netherlands	
Year <sup>a</sup>	Spring 2016 autumn 201	2017	2014	2013	
Sample type	Pooled paddock faecal samples	Pooled paddock faecal samples	Recto-anal mucosal swabs from waste milk fed dairy calves	Faecal samples from calves, young stock, dairy cattle and pooled faecal pat samples from the floor where dairy cows were housed.	

Ref.	[195]	[194]	[6]	[7]	[192]	
Study population	A convenience sample from 53 dairy farms. Monthly sampling over 12 months.	Cross-sectional study on 101 randomly selected dairy farms.	Cross-sectional study of dairy cattle $(n=82)$ , beef cattle $(n=104)$ and dairy sheep $(n=114)$ herds.	Cross-sectional study of 100 randomly selected conventional dairy farms.	Cross sectional study of 90 ran- domly selected dairy herds.	
Prevalence <sup>c</sup>	<ul> <li>9.3% of samples (384 of 4,145) representing 88.7% (47 of 53) dairy farms were positive for <i>E. coli</i> growth on agar containing CTX. The sample level prevalence was 5.4% (224 of 4,145) for bla<sub>CTX-M</sub> positive <i>E. coli</i>.</li> </ul>	85% of farms (86 of 101) had at least one sample positive for AmpC-/ESBL-producing $E$ . $coli$ . The likelihood of isolating AmpC- /ESBL-producing $E$ . $coli$ was higher from calves compared to adult dairy cows or the manure pit.	9.6% (10 of 104) of beef cattle, 32.9% (27 of 82) of dairy cattle and 7.0% (8 of 114) of dairy sheep were positive for third-generation cephalosporin resistant $E. coli.$	41% (41/100) herds positive for ESBL-producing $E. \ coli$ .	13% (12/90) of slurry samples from 90 herds were positive for ESBL- and AmpC-producing $E.$ coli.	
Detection method <sup>b</sup>	Faeces were plated on tryptone bile X-glucuronide agar (TBX) supplemented with 16 mg/L cephalexin. E. coli from these plates were sub-cultured on TBX agar containing 2 mg/L CTX.	Samples enriched in BPW and plated on MC + CTX.	Faeces enriched in mTSB with novobiocine, followed by enrich- ment in MC broth $+$ CTX and plated on MC $+$ CTX.	Enriched in LB broth with 1 mg/L CTX and plated on MC + CTX.	Faecal samples streaked on MC + CTX and the swab also in- oculated in LB broth supple- mented with 1 mg/L CTX.	
Country	England	Canada	Spain	Netherlands	Netherlands	
$\operatorname{Year}^{\mathrm{a}}$	2017-18	2017	2014-16	2011-12	2011	
Sample type	Faecal contaminated environments and pastureland (adult dairy cows, dry cows, heifers, calves and samples where no animals were present) and publicly accessible farmland (footpaths).	Pooled faecal samples from calves, dairy cows and the manure pit.	Rectal faecal samples	Pooled slurry samples	Slurry samples from organic farms	

Ref.	[198]	[6]	[199]	[200]	[201]	[202]	
Study population	One CTX-M-15 positive dairy farm was sampled over two years.	Cross-sectional study of 11 dairy farms. Farms were selected ac- cording to herd size criteria and convenience sampling.	Randomly selected 45 dairy farms and 10 control farms. The control group was farms which reported no antimicrobial usage in the previous six months before the study commenced.	Herds and samples included in the study were farms which sent diarrhoea samples from cattle and calves to a clinical diagnos- tic laboratory.	Four dairy farms sampled.	182 dairy farms sampled, investi- gating prevalence in healthy cat- tle, calves with enteric disease and healthy cattle linked to the sick calves.	
Prevalence <sup>c</sup>	Within herd proportion of CTX-M $E. coli was 17\% (219/1292)$ , with a higher proportion of milking cows positive (30.3\%, 200/660) compared to non-milking cows (3.0\%, 19/632).	54.5% (6/11) of dairy farms were positive for ESBL-producing <i>Es-</i> <i>cherichia</i> spp.	86.7% (39/45) farms were positive for ESBL-producing <i>E. coli</i> .	5.2% (20/381) of herds were positive for CTX-M-type ESBL-E.	46.6% (98/210) of samples were positive for ESBL-producing <i>E.</i> <i>coli</i> .	CTX-M type ESBL-producing $E.$ coli were detected in 5% (13/271) of faecal swabs.	
Detection method <sup>b</sup>	Enriched in BPW and plated on Chromagar CTX.	Enriched in BPW and plated on Brilliance <sup>TM</sup> ESBL agar.	Enriched in LB broth and plated on MC + CTX.	Faeces streaked on BTB-lactose agar supplemented with 2 mg/L CTX and 8 mg/L vancomycin.	Enriched in BPW and plated on $Brilliance^{TM}$ ESBL agar.	Faeces plated on ESBL screen- ing agar plates containing Dri- galsky medium supplemented with either 4 mg/L CTX or 4 mg/L CAZ.	
Country	United Kingdom	Germany	Germany	Japan	Egypt	France	
$\operatorname{Year}^{\mathrm{a}}$	2008-09	2012	2011-12	2007, 2009	2014	2009, 2010	
Sample type	Floor faecal samples	Pooled faecal samples from dairy and beef cattle	Pooled faecal samples from dairy and beef cattle	Individual faecal sam- ples	Rectal faecal swabs	Faecal swabs	

Ref.	[196]	[203]	[182]	[183]	[204]
Study population	Samples were collected as part of the National Microbiological Database programme, in which samples from processing plants were collected for a one month period in 2009 and a three month period in 2010.	Faecal samples (n=491) were collected from 12 slaughter houses over a six month period.	Ten calf faecal swabs were taken post-slaughter from 100 slaugh- ter batches (from five slaughter houses), which originated from different herds.	Samples were stored from a lon- gitudinal study (1997-2010) from randomly selected farms which were participating in a study of foodborne zoonotic pathogens.	Three veal-calf farms were stud- ied over a ten week period in 2009. Farms were selected ac- cording to criteria set by the au- thors.
Prevalence <sup>c</sup>	Prevalence of ESBL-producing $E$ . <i>coli</i> was 0% (0/300) from isolates obtained from swab rinsates of young calf carcasses.	Prevalence of ESBL-E was $29.4\%$ (144/489) in veal faecal samples.	66% of herds (66/100) were positive for CTX resistant $E. \ coli.$	The prevalence of CTX resistant $E.\ coli$ showed an increasing trend in the study population, ranging from 6% (3/49) in 1997 to 39% (71/182) in 2010.	When the calves arrived at the farms, the prevalence of calves positive for ESBL- and AmpC-producing $E$ . $coli$ was 18-26%. However, this prevalence decreased during the study period.
Detection method <sup>b</sup>	ASTs were carried out on <i>E. coli</i> isolated from swab rinsates on petrifilms.	Faeces diluted in PBS and plated on MC agar and Chro- mID ESBL agar.	Samples enriched in LB broth with 1 mg/L CTX and plated on MC + CTX.	Samples enriched in LB broth with 1 mg/L CTX and plated on MC + CTX.	Plated on MC + CTX and in LB broth supplemented with 1 mg/L CTX.
Country	ZN	France	Netherlands	Netherlands	Netherlands
$\operatorname{Year}^{\mathrm{a}}$	2009-10	2012	2011	1997 - 2010	2009
Sample type	Swab rinsates from very young calf car- casses.	Faecal samples from veal calves.	Faecal swabs from veal calves.	Pooled faecal samples from veal calves.	Rectal faecal swabs from veal calves.

Ref.	[184]	[10]	[189]	[190]	[205]
Study population	A prospective cohort study fol- lowing four veal calf cohorts on separate farms.	One dairy farm (it is unclear how the farm was selected). Fae- cal samples were collected from calves $(n=13)$ 1-63 days old pe- riodically and their respective mothers $(n=13)$ prior to calving.	Conventional dairy farms (n=18). The calves were 3 days - 2 months old.	Cross-sectional study on dairy $(n=52)$ and beef $(n=60)$ cattle production units.	Calves on dairy farms (n=8). Farms were recruited by conve- nience sampling.
Prevalence <sup>c</sup>	<i>E. coli</i> resistant to CTX and cefepime were: in farm faecal samples $91\%$ ( $127/140$ ) and $24\%$ ( $33/140$ ); pre-evisceration swabs $34\%$ ( $55/164$ ) and $6.7\%$ ( $11/164$ ) and final carcass swabs $19\%$ ( $23/122$ ) and $0.8\%$ ( $1/122$ ), respectively.	No ESBL-producing $E. coli$ were identified and 94% (n=87) CTX resistant $E. coli$ were AmpC pos- itive. CTX resistant $E. coli$ were isolated from all calves (n=13) and only two dairy cows (15.3%, 2 of 13).	11.1% of samples (20 of 180) were positive for $\operatorname{AmpC/ESBL}$ -producing $E. \operatorname{coli}$ .	At least one sample was positive for CTX resistant $E$ . coli from 70% beef cattle units (42 of 60) com- pared to 85% of dairy cattle pro- duction units (44 of 52).	81.2% of calves (396 of 488) had <i>E. coli</i> with reduced sus- ceptibility to extended-spectrum cephalosporins.
Detection method <sup>b</sup>	Enrichment in MC broth with 2 mg/L CTX and subsequently plated on MC agar supple- mented with either 8 mg/L ce- foxitin or 4 mg/L cefepime.	Selective media: MC agar with 2 mg/L CTX	Faecal swab was mixed with saline and plated on MC + CTX.	LB broth suspension prepared for faeces and boot swabs and dust swabs mixed with PBS. The samples were inoculated in LB broth overnight. Enriched samples were plated on MC + CTX.	Swabs enriched in BPW fol- lowed by <i>E. coli</i> broth which was subsequently plated on tryptic soy agar with 5% sheep blood containing vancomycin (6 mg/L), CAZ (2 mg/L), ampho- tericin B (2 mg/L), and clin- damycin (1 mg/L).
Country	USA	Czech Republic	Latvia	Germany	Canada
Year <sup>a</sup>	Unknown	2014	2016	2011-12	2014-15
Sample type	A combination of both farm and har- vest faecal samples, pre-evisceration swabs and carcass swabs.	Faecal samples from calves and their re- spective mothers.	Rectal swabs from calves	Pooled faecal sam- ples (n=3) from sta- ble floor, dust sample and boot swabs from the feed alley.	Rectal swabs

Sample type	$\mathrm{Year}^{\mathrm{a}}$	Country	Detection method <sup>b</sup>	Prevalence <sup>c</sup>	Study population	Ref.
Faecal samples	2018-19	Germany	Faecal samples were mixed with NaCl-Trytpone broth, diluted and plated on <i>Brilliance</i> <sup>TM</sup> ESBL agar.	96.5% of calves (193 of 200) were positive for ESBL-E.	Ten dairy farms visited twice during the study. Calves were less than two weeks old.	[181]
Farm slurry	2014	UK, Scot- land and Wales	Slurry enriched in BPW and plated on CHROMagar ESBL and CHROMagar CTX.	28% (27 of 97) slurry samples were positive for presumptive ESBL- producing <i>E. coli</i> .	Slurry samples collected from dairy farms after milking and be- fore cleaning and were collected from five sites on each farm.	[206]
Untreated farm waste and effluent samples	2019-20	Ireland	Samples were filtered and the filter enriched in BPW. Enrichments were plated on <i>Brilliance</i> <sup>TM</sup> ESBL agar.	Of the antibiotic resistant bacteria isolated, 36% on dairy and 4.5% on dairy and poultry farms were ES- BLs. Across the different farms, a higher proportion of antibiotic re- sistant bacteria were isolated from influent compared to effluent sam- ples.	Monthly sampling over 12 months of effluent samples collected before (influent) and after treatment (Integrated Constructed Wetland) from four different farm production systems (suckler, dairy farm, dairy and poultry and pig).	[188]
Faecal samples	2016	China	Faeces enriched in LB broth and plated on ChromID ESBL agar plates.	43.6% cows (284 of 651) were positive for ESBL-producing <i>E. coli</i> .	Ten randomly selected dairy farms.	[126]
<ul> <li><sup>a</sup> The year(s) the sample</li> <li><sup>b</sup> BPW, Buffered peptor</li> <li>+ 1 mg/L ceftazidime. T</li> <li><sup>c</sup> ESBL, Extended-spect</li> </ul>	es were collect 1e water. MC SB, Tryptic ξ rum β-lactam	ted. ', MacConkey ≀ Soy Broth. TB ıase. ESBL-E,	адаг. MC + CTX, MacConkey ag. X, Tryptone Bile X-Glucuronide n ESBL-producing <i>Enterobacteriace</i>	çar supplemented with 1 mg/L cefota: medium. LB, Luria-Bertani Broth. PI :ae.	xime. MC + CAZ, MacConkey aga 3S, Phosphate Buffered Saline.	5ar

## 2.6.2 Prevalence of AmpC- and ESBL-E in milk

The prevalence of ESBL-E in bulk tank milk and milk from mastitic cows has not been investigated in NZ, but varies widely between studies undertaken overseas (Table 2.4). The prevalence in milk from mastitic cows was low (1.5%) in a study in Switzerland [207] and high in milk samples (25.53%) from Chinese dairy farms [55]. The prevalence of ESBLproducing E. coli from milk samples from individual healthy lactating cows in Malaysia was also comparatively high (16.9%) [208] and this may be due to environmental contamination on the udder prior to milk collection. In addition, the milk samples analysed were handmilked from individual lactating cows [208], whereas bulk tank milk is often used in other studies. The prevalence of ESBL-E in bulk tank milk was relatively low (0-22.6%; Table 2.4), and although pasteurisation of milk using established heat-treatment processes will inactivate Enterobacteriaceae [209] these data indicate that raw milk consumption may pose a public health risk. Ceftazidime resistant E. coli were also isolated from retail milk samples in the USA, however some of these products for sale were unpasteurised [210]. Another study identified a higher proportion of *E. coli* from bulk tank milk samples (34.4%) in Northern China, some of which carried  $\beta$ -lactamase genes (Table 2.4) [211], however, culture-based selection methods for AmpC- and ESBL-E were not utilised in this study and therefore the prevalence in these samples may be under-reported.

Ref.	[55]	[207]	[212]	[209]	[213]	[208]	[191]
Study population	A total of 1,252 mastitis milk samples were collected when animals were ill from 61 com- mercial dairy herds.	Bulk tank milk from a dairy manufacturing plant repre- senting 100 dairy farms was collected and 67 <i>E. coli</i> iso- lates from mastitis milk were analysed.	40 randomly selected dairy farms were sampled, includ- ing 35 conventional and 5 or- ganic farms	Bulk tank milk samples were obtained from a pre- established and mandatory milk sampling program.	80 randomly selected dairy farms.	Ten dairy farms.	The samples were collected from 62 bulk tanks across 1,252 farms.
Prevalence	Of 153 <i>E. coli</i> isolates from 1,252 mastitis milk samples, 23.53% (36/153) were ESBL- producers.	Zero out of 100 bulk tank milk samples were positive for ESBL-E and $1.5\%$ $(1/67)$ $E.$ <i>coli</i> isolates from mastitis milk displayed an ESBL phenotype.	An ESBL phenotype was displayed in $0.7\%$ (2/270) of <i>E. coli</i> isolates.	9.5% (82/866) of bulk tank milk samples on farms were positive for ESBL-E.	8.75% (7/80) of bulk milk tank samples were positive for ESBL-producing K. pneumo- niae.	16.9% (12 of 71) of milk samples from individual healthy cows were positive for ESBL-producing $E. \ coli.$	22.6% of milk samples were positive for ESBL-producing $E. \ coli$
Detection method <sup>b</sup>	Direct plating (10 $\mu$ L) onto MC + CTX.	Enriched in <i>Enterobac-</i> <i>teriaceae</i> enrichment broth and plated on <i>Brilliance</i> <sup>TM</sup> ESBL agar.	Enrichment in BPW and plated on MC agar supple- mented with 2 mg/L CTX and TBX agar.	Enriched with 1 mg/L CTX and plated on MC + CTX.	Enriched with 1 mg/L CTX and plated on MC + CTX.	Selective isolation using Chromocult <sup>®</sup> Coliform Agar and subsequently CHROMagar <sup>TM</sup> ESBL.	Enriched in BPW and plated on ChromID ESBL plates.
Country	China	Switzerland	Czech Republic	Germany	Indonesia	Malaysia	Turkey
Year <sup>a</sup>	Jan. 2015 - May 2016	Oct. 2009, Nov. 2010 to March 2011	2010-13	Nov/Dec. 2011, Aug. 2012	2011	NA	2016
Sample type	Milk from cows with mastitis	Bulk tank milk and milk from cows with mastitis	Bulk tank milk	Bulk tank milk	Bulk tank milk	Milk samples were hand-milked from an individual lactating cow.	Bulk tank milk

Table 2.4: Prevalence of AmpC- and ESBL-producing *Enterobacteriaceae* in milk

Sample type	Year <sup>a</sup>	Country	Detection method <sup>a</sup>	Prevalence	Study population	Ref.
etail milk samples: npasteurised milk, asteurised milk via at pasteurisation, igh-temperature- nort-time pas- urisation, and ltra-pasteurisation ilk.	2017	USA	Presumptive <i>E. coli</i> iso- lated during bacterial quantification of samples and ASTs completed for these isolates.	60% of <i>E. coli</i> displayed a CAZ resistant phenotype (57 of 95).	2,034 retail milk samples purchased.	[210]
ulk tank milk	2015	China	Enriched in tryptic soy broth and plated on Eosin Methylene Blue agar plates.	34.4% (67 of 195) of milk samples were positive for $E$ . coli and $\beta$ -lactamase genes were identified in 34.3% of $E$ . coli, specifically the bla <sub>SHV</sub> (1.5%), bla <sub>TEM</sub> (20.9%), bla <sub>CUYY</sub> (10.4%), bla <sub>CUYX</sub> (1.5%) genes.	195 dairy farms in Northern China	[211]
The year(s) the sample BPW, Buffered pepton	ss were collected. N 1e water. MC, Mac	IA, Not available. :Conkey agar. MC	+ CTX, MacConkey agar sup	${ m pplemented}$ with 1 mg/L cefotaxir	me. MC + CAZ, MacConkey a	gar

1 mg/L cettazidime. +

# 2.7 Shotgun metagenomics for the detection of ARGs

Culture-independent metagenomic methods targeting total DNA from an environmental sample provides a holistic approach for the detection of antimicrobial resistant bacteria and their genetic determinants. Culture-independent methods include quantitative real-time PCR for targeted amplification of ARGs and shotgun metagenomics involving sequencing of the collective genetic material in an environment [214]. Specific ARG databases such as the Comprehensive Antibiotic Resistance Database (CARD) [215] and ResFinder [216] are available to assign reads to DNA sequences to infer resistance. Databases such as PointFinder [217] can be used to identify chromosomal point mutations associated with AMR, however due to sequencing errors, ARGs identified in metagenomic sequencing data due to point mutations should be further validated. Metagenomic methods allow for community analysis to determine the abundance and diversity of microbes present in a sample, in contrast to culture-based methods which are often limited to specific bacteria according to the isolation methods used.

Culture-independent methods may be limited by the presence of low read numbers of ARGs sequences [218] and databases may only classify ARG to certain phyla, rather than to the species level [218], however shotgun metagenomics provides improved taxonomical classification at the genus level [219]. The classification method and reference database used can influence the taxonomic classification of short-read shotgun metagenomic sequencing data. especially at the species level. For example, the number of faecal samples classified as positive for S. enterica from feedlot cattle was discordant between culture and PCR methods compared to metagenomic sequencing and it was found that plasmids which were associated with S. enterica genomes in the reference database were influencing the taxonomic results. One limitation of culture-independent methods for resistome analysis is that ARG detection relies on gene homology and often does not consider flanking sequences important for gene function such as promoters [220], and this may overestimate the burden of AMR. In addition, culture-independent methods detect DNA and therefore it is not known whether a specific ARG originated from live or dead bacteria. Lastly, even though a gene is known to confer resistance to antimicrobials in one organism, it may not be functional in another [214]. Thus, the complementary use of both culture-based and metagenomic methods in parallel may provide a more thorough AMR surveillance approach.

Several factors require consideration for ensuring the accuracy of the sequencing data compared to the true microbial population composition during metagenomic studies. Such factors include sample storage, DNA extraction, read length, sequencing depth and bioinformatic analysis [30, 221, 222]. Consistent sample storage conditions within a study are crucial to minimise changes to the microbial population prior to DNA extraction. Various storage methods may be used, however it is recommended that faecal samples should not be stored for more than 3 hours at room temperature prior to low-temperature storage [223]. Storage of soil, human skin and faeces at -80°C prior to DNA extraction has been shown to have a minimal effect on the microbiota composition [223, 224], however in one of these studies, a relatively low sequencing depth per sample was used [224]. In comparison, storage temperature and time has been shown to affect bacterial community composition in soil samples [225]. Similarly, the repeated freeze-thawing of samples should also be avoided to prevent DNA degradation [223].

Across all sample matrices, the same DNA extraction kit should be used due to possible kit reagent contamination [226] and the effect kits may have on the downstream sequencing results [227, 228]. Numerous studies have compared the efficacy of DNA extraction kits across specific sample matrices [227, 229, 230, 228, 231, 232, 233], however no study has compared all matrices used in this study. To mitigate the effect of contamination introduced via DNA extraction kits, blank extraction controls should be concurrently sequenced which is especially important for low biomass samples [234, 235, 226]. In addition, the use of mock communities (samples with a known composition) provides a good control to standardise analyses and/or determine any biases [221].

Sequencing depth is another crucial step when designing a shotgun metagenomic study. The effect of sequencing depth on the characterisation of the microbiome and resistome has been investigated in composite faecal samples from beef feedlot cattle [236]. The proportion of read assignments to phyla and antimicrobial classes was relatively unaffected by sequencing depth, however a significant increase in the number of sequencing reads assigned to ARGs and microbial taxa was correlated with increasing sequencing depth [236]. A sequencing depth of 59 million sequence reads per sample was suggested as suitable to accurately describe the cattle faecal microbiome and resistome [236]. Subsampling of deep shotgun metagenomic sequencing data from the Human Microbiome Project [237] across multiple sample matrices found that a depth of 0.5 million sequences was sufficient to identify the same trends in species-level  $\alpha$ - and  $\beta$ -diversity as was observed in the full dataset [238]. However, the Human Microbiome Project did not compare the detection of ARGs, which would likely be present at a lower proportion in the samples. A study of human effluent and pig caeca samples found that the number of AMR gene families stabilised at 80 million reads per sample [239]. However, the human effluent sample had the most diverse range of ARGs and the detection of ARG allelic variants had not stabilised at a sequencing depth of 200 million reads per sample [239]. In summary, the required sequencing depth depends on the study objectives.

The proportion of contaminating host DNA (e.g. bovine DNA from a milk sample) versus microbial DNA can also affect the desired sequencing depth. For example, if a large proportion of sequencing read pairs are host DNA and thus removed, the total sequencing depth is reduced. Treatment of DNA with propidium monoazide (PMA) has been suggested as an effective method to remove host DNA from human saliva samples [240], and PMA treatment has been used on cattle effluent samples for subsequent quantitative PCR and 16S rRNA sequencing [241]. A recent study found that pre-extraction methods were the most effective for host depletion of *Bos taurus* reads from bulk tank milk and in-line milk filter homogenate samples, compared to post-extraction methods [242]. Regardless of sequencing depth and treatment, the bulk tank milk samples (82.5 - 99.4%) had a higher proportion of host contamination compared to the milk filter homogenate samples (24.7 - 91.5%) and the authors therefore suggested that milk filters may be a suitable sample matrix to study as a proxy for bulk tank milk [242]. The detection of ARGs in milk filters may infer the presence in bulk tank milk, however this can not be confirmed. Reducing the number of contaminating host reads resulted in a higher detection of microbial taxa and ARGs [242]. Due to the potentially high level of host contamination in bulk tank milk samples, a high sequencing depth is required to detect ARGs which are hypothesised to be present in a low abundance in these samples [242]. A low percentage (average <1%) of contaminating *Bos taurus* DNA has been identified in cattle faeces [243], effluent and soil [244] compared to a higher proportion in bulk tank milk or milk filter homogenate samples (24.7 - 99.4%) [242]. Free DNA may also be encountered in soil samples [245].

Metagenomic studies assessing AMR often focus on ARG abundance, however there is growing evidence to suggest that the risk associated with ARGs in the resistome should also be considered [53]. Key considerations in an ARG risk ranking framework include (i) whether the ARG is part of the intrinsic resistome, (ii) fitness cost of the resistance gene and (iii) the environment in which the ARG was detected; for example, high risk ARGs found in humans poses a higher risk than if the gene was detected in a remote environment [53]. ARGs classified as the highest risk are those which can be identified on mobile genetic elements, are known to contribute to the failure of antibiotic treatment and have previously been found in human bacterial pathogens [53]. Therefore, to estimate the burden of AMR in an environment or ecosystem, metagenomic studies should evaluate not only the abundance of ARGs but they should also assess the risk each ARG poses. A similar risk framework approach has been applied to assess ARGs from human gut microbiome genomes and only a small proportion of ARGs were ranked as the highest risk group in the study (3.6%) [246]. A limitation of the current proposed ARG risk frameworks is that they are often human centric [53, 247, 246] with the risk described in relation to human health. Future modifications to risk frameworks should take a "One Health" approach and include risk assessments which collectively consider human, animal and environmental health.

## 2.8 AMU on dairy farms

There has been a global movement to reduce AMU in livestock animals to prevent the development and transmission of AMR, to improve AMR surveillance and to encourage prudent AMU and develop and adopt antimicrobial alternatives [248, 249, 250]. For example, to encourage both prudent use and the reduction of antimicrobials in animal husbandry, the NZVA has aspired that by 2030 NZ will not require antimicrobials for the maintenance of animal health and wellness [42] but rather reserve their use for the treatment of disease. A sample of dairy herds in NZ (n=477 herds) demonstrated that the antibiotic use rate was

generally low compared to international standards, but herd-to-herd variation occurred, with some high-use herds identified [43]. The ubiquity of a largely pasture-based system may be a contributing factor to the low antimicrobial use in NZ dairy cattle. A phone survey of 1,000 randomly selected adults in the USA found that 90.7% of participants (892 of 983) were concerned that AMU on dairy farms posed a threat to human health and 71.5% (580 of 811) of consumers were prepared to pay more for milk produced on a farm without antibiotics [15]. The participant responses were associated with demographic factors and worryingly only 51.5% of survey respondents were aware that antibiotics were used to treat bacteria only [15]. This suggests that increased education for the general public regarding prudent antimicrobial stewardship and the development of AMR is required.

Dairy farm management practices involving AMU to maintain animal welfare and reduce the prevalence of foodborne pathogens, and antimicrobial resistant bacteria, are interrelated [251]. Regulations for AMU in dairy cattle vary considerably, with some countries (e.g. USA and China) permitting the use of antimicrobials for growth promotion in addition to prophylactic use for the treatment, control and prevention of animal disease without a veterinary prescription [252]. In comparison the use of antimicrobials for growth promotion is not permitted in NZ [41] and European Union (EU) countries, and their prophylactic use is only permitted in animals with a veterinary prescription [252]. There is currently a lack of reliable surveillance data on AMU in livestock globally [253] and in many developing countries AMU is unregulated in food-producing animals [252].

AMU also differs between conventional and organic farm management systems. In contrast to conventional dairy farms, organic farms must adhere to AMU regulations aimed at improved environmental based management practices. For example antimicrobial therapy is banned in dairy cattle on USA organic farms; if an animal requires treatment they must be removed from the herd [251, 254]. In the EU however, organic dairy farms are permitted to use antimicrobials for disease treatment with an increased withholding milk period compared to conventional farms [251]. In NZ the regulations are designed to meet export market requirements, therefore organic dairy farms are banned from using antimicrobials (with some specified exceptions) to enable organic milk products to be exported to countries including the USA and the EU members [255, 256].

The range of antimicrobials approved for veterinary use differs between countries and there is some overlap between those approved for use in both human and animal health. In dairy cattle, antimicrobials are used to treat diseases such as mastitis, endometritis, neonatal diarrhoea and for dry cow therapy (DCT) [21]. Treatment with DCT (a longlasting intra-mammary antibiotic preparation) at the end of lactation is intended to treat both existing mastitis infections and prevent new cases during the dry period. When DCT is applied to the whole herd, it is a farm management practice which may contribute to the development of antimicrobial resistant bacteria [257]. In NZ, antimicrobial DCT was not economically viable until the introduction of somatic cell count (SCC) penalties (as SCC is an indicator for intra-mammary infections) in 1993 [258, 259]. According to antibiotic sales data from the dairying Waikato region of NZ (n=477 herds), 86% of the total antibiotics used in June 2009 - 2012 (measured as daily doses/animal/year) were for DCT and mastitis treatment [21]. Analysis of antimicrobial sales data for 2018 in NZ found that almost half of the antibiotics sold (47%) were registered for use in pigs and poultry, compared to 14% registered for use in cattle [20]. However, some products are registered for multiple species (production and companion animals) and multiple production animals, therefore the use of antibiotics in these categories cannot be distinguished using sales data [20]. In 2018, antibiotics for DCT represented 9.5% of overall sales and this represented a 16% decrease in antibiotic mass sold for DCT compared to 2017 [20]. The sale of critically important antibiotics such as third- and fourth-generation cephalosporins (23% and 27%, respectively), macrolides (15%) and fluoroquinolones (4%) decreased in 2018 [20]. Colistin, which belongs to the polymyxin class and is a critically important antibiotic, is not registered for use in NZ dairy cattle [20]. The reported antimicrobial use on 26 NZ dairy farms was 4.39 - 20.92 mg/population correction unit (PCU, which accounts for the animal population and estimated animal weight at the time of treatment) according to sales data, and  $\beta$ -lactams were the most frequently used antibiotic group [26]. A study of dairy farms across four regions in NZ also found that penicillins were the most commonly used antibiotic [260]. A convenience sample of dairy farms (n=358) in Great Britain found that the AMU ranged from 0.36 - 97.79 mg/PCU (mean 20.62 mg/PCU) according to antimicrobial sales data, and some high use farms were identified [261].

Caution should be used when comparing AMU between studies and countries as the methods used to assess AMU may differ [261]. This is of particular concern when comparing mg/PCU between countries as the population structure (e.g. animal weight) and recommended dosing regimes can differ [262]. It is essential to also consider the use of specific antimicrobials, as some critically important antimicrobials such as the third- and fourthgeneration cephalosporins and fluoroquinolones are used at low doses [262, 261], therefore the use of these antimicrobials (compared to other antibiotics recommended at a higher dosage) would reduce the total amount of antimicrobials used but would increase the use of critically important antimicrobials. This could lead to a misinterpretation of AMU on farm. AMU calculated based on farm records (e.g. individual animal treatments) may potentially under-report the actual usage due to reporting bias or errors in updating records. Using antibiotic sales data as a proxy for true antimicrobial use can also be unreliable as farmers may receive bulk antibiotic prescriptions during a season, therefore it is possible the total antibiotics sold are not always used in one season. Standardised reporting protocols and metrics are essential for national AMU monitoring and comparison between farms and countries [261]. Further research is required to gather more accurate AMU data across dairy farms, including detailed information regarding treatments, frequency and time of use during the farming season. This detailed AMU data would help determine the baseline levels of AMU which could be used as targets to reduce AMU and it would be useful for programmes targeting improved antimicrobial stewardship. A systematic review analysed the association between antimicrobial restriction and the prevalence of ARGs and found that generally restricted use of antimicrobials in food animals was associated with a lower

or equal prevalence of ARGs in the study population, however the specific ARG, host species and antibiotic all influenced this effect [263]. For example, the prevalence of the  $bla_{\rm CTX-M}$  gene was lower in intervention groups, however no effects were observed for other  $\beta$ -lactamases [263]. These data suggest mitigation strategies aimed at preventing mastitis and targeted antimicrobial use for mastitis treatment may be the most effective measure to reduce AMU on NZ dairy farms.

The NZVA has set high aspirations to reduce the reliance on antimicrobials for the maintenance of animal health and wellness in NZ by 2030 [42]. Veterinarians play a vital role in promoting prudent antimicrobial stewardship on NZ dairy farms and are a trusted source of information for farmers. A survey of NZ farmers and veterinarians determined that AMU was influenced by veterinarian advice and personal experience, with farmers generally having limited knowledge of the development and spread of AMR [264]. A study of Swedish dairy farms (n=7) also found that advice from the local veterinarian was valued and important [265]. Increased awareness and education provided to dairy farmers by their local veterinarian regarding prudent antimicrobial stewardship and the development of AMR in agricultural environments will likely have a beneficial role on the use of critically important antimicrobials in NZ dairy farms. The identification of some farms with higher AMU [43, 261] suggests that involving them in AMU reduction strategies will help reduce AMU across the dairy farm production system.

In the dairy industry alternatives to antimicrobials for the treatment and prevention of mastitis have been recommended including good hygienic practice and the use of teat sealants for the prevention of new mastitis infections during the non-lactation period [266]. Whole herd application of teat sealants during the dry off period has been shown to have a protective effect against the development of ESBL-E positive herd-level status [7]. Other alternatives to antimicrobials for mastitis treatment and prevention [58] include the use of phage therapy and vaccines. These methods are more commonly utilised in the more intensive farming systems of the northern hemisphere where there are higher rates of mastitis. These alternatives have limitations due to the complexity and wide range of aetiological agents causing mastitis in dairy cattle.

# 2.9 Other farm management practices which may influence AMR on dairy farms

In addition to AMU, a number of other farm management practices are likely to impact the prevalence of foodborne pathogens, antimicrobial resistant bacteria and ARGs in agricultural environments [251]. Other farm management practices (Table 2.5) such as the use of biocides or disinfectants (e.g. quaternary ammonium compounds, triclosan) and the presence of heavy metals (e.g. copper, zinc) may influence the prevalence of AMR by co-selecting for ARGs. Co-selection for resistance genes can occur via co-resistance which occurs when the selection of one gene aids the maintenance of another resistance gene, or through cross-resistance, which occurs when one gene confers resistance across multiple toxic compounds such as antibiotics, biocides or heavy metals [267]. A study analysing publicly available bacterial genomes and plasmids identified heavy metal and biocide resistance genes in 86% of the genomes and 17% of these co-harboured ARGs [29]. Interestingly, plasmids hosted by *Escherichia, Staphylococcus, Salmonella* and *Klebsiella*, which are often associated with human and animal pathogens, were more likely to co-harbour antimicrobial, heavy metal and biocide resistance genes compared to other bacteria in the study [29]. Plasmids co-harbouring ARGs and heavy metal/biocide resistance genes were uncommon from external environments (<0.7%) but were more common in plasmids from either human (5%) or domestic animal (7%) origins [29], suggesting that drivers in human and animal populations (e.g. antibiotics) may be co-selecting for these resistance genes.

Other farm management factors, such as feed type and herd size may influence the prevalence of diseases such as clinical mastitis, as a higher prevalence of certain pathogens have been associated with these more intensive farming practices. Intensive farming practices (e.g. supplementary feed and indoor housing) within northern hemisphere farming systems have been associated with an increase in livestock pathogens including coliform mastitis which may result in increased AMU to prevent and treat such infections [251, 268]. Cows which were confined in a more contaminated area were observed to have a higher incidence of coliform mastitis compared to pasture-based cows [269]. Northern hemisphere dairy farming systems often employ indoor barns, such as tie-stall or freestall barns to house cattle during the winter, with limited access to pasture [251, 254, 270]. In comparison, NZ dairy farming systems employ a majority pasture-based system where cattle are routinely grazed on ryegrass/clover [18], however there are examples of NZ farms where off pasture systems and open feed-pads are becoming more common. Feed regimes may also influence teat infections and consequently the use of antimicrobials to treat such infections; cows fed a total mixed ration (silage and concentrates) diet had a higher incidence of clinical mastitis compared to all-pasture based diets [269]. In contrast, a metagenomic study of the rumen gut microbiome found no correlation between the prevalence of ARGs in feedlot steers fed antimicrobial feed additives compared to steers where antimicrobial additives were withheld [219]. Farms operating an open herd policy (bringing new animals into the farm) may also increase the possibility of animals carrying antimicrobial resistant bacteria entering the farm. However, the role and significance each management factor has on driving AMR in the dairy farm environment remains to be fully elucidated.

Climate change may also have an impact on AMR in agricultural environments. Climate change is likely to increase the frequency of extreme weather events, heavy rainfall, heat stress in animals, result in changes to annual pasture and crop yields and potentially increase the incidence of some disease such as mastitis, which would necessitate further AMU and the persistence of pathogenic and antimicrobial resistant bacteria in the dairy farm environment [271, 272, 273]. All of these impacts of climate change would negatively influence agricultural sectors and cause challenges to these production systems. Some

proposed mitigation strategies to limit climate change would also have a beneficial impact on AMR. For example, smaller dairy cattle herd sizes to reduce methane emissions (in conjunction with other approaches such as feed management and selective breeding to increase animal production) would result in less intensive farming practices [274], that have been associated with a lower odds of detecting specific antimicrobial resistant bacteria [275, 189].

	[[		5, 277]	2	), 280,
Re	[27(	[2]	[150	[278	[275 11]
tudy type	bn-farm cohort sudy.	bn-farm cross- ectional study.	n-farm longi- ıdinal study nd an <i>in vitro</i> nodel.	n-farm ran- omized control ial.	n-farm cross- ectional studies.
Bacterial group Studied	$E. \ coli$ and $Klebsiella$ spp. $\overrightarrow{\mathrm{O}}$ st	E. coli O	E. coli t1 m	Intra-mammary infections C caused by <i>S. aureus, Strep.</i> di <i>uberis, Corynebacterium</i> tr spp. and coagulase neg- ative staphylococci were reduced.	Culture-independent C methods (PCR) detecting st specific ARGs and shotgun metagenomic sequencing.
Importance	Herd-level associations between AMU and AMR in $E. \ coli$ were observed for certain antimicrobials, although no associations were observed for $Klebsiella$ spp.	The use of third- and fourth-generation cephalosporins is a risk factor for a dairy herd testing positive for ESBL and/or AmpC-producing $E. \ coli.$	Higher amoxicillin/clavulanate use on dairy farms has been associated with an increased risk of detection AmpC- producing $E$ . coli with mutations in the promoter region of the $ampC$ gene. In vitro studies also showed an association between amoxicillin use and AmpC-producing $E$ . coli arising from mutations in the promoter region of the $ampC$ gene.	Post-milking disinfection using a teat spray reduces both the incidence of new intra-mammary infections caused by specific bacterial pathogens and the somatic cell count in bulk tank milk.	A significant correlation between some heavy metals and ARGs has been observed in agricultural soil and manure. A correlation between low levels of heavy metals and ARGs from soil microbes has also been detected. Network inference mod- elling found a strong correlation between ARGs and heavy metal and biocide resistance genes from shotgun metagenomic sequencing data, suggesting co-selection of these resistance genes in dairy calves.
Country <sup>a</sup>	Canada	Netherlands	England, NA	ZN	China, Scotland, USA
Factor	Total AMU	Specific AMU (third- and fourth-generation cephalosporins)	Specific AMU (amox- icillin)	Post-milking teat dis- infectants	Heavy metals (e.g. zinc, copper, mer- cury)

Factor	Country <sup>a</sup>	Importance	Bacterial group studied	Study type	Ref.
Use of biocides and disinfectants	Spain, NA	A significant positive correlation between biocide tolerance and AMR to certain antimicrobials was observed in a study of bacteria isolated from seafood, suggesting that the use of biocides may co-select for AMR. A review of current evidence found a number of laboratory based studies suggested the use of biocides can co-select for AMR. Field studies (e.g. on farm) providing sufficient evidence for this correlation are currently lacking.	A number of bacterial genera were investigated including <i>Pseudomonas</i> , <i>Acinetobacter</i> , <i>Listeria</i> and <i>Shewanella</i> .	Laboratory analysis of bac- terial isolates from seafood products and a review study.	[281, 282]
Open herd policy	England/ Wales, Latvia, Great Britain	Operating an open herd policy such as buying in calves, bulls or fattening cattle has been identified as a risk factor for third-generation cephalosporin resistant $E. \ coli$ .	E. coli, Enterobacteri- aceae, Enterococcus spp.	On-farm cross- sectional studies.	[283, 189, 284]
Waste milk disposal	USA	A PCR-based study found a higher number of faecal $E$ . <i>coli</i> isolates from calves fed pasteurized waste milk were phenotypically resistant to some $\beta$ -lactams including the third-generation cephalosporin ceftiofur, however resistance to other antimicrobials including sulphonamides, tetracy- clines and aminoglycosides was high in calves fed both feed types.	PCR-based methods.	On-farm cohort study.	[285]
Herd size	USA, Latvia	Larger farming operations (>500 cattle) were associated with a 58% higher likelihood of detecting cefotaxime resistant bacteria from faecal samples in a cross-sectional of grazing beef cattle farms. A study of calves from dairy farms in Latvia found significantly higher levels of AMR to penicillins, cephalosporins and fluoroquinolones in <i>E. coli</i> and resistance to fluoroquinolones and macrolides in <i>Enterococcus</i> spp. in large farms (>201 milking cows).	E. coli and Enterococcus spp.	On-farm cross- sectional studies.	[275, 189]

# 2.10 "One Health" approach

The One Health High Level Expert Panel recently formed a definition of "One Health" as being "an integrated, unifying approach that aims to sustainably balance and optimize the health of people, animals and ecosystems" [287]. "One Health" often uses multi- and transdisciplinary methods. A "One Health" approach is required to tackle complex issues such as AMR, as humans, animals and ecosystems are intertwined and the development and dissemination of AMR within each sector is interdependent and is not distinct [99, 287]. Thus, selection pressures and the development of AMR in one sector may be reflected across other sectors [99]. Recent prevalence studies of ESBL-producing *E. coli* [206, 288] and AMR source attribution studies [289] have utilised a "One Health" approach and analysed *E. coli* from human, animal and retail food sources. However, these studies often have not included environmental samples nor samples from other vectors (e.g. birds, rodents, mammalian pests); the role of these sources in the development and dissemination of AMR remains largely unknown. Future studies focusing on the development and transmission of AMR should utilise a holistic "One Health" approach to fully understand the transmission pathways and reservoirs between humans, animals and the environment.

# 2.11 Transmission of antimicrobial resistant bacteria and ARGs in the dairy farm environment

Several potential pathways for the transmission of AMR within the dairy farm environment and humans have been identified (Figure 2.1) including contact with faeces-contaminated environments, via direct contact (for example, farmers may have a high risk during close contact with animals during milking or calving) or through the food chain from animalderived products including raw milk and veal [290]. However, the actual contribution and direction of each pathway is undetermined.

Within agricultural environments, the ruminant digestive tract has been suggested as a reservoir for the development and spread of AMR due to the diverse range of bacterial species and ARGs present in both the bovine and ovine ruminant gut [286, 291]. Antimicrobial resistant bacteria and resistance genes can be excreted in cattle faeces [292, 293] which can subsequently contaminate the dairy farm environment, such as faecal contamination of pasture/soil during the application of manure or dairy effluent as a fertilizer to pastures [294].

Evidence for AmpC and ESBL-E and ARG transfer from food-producing animals to humans remains equivocal. A German study assessing the risk of animal-human direct contact for the cross-transmission of ESBL-E identified two isolates, one from dairy cattle and the other from a farm worker, that shared identical multi-locus sequence types (MLST) and CTX-M alleles, potentially indicating zoonotic transfer [6]. Whereas a comparison of ESBL- and AmpC-producing E. coli isolated from animal-derived food products, animal gut microbiota and human clinical cases such as urinary tract infections, found little similarity between animal and human isolates at the level of MLST and virulence and AMR gene profiles [45]. Only 1.2% (3 of 258) of the animal isolates were >70% similar to the human isolates in relation to MLST clonal complex and microarray gene profiles [45]. Specific cases were observed however in which animal and human derived isolates were similar, with a UK cattle isolate and Dutch human isolate sharing >90% similarity in virulence and AMR microarray gene profiles, however such similarities only indicate genetic similarity and do not provide sufficient epidemiological evidence supporting transfer between humans and cattle [45]. A systematic review of the role of food-producing animals as a source of human infections caused by expanded-spectrum cephalosporin resistant extra-intestinal E. *coli* described studies both supporting and rejecting food-producing animals as a source of infection [295]. Where there are associations between resistance in food-producing animals and humans this appears to be sporadic and limited to specific strains. For example, WGS identified the recent emergence of a distinct lineage of third-generation cephalosporin resistant Salmonella enterica serovar Typhimurium in Australia, from both human infections and dairy cattle, suggesting transmission between dairy cattle and humans [296]. A large study comparing plasmid replicon types and ESBL/pAmpC genes combinations from farm animals, food and humans in Sweden found there was limited dissemination of ES-BL/pAmpC genes and plasmids carrying these genes between sectors [297], but a limited number of bovine isolates (n=9 from preweaned calves) were included in the study. Currently, there are no published studies investigating the transmission and zoonotic transfer of AmpC- and ESBL-E in NZ environments. The evidence for direct contact as a transmission route of antimicrobial resistant bacteria from cattle to humans is inconsistent partly due to the number of potential transmission pathways, the methodologies used to infer transmission as well as the difficulty proving the direction of transmission.

AmpC- and ESBL-E have been detected in numerous animal-derived food products overseas including in bulk tank milk from dairy farms [209, 213, 298], raw milk and cheese during the production process [298] and veal calves and carcasses [203, 182, 184] indicating their occurrence along the food chain. A recent study of veal calves on-farm and pre- and post-slaughter isolated antimicrobial resistant  $E. \ coli$  from both farm faecal and pre-evisceration samples sharing the same pulsed-field gel electrophoresis patterns and resistance phenotypes [184], highlighting a potential pathway for the transfer of antimicrobial resistant bacteria to the human food production chain.

Human-human spread is also an important route of transmission of AmpC- and ESBL-E [206, 299]. Using a source attribution model, it was identified that most community acquired carriage of plasmid-mediated AmpC- or ESBL-producing *E. coli* in the Netherlands were attributed to human to human transmission [299]. A recent study utilised a "One Health" approach to investigate source attribution of ESBL-producing *E. coli* isolated from human faecal samples and bloodstream infections, human sewage, food, dairy farm slurry and veterinary diagnostic samples from food-producing animals across England, Wales and Scotland [206]. Across human-associated samples (faeces, bloodstream infections and sewage) the predominant sequences types were ST131 followed by ST38 and ST648 with  $bla_{\text{CTX-M-15}}$  being the most common ESBL variant [206]. In the food products and veterinary isolates, the most common sequence types were ST602, ST23 and ST117 with  $bla_{\text{CTX-M-1}}$  being the most common ESBL variant. ST10 E. coli were isolated from both humans and animal-associated samples, however the ST10 isolates belonged to a diverse range of serotypes [206]. E. coli isolated from a cross-sectional study of livestock (cattle, pig and poultry farms) and retail meat (beef, chicken, pork, venison, veal and turkey) were compared with E. coli from human bloodstream infections in the UK [288]. WGS and core genome comparisons demonstrated that the livestock associated and human isolates were distinct (median 41,658 SNPs), however genetically similar E. coli (0-5 SNPs) were isolated from animals on different farms [288], perhaps suggesting animal to animal transmission. There was a small number of human and livestock associated isolates which were highly similar, with two human isolates being within 15 SNPs of two pig and one turkey isolate [288]. The predominant sequence types from the study in humans were ST73, ST131 and ST95 compared to ST10, ST117 and ST602 in livestock. These findings suggested that it was unlikely that livestock and retail meat are a major source of the E. coli causing bloodstream infections in the UK [288]. Recently, a study utilised shotgun metagenomic sequencing and Random Forests models for AMR source attribution in pigs, veal calves, broilers, turkey and humans occupationally exposed to livestock in Europe and identified some country-specific and reservoir-specific resistome markers which could potentially be used for source attribution studies [289]. However, additional sampling of the general human population, collecting human samples from more countries and including additional livestock species will strengthen these models and potential markers [289].

Numerous potential AMR transmission pathways have been identified, however the importance of each of these routes still remains unclear. Variation in the food production chains, farm management practices and policy regarding AMU in agricultural environments may influence the transmission pathways of antimicrobial resistant bacteria from animals to humans and vice versa [295]. Future work using short- and long-read WGS and shotgun metagenomics are required to determine the transmission pathways of antimicrobial resistant bacteria and ARGs at the dairy farm, human and environmental interface.



Orange arrows indicate drivers of AMR, green arrows indicate potential transmission pathways and arrow orientation indicates the direction of the relationship. Figure adapted from Collis et al., (2019)

## 2.12 Conclusion

AMU in dairy farm environments and the consequent emergence and transmission of antimicrobial resistant bacteria and ARGs is a concern for both public, animal and environmental health. Antimicrobial resistant bacteria of concern to human health such as AmpC- and ESBL-E have been detected both overseas and in the NZ dairy farm environment. The prevalence of AmpC- and ESBL-E in calves and cattle, and associated food products and the wider farm environment varies between studies and is likely due to contrasting farm management practices, national and local veterinary AMU regulations, food processing standards, pathogen surveillance and the sensitivity and specificity of detection methods to identify antimicrobial resistant bacteria. Therefore, caution should be used when comparing data between studies.

Evidence of transmission of antimicrobial resistant bacteria from livestock to humans and vice versa remains equivocal, but some studies focusing on ESBL-producing *E. coli* in Europe found that the predominant sequence types of E. coli from human bloodstream infections and livestock were genetically distinct. However, further sampling of livestock animals of various ages, environmental sources, vectors (e.g. birds, rodents and mammalian pests) and healthy humans are required to further assess any potential transmission pathways. Enhanced knowledge of the prevalence, distribution and transmission of antimicrobial resistant bacteria and ARGs in the dairy farm environment, including pasture-based farms and those with low AMU such as in NZ, is required to understand the transmission pathways and risk factors for subsequent development of mitigation strategies to reduce antimicrobial use and AMR in agricultural environments. Additional high-resolution WGS (e.g. short- and long-read sequencing) or metagenomic studies would provide important transmission and prevalence data. Studies using shotgun metagenomic sequencing to assess ARG abundance should also apply an ARG risk framework approach to consider the bacterial host, genomic context and risk each ARG poses to fully understand the burden of AMR in agricultural environments.

Lastly, NZ has policies and guidelines for the judicious use of antimicrobials in animal health and is a comparatively low user of such antimicrobials. However, education, outreach and stakeholder engagement (by both veterinary professionals and farmers) is required to meet the targets set by the NZVA and to promote prudent antimicrobial stewardship. Addressing the rising threat of AMR requires a holistic "One Health" approach across multiple sectors and further studies are required to understand the role of the dairy farm environment in the development, persistence and transmission of antimicrobial resistant bacteria and ARGs.

Studies investigating AMR in NZ dairy farm environments have generally taken a culturebased approach, focusing on specific target organisms and resistant phenotypes [22] while utilising a cross-sectional study design [25, 26]. This research, using a longitudinal approach, addresses the hypothesis that antimicrobial use on NZ dairy farms influences the prevalence of AMR in dairy farm environments, taking into consideration seasonality and contrasting farm management practices. This study utilises a combination of phenotypic, molecular and next-generation sequencing techniques to study AMR in two NZ dairy farm environments.

# Chapter 3

# Prevalence and distribution of extended-spectrum $\beta$ -lactamase and AmpC-producing *Escherichia coli* in two New Zealand dairy farm environments

## 3.1 Introduction

Antimicrobial resistance (AMR) is a complex "One Health" issue which affects human, animal and environmental health [300]. Antimicrobial exposure, particularly the misuse and over-use of antimicrobials, in both human and animal health has been suggested as the main driver of AMR [1, 2]. However, other factors including heavy metal [280, 279, 301, 302] and biocide use [281], and in the dairy farm environment, management practices such as waste milk disposal and feed type, may also influence the development and dissemination of AMR [28].

Multi-drug resistant Enterobacteriaceae, particularly AmpC- and extended-spectrum  $\beta$ lactamase-producing Enterobacteriaceae (ESBL-E) are of concern to human health. AmpC and ESBLs are  $\beta$ -lactamase enzymes which confer resistance to first- and third-generation cephalosporins, penicillins and monobactams. The AmpC  $\beta$ -lactamase enzymes also confer resistance to the second-generation cephalosporins and cephamycins [92]. ESBL-E can cause both hospital- [118] and community-acquired infections [119] and have been classified as critical on the World Health Organisation's "Priority Pathogens List" [54]. ESBL genes are often encoded on plasmids which can carry multiple antimicrobial resistance genes (ARGs), resulting in a multi-drug resistance phenotype [50, 48]. In *E. coli*, an AmpC phenotype can arise from mutations in the promoter region of the chromosomal *ampC* gene resulting in putative AmpC hyperproducers [138], or from plasmid-mediated AmpC  $\beta$ -lactamase (pAmpC) genes [177]. Plasmid-mediated resistance is of particular concern as this phenotype can be shared by horizontal gene transfer (HGT) between bacteria.

Globally, AmpC- and ESBL-E have been detected in various agricultural environments

including dairy [7, 192, 194, 156], poultry [303, 304, 305], swine [306, 307] and aquaculture [308]. The prevalence of AmpC- and ESBL-E varies between studies which could partially be due to variation in selection methods, farm management practices and antimicrobial use. Risk factors for AmpC- and ESBL-E on dairy farms include third- and fourth-generation cephalosporin use, increased antimicrobial use in calves [7] and amoxicillin use for AmpC-producing *E. coli* [156]. AmpC- and ESBL-E have also been detected in organic dairy herds with low antimicrobial usage (AMU) [192]. A study using mixed effects logistic regression and population-attributable fractions found that AMU could only explain 22 - 36% of the AmpC/ESBL positive samples on dairy farms in the Netherlands, although this study investigated the selection of AMR rather than introduction [309]. These findings highlight the complexity and multitude of factors involved in the development and transmission of AMR in agricultural environments.

Few studies have investigated the prevalence of AmpC and ESBL-E in pasture-based dairy farm environments such as those found in New Zealand (NZ) where there is relatively low use of antimicrobials in food-producing animals [16, 17]. One regional cross-sectional study of dairy farms in NZ found a low prevalence of ESBL-producing *E. coli* (1 of 116; 0.9%) in pooled faecal samples [25]. No *E. coli* with pAmpC genes were isolated but 7.9% (9 of 114) of faecal samples were positive for putative AmpC hyperproducers with mutations in the promoter region of the *ampC* gene [25]. Similarly, a nationwide cross-sectional study (n=26 farms) in NZ did not detect any ESBL-producing *E. coli* [26] and chromosomal AmpC-producing *E. coli* were isolated from 14% (11 of 78) of pooled faecal enrichments originating from seven farms [26]. Additionally, an NZ study between 2009 - 2010 found no ESBL-producing *E. coli* (0 of 300) from bobby calf carcasses (4 to 10 days old) [196]. Hence, the aims of this study were to utilise culture-based methods to investigate the prevalence of AmpC and ESBL-E from two NZ dairy farm environments over a 15 month period, taking into consideration seasonal variation and farm management practices.

## 3.2 Methods

### 3.2.1 Study population and sample collection

The Massey University research farms No. 1 Dairy Farm (referred to hereafter as Dairy 1) and No. 4 Dairy Farm (referred to hereafter as Dairy 4) were recruited for inclusion in this study. The two dairy farms are located in Palmerston North, New Zealand, are <5 kilometres apart and both operate a closed dairy farm system (animals are not introduced into the herd). Dairy 1, located next to the Manawatū River, is a small-scale 142.7 hectares (ha) farm with 261 cows during the study period. Dairy 4 is a larger scale farm of 250 ha and a herd size of 584 cows during the study period. The two farms are pasture-based, with the use of supplementary feed such as silage (pickled pasture) and baleage when required. Both farms have stand-off pads (Dairy 1 with a 200 cow capacity, compared to Dairy 4

which has two, of 150 and 200 cow capacities) and Dairy 4 has a freestall barn (200 cow capacity). The predominant breeds on both farms are Friesian, Jersey and Friesian/Jersey cross breeds. Both dairy farms have a spring calving system, and Dairy 1 uses once a day milking whereas Dairy 4 milks twice a day (morning and afternoon). The two farms use selective dry cow therapy (DCT) and teat sealants are applied to the whole herd. On Dairy 4, farm dairy effluent (FDE) is applied to the paddocks using a traveling irrigator system, which is a common practice on NZ dairy farms. The FDE management strategy on Dairy 1 changed during the study period. From October 2018, the FDE was filtered using the dairy effluent recycling system by Forsi Innovations [310] and applied to paddocks. Due to technical issues, from July 2019 onwards the FDE was not applied to paddocks and was discharged into the Palmerston North sewage system.

Collection of farm environmental samples was on a monthly basis from October 2018 to December 2019 inclusive, spanning a 15 month period. Both farms were sampled on the same day. On each sampling occasion, soil, composite cow faeces from cow pats in a recently grazed paddock, FDE, bulk tank milk (referred to hereafter as milk), milk filters and supplementary feed were collected. Samples were transported to the Hopkirk Research Institute (Massey University, Palmerston North, New Zealand) in a chilly-bin with ice-bricks. Samples were stored at 4°C until processing and were processed within eight hours of collection. The collection method for each sampling type is described in Table 3.1. Farm management practice metadata such as the use of supplementary feed and hygiene practices were collected during sampling visits. AMU was reported as individual animal antimicrobial treatments recorded on farm. The total amount administered for each treatment (mg) was calculated according to the concentration of the product (mg/mL), number of doses and volume (mL). AMU was reported as mg/population correction unit (PCU) calculated using the total active ingredient weight (mg) / herd size / average cow size in NZ (453 kg).

## 3.2.2 Sample processing

To aid in the growth of *Enterobacteriaceae*, the environmental samples were enriched in buffered peptone water (BPW) (BD Difco<sup>TM</sup>, Fort Richard Laboratories, Auckland, New Zealand). The enrichment methods are detailed in Table 3.2. Once inoculated in BPW, each sample was mixed by vortex and incubated at 35°C for 18 hours. After incubation, enrichments were mixed by vortex and 1 mL was mixed with glycerol (30% [v/v]) and stored at -80°C.

TANING .T.O OTANT			
Sample type	Source	Collection method	
Soil	Recently grazed paddock (grazed the night prior to sampling).	Four soil cores (10 cm depth; 3 cm diameter) collected from near the centre of the paddock. Each soil core taken in a transect line, approximately 1 m apart. The collection of pasture and faeces was avoided.	
Faeces	Recently grazed paddock (grazed the night prior to sampling).	Sixteen fresh faecal samples randomly collected from the paddock and pooled as groups of four.	
FDE	Dairy 1, two sample collection points were used. Collection point one was the effluent sump by the cow-shed (October 2018 - May 2019) and collection point two was from a grate in the cow-shed (July 2019 - December 2019). For Dairy 4, the FDE was collected from the effluent pond.	Approximately 500 mL FDE collected in two sterile 250 mL Schott bottles attached to the end of a pole.	
Milk	Milk vat.	Approximately 1 L collected from the milking vat. On Dairy 4, milk collected from the morning milk- ing. The milk sample collected in October 2019 from Dairy 1 was waste milk (DF0167), rather than bulk tank milk. The sample collection issue was dis- covered during the analysis of the shotgun metage- nomic sequencing data from this sample.	
Milk filter	Cow-shed.	Collected by farm staff prior to vat cleaning and stored in a clean bag until collection.	
Silage	Silage nit.	Annroximately 30 g collected.	

Table 3.1: Sample collection methods

Sample type	Sample volume or weight	BPW <sup>a</sup> (mL)	Additional steps
Soil	1 g	9	Soil samples homogenised by hand in the WhirlPak bag.
Faeces	Pea-size amount per sample	15	The prevalence of ESBL-producing $E$ . coli was expected to be low, therefore four faecal samples were pooled. A pea-size amount of each faecal sample was transferred using a sterile cotton swab into BPW.
FDE	100 µL re- suspended pellet	9.9	100 µL of the re-suspended pellet from the DNA extraction (refer to section 5.2.1).
Milk	100 µL re- suspended pellet	9.9	100 µL of the re-suspended pellet from the DNA extraction (refer to section 5.2.1).
Milk filter	Sponge swab	25	The milk filters were cut open with flame sterilised scissors. The filter was swabbed with a sponge swab, then stomached for 2 min to extract and wash any microbes present into the BPW.
Silage	10 g	50	Samples stomached for 2 min in BPW.

Table 3.2: Sample processing methods

<sup>a</sup> BPW, Buffered peptone water.

## 3.2.3 Microbiological methods

Stored frozen enrichments were removed from -80°C and kept on ice. A loopful of frozen enrichment was directly plated on (i) MacConkey (MC) Agar plates (Fort Richard Laboratories, Auckland, New Zealand) as a positive control to ensure growth of Enterobacteriaceae, (ii) MC agar (BD Difco<sup>TM</sup>, Fort Richard Laboratories, Auckland, New Zealand) with 1 µg/mL cefotaxime sodium (Sigma-Aldrich, St. Louis, MO, USA), (iii) MC agar (BD Difco<sup>TM</sup>) with 1 µg/mL ceftazidime pentahydrate (Sigma-Aldrich) and (iv) CHROMagar<sup>TM</sup> ESBL (CHROMagar, Paris, France). Agar plates were incubated aerobically at 35°C for 18 hours. After incubation, a minimum of two presumptive Enterobacteriaceae representing all distinct colony morphologies from plates (i-iii) were sub-cultured onto Columbia Sheep Blood agar (5% blood) (Fort Richard Laboratories). From CHROMagar<sup>TM</sup> ESBL, if present a minimum of two presumptive E. coli (dark pink to reddish) and Klebsiella (metallic blue) colonies were sub-cultured onto fresh CHROMagar<sup>TM</sup> ESBL to ensure purity using colour differentiation as a guide. The colonies were subsequently sub-cultured onto Columbia Sheep Blood agar (5% blood). The AmpC-producing E. coli NZRM4402, the ESBL-producing Klebsiella pneumoniae NZRM3681 and the susceptible E. coli NZRM916 were used as reference strains for quality control of the selective MC agar and CHROMagar<sup>TM</sup> ESBL plates. Presumptive Enterobacteriaceae pure isolates were stored in brain heart

infusion broth (Oxoid, Hampshire, United Kingdom) containing glycerol (30% [v/v]) at -80°C. Isolates were identified by matrix-assisted laser desorption/ionization-time of flight (MALDI-TOF) mass spectrometry (MS) (Bruker Daltonics, Billerica, CA, USA) using the previously described "on slide formic acid extraction" method [311]. For MALDI-TOF MS analysis, pure cultures were streaked on Columbia Sheep Blood agar (5% blood) either from glycerol stocks stored at -80°C or culture plates stored at 4°C and incubated at 35°C for 18 hours. Briefly, using a sterile toothpick, a sample of a single colony was collected and smeared on the MALDI-TOF MS target plate. Next, 1 µL of 70% formic acid was applied to each spot on the target plate and once dried, 1 µL HCCA matrix was added. The target plate was dried at room temperature and the preparation was checked to ensure a homogeneous appearance for each spot. The target plate was read using MALDI-TOF MS using Bruker Daltonics flexControl (v3.4) and Bruker Daltonics MALDI Biotyper Compass (v4.1.100). Each sample was analysed in duplicate and a score value  $\geq 2$  indicated species identification.

#### 3.2.4 Antimicrobial susceptibility tests

Confirmed E. coli strains isolated from selective agars (ii - iv) outlined above were subjected to Kirby-Bauer disc diffusion antimicrobial susceptibility tests (ASTs) for six antimicrobials of interest and interpreted according to either CLSI [312] or EUCAST [165] guidelines (Table 3.3). Bacterial cultures stored in glycerol broth (section 3.2.3) were inoculated on Columbia Sheep Blood agar (5% blood) and incubated at 35°C for 18 hours. After incubation, an individual colony was subsequently streaked onto Columbia Sheep Blood agar and incubated at 35°C for 18 hours. A bacterial suspension was prepared in sterile saline equivalent to a 0.5 McFarland standard by mixing pure colonies with a cotton swab. The bacterial suspension was mixed with a new cotton swab, and a Mueller Hinton agar plate (Fort Richard Laboratories) was swabbed in three directions to achieve a lawn of growth. Antibiotic discs were applied within 15 min and the plates incubated at 35°C for 16 - 20 hours. An AmpC and ESBL positive phenotype was confirmed for isolates resistant to either cefoxitin and cefotaxime and/or cefpodoxime (Figure 3.1) using either a three-disc (D69C AmpC disc test, Mast Group Ltd., Liverpool, United Kingdom) or double-disc comparison assay (D62C cefotaxime and D64C ceftazidime ESBL disc tests, Mast Group Ltd., Liverpool, United Kingdom), respectively. AmpC  $\beta$ -lactamase enzymes can hydrolyse cefotaxime and ceftazidime and are less effected by ESBL inhibitors such as clavulanic acid [131] which is used in the D62C/D64C ESBL double-disc comparison assays. Therefore, isolates which were AmpC positive and had a zone size which could not be differentiated as either positive or negative ( $\geq 2 - \leq 5$ mm) for either ESBL double-disc comparison assay were further tested using an assay containing cefepime, which is stable in the presence of *Enterobacteriaceae* with chromosomal ampC expression (D63C cefepime ESBL disc test, Mast Group Ltd., Liverpool, United Kingdom). The AmpC-producing E. coli NZRM4402 and the ESBL-producing Klebsiella pneumoniae NZRM3681 were used as positive controls in the AmpC and ESBL confirmatory disc assays, respectively and the susceptible E. coli NZRM916 was used as a negative control for both assays.

	Antibiotic class	${ m Con.^a}\ (\mu { m g})$	Guideline	Susceptible	$\mathbf{Resistant}$
Antibiotic disc				zone	zone
				size (mm)	size $(mm)$
Cefotaxime (CTX)	$\beta$ -lactam	30	CLSI	$\geq 26$	$\leq 22$
Cefoxitin (FOX)	$\beta$ -lactam	30	CLSI	$\geq 21$	$\leq 17$
Cefpodoxime (CPD)	$\beta$ -lactam	10	CLSI	$\geq 18$	$\leq 14$
Tetracycline (TET)	Tetracycline	30	CLSI	$\geq 15$	$\leq 11$
Streptomycin (STR)	Aminoglycoside	10	CLSI	$\geq 15$	$\leq 11$
Ciprofloxacin (CIP)	Fluoroquinolone	5	CLSI	>26	<21

Table 3.3: Discs for antimicrobial susceptibility tests.

<sup>a</sup> Con, Concentration.

#### 3.2.5 Molecular characterisation

Crude DNA was extracted from pure isolates by adding 3-4 colonies to 400 µL sterile molecular biology-grade water. The bacteria were heated at 100°C for 10 min in a heating block and stored at -20°C. The supernatant was used for subsequent PCR reactions. PCR reactions, control strains and gel electrophoresis conditions are detailed in Appendix A. Briefly, the *E. coli* phylogroup was determined using the Clermont quadruplex PCR Typing method [76]. *E. coli* with an AmpC phenotype were tested for pAmpC gene families using a multiplex PCR [177] and a PCR targeting the  $bla_{\rm CMY}$  gene family [313] was used on boiled DNA preparations from isolates positive for the CITM primer set (Table 3.4), indicative of CMY-positive *E. coli*. *E. coli* in which no pAmpC genes were identified were further analysed to identify mutations in the promoter region of the *ampC* gene [138]. The ESBL  $bla_{\rm CTX-M}$  group was confirmed using the CTX-M-1-group PCR [7] for the ESBL-producing *E. coli* which did not undergo whole genome sequencing (WGS).

PCR products were purified using the QIAquick PCR purification kit (QIAGEN, Hilden, Germany) and primers corresponding to the PCR product being sequenced were used in the sequencing reaction (Appendix A). Briefly, separate sequencing reactions were performed in 10 µL reaction volumes, using the same forward and reverse primers, with 1.75 µL buffer, 0.5 µL Big Dye<sup>TM</sup> Terminator v3.1 (ThermoFisher Scientific Inc., Waltham, MA, USA) and 1 µL primer (3.2 pmol/µL). For sequencing of the *ampC* gene promoter region PCR product (191 bp), 5.75 µL sterile molecular biology-grade water and 1 µL DNA template (3 ng) was used. For sequencing of  $bla_{CMY-2}$  and  $bla_{CTX-M-1}$  PCR products (1,138 and >1000 bp, respectively), 4.75 µL sterile molecular biology-grade water and 2 µL DNA template (12 ng) was used. The sequencing PCR reactions were undertaken on a Bio-Rad T100 Thermal Cycler (Bio-Rad, Waltham, MA, USA) using the following conditions: 95°C for 1 min, then 25 cycles of 95°C for 10 sec, 50°C for 10 sec, and 60°C for 90 sec. Capillary separation of sequencing reactions was undertaken by the Massey Genome Service using an ABI3730 DNA analyser (Massey University, Palmerston North, New Zealand). DNA sequence chromatograms were trimmed at the proximal 5' and distal 3' end to remove poor


quality sequence using Geneious Prime v2019.1.1 [314, 315] and forward and reverse traces were aligned to form a contiguous DNA sequence representing the complete PCR product. The PCR products were analysed using BLASTN [316].

PCR reaction	Target	Primer set	${\bf Primer\ sequence\ (5^{\prime}\rightarrow 3^{\prime})}$	Product size (bp)	Reference
AmpC promoter region	Chromosomal $ampC$	${ m AmpC1}^{71}$ ${ m AmpC2}^{120}$	AATGGGTTTTCTACGGTCTG GGGCAGCAAATGTGGAGCAA	191	[138]
pAmpC	MOX-1, MOX-2, CMY-1, CMY-8 to CMY-11	MOXMF MOXMR	GCTGCTCAAGGAGCACAGGAT CACATTGACATAGGTGTGGTG	520	[177]
pAmpC	LAT-1 to LAT-4, CMY-2 to CMY-7, BIL-1	CITMF CITMR	TGGCCAGAACTGACAGGCAAA TTTCTCCTGAACGTGGCTGGC	462	[177]
pAmpC	DHA-1, DHA-2	DHAMF DHAMR	CCGTACGCATACTGGCTTTGC AACAGCCTCAGCAGCCGGTTA	405	[177]
pAmpC	ACC	ACCMF ACCMR	AACAGCCTCAGCAGCCGGTTA TTCGCCGCAATCATCCCTAGC	346	[177]
pAmpC	FOX-1 to FOX-5b	FOXMF FOXMR	CAAAGCGCGTAACCGGATTGG AACATGGGGTATCAGGGAGATG	190	[177]
CMY-group	CMY-group	CMY-2-F CMY-2-R	ATGATGAAAAAATCGTTATGCTGC GCTTTTCAAGAATGCGCCAGG	1,138	[313]
CTX-M-1	CTX-M-1-group	CTX-1-SEQ-F CTX-1-SEQ-R	CCCATGGTTAAAAATCACTGC CAGCGCTTTTGCCGTCTAAG	>1,000	[313]
Quadruplex	chuA	chuA.1b chuA.2	ATGGTACCGGACGAACCAAC TGCCGCCAGTACCAAAGACA	288	[76, 75]
Quadruplex	yjaA	yjaA.1b yjaA.2b	CAAACGTGAAGTGTCAGGAG AATGCGTTCCTCAACCTGTG	211	[76]
Quadruplex	TspE4.C2	TspE4C2.1b TspE4C2.2b	CACTATTCGTAAGGTCATCC AGTTTATCGCTGCGGGGTCGC	152	[76]
Quadruplex	arpA	${ m AceK.f} { m ArpA1.r}$	AACGCTATTCGCCAGCTTGC TCTCCCCCATACCGTACGCTA	400	[76, 317]
Group C	trpA	trpAgpC.1 trpAgpC.2	AGTTTTATGCCCAGTGCGAG TCTGCGCCGGTCACGCCC	219	[318]
Group E	arpA	ArpAgpE.f ArpAgpE.r	GATTCCATCTTGTCAAAATATGCC GAAAAGAAAAAGAATTCCCAAGAG	301	[318]

Page 57

# 3.2.6 DNA extraction, library preparation, and whole genome sequencing

A subset of AmpC- and ESBL-producing *E. coli* were selected for WGS according to phenotype, resistance profile, E. coli phylogroup and metadata (including farm, source and collection date). For samples where multiple isolates were identified, only one isolate per molecular and resistance profile combination was included. Bacterial isolates from glycerol broths stored at -80°C (section 3.2.3) were inoculated on Columbia Sheep Blood agar (5% blood) and incubated for 18 hours at 35°C. An individual colony was subsequently sub-cultured onto a fresh Columbia Sheep Blood agar plate (5% blood) and incubated at 35°C for 18 hours. An individual colony was inoculated in 4 mL Luria-Bertani Miller broth (Fort Richard Laboratories) and incubated at 35°C for 15 hours at 200 rpm. Genomic DNA (for both Illumina and MinION sequencing methods) was extracted using the Wizard<sup>®</sup> Genomic DNA Purification Kit (Promega, Madison, WI, USA), according to the manufacturer's instructions for gram negative bacteria. Several modifications were included to optimise the protocol for E. coli. Briefly, 2 mL overnight culture was centrifuged at 13,000 x q for 2 min to pellet the cells which were treated with RNAse at 37°C for 1 hour. After the Protein Precipitation Solution was added to the cell lysate and incubated on ice for 5 min, the sample was centrifuged at 13,000 x q for 3 min, the supernatant transferred to a clean tube and centrifuged again at  $13,000 \ge q$  to reduce any residual protein contamination. The DNA was re-hydrated overnight at 4°C in 100µl 10mM Tris-HCl (Geneaid Biotech Ltd, New Taipei City, Taiwan).

The DNA concentration was quantified using a Qubit 4.0 fluorometer (ThermoFisher Scientific Inc., Waltham, MA, USA) and  $A_{260/280}$  and  $A_{260/230}$  ratios determined using the Nanodrop microvolume spectrophotometer (Nanodrop 2000c, ThermoFisher Scientific Inc., Waltham, MA, USA). DNA integrity and size was visualised on a 0.8% [w/v] agarose gel using a high molecular weight Lambda DNA/Hind III ladder (ThermoFisher Scientific Inc., Waltham, MA, USA). The libraries were prepared using the Nextera XT DNA library preparation kit (Illumina, San Diego, CA, USA), and sequencing performed using an Illumina MiSeq v3 with 2 x 300 bp paired-end reads (Massey Genome Service, Massey University, Palmerston North, New Zealand). The Nanopore MinION sequencing was performed at the Molecular Epidemiology and Public Health Laboratory (Massey University, Palmerston North, New Zealand) using a R9.4.1 flow cell (Oxford Nanopore Technologies, UK). The libraries were prepared using the Rapid Barcoding Sequencing kit (SQK-RBK004; Oxford Nanopore Technologies, UK) according to the manufacturer's instructions ("Library Preparation"), with minor modifications for E. coli. Briefly, 600ng DNA template was used as input. After the beads were washed twice with 70% ethanol, the residual ethanol was removed and the pellet dried for 15 min. The pellet was then re-suspended in 10 µL 10mM Tris-HCl/50mM NaCl and the solution containing the beads and DNA was then incubated at 50°C for 10 min. The flow cell was primed and loaded according to the manufacturer's instructions and run for 24 - 48 hours (SQK-RBK004, "Priming and loading the SpotON flow cell", Oxford Nanopore Technologies, UK).

#### 3.2.7 Bioinformatic analysis

Due to a high sequencing depth, the Illumina MiSeq sequencing reads were randomly subsampled down to a 100X genome coverage using Rasusa v0.6.0 [319]. The Illumina MiSeq sequencing reads were subsequently processed using the Nullarbor v2.0 [320] pipeline with default parameters. In summary, adapters were removed from raw reads using Trimmomatic v0.39 [321], species identification by k-mer analysis performed using the Kraken v1.1.1 database [322], the genomes were assembled using SKESA v2.4.0 [323] and annotated with Prokka v1.14.6 [324]. The sequence type was determined using mlst v2.19.0 [325] with information downloaded from PubMLST [326], the resistome profile identified with ABRicate v1.0.1 [327] using the Resfinder 4.0 database [216, 328], and the Centre for Genomic Epidemiology website [329] was used to detect the virulence genes and serotype in assembled genomes using the VirulenceFinder 2.0.3 database (v2020-05-29) [73] and the SerotypeFinder 2.0.1 database (v1.0.0), respectively. The presence/absence data from 37 virulence associated genes identified in the whole genome sequences (n=12) was used to construct a hierarchical cluster tree using Jaccard distances and the tree was further annotated using the Interactive Tree of Life webserver [330]. The core single nucleotide polymorphism (SNP) variation was assessed using Snippy v4.4.3 [331] with a ST131 ESBLproducing E. coli JJ1887 as the reference sequence (Genbank accession: CP014316). A maximum-likelihood tree was generated from the core SNP alignment using a general time-reversible model with the Randomised Axelerated Maximum Likelihood (RAxML) Next-Generation tool [332] and visualised in GrapeTree [333].

For long-read data, the MinION fast5 sequencing read files were basecalled using Guppy v4.2.2. The reads were de-multiplexed using qcat v1.1.0 [334] and adapters removed with Porechop v0.2.4 [335] using default settings. Filtlong v0.2.0 [336] was used to trim the reads with a minimum length of 1 kb, the lowest quality bases filtered (10%) and the best reads kept up to 500 Mb. Hybrid assemblies were generated using Unicycler v0.4.9b [337] with default settings. Plasmids from hybrid assemblies were re-constructed and typed using MOB-suite v1.4.9.1 [338, 339] and annotated with Prokka v1.14.6 [324] using a custom database consisting of the best-match "nearest neighbour" plasmids as identified by MOB-suite v1.4.9.1 (Genbank accessions: CP009566, CP015997, CP018107, CP016585 and KF362121). For pMLST, variants within IncI group plasmids and Inc groups for short-read data were identified using PlasmidFinder v2.1 [340]. Plasmid core genome variation for selected isolates was assessed using Snippy v4.6.0 [331] with pDF0049.2e\_1 as the reference (Table 3.7). Clusters of Orthologous Groups (COG) were identified using eggNOG-mapper v2 [341, 342, 343] and plasmid annotations visualised using Mauve [344] and Geneious Prime v2019.1.1 [314, 315]. The plasmid oriC region was identified using DoriC v10.0 [345] and the figures were edited using Inkscape v0.92.1 [346].

#### 3.2.8 Statistical tests

Statistical tests were performed in  $Minitab^{\textcircled{R}}$  19.1.1 [347] using a one-way analysis of variance (ANOVA) with a 95% confidence interval for sample prevalence comparisons.

### 3.3 Results

Of the two dairy farms visited, 101 enrichments from Dairy 1 and 103 enrichments from Dairy 4 were resuscitated on four agar plates: MC agar, MC supplemented with either 1  $\mu$ g/mL cefotaxime or 1  $\mu$ g/mL ceftazidime and CHROMagar<sup>TM</sup> ESBL plates. From the selective agar plates, 52 putative third-generation cephalosporin resistant *Enterobacteriaceeae* (n=24 Dairy 1; n=28 Dairy 4) were isolated and the species confirmed as *E. coli* using MALDI-TOF MS (Appendix B). No *Klebsiella pneumoniae* were isolated from the selective agar plates.

#### 3.3.1 Antimicrobial resistance profiles

The susceptibility of *E. coli* isolated in this study (n=52) to six clinically relevant antimicrobials (Table 3.3), including three  $\beta$ -lactams, was examined (Figure 3.2; Appendix C). All of the *E. coli* (n=52) were resistant to cefpodoxime, 46 of 52 (89%) to cefoxitin and 34 of 52 (65%) were resistant or intermediate (15 of 52, 29%) to cefotaxime. The high levels of resistance to the second- and third-generation cephalosporins reflects the selective methods used to isolate AmpC- and ESBL-E in this study. Numerous isolates were resistant to streptomycin (41 of 52; 79%) and tetracycline (33 of 52, 64%) and all isolates (n=52) were susceptible to ciprofloxacin, which is a critically important antimicrobial in human medicine [155]. According to phenotypic testing, 33 of 52 *E. coli* (64%) were multi-drug resistant, but phenotypic testing was only performed using antibiotics representing four classes ( $\beta$ -lactams, tetracyclines, aminoglycosides and fluoroquinolones).



Figure 3.2: Resistance profiles of *E. coli* (n=52) isolated across 14 farm samples. Isolates with intermediate resistance to cefotaxime (n=15) were grouped as resistant in the Upset plot.

<sup>a</sup> CPD, cefpodoxime; FOX, cefoxitin; STR, streptomycin; CTX, cefotaxime; TET, tetracycline; CIP, ciprofloxacin

All 52 *E. coli* were tested for ESBL production (Figure 3.1) and two strains isolated from the same FDE sample were confirmed as ESBL producers (DF0183c and DF0183g). All 52 *E. coli* were tested for AmpC production and 46 AmpC-producing *E. coli* were identified (46 of 52, 88.5%). The two ESBL-producing *E. coli* were AmpC negative and four isolates from the same sample (DF0102.4e-h) were both ESBL and AmpC negative according to the phenotypic confirmation tests.

### 3.3.2 Molecular characterisation of AmpC- and ESBL-producing *E. coli*

*E. coli* isolates (n=52) were differentiated into phylogroups using the Clermont Typing method [76]. The predominant phylogroups were B1 (18 of 52, 35%) and C (17 of 52, 33%), followed by E (6 of 52, 12%), D (5 of 52, 10%), A (4 of 52, 8%) and F (2 of 52, 4%). The  $bla_{\rm CTX-M-15}$  gene was identified in the ESBL-producing *E. coli* (DF0183c and DF0183g). *E. coli* with an AmpC phenotype (n=46) were investigated for pAmpC genes, and the  $bla_{\rm CMY-2}$  gene was detected in 28 isolates (28 of 46, 61%). For the remaining 18 AmpC-producing *E. coli*, mutations were identified in the promoter region of the *ampC* gene. Interestingly, 27 of 28 (96%) isolates with the  $bla_{\rm CMY-2}$  gene were isolated from

Dairy 4. All putative AmpC hyperproducers (n=18) were isolated from Dairy 1. The 18 isolates were all phylogroup C and compared to a susceptible *E. coli* reference strain (ATCC<sup>®</sup> 25922<sup>TM</sup>), had mutations in the promoter region of the *ampC* gene at positions -42 (C  $\rightarrow$  T), -18 (G  $\rightarrow$  A), -1 (C  $\rightarrow$  T) and +58 (C  $\rightarrow$  T), excluding DF0025c in which position -42 could not be determined (positions relative to the *E. coli* K12 transcriptional start base (+1) [348]). No pAmpC or ESBL genes were detected in the four *E. coli* which were resistant to cefotaxime and cefpodoxime but were AmpC and ESBL negative in the phenotypic testing (DF0102.4e-h). Therefore, the mechanism for resistance to third-generation cephalosporins was unassigned for these isolates (designated as unknown in subsequent figures).

Interestingly, the 15 strains with intermediate resistance to cefotaxime were all putative AmpC hyperproducers, with mutations in the promoter region of the ampC gene. The remaining isolates shared identical mutations (n=3) yet were all susceptible to cefotaxime. In comparison, all AmpC-producing *E. coli* with pAmpC genes (n=28) were resistant to cefotaxime. The difference in AST cefotaxime zone sizes (mm) from 15 of 46 distinct AmpC-producing *E. coli* clones (excluding *E. coli* with an identical phylogroup and AST profile which were isolated from the same sample) were compared and plasmid-mediated AmpC-producing *E. coli* had larger zone sizes (Appendix D), however a larger sample size is required to perform any statistical comparisons.

#### 3.3.3 Population structure and comparative genomics

A subset of AmpC- and ESBL-producing *E. coli* were analysed using WGS to understand the genomic epidemiology and transmission dynamics of these bacteria on farm (Table 3.5). The *E. coli* were selected for WGS according to phenotype, resistance profile (focusing on isolates with pAmpC genes and ESBL producers), *E. coli* phylogroup and metadata (including farm, source and collection date). *E. coli* isolates with a plasmid-mediated AmpC-producing phenotype (n=5), ESBL-producing (n=1) and an unassigned mechanism for resistance to third-generation cephalosporins (n=1) were also selected for MinION longread sequencing, with a focus on generating complete genomes and examining plasmids in detail. For samples where multiple isolates were identified, only one isolate per molecular and resistance profile combination was selected for WGS.

The population structure of the AmpC- and ESBL-producing *E. coli* which were sequenced in this project (n=12) was diverse and the isolates belonged to eight sequence types including ST56 (3 of 12, 25%), ST57 (2 of 12, 17%), ST88 (2 of 12, 17%) and singletons for ST442, ST973, ST2541, ST4553 and ST5135 (1 of 12, 8%) (Table 3.5). The three ST56 and two ST88 strains had identical serotypes (O8/O40:H21 and O8/O32:H19, respectively). The remaining isolates (n=7), including the two ST57 strains, had different serotypes. According to SerotypeFinder, the O serogroup was non-typeable for three strains. All isolates had a genome size ranging from 4,839,855 - 5,431,661 bp and a GC content between 51.7

#### - 52.0%.

Core genome SNP analysis separated the strains into eight clusters, each representing a different ST (Figure 3.3). The separation of clusters according to ST was reflected in the large number of SNPs identified in the core genome of these strains (102,841 SNPs), accounting for approximately 2.0% of the *E. coli* genome. However, within two clusters genetically similar isolates representing two STs were identified. Three *E. coli* with the  $bla_{\rm CMY-2}$  gene were isolated in December 2019 on Dairy 4 from different samples and all belonged to ST56, were serotype O8/O40:H21 and were genetically similar (58 - 65 SNPs; isolates DF0181.1c, DF0181.3c and DF0183e). Similarly, two putative AmpC hyperproducers had the same serotype, were isolated in December 2018 and January 2019 from Dairy 1 and differed by 82 SNPs (DF0031.1c and DF0047c). In each case, the *E. coli* were isolated from both faeces (ST56 n=2; ST88 n=1) and FDE (ST56 n=1; ST88 n=1), but the number of SNPs is too high to infer recent within farm transmission. In contrast, the two isolates belonging to ST57 were isolated from separate farms and were genetically distinct (6,750 SNPs).



Figure 3.3: Maximum-likelihood tree of core genome single nucleotide polymorphism (SNP) analysis of AmpC and ESBL-producing *E. coli* (n=12). *E. coli* CP014316 (ST131,  $bla_{\text{CTX-M-15}}$  positive) was used as the reference and nodes are coloured by sequence type, as indicated in the figure legend. The scale bar indicates the proportion of the core genome alignment over which core SNPs have been calculated. The number of SNPs between isolates in the same cluster is indicated on the figure.

Isolate <sup>a</sup>	Farm	Source	Collection date	Phylo-	$\mathbf{ST}$	Serotype <sup>b</sup>	Phenotype <sup>c</sup>	Genome size (bp)	Contigs <sup>d</sup>	<b>G</b> C (%)	Plasmids
				group							
DF0031.1c	1	Faeces	12/2018	C	88	O8/O32:H19	$\operatorname{AmpC}$	5,180,929	91	51.9	IncFIB, IncFII, IncQ1
$\rm DF0047c$		FDE	01/2019	C	88	O8/O32:H19	AmpC	5,157,783	90	51.9	IncFIB, IncFII, IncQ1
$DF0049.2e^{a}$		Faeces	01/2019	ы	57	O27:H18	pAmpC	5,047,883	$86(2^*)$	51.8	IncI1
$\rm DF0059.2e^{a}$	4	Faeces	02/2019	D	5135	ONT:H26	pAmpC/ESBL	5,437,661	$99 (4^*)$	51.7	IncI1, IncY, Inc- FIA/IncQ1
DF0102.4g <sup>a</sup>		Faeces	05/2019	D	973	ONT:H15	Unknown	5,041,356	81 (4)	51.7	IncI1, IncI1/IncFII
$\rm DF0159.2g^{a}$	4	Faeces	10/2019	A	2541	ONT:H7	pAmpC	4,869,780	80(5)	51.9	IncI1
$\rm DF0181.1c^{a}$	4	Faeces	12/2019	B1	56	O8/O40:H21	pAmpC	5, 199, 529	192(3)	51.9	IncI1
DF0181.3c	4	Faeces	12/2019	B1	56	O8/O40:H21	pAmpC	5,202,415	194	52.0	IncI1
DF0181.4c	4	Faeces	12/2019	E	57	O124/O164:H25	pAmpC	4,961,348	83	51.9	IncI1
$\rm DF0183e^{a}$	4	FDE	12/2019	B1	56	O8/O40:H21	pAmpC	5,201,499	$194(2^*)$	51.8	IncI1
$\rm DF0183g^{a}$	4	FDE	12/2019	ĹŦł	4553	O83:H42	ESBL	5,001,896	$70(1^*)$	51.8	None
DF0183i	4	FDE	12/2019	B1	442	O146:H21	pAmpC	4, 839, 955	69	51.8	$\operatorname{IncI1}/\operatorname{IncFIB}$
<sup>a</sup> Isolates sequ	tenced u	sing both	Illumina MiS	eq and O	kford N	anopore Technolog	jies.				
<sup>b</sup> ONT, O noi	1-typeak	Je.									

Table 3.5: Genome composition of E. coli (n=12) sequenced in this study

ESBL- and AmpC-producing E. coli on two NZ dairy farms

<sup>c</sup> pAmpC, Plasmid-mediated AmpC  $\beta$ -lactamase; AmpC, E. coli with mutations in the promoter region of the ampC gene.

<sup>d</sup> Hybrid assembly contig numbers in brackets. Closed genomes indicated by an asterisk.

#### 3.3.4 Antimicrobial resistance genes

Analysis of WGS data of 12 E. coli confirmed the presence of the  $\beta$ -lactam resistance genes  $bla_{\text{CMY-2}}$  (8 of 12, 66.7%),  $bla_{\text{OXA-1}}$  (2 of 12, 16.7%),  $bla_{\text{CTX-M-1}}$  and  $bla_{\text{CTX-M-15}}$  (1 of 12) each, 8.3%) (Table 3.6). Surprisingly, the  $bla_{CTX-M-1}$  gene was detected in the sequence data for isolate DF0059.2e, which was negative using the double-disc ESBL confirmation test (D62C cefotaxime and D64C ceftazidime ESBL disc tests) as described in sections 3.2.4 and 3.3.1. However, DF0059.2e was AmpC positive and harboured the  $bla_{CMY-2}$  gene. AmpC  $\beta$ -lactamases are not inhibited by ESBL inhibitors such as clavulanic acid (Table 2.1), which may explain the negative ESBL confirmation test. Subsequent ESBL confirmation tests using an assay containing cefepime, which is stable to hydrolysis by AmpC  $\beta$ -lactamases, were carried out for isolate DF0059.2e and other AmpC-producing E. coli that had a zone size which could not be differentiated as either positive or negative using the ESBL double-disc tests (n=3) (Figure 3.1). The other three *E. coli* were confirmed as ESBL negative, whereas DF0059.2e was ESBL positive. Therefore, isolate DF0059.2e was confirmed as both AmpC and ESBL positive using phenotypic and genotypic testing and co-harboured the  $bla_{\rm CMY-2}$  and  $bla_{\rm CTX-M-1}$  genes. The remaining genotypic results were in agreement with the confirmed ESBL and AmpC phenotypes. All isolates carried the  $bla_{\rm EC}$ gene (a synonym for the ampC gene [132]) and gene-specific variation was observed which was broadly associated with the different *E. coli* phylogroups.

Other ARGs including mph(A), catA1, dfrA5, dfrA17, sul1 and sul2 were identified during analysis of the assembled genomes but representative phenotypes were not established. Initial PCR methods failed to identify any ESBL and AmpC genetic determinants associated with the four DF0102.4e-h isolates that were resistant to third-generation cephalosporins (cefotaxime and cefpodoxime). Further analysis of DF0102.4g using WGS also failed to determine the genetic basis for the AMR phenotype, therefore the resistance mechanisms is currently unassigned, but this isolate carried the  $bla_{OXA-1}$  gene which encodes a narrowspectrum Class D  $\beta$ -lactamase.

Isolate	$ST^{a}$	$\mathbf{AST^b}$	<i>B</i> -lactam	Other ABCs <sup>d</sup>
isolate	51	phenotype		Other Artos
			genes <sup>c</sup>	
DF0031.1c	88	CPD, FOX,	$bla_{\rm EC-13},$	aph(3")-Ib, $aph(3')$ -Ia, $aph(6)$ -Id,
		TET, STR	bla <sub>TEM-1</sub>	dfrA5, sul1, sul2, tet(A)
DF0047c	88	CPD, FOX,	$bla_{\rm EC-13}$	aph(3")-Ib, $aph(3')$ -Ia, $aph(6)$ -Id,
		TET, STR		sul2, tet(A)
DF0049.2e	57	CTX, CPD,	$bla_{\rm EC},$	None
		FOX	$bla_{\rm CMY-2}$	
DF0059.2e	5135	CTX, CPD,	$bla_{\rm EC},$	aac(3)-IId, $aadA5$ , $aph(3")$ -Ib,
		FOX, TET,	$bla_{\rm CMY-2},$	aph(3')-Ia, $aph(6)$ -Id, $catA1$ , $dfrA17$ ,
		STR	$bla_{\rm CTX-M-1}$	mph(A), sul1, sul2, tet(B)
DF0102.4g	973	CTX, CPD,	$bla_{\rm EC-8},$	aadA1, sul1
		STR	$bla_{\text{OXA-1}}$	
DF0159.2g	2541	CTX, CPD,	$bla_{\rm EC-18},$	aadA1, sul1
		FOX, STR	$bla_{\rm CMY-2},$	
			$bla_{\text{OXA-1}}$	
DF0181.1c	56	CTX, CPD,	$bla_{\rm EC-18},$	aph(3")-Ib, $aph(6)$ -Id, $sul2$ , $tet(B)$
		FOX, TET,	$bla_{\rm CMY-2}$	
		STR		
DF0181.3c	56	CTX, CPD,	$bla_{\text{EC-18}},$	aph(3")-Ib, $aph(6)$ -Id, $sul2$ , $tet(B)$
		FOX, TET,	$bla_{\rm CMY-2}$	
		STR		
DF0181.4c	57	CTX, CPD,	$bla_{\rm EC},$	None
		FOX	$bla_{\rm CMY-2}$	
DF0183e	56	CTX, CPD,	$bla_{\text{EC-18}},$	aph(3")-Ib, $aph(6)$ -Id, $sul2$ , $tet(B)$
		FOX, TET,	$bla_{\rm CMY-2}$	
		STR		
DF0183g	4553	CTX, CPD	$bla_{\rm EC-19},$	None
			bla <sub>CTX-M-15</sub>	
DF0183i	442	CTX, CPD,	$bla_{\rm EC-18},$	None
		FOX	$bla_{\rm CMY-2}$	

Table 3.6: Resistance profiles of AmpC- and ESBL-producing *E. coli* (n=12)

<sup>a</sup> Sequence type.

<sup>b</sup> Antimicrobial susceptibility testing; CPD, cefpodoxime; FOX, cefoxitin; STR, streptomycin; CTX, cefotaxime; TET, tetracycline.

<sup>c</sup>  $bla_{\rm EC}$  is a synonym for the ampC gene [132].

<sup>d</sup> Antimicrobial resistance genes defined in Appendix E.

#### 3.3.5 Virulence factors and *E. coli* pathotypes

The *E. coli* sequenced in this study carried a range of virulence factors identified with VirulenceFinder [73], which were mainly involved in adhesion, protection/serum resistance, iron uptake and toxins, hemolysins, proteases or autotransporters (Figure 3.4). *E. coli* strains of the same ST clustered together. The number of virulence factors each *E. coli* harboured varied, with strains DF0049.2e and DF0183i carrying the fewest virulence factors (n=4 each) and DF0031.1c and DF0047c carrying the most (n=27 each). All *E. coli* harboured *terC*, which is involved in tellurite resistance [349].

Five E. coli strains were putatively classified as avian pathogenic E. coli (APEC), a subgroup of extraintestinal pathogenic E. coli due to the presence of the following virulence factors: salmochelin siderophore receptor (iroN), outer membrane protease (ompT) and increased serum survival (iss) (in DF0181.1c, DF0181.3c and DF0183e) as well as ferric aerobactin receptor (iutA) and avian hemolysin (hlyF) which were also detected in strains DF0031.1c and DF0047c. These virulence factors have been associated with APEC and are often encoded on plasmids [350]. No E. coli harboured the Shiga toxin genes (stx1 or stx2) or the locus of enterocyte effacement pathogenicity island, indicative of Shiga-toxin producing E. coli (STEC) or enteropathogenic E. coli (EPEC), respectively [81]. Three strains carried the gene for the enteroaggregative immunoglobulin repeat protein Air (air) which is commonly found in enteroaggregative E. coli (EAEC), however other virulence genes associated with this pathotype were not detected (EAST-1 and pic) [350]. Five strains harboured enterotoxigenic E. coli (ETEC) associated fimbriae F17 (f17A and f17G) encoding genes, although no other ETEC-related toxin or fimbriae genes were detected in these strains; two of these strains, DF0031.1c and DF047c, were previously identified as APEC.





#### 3.3.6 Plasmid characteristics

A large proportion of the AmpC- (8 of 10, 80%) and ESBL-encoding (1 of 2, 50%) genes identified from *E. coli* isolated within this study are plasmid-associated. To examine the plasmids in more detail, seven isolates were also sequenced using MinION long-read sequencing technology (Table 3.5). If present, the plasmids were re-constructed and analysed from *E. coli* hybrid assemblies (n=7) using MOB-suite [338, 339].

Four isolates carried one plasmid, one isolate carried two and one isolate carried three plasmids (Table 3.7). No plasmids were identified from the assembled sequence data of DF0183g, instead the  $bla_{\rm CTX-M-15}$  gene was chromosomally encoded. All plasmids contained a relaxase and mate-pair formation marker, and were therefore classified as conjugative except for plasmid pDF0059.2e 2 which belonged to the IncY group and was non-mobilisable. DF0059.2e also carried an IncFIA/IncQ1 plasmid of unknown pMLST, which was 244,307 bp in size (the nearest plasmid neighbour was the E. coli strain T23 multi-drug resistant plasmid pEQ1; Genbank accession: KF362121). The pDF0059.2e 3 plasmid was annotated and is included in Appendix F. This plasmid carried multiple ARGs (n=12), potentially conferring resistance to aminoglycoside,  $\beta$ -lactam, phenicol, trimethoprim, macrolide, sulfonamide and tetracycline antibiotic classes (Table 3.7). Physical linkage of ARGs and the location of mobile genetic elements on pDF0059.2e 3 is shown in Figure 3.6. In addition, a partial copy of the  $bla_{\text{TEM-105}}$  gene (279 bp; 31.82% coverage), the catB4 gene (106 bp; 19.49% coverage) and the aadA1 gene (166 bp; 17.18% coverage) were detected in this plasmid using ABRicate v1.0.1 [327] and the Resfinder 4.0 database [216, 328]. However, the partial copy of the catB4 and aadA1 genes could not be confirmed by visual inspection of the annotated plasmid. The genetic region surrounding the partial  $bla_{\text{TEM-105}}$  gene and the upstream IS26 transposase was extracted (2,438 bp) from the pDF0059.2e\_3 plasmid and aligned to the complete  $bla_{\text{TEM-105}}$  reference gene sequence (Genbank accession number: NG 050150, 966 bp). Pairwise alignment showed that 274/279 bp of the partial  $bla_{\text{TEM-105}}$  gene was an identical match to the distal end of the reference gene sequence, with the IS26 transposase disrupting the upstream region of the  $bla_{\text{TEM-105}}$  gene (Figure 3.5). We can hypothesise that plasmid pDF0059.2e\_3 once harboured a complete copy of the  $bla_{\text{TEM-105}}$  gene and a recombination event involving IS26 transposase disrupted this gene sequence.



The complete *bla*<sub>TEM-105</sub> gene reference sequence (accession number: NG\_050150; 966 bp) was aligned to the genetic region surrounding the partial blaTEM-105 gene in the pDF0059.2e\_3 plasmid. The extracted region was 2,483 bp in size and encompassed the partial copy of the blaTEM-105 gene and the IS26 transposase directly upstream. The bases are shaded according to nucleotide sequence similarity, with black indicating a perfect match. Figure 3.5: The 279 bp partial region of the *bla*<sub>TEM-105</sub> gene was located directly downstream of a IS26 transposase.

Five strains carried IncI1 plasmids (also called  $IncI\alpha$ ) that were pMLST 23, had a similar number of coding sequences (101-109), encoded the  $bla_{\rm CMY-2}$  gene and ranged in size from 89,859 - 94,357 bp. Using the MOB-suite database (consisting of 17,779 complete plasmids) the most similar reference to the IncI1 plasmids identified using Mash distances, was the Salmonella enterica subspecies enterica serovar Newport plasmid pCVM22462 (Genbank accession number: CP009566.1). Core genome SNP analysis of these five plasmids, using pDF0049.2e 1 as the reference, suggested they were genetically similar (Table 3.8). Strikingly, there were no SNP differences between the plasmid-core genome of DF0181.1c and DF0183e, which were isolated from composite faeces and FDE respectively on Dairy 4 in December 2019 (Table 3.5). The core genome of these isolates was also genetically similar, differing by only 65 SNPs (Figure 3.3). E. coli DF0181.3c was isolated from the same sampling month as DF0181.1c and DF0183e and had an identical serotype and resistance profile. Incl1 plasmid incompatibility factors were detected in the draft genome for DF0181.3c, and therefore this isolate may carry a similar plasmid as the aforementioned strains, however this plasmid could not be reconstructed and compared in detail in the absence of long-read sequencing data (Figure 3.5). Plasmid DF0049.2e 1 was annotated as a representative of the IncI1 plasmids and is included in Appendix G and the location of the  $bla_{\rm CMY-2}$  gene, mobile genetic elements and the tra genes involved in the conjugal transfer system is shown in Figure 3.6.

According to the plasmid core genome, plasmid pDF0059.2e\_1 was the most genetically dissimilar to the other four plasmids (19 - 25 SNPs); this plasmid is a different pMLST, size and Inc group compared to the other plasmids analysed in this study (Table 3.7). Plasmid pDF0049.2e\_1 was associated with an *E. coli* isolated from Dairy 1 whereas the remaining plasmids originated from *E. coli* isolated on Dairy 4. Interestingly, pDF0049.2e\_1 was genetically similar (1 - 26 SNPs) to the other four  $bla_{CMY-2}$  positive pMLST 23 plasmids (Table 3.8). Of the plasmid core SNPs with annotations (n=7), five resulted in missense mutations in either the conjugal transfer proteins (n=2), pilus assembly protein PilO (n=2) or in a methyltransferase (n=1) (Appendix H). One synonymous SNP was detected in a hypothetical protein and plasmid pDF0059.2e\_1 had a complex mutation (mixture of SNPs and multiple nucleotide polymorphisms) resulting in an absent start site in a DNA primase CDS. Annotated comparison of the five plasmids (Appendix I) reflects the similarity and gene synteny among these plasmids. All plasmids showed variation near the shufflon protein (Appendix J), a genetic region involved in conjugation of IncI1 plasmids [351].

Illumina WGS of two chromosomal mediated AmpC-producing *E. coli* identified additional ARGs which potentially confer resistance to aminoglycoside, trimethoprim, sulfonamide and tetracycline antibiotics (trimethoprim and sulfonamide resistance was not phenotypically confirmed) as well as plasmid incompatibility factors. These findings suggest that these isolates harbour additional ARGs, some of which are likely encoded on plasmids due to the co-location of ARGs (aph(3")-Ib, aph(6)-Id, sul2) and plasmid incompatibility factors (IncQ1) on the same contig and therefore could potentially spread via HGT. IncQ is a group of mobilisable plasmids with a broad host range that are not often reported in

the literature [47]. Interestingly, IncQ plasmids harbouring the sul2-strA-strB gene cluster (synonyms for aph(3")-Ib and aph(6)-Id, respectively) were detected in S. enterica servar Typhimurium phage type 9 isolates from bovine and human sources in Australia [352]. The plasmids and ARGs in isolates DF0031.1c and DF0047c could be further elucidated by the addition of long-read sequencing data which would enable a hybrid assembly to be generated, allowing for a high resolution analysis and clarification of this issue.

VirulenceFinder was used to detect virulence genes carried on the plasmids. The five IncI1 plasmids harboured virulence genes encoding colicins, namely *cia* (n=4, *bla*<sub>CMY-2</sub> positive plasmids) and *cib* (n=1, pDF0102.4g\_1). Plasmid pDF0102.4g\_2 harboured the *traT* gene, which encodes an outer membrane protein involved in complement resistance, although the *traT* gene is also a part of the transfer operon in conjugative plasmids and is involved in surface exclusion between identical or closely related plasmids by preventing stable mating aggregates [353]. Plasmids pDF0059.2e\_2 and pDF0059.2e\_3 did not carry any known virulence genes.



Figure 3.6: Representative IncI1 complete circular plasmid pDF0049.2e\_1 and IncFI-A/IncQ1 plasmid pDF0059.2e\_3 assembled from MinION and Illumina sequencing data showing physical linkages of antimicrobial resistance genes and mobile elements. The *tra* genes are involved in the conjugal transfer system.

Isolate	Plasmid	Size (bp)	pMLST	Inc group	$CDS^{a}$	ARGs <sup>b</sup>	Partial ARGs <sup>c</sup>
DF0049.2e	$pDF0049.2e_1$	89,859	23	Incl1	102	bla <sub>CMY-2</sub>	None
DF0059.2e	$pDF0059.2e_1$	94,357	23	Incl1	109	blaCMY-2	None
DF0059.2e	$pDF0059.2e_2$	97,734	1	IncY	108	None	None
DF0059.2e	$p\mathrm{DF0059.2e}_{-3}$	244,307	[F-:A8:B-]	IncFIA/IncQ1	277	aac(3)-IId, $aadA5$ , $aph(3")$ -Ib, aph(3")-Ia, $aph(6)$ -Id, $catA1$ , $bla_{CTX-M-1}$ , $dfrA17$ , $mph(A)$ , sul1, $sul2$ , $tet(B)$	aadA1 (17.2%, 96.0%) catB4 (19.5%, 100%) blaTEM-105 (31.8%, 100%)
DF0102.4g	$pDF0102.4g_1$	97,225	26	IncI1	109	$bla_{OXA-1}, \ sull, \ aadAI$	aadA1 (17.4%, 97.1%) catB4 (19.5%, 100%)
DF0102.4g	$pDF0102.4g_2$	58,773	[F35:A-:B-]	$\mathrm{IncI1}/\mathrm{IncFII}$	64	None	None
DF0159.2g	$pDF0159.2g_1$	89,920	23	IncI1	101	bla <sub>CMY-2</sub>	None
DF0181.1c	$pDF0181.1c_1$	90,151	23	IncI1	103	blacMY-2	None
DF0183e	$pDF0183e_1$	92,691	23	IncI1	106	$bla_{ m CMY-2}$	None
<sup>a</sup> CDS, Codi <sup>b</sup> ARGs, An	ing sequence. timicrobial resista	nce genes def	ined in Appendix I				

Γ

Toddy Η ) IO

<sup>c</sup> Partial ARGs, ARGs in which the coverage is less than 80%. Numbers in brackets represent the percentage of coverage and identity, respectively.

Plasmid	$\mathrm{pDF0049.2e}\_1$	$pDF0059.2e\_1$	$\mathrm{pDF0159.2g\_1}$	pDF0181.1c_1	pDF0183e_1
pDF0049.2e_1	0	26	1	9	9
$\rm pDF0059.2e\_1$	26	0	25	19	19
pDF0159.2g_1	1	25	0	8	8
pDF0181.1c_1	9	19	8	0	0
$pDF0183e_1$	9	19	8	0	0

Table 3.8: Core genome single nucleotide polymorphism analysis of IncI1 plasmids carrying the  $bla_{\rm CMY-2}$  gene identified in this study

# 3.3.7 Prevalence of AmpC- and ESBL-producing *E. coli* from farm environmental samples

Pooled faecal samples were investigated for AmpC- and ESBL-producing *E. coli* (Table 3.9). One out of 60 (1.7%) pooled faecal and one out of 15 (6.7%) FDE samples from Dairy 4 were positive for ESBL-producing *E. coli*, with one and two isolates cultured from these samples, respectively. No ESBL-producing *E. coli* were isolated from pooled faeces and FDE from Dairy 1 nor from soil or milk samples from either farm. AmpC-producing *E. coli* were isolated from faeces (2 of 60, 3.3%; 5 of 60, 8.3%) and FDE (5 of 13, 38.5%; 1 of 15, 6.7%) on Dairy 1 and Dairy 4, respectively and none were isolated from soil (0 of 30) or milk (0 of 26) from either farm. The sample level prevalence of AmpC-producing *E. coli* isolated from Dairy 1 and Dairy 4 (p = 0.526) and between sample types (faeces or FDE; p = 0.408) was not statistically significant. Across the study, multiple isolates were obtained from the same sample on the selective agar plates and some of which were examined in more detail to determine if they were clonal.

Farm	$\begin{array}{c} \mathbf{Sample} \\ \mathbf{type}^{\mathbf{a}} \end{array}$	ESBL-producing	E. coli <sup>b</sup>	AmpC-producing	E. coli
		No. samples $(\%)$	No. isolates	No. samples $(\%)$	No. isolates
Dairy 1	FDE	0/13	0	5/13 (38.5%)	14
	Faeces	0/60	0	2/60 (3.3%)	6
	Milk	0/13	0	0/13	0
	Soil	0/15	0	0/15	0
Dairy 4	FDE	1/15~(6.7%)	2	1/15~(6.7%)	4
	Faeces	1/60~(1.7%)	1	5/60 (8.3%)	22
	Milk	0/13	0	0/13	0
	Soil	0/15	0	0/15	0

Table 3.9: Number of positive AmpC- and ESBL-producing E. coli samples and isolates

<sup>a</sup> FDE, Farm dairy effluent.

 $^{\rm b}$  One isolate was both AmpC and ESBL positive (DF0059.2e) and has been included in both columns.

Low numbers of ESBL-producing *E. coli* were isolated from Dairy 4 and none were isolated from Dairy 1. Dairy 4 had a higher prevalence of AmpC-producing *E. coli* from faeces (8.3%), whereas Dairy 1 had a higher prevalence in FDE (38.5%). The effluent management strategy on Dairy 1 changed during the study period (section 3.1) and compared to the first collection point where four samples were positive for AmpC-producing *E. coli* (4 of 7; 57.1%), only one sample (1 of 6; 16.7%) from the second collection point was positive.

## 3.3.8 Estimated antimicrobial use on dairy farms between October 2018 and December 2019

Between October 2018 and December 2019 the estimated AMU on Dairy 1 and Dairy 4 was 17.09 mg/PCU and 5.36 mg/PCU, respectively (Appendices K and L). The antimicrobial products used on Dairy 1 and Dairy 4 are detailed in Appendix M. The predominant route of antibiotic administration for both Dairy 1 and Dairy 4 was parenteral other/other (Appendix N). Months with higher AMU are consistent with spring calving in NZ. The NZVA has classified antimicrobials as green, yellow or red tier according to the World Health Organisation classes [42]. Of the total AMU used on Dairy 1 and Dairy 4, the majority of antimicrobials were green tier (91.1% and 30.5%), followed by yellow tier (6.2% and 67.9%) and red tier antimicrobials were infrequently used (2.75% and 1.6%), respectively.

Antimicrobial class	Dairy 1 <sup>a</sup> (mg/PCU)	Dairy 4 (mg/PCU)
Aminoglycosides	0.68~(4.0%)	0.19~(3.5%)
Aminopenicillins	0.05~(0.3%)	0.02~(0.4%)
First-generation cephalosporins	0.23~(1.3%)	2.83~(52.9%)
Third-generation cephalosporins	0.12~(0.7%)	0.04~(0.8%)
Penicillins	$15.50\ (90.7\%)$	2.18 (40.7%)
Quinolones	0.35~(2.1%)	0.03~(0.6%)
Tetracyclines	0.17~(1.0%)	0.04~(0.7%)
Macrolides	0	0.02~(0.4%)
Multiple classes	0	$< 0.01 \ (0.02\%)$

Table 3.10: Amount of the antimic robial classes used on Dairy 1 and Dairy 4 between October 2018 and December 2019

<sup>a</sup> mg/PCU, mg per active ingredient per population correction unit (% total PCU).

As shown in Figure 3.7, AmpC-producing *E. coli* were isolated after months of both high and low AMU. On Dairy 1, the highest AMU occurred in October 2018 (12.3%) and August to November 2019 (8.7 - 18.6%); whereas on Dairy 4 the highest AMU predominantly occurred in September 2019 (56.7%) (Figure 3.7 and Appendices K - L). AMU treatments varied between farms, with the predominant reasons for treatment on Dairy 1 including clinical or subclinical mastitis (65.5%), DCT (9.7%) and between claw/footrot (8.7%). On Dairy 1, 70 cows were treated for mastitis during the study period and half of the cows (35 of 70; 50%) received treatment on multiple occasions which may account for the higher AMU related to mastitis. In comparison, less than 20% of cows on Dairy 4 were treated multiple times for mastitis (7 of 36; 19.4%). The predominant reasons for treatment on Dairy 4 was for a metri-check (used for early identification of cases of endometritis; 35.2%), clinical or subclinical mastitis (22.5%) and DCT (19.7%). Further work is required, including comparing antimicrobial sales data and discussions with the farmers and veterinarians, to further understand the AMU patterns on farm.



Figure 3.7: (A) Collection dates of samples positive for AmpC- and/or ESBL-producing E. *coli* over the 15 month study period. The phenotype is represented by shape as indicated in the figure legend. (B) Antimicrobial use per month (mg active ingredient per population correction unit) on Dairy 1 and Dairy 4 during the 15 month study period.

#### 3.4 Discussion

The sample level prevalence of ESBL-producing  $E. \ coli$  from faeces and FDE was low on Dairy 4 and they were not detected on Dairy 1 (Table 3.9). These results are consistent with a previous regional-based cross-sectional study of NZ dairy farms, which found a low prevalence of ESBL-producing *E. coli* in pooled faecal samples and no *E. coli* with pAmpC genes were identified [25]. Similarly, a nationwide cross-sectional study did not detect any ESBL- or plasmid-mediated AmpC-producing *E. coli* [26]. Overseas studies have found a higher prevalence of AmpC- and/or ESBL-producing *E. coli* from dairy farms, with herd level prevalence estimates ranging from 13% on organic dairy farms in the Netherlands [192] to 5.2 - 86.7% on conventional farms [7, 200, 6, 199, 8]. A large study across 53 UK dairy farms sampled a range of faecal contaminated environments and pastureland and found the sample level prevalence to be 5.4% (224 of 4,145) for  $bla_{\rm CTX-M}$  positive *E. coli* [195]. In this study, ESBL-producing *E. coli* were isolated from faeces and FDE samples collected in February and December 2019 respectively, which is during summer in NZ. A recent study suggested that lower monthly ambient temperatures were associated with a lower odds of identifying  $bla_{\rm CTX-M}$  positive *E. coli* samples [195]. Despite this finding, the small number of samples (n=2) positive for ESBL-producing *E. coli* in this study does not allow for any associations between seasonality and ESBL positive samples to be observed.

The sample level prevalence of AmpC-producing E. coli in this study was also relatively low (Table 3.9). AmpC-producing E. coli were isolated after months of both high and low AMU, highlighting that additional factors other than total AMU may also play a role in the development and transmission of antimicrobial resistant bacteria in the dairy farm environment. E. coli with pAmpC genes were infrequently identified on Dairy 1 (1 of 18; 5.6%) and all AmpC-producing E. coli isolated from Dairy 4 were plasmid-mediated (27 of 27; 100%;  $bla_{\rm CMY-2}$ ). This suggests that although Dairy 1 had a higher sample level prevalence of AmpC-producing E. coli, particularly in FDE, these putative AmpC hyperproducers predominantly spread by vertical transmission. This is less concerning compared to resistance genes located on mobile genetic elements that can spread via HGT. The differences in sample prevalence from FDE may be due to different effluent management strategies between farms, with Dairy 4 storing FDE in a large pond prior to spraying onto paddocks which likely results in the dilution of bacteria. Univariable analysis on British beef farms found that spreading of farm manure was significantly associated with an animal testing positive for AmpC-producing E. coli [284]. The FDE management strategy on Dairy 1 changed during the study period and therefore two FDE sample collection points were used (Table 3.1). On Dairy 1, the majority of the AmpC-producing *E. coli* were isolated from samples collected from the first sampling site (4 of 5 AmpC positive samples; 80%), which were collected from the FDE sump that is smaller in size and may allow for concentration of bacteria. In addition, there may have been pump issues feeding into the sump. On Dairy 1, raw FDE is not applied to paddocks which reduces the risk of transmission of these putative AmpC hyperproducers within the farm environment.

Of the of AmpC-producing *E. coli* isolated during this study (n=46), 18 (39.1%) were putative AmpC hyperproducers. A study of 2,304 extended-spectrum cephalosporin resistant *E. coli* from livestock in the Netherlands found that mutations in the promoter region of the *ampC* gene accounted for a small proportion of the total resistant *E. coli* (217 of 2,034; 9.4%) [354]. However the proportion of putative *ampC* hyperproducing *E. coli* may depend on the study population as a higher proportion  $(46.2\% \text{ of cefotaxime-resistant } E. \ coli)$  was identified across 53 farms in the UK [156, 195]. Higher use of amoxicillin/clavulanate and sampling of faecal samples from the environment of young calves has been associated with an increased risk of identifying putative ampC hyperproducers on dairy farms [156] and in vitro studies have shown an association between amoxicillin use and AmpC-producing E. *coli* arising from mutations in the promoter region of the ampC gene [277, 355]. Amoxicillin/clavulanate are classified as yellow tier antimicrobials by the NZVA, indicating their use should be restricted in NZ veterinary practices [42]. Amoxicillin/clavulanate was not used on Dairy 1 and Dairy 4 during the study period and this study did not include samples from the calf environment. Cephalosporin use (including third-generation cephalosporins) was not identified as a risk factor for putative ampC hyperproducing E. coli [156] and third-generation cephalosporin use on Dairy 1 and Dairy 4 was very low during the 15 month study period (<1% total mg/PCU). The lack of known risk factors for putative AmpC hyperproducers on Dairy 1 highlights the complexity of factors involved in the development of AMR and suggests additional studies are required to identify risk factors for AmpC hyperproducers, particularly in pasture-based dairy farms.

Infections caused by AmpC/ESBL positive *Enterobacteriaceae* pose significant treatment option challenges for clinicians, with cefepime (a fourth-generation cephalosporin) or carbapenems being suggested as the main treatment options [356], both of which are critically important antimicrobials for human medicine [98]. The co-occurrence of plasmid-mediated AmpC and ESBL positive *Enterobacterales* has been reported to be relatively uncommon in clinical studies [356], although the prevalence likely differs between study populations. For example, a higher proportion of ESBL-producing *E. coli* and *Klebsiella* spp. were also AmpC positive (30%; 69 of 230) from human clinical samples in Iran [357]. One E. coli isolated from faeces in this study (1.9%, 1 of 52) was both AmpC and ESBL positive, carrying the  $bla_{\rm CMY-2}$  and  $bla_{\rm CTX-M-1}$  genes on two distinct plasmids (Table 3.7). A low proportion of *E. coli* isolates displayed an AmpC/ESBL phenotype in a study of dairy and beef cattle and sheep farms in Spain (5.2%, 7 of 135 isolates) [9]. One AmpC/ESBL positive isolate was sequenced in the aforementioned study, and this isolate co-harboured the  $bla_{\text{CTX-M-14}}$  gene and a mutation in the ampC gene promoter region, confirming the AmpC/ESBL phenotype [9]. A cross-sectional study of dairy farms in Canada (n=102) also found a low proportion of AmpC/ESBL positive E. coli (2%) in comparison to AmpC(51%) and ESBL (46%) phenotypes [194]. These findings suggest that E. coli displaying both an AmpC and ESBL phenotype are infrequently isolated from dairy farm environments and that the AmpC/ESBL genetic determinants co-harboured by such E. coli may differ.

The ESBL phenotype for isolate DF0059.2e was not detected in the initial ESBL confirmation tests which is likely due to the production of AmpC  $\beta$ -lactamases which is less affected by ESBL inhibitors such as clavulanic acid. The ESBL gene was identified using WGS, and the subsequent ESBL phenotype confirmed using cefepime (fourth-generation cephalosporin) which is stable to AmpC hydrolysis. This highlights the importance of using both culture-based and culture-independent methods in parallel, in particular using genotypic methods to confirm a resistant phenotype and vice versa. These findings suggest that although relatively uncommon in dairy farm environments, additional ESBL confirmation tests using a fourth-generation cephalosporin may be required for *E. coli* with plasmid-mediated AmpC  $\beta$ -lactamases which may mask ESBL production in confirmation tests that rely on third-generation cephalosporin and clavulanic acid combinations.

AmpC- and ESBL-producing *E. coli* were not detected in soil or bulk tank milk samples on either farm. Despite a small sample size, the lack of detection from these matrices over a 15 month period indicates that they are less likely sources of AmpC- and ESBL-producing E. *coli* in the NZ dairy farm environment, whereas faeces and FDE are more probable sources. At the sample level,  $bla_{\text{CTX-M}}$  or  $bla_{\text{CMY-2}}$  positive *E. coli* were infrequently detected from soil samples across 17 commercial beef farms in the USA (3.89%; 3 of 77) [174]. The prevalence of AmpC- and ESBL-E from bulk tank milk was more varied. A study of bulk tank milk samples representing 100 dairy farms in Switzerland found no ESBL-E positive samples (0%; 0 of 100) [207], and bulk tank milk samples from conventional and organic dairy farms in the Czech Republic found that 0.7% of E. coli (2 of 270) displayed an ESBL phenotype [212]. Some studies have reported ESBL-E at a higher prevalence in bulk tank milk; analysis of a large number of bulk tank milk samples in Germany found 9.5% (82 of 866) were positive for ESBL-E [209], 8.75% (7 of 80) bulk tank milk samples were positive for ESBL-producing K. pneumoniae in Indonesia [213] and 16.9% (12 of 71) of milk samples from individual healthy cows were positive for ESBL-producing E. coli in Malaysia [208]. In the Malaysian study, 229 faecal samples collected from cattle of various ages across ten farms identified only one faecal sample (1 of 229; 0.4%) which was positive for ESBL-producing *E. coli*, indicating a higher prevalence in milk samples compared to faeces in this study population. The increased detection of ESBL-producing E. *coli* from milk than faeces is unexpected, and may be due to environmental contamination on the udder prior to milk collection. In addition, the milk samples analysed were handmilked from individual lactating cows [208], whereas bulk tank milk is often used in other studies. Kamaruzzaman et al., [208] also isolated ESBL-producing E. coli from the farm environment including drinking water, house flies and water sources but not floor, feed or water trough swabs or supplementary feed. A recent study analysing raw bulk tank milk from dairy farms (n=195) in Northern China found 34.4% (67 of 195) of milk samples were positive for E. coli and  $\beta$ -lactamase genes were identified in 34.3% of E. coli, specifically the  $bla_{\rm SHV}$  (1.5%),  $bla_{\rm TEM}$  (20.9%),  $bla_{\rm CMY}$  (10.4%) and  $bla_{\rm CTX-M}$  (1.5%) genes [211], however only a limited number of  $\beta$ -lactamase resistance genes were tested. In addition, E. coli with the  $bla_{\rm CMY}$  and  $bla_{\rm CTX-M}$  genes were not phenotypically confirmed as AmpCor ESBL-producing *E. coli*, respectively. The lack of detection of ESBL-E in bulk tank milk in NZ is not unexpected due to the stringent hygiene and food safety standards for dairy farming and milk storage in NZ [358].

It is important to consider variations in study design (sample size and animal age/health status), sample matrices and culture selection methods when comparing between prevalence

studies. For example, this study did not use a pre-enrichment step for AmpC- and ESBLproducing E. coli prior to plating on selective agar, which is a technique used in some prevalence studies. The age of the study population is also a crucial factor to consider. A study of 101 dairy farms in Canada detected AmpC- or ESBL-producing E. coli at least once during the study in 85% of farms, although the majority were isolates from calves [194]. A longitudinal study of a dairy farm previously identified as ESBL positive in the UK found a higher proportion of E. coli positive for the  $bla_{\text{CTX-M}}$  gene in milking cows (30.3%) compared to non-milking cows (3.0%) [198], although this study did not look for other ESBL enzyme types. Antimicrobial use, including products containing third- and fourth-generation cephalosporins, was higher in the site used to house the lactating and calving cows, however, other management factors also differed between the cattle groups studied including location on the farm [198]. This study focused only on adult dairy cattle and did not examine the prevalence in young calves or non-milking cows. It is hypothesised that a higher prevalence of AmpC- and ESBL-producing E. coli would be present in faces from young calves compared to adult dairy cows. A study analysing recto-anal mucosal swabs from waste milk fed dairy calves (n=40) in NZ did not identify any ESBL-producing E. coli and 25% of enrichments (10 of 40) were positive for AmpC-producing E. coli, all of which were putative AmpC hyperproducers [27]. Given the higher prevalence of AmpCand ESBL-E in young calves globally, this could be an area of further research in NZ.

Variation in farming systems between countries is also important when comparing studies. Intensive farming systems, particularly indoor housing, have been associated with a higher prevalence of mastitis [269], which can lead to higher AMU and subsequently increased levels of AMR. It has been proposed that the NZ pasture-based farming system, in conjunction with low AMU in food-producing animals, may contribute to lower levels of AMR [28]. The low sample level prevalence of AmpC- and ESBL-producing *E. coli* from Dairy 1 and Dairy 4 supported this hypothesis. A cross-sectional study of ruminant farming systems in Spain found beef cattle herds and sheep flocks, which have a semi-intensive pasture-based farming system, had significantly lower herd level prevalence of cefotaximeresistant E. coli compared to dairy cattle, which are generally housed inside pens [9]. Similarly, a cross-sectional study of 60 beef and 52 dairy cattle herds in Germany found a lower sample level prevalence of cefotaxime-resistant E. coli in beef cattle (35%, 161 of 455) compared to dairy cattle (48%, 156 of 323) [190]. Factors linked to less intensive farming practices were associated with fewer positive samples for both beef and dairy cattle [190]. A cross-sectional study of grazing beef cattle farms (n=17) in the USA found that larger farming operations (>500 cattle) were associated with a 58% higher likelihood of detecting cefotaxime resistant bacteria from faecal samples [275]. Interestingly, soil, water (ponds and water troughs) and forage samples had a higher prevalence of cefotaxime resistant bacteria compared to faecal samples, although bacteria isolated from MC agar containing  $4 \ \mu g/mL$  cefotaxime were classified as cefotaxime resistant bacteria and the cefotaxime minimum inhibitory concentration of only 87 of 3,175 isolates was confirmed [275]. In addition, the bacteria were not identified to the species level [275], therefore a high abundance of bacteria with intrinsic resistance to cefotaxime, such as M. morganii and S. marcescens which have an inducible chromosomal AmpC  $\beta$ -lactamase [134], may have accounted for the higher prevalence from environmental samples. Another study found samples collected from pasture (collected by traversing pasture areas where animals were grazing or publicly accessible farmland with sterile over-boot socks) had a lower odds of *E. coli* positive for  $bla_{\rm CTX-M}$ , compared to samples collected from the rest of the farm environment (collected by traversing various faeces contaminated environments of lactating cows, dry cows, heifers and calves with sterile over-boot socks) [195].

Both farms in this present study operated a closed herd system, which may reduce the risk of introductions of antimicrobial resistant bacteria into the herd by not introducing animals from outside sources to the farm. For example, on British beef farms, buying bulls or fattening cattle have been identified as risk factors for  $bla_{CTX-M}$  positive *Enterobacteriaceae* or AmpC-producing *E. coli*, respectively [284]. However, the impact that introductions of antimicrobial resistant bacteria has compared to AMR selection in agricultural environments is unknown.

Interestingly, Dairy 1 had a higher total AMU during the study period, yet no ESBLproducing E. coli were detected on Dairy 1 (Appendices K and L). However, a significant proportion of the total AMU on Dairy 1 was classified as green tier antimicrobials by the NZVA with Dairy 4 using a higher proportion of yellow tier antimicrobials. The variation, predominantly green on Dairy 1 and yellow tier classes on Dairy 4, is likely associated with the main treatments per farm. For example, Dairy 4 had a higher proportion of endometritis treatments in which cephapirin (a first-generation cephalosporin) was used (yellow tier; Appendix M). Other factors may have an impact on the AMU between farms, such as the average age of the cows as the incidence of clinical mastitis is higher in older cows [359]. The estimated use of antimicrobials in all food-producing animals in NZ in 2018 was 10.21 mg/PCU [17]. The representative total AMU during the study period on Dairy 1 and Dairy 4 was slightly higher (17.09 mg/PCU) and lower (5.36 mg/PCU) than this estimate, respectively (Figure 3.7). The AMU on D1 and D4 was also within the range reported in a cross-sectional study of 26 dairy farms across NZ (4.39 - 20.92 mg/PCU) [26], albeit at the higher and lower end of the spectrum, respectively. Although the total AMU was estimated using sales data in the aforementioned study, individual antimicrobial treatments were used for calculating the total AMU in this study, which makes comparisons difficult. The use of third-generation cephalosporins (<1% total mg/PCU), which have been identified as a risk factor for ESBL-E [7], and NZVA red tier classified antimicrobials was very low on both farms during the study period (Table 3.10 and Appendices M - L). AMU between farms likely differs due to a number of management factors including average age of the herd, disease outbreaks, hygiene practices, the use of teat sealants as well as the farmers perception towards antimicrobial stewardship.

Multiple distinct *E. coli* clones were isolated from one composite faecal sample (DF0181.1) and one FDE sample (DF0183), both collected in December 2019 from Dairy 4. From sample DF0181.1, one *E. coli* isolate had a different AST profile compared to the other

isolates (1 of 6 *E. coli*) and was susceptible to tetracycline and streptomycin (Appendix C), suggesting that two distinct *E. coli* clones were isolated from this sample. Two ESBL- and four AmpC-producing *E. coli* were isolated from the FDE sample DF0183. However, the AST profile of one AmpC-producing *E. coli* differed compared to the other *E. coli* (n=3) as this strain was susceptible to tetracycline and streptomycin whereas the other strains were resistant to these antimicrobials, indicating three distinct *E. coli* clones were isolated from sample DF0183. These findings highlight the importance of using multiple selective agar plates to enrich and isolate AmpC and ESBL-producing *E. coli* and suggests that multiple colonies from each agar plate should be isolated and analysed in prevalence studies. Analysing only one presumptive AmpC- and ESBL-E per sample may underestimate the true prevalence.

The AmpC- and ESBL-producing E. coli which were sequenced in this study belonged to a diverse range of STs and serotypes. Similar findings have been reported for  $bla_{\rm CMY-2}$ positive E. coli from human clinical cases, livestock and food matrices in which a diverse range of STs were reported [360]. In contrast E. coli ST131, which are frequently multi-drug resistant and harbour bla<sub>CTX-M</sub> genes, are widely disseminated in humans globally [361, 362]. The two sequenced putative AmpC hyperproducers belonged to ST88 (phylogroup C), which is consistent with previous findings that AmpC hyperproducers predominantly belong to this sequence type [156, 203]. In contrast, three putative AmpC hyperproducing E. coli isolated from dairy cattle in NZ belonged to ST1148 (n=2) and ST298 (n=1) [25]. In comparison to previous studies in which AmpC hyperproducers often belonged to E. coli phylogroup A [363, 157], the 18 E. coli in this study with mutations in the promoter region of ampC gene were phylogroup B1 (n=1) or C (n=17). AmpC hyperproducing E. *coli* belonging to phylogroups B1 and C have also been reported from faecal samples on dairy farms in the UK [156]. Identical mutations in the promoter region of the ampC gene to those seen in the AmpC-producing E. coli in this study (n=18) have been previously described [348, 138]. The mutations at positions -42 and -18 change the secondary structure of the -35 and -10 boxes, respectively and increase the separation distance between the two boxes to 17 bp, resulting in an alternative displaced stronger promoter and thus putative AmpC  $\beta$ -lactamase hyperproduction and increased resistance [140, 139]. The mutation at the -42 position  $(C \rightarrow T)$  has been shown to be homoplastic and was identified in 24 E. coli across five distinct phylogroups [364]. This mutation was also significantly associated with cefotaxime resistance and almost exclusively identified in the absence of plasmid-mediated ampC genes [364, 139]. The two mutations at positions -1 and +58 are commonly found together, and have been proposed to represent strain-specific polymorphisms [139].

*E. coli* which have been phenotypically confirmed as AmpC positive should undergo further molecular characterisation to deduce whether this phenotype is a result of a mutation in the promoter region of the ampC gene (chromosomal) or plasmid-mediated ampC genes, which has implications for the dissemination of AMR. Cefoxitin resistance combined with resistance to cefotaxime and/or ceftazidime is recommended for the detection of AmpC-producing *Enterobacteriaceae* [170]. In this study, differences in the AST zone

sizes (mm) for cefoxitin and cefotaxime differed between putative AmpC hyperproducers and plasmid-mediated AmpC-producing E. coli, however the sample size is small which limits any statistical comparisons. These results suggest that cefoxitin and/or cefotaxime zone size may be a useful preliminary method to distinguish between putative AmpC hyperproducers and plasmid-mediated AmpC-producing E. coli, followed by subsequent molecular confirmation using PCR and sequencing. However, the resistance spectrum of an isolate is complex and other factors such as porins and efflux pumps will contribute to the level of resistance [131]. A study of clinical AmpC-producing E. coli isolated from a French hospital found that phenotypic testing (broth microdilution and minimum inhibitory concentration) could not distinguish between putative AmpC hyperproducers and plasmid-mediated AmpC-producing E. coli. However discrepancies in putative AmpC hyperproducers susceptibility to some  $\beta$ -lactame varied, which may be due to differences in mutations in the ampC gene promoter region and the varying effects on promoter strength [132]. Despite the small sample size in this study (n=14 distinct AmpC-producing E. coli clones), these results warrant further investigation of the cefoxitin and cefotaxime zone sizes of a larger number of AmpC-producing E. coli isolated from various sources and a range of resistance mechanisms (including different mutations in the ampC gene promoter region as well as numerous pAmpC types).

ESBL-producing *E. coli* were isolated from two samples in this study (DF0059.2e and DF0183). Isolate DF0183g, belonging to ST4553, harboured a chromosomally encoded  $bla_{\rm CTX-M-15}$  and ampC gene (no mutations in the promoter region), but did not harbour any additional ARGs. Four ST4553 *E. coli* positive for the  $bla_{\rm CTX-M-15}$  gene have been detected in dog faeces (n=1) [365] and storm water (n=3) in NZ [366]. Three isolates did not harbour any other ARGs, and one isolate from storm water co-harboured the  $bla_{\rm CMY-2}$  gene. *E. coli* DF0059.2e belongs to ST5135 and to the best of our knowledge, this is the first *E. coli* ST5135  $bla_{\rm CTX-M-1}$  to be reported. Few *E. coli* ST5135 have been reported on Enterobase, and those identified were isolated from human, livestock and poultry samples (accessed 27th July 2021). *E. coli* ST57 have been reported from chickens and pigs ( $bla_{\rm CMY-2}$  positive) [360] and from veal calves ( $bla_{\rm CTX-M-14}$  positive) [309]. The remaining STs identified in this study (ST56, ST442, ST973, ST2541) have been infrequently associated with AmpC or ESBL-producing *E. coli* in the published literature.

It is important to assess the virulence potential of AmpC- and ESBL-producing  $E.\ coli$  to ascertain the public and animal health risk posed. The majority of the virulence factors identified in the AmpC- and ESBL-producing  $E.\ coli$  sequenced in this study were involved in adhesion and protection/serum resistance (Figure 3.4). Two isolates (DF0031.1c and DF0047c) carried more virulence factors compared to the other  $E.\ coli$  sequenced and the remaining strains carried much fewer virulence genes (4 - 13; average 9). Consistent with this study, genes associated with the diarrheagenic  $E.\ coli$  pathotypes EIEC, ETEC (toxin related genes) and STEC have not been identified in other studies investigating AmpCand ESBL-producing  $E.\ coli$  isolated from food animals [117], suggesting these pathotypes infrequently confer AmpC and ESBL phenotypes in healthy adult cattle. Five E. coli sequenced in this study were identified as APEC, harbouring a number of virulence factors associated with this E. coli pathotype. According to core genome SNP analysis, these five strains formed two clusters: (i) DF0031.1c and DF0047c and (ii) DF0181.1c, DF0181.3c and DF0183e (Figure 3.3). Unsurprisingly, within these clusters the strains are epidemiologically related, genetically similar and are identical STs and genotypes (Table 3.5). These five *E. coli* strains were all serogroup O8, which has been frequently detected in APEC strains [367]. However one O serogroup may be associated with multiple E. coli pathotypes [72]. The assignment of strains to specific E. coli pathotypes can also be complicated due to the genetic fluidity of *E. coli*, their ability to readily acquire new genetic material via HGT and some genetic markers for specific pathotypes being non-exclusive [81]. Despite APEC being associated with avian sources, these E. coli were isolated from bovine faces (n=3) and FDE (n=2). A global comparison of AmpC- and ESBL-producing E. coli from healthy food animals (broiler chickens, beef cattle and pigs) across Europe found that of the 99 APEC identified, 19.6% were from non-avian sources and consisted of a diverse range of STs [117]. Plasmids harbouring the  $bla_{\rm CMY-2}$  gene were reconstructed for three APEC strains in this study (pDF0181.1c\_1, pDF0181.3c\_1 and pDF0183e\_1) and no APEC associated virulence genes were identified on these plasmids. The remaining two APEC strains were putative AmpC hyperproducers and hybrid assemblies were not generated, although plasmid incompatibility factors IncFIB, IncFII and IncQ1 were detected in these whole genome sequencing data.

Bacteria with the  $bla_{\rm CMY}$  and  $bla_{\rm CTX-M}$  genes have been isolated from migratory and non-migratory birds on Ohio dairy farms, suggesting that birds may represent vectors for transmission of third-generation cephalosporin resistant *Enterobacteriaceae* [368]. It is plausible that birds may also be vectors for pathogenic and/or antimicrobial resistant *E. coli* on NZ dairy farms. Bird faecal samples were collected from the two dairy farms in this study, however, the samples have not yet undergone culture-based analysis for AmpC- and ESBL-E. These enrichments could be examined to determine whether birds on NZ dairy farms may be a vector of AmpC- or ESBL-E in NZ dairy farm environments.

The resistance mechanism for third-generation cephalosporins is currently unassigned in four *E. coli* (DF0102.4e-h) as they were both AmpC and ESBL negative. The *E. coli* (DF0102.4e-h) were isolated from a composite faecal enrichment on MC agar with 1 µg/mL cefotaxime (DF0102.4e-f) and CHROMagar<sup>TM</sup> ESBL plates (DF0102.4g-h). These *E. coli* were resistant to cefotaxime, cefpodoxime and streptomycin and intermediate to cefoxitin (Table 3.6). Resistance to cefepime was confirmed in isolates DF0102.4f and DF0102.4h, but resistance to cefepime has not been assessed for DF0102.4e or DF0102.4g. *E. coli* DF0102.4g underwent whole genome sequencing using both Illumina and MinION sequencing methods. Two plasmids were identified and one plasmid harboured the  $bla_{OXA-1}$ , sul1and aadA1 genes as well as partial copies of the aadA1 and catB4 gene (Table 3.7) and this plasmid was predicted to be mobile. The  $bla_{OXA-1}$  gene encodes a narrow-spectrum  $\beta$ -lactamase which traditionally confers resistance to aminopenicillins, carboxypenicillins

and ureidopeniclins [369], however overexpression of this gene has been shown to confer reduced susceptibility to fourth-generation cephalosporins such as cefepime and susceptibility to cefotaxime and ceftazidime when coupled with porin loss (OmpC and/or OmpF) [370]. In addition, other resistance mechanisms such as decreased expression of porins, overexpression of efflux pumps, mutations in the marR gene and the presence of the  $\beta$ -lactamase TEM-1 may also be involved in conferring resistance to third- and/or fourth-generation cephalosporins among Enterobacteriaceae [369, 156]. Three AmpC-producing E. coli isolated from UK dairy farms also carried the  $bla_{OXA-1}$  gene and showed reduced susceptibility to cefepime [156]. Salmonella enterica (serovar Typhimurium) isolated from human clinical samples in Kenya, Malawi and Ireland carried the  $bla_{OXA-1}$  gene and were shown to have elevated cefepime MIC values [371]. OXA-1-like  $\beta$ -lactamase enzymes do not confer resistance to ceftazidime [153] and no third-generation cephalosporin resistant E. coli were isolated from MC agar with 1  $\mu$ g/mL ceftazidime from the faecal enrichment DF0102.4, compared to E. coli isolated from both MC agar with 1  $\mu$ g/mL cefotaxime (n=2) and CHROMagar<sup>TM</sup> ESBL plates (n=2). This further strengthens the hypothesis that the overexpression of the  $bla_{OXA-1}$  gene and/or porin modification is involved in the resistance of E. coli isolates DF0102.4e-h to third- and fourth-generation cephalosporins. Further work is required to identify the resistance mechanism of  $E. \ coli$  DF0102.4g to third- and fourthgeneration cephalosporins (confirmation of resistance to fourth-generation cephalosporins is still required). Possible resistance mechanisms warranting further investigation include the overexpression of the  $bla_{OXA-1}$  gene coupled with porin loss assessed using expression studies and proteomics, respectively or modification and mutations in other genes such as the mar genes. The identification of the mechanism conferring resistance to third- and fourth-generation cephalosporins within an isolate is essential for AMR surveillance and risk assessment as well as for epidemiological studies.

AmpC-producing *E. coli* were not routinely screened for reduced susceptibility to cefepime in this study, therefore, further work is required to investigate whether any AmpCproducing *E. coli* isolated in this study produced ESAC  $\beta$ -lactamases by amplifying and sequencing the *ampC* gene to identify putative ESAC associated AA changes [154] and/or performing cloning experiments where feasible [203]. In particular, any ESBL negative *E. coli* isolated from CHROMagar<sup>TM</sup> ESBL plates (e.g. DF0159.2g) should be investigated as ESAC  $\beta$ -lactamase activity may enable these isolates to grow on this medium. The exact composition of CHROMagar<sup>TM</sup> ESBL plates is unknown which makes it difficult to ascertain how the ESBL negative *E. coli* were able to grow on these agar plates. ESAC  $\beta$ -lactamase activity, overexpression of the  $bla_{OXA-1}$  gene and/or porin loss or modification [372] or the overexpression of efflux pumps [373] may have enabled these isolates to grow on CHROMagar<sup>TM</sup> ESBL plates in the absence of ESBL enzymes. The isolation of ESBL negative *E. coli* from CHROMagar<sup>TM</sup> ESBL plates highlights the importance of confirming putative ESBL producers using phenotypic tests to reduce false positive results. The detection of ARGs on plasmids is of particular concern due to the potential for dissemination within bacterial populations, particularly to pathogenic bacteria. Within plasmid groups, plasmid backbones are comparatively conserved but the gene content can be variable [374]. Plasmids play a major role in the dissemination of AMR in *Enterobacteriaceae* [47]. ESBL genes, often encoded on plasmids, can carry several ARGs, resulting in a multi-drug resistance phenotype [50, 48]. Isolate DF0059.2e harboured three plasmids, two of which encoded ARGs including pDF0059.2e\_2 which carried the  $bla_{CMY-2}$ gene. pDF0059.2e\_1 belonged to the IncY group, which are prophages that autonomously replicate and have been reported to be co-associated with other plasmid types including IncF and/or IncI [47], as was seen in this isolate. Although no ARGs were encoded on pDF0059.2e\_1, IncY plasmids have been reported to carry resistance genes including  $bla_{CTX-M-15}$  in *Klebsiella pneumoniae* [375] and *mcr-1* in an *E. coli* isolated from a pig farm in China [376].

The third plasmid, pDF0059.2e\_3, harboured 12 ARGs including  $bla_{\rm CTX-M-1}$ . All 12 of these genes were physically linked on the plasmid and mobile genetic elements were identified surrounding many of these genes (Appendix F). Interestingly, pDF0059.2e\_3 also harboured partial copies of three ARGs:  $bla_{\rm TEM-105}$ , catB4 and the aadA1, although partial copies of the catB4 and aadA1 genes could not be confirmed in the annotated plasmid, perhaps due to the low coverage (19.49% and 17.18%, respectively) and length (106 bp and 166 bp, respectively) of these partial genes. The 279 bp partial region of the  $bla_{\rm TEM-105}$  gene was located directly downstream of an IS26 transposase (Figure 3.5).

Several copies of transposases surrounding ARGs including transposon Tn7 transposition proteins TnsA and TnsB were detected in pDF0059.2e\_3. The "core machinery" of the Tn7 trasposon includes proteins TnsA, TnsB and TnsC. Target-site selecting proteins TnsD and TnsE are involved in preferential insertion into the bacterial chromosome (vertical transmission) and conjugable plasmids (horizontal transfer), respectively [377]. Tn7-like transposons containing ARGs have also been been identified in *Enterobacterales* isolated from samples collected on chicken and swine farms in China [378]. Detection of Tn7 transposition proteins TnsA and TnsB in pDF0059.e\_3 suggests that this plasmid has undergone recombination events, potentially contributing to the development of a multidrug resistant plasmid in this isolate. Such recombination events can make plasmid replicon or MOB-typing difficult [374] and may explain why the scheme could not assign a pMLST type for pDF0059.2e\_3. Multi-drug resistance plasmids, such as pDF0059.2e\_3, pose a significant public and animal health concern.

Plasmid associated  $\beta$ -lactamase gene types have been shown to have a highly homologous genetic environment, regardless of source [174]. In this study, five strains harboured highly similar IncI1 plasmids which encoded the  $bla_{CMY-2}$  gene (Table 3.8). IncI plasmids are conjugative, have a low copy number and a narrow host range [47]. A high proportion of AmpC and ESBL genetic determinants have also been found on IncI1 plasmids in other studies [9, 117], with IncI1 plasmids being frequently identified from extended-spectrum

cephalosporin resistant E. coli isolated from healthy livestock in the Netherlands over a ten vear period [354]. Although the plasmid and AmpC/ESBL gene combinations in E. coli isolated from dairy cattle and veal calves was more variable compared to the plasmid and gene combinations identified in *E. coli* isolated from other livestock sectors [354]. Gene synteny for the five IncI1 plasmids was highly similar, excluding the Shufflon A gene region where there was variation between the strains (Appendix J). The shufflon gene region is involved in conjugation of IncI1 plasmids [351]. IncK plasmids harbouring the  $bla_{\rm CMY-2}$ gene originating from human and poultry sources showed high nucleotide sequence identity when the shufflon region was excluded [360]. In the same study, IncI1 plasmid backbones also showed a high level of sequence similarity, but were a different pST to the plasmids characterised in this study [360]. Genetically similar E. coli harbouring IncI1 plasmids were isolated from multiple samples collected during the same sampling month on one farm in this study (faeces n=2; FDE n=1). This finding suggests that these E. coli may have been circulating within the farm environment. Composite faeces were randomly collected from faecal pats deposited throughout the paddock, and it cannot be discounted that the faeces were excreted from the same animal. Highly similar plasmids have also been detected in multiple isolates across other farm studies. For example, genetically similar plasmids harbouring identical co-located ARGs ( $bla_{CTX-M-1}$ , sul2 and tetA) were identified in six E. *coli* isolated from the same beef cattle farm [284].

Plasmids and HGT likely play a key role in the dissemination of third- and fourthgeneration cephalosporin resistance in *Enterobacteriaceae*. Core genome SNP analysis of *bla*<sub>CTX-M</sub> or *bla*<sub>CMY-2</sub> positive *E. coli* (n=59) showed clustering according to farm location, however, there was some incidences where isolates from different farms and sources (forage and cattle faces) clustered together [174]. SNP-based analysis of  $bla_{\rm CMY-2}$  positive E. coli isolated from human clinical cases, livestock and food samples identified examples where E. coli isolates of the same ST but different sample origin were genetically very similar and were separated by a small number of SNPs [360]. For example, two human clinical ST131 isolates were genetically similar to a chicken isolate and differed by 27-28 SNPs [360]. However within a ST, generally livestock and food isolates were less genetically similar (according to SNP distances) compared to human clinical isolates [360]. A diverse range of STs and phylogenetic groups were detected among third-generation cephalosporin resistant Escherichia spp. isolated from healthy cattle, pigs and chickens. This suggested that the clonal spread of single lineages is unlikely [117]. Together, these results and the results of this study highlight the importance of horizontal plasmid transfer in the dissemination of third-generation cephalosporin resistance among *Enterobacteriaceae*.

## 3.5 Conclusion

In conclusion, this study assessed the sample level prevalence of AmpC- and ESBLproducing *E. coli* on two NZ dairy farms over a 15 month period. No AmpC- or ESBLproducing *E. coli* were isolated from bulk tank milk or soil in this study, suggesting they are infrequent reservoirs in the NZ dairy farm environment. ESBL-producing *E. coli* were infrequently identified during the study, suggesting they are present at a low prevalence in these two dairy farms. Plasmid-mediated AmpC-producing *E. coli* were isolated at a low prevalence in faeces and FDE, but were isolated at a higher frequency on the higher intensity farm (Dairy 4) compared to Dairy 1. In contrast, AmpC-producing *E. coli* with mutations in the promoter region of the ampC gene were detected at a higher prevalence, particularly in FDE, but only on Dairy 1. The detection of AmpC-producing *E. coli* with mutations in the promoter region of the ampC gene is less concerning for human and animal health, as the chromosomal ampC gene cannot be transferred by HGT. These findings highlight the necessity to confirm the genotype of an AmpC phenotype in *E. coli*.

Both AmpC- and ESBL-producing *E. coli* were isolated at various time points throughout the 15 month study period and their detection was not associated with periods of elevated AMU on NZ dairy farms. Dairy 1 had higher AMU (mg/PCU) than Dairy 4 during this study period, however both farms used a low proportion of third- and fourth-generation cephalosporins which are known risk factors for ESBL-E. In addition, the farms used low levels of antimicrobials classed as red tier by the NZVA and these results support the hypothesis that AMU on NZ dairy farms influences the prevalence of AMR in the dairy farm environment. Additional studies should focus on high risk animals such as young calves and a larger number of dairy farms throughout NZ, including farms with known risk factors or management practices which may influence AMR such as high use of third- and fourth-generation cephalosporins, an open herd system, feeding waste milk to calves and the use of blanket dry cow therapy.
## Chapter 4

# Impact of systemic antimicrobial therapy on the faecal microbiome in dairy cows

#### 4.1 Introduction

Antimicrobial therapy, either for the treatment of disease or prophylactically as a feed additive, can alter both the bacterial community composition and faecal resistome in cattle, swine and chicken faeces [292, 379, 380]. Such perturbations in the microbiome can either be transient and the bacterial composition recovers, or microbiome changes may be longlasting [381]. Broad-spectrum antibiotics generally reduce the microbiome diversity [381], however the recovery and resilience of the microbiome can differ depending on the spectrum of the antimicrobial used, the treatment duration [382] and due to other host-dependent factors [383]. The ecological balance of the bovine faecal microbiome is complex and can also be affected by factors other than antimicrobial therapy such as feed type and concentrates [219, 286], gastrointestinal disease [384] and other farm management factors such as housing (e.g. freestalls or indoor barns) [385].

There is growing concern that antimicrobial use in food-producing animals may increase the prevalence of antimicrobial resistant bacteria and ARGs, as well as increasing the potential for HGT of such resistance genes to pathogenic bacteria [381]. Treatment with subtherapeutic levels of antimicrobials has been associated with an increase in the prevalence of antimicrobial resistant *E. coli* in cattle faeces [243]. Third-generation cephalosporins, such as ceftiofur, have a broad spectrum of activity including against a variety of Gramnegative pathogens [386]. Ceftiofur is the active ingredient in a number of products such as Excede<sup>®</sup> LA and Excenel<sup>®</sup> RTU which are used to treat infections on dairy farms, particularly infections other than mastitis such as metritis [387]. However, the use of third- and fourth-generation cephalosporins is a known risk factor for detecting ESBL- and AmpC-producing *E. coli* on dairy farms [7]. Ceftiofur is also classified as red tier by the NZVA [388], therefore its use should be limited to the treatment of specific susceptible bacteria or resistant infections. Using metagenomic sequencing it has been demonstrated that faeces from cows (n=3) injected with ceftiofur had a higher proportion of  $\beta$ -lactam resistance genes compared to control cow faeces (n=3), but this was admittedly based on

a small sample size. This finding was not replicated with ARGs belonging to all antibiotic classes [292]. Another study found no significant changes in the abundance of  $\beta$ -lactam resistance genes in the faeces from cows injected with ceftiofur [243], however there was a significant increase in the abundance of tetracycline resistance genes in the faeces of cattle fed chlortetracycline [243]. Cows treated with ceftiofur also had an increased proportion of gene sequences associated with mobile genetic elements (plasmids, phage, prophage and transposable elements) [292] which suggests that the treatment with ceftiofur may increase the potential for of HGT and thus, the spread of ARGs. Similarly, the use of in-feed antibiotics in swine has been associated with an increase in prophage induction [389], which may mediate gene transfer.

The diversity and composition of the bacterial microbiome can influence the resistome. For example, some bacteria have higher levels of intrinsic resistance whilst others, such as clinically relevant bacteria from the phylum Proteobacteria, have been associated with a number of ARGs [390]. Therefore, in order to understand the impact of antimicrobial therapy on the faecal microbiome in cattle, it is important to first investigate the changes to the microbiome composition and diversity pre-, during and post-treatment, which can have an important impact on the resistome structure. For example, in the study by Chambers *et al.*, [292] the increase in  $\beta$ -lactam resistance genes may be partially attributed to an increase in *Bacteroidia* in faeces from treated cows compared to control cows, as the most abundant genes within the  $\beta$ -lactam class, cfxA2 and cfxA3 encode class A  $\beta$ -lactamases that are often found in *Prevotella* spp. [391] and *Capnocytophaga* spp. [392].

Of the limited studies conducted in food-producing animals, including cattle, there has been a focus on the use of specific antimicrobial treatments or feed additives in controlled environments or experiments, often in the absence of disease. Therefore, the aim of this study was to assess the impact of systemic antimicrobial therapy on the faecal microbiome in dairy cows in the natural farm environment, by analysing faecal samples from cattle impacted by a number of different clinically-defined conditions with corresponding antimicrobial treatments.

### 4.2 Methods

#### 4.2.1 Study population and sample collection

Animals were recruited in the study and faecal samples were collected over a 14 month period from  $1^{\text{st}}$  July 2019 –  $30^{\text{th}}$  September 2020, however samples were not collected between  $23^{\text{rd}}$  March 2020 –  $13^{\text{th}}$  May 2020 due to the Covid-19 pandemic restrictions in NZ. Dairy cows requiring systemic antimicrobial treatment from Massey University Dairy 1 and Dairy 4 farms were recruited in the study on an opportunistic basis at the farm managers' discretion. Faecal samples were collected from cows receiving systemic

antimicrobial therapy pre- (S1), during (S2) and post-treatment (S3) (Figure 4.1). S1 samples were collected directly after diagnosis and before receiving the first antimicrobial treatment. Sample S2 was collected during treatment and was dependent on the prescribed treatment course (median one day after first treatment) and sample S3 was collected posttreatment once the cow re-entered the milking herd (median six days). Samples from healthy cows within the herd were also collected at the same sampling period as a untreated control to account for temporal variation in farm management factors such as feed type and breed as well as any farm-specific variation throughout the duration of the experiment. Where possible, untreated control cows were matched with more than one treated cow if the treatment was started on the same day. Cows which had received systemic antimicrobial therapy within 6 months prior to sample collection were excluded from the study. Each treated cow was assigned a number (1 - 30) and the respective control cow was designated using the same numbering system (1C - 30C). Sample collection was performed by the farm staff. Briefly, dairy cows were safely restrained and using a disposable glove and lubricant, faeces were collected and the first portion discarded due to potential contamination and approximately 5 g of faeces was collected and stored in a sterile pottle. Samples were stored at 4°C and processed within 24 hours of collection. A questionnaire regarding animal health, prescribed treatment course and antimicrobial treatment history was completed by farm staff per sick cow and healthy control (Appendix O). This research was approved by the Massey University Animal Ethics Committee (Protocol number 18/123).



Figure 4.1: Timeline for faecal sampling. Experimental sampling days are indicated on the figure (S1, S2 and S3) and the number of treated and control (untreated) cows is shown at the top of the figure. Where possible control cows were matched with more than one treated cow if treatment was started on the same day, hence the differing numbers between treatment groups. Antimicrobial treatment is indicated by the syringe and the cows reentered the milking herd at sample S3. Healthy control cow samples were collected at the same time as the treated cows.

#### 4.2.2 Sample processing

Faecal samples were homogenised and 0.2 g was stored in triplicate in cyrovials (Greiner Bio-One, Austria) at -80°C until DNA extraction. A pea-sized amount of homogenised faecal material was collected with a cotton swab and vortexed and enriched in 10 mL phosphate-buffered peptone water (BD Difco<sup>TM</sup>, Fort Richard Laboratories, Auckland, New Zealand) at 35°C for 18 hours. After incubation, enrichments were mixed by vortex and 1 mL of the enrichment was mixed with 429 µL of glycerol (30%[v/v]) and stored at -80°C. In addition, 1 mL of the enrichment was pelleted at 17,000 x g. The supernatant was removed and the final pellet re-suspended and washed twice with phosphate-buffered saline (PBS) (10 mM). The remaining pellet was re-suspended in 1 mL molecular biology grade water, vortexed and heated in a heating block at 100°C for 10 min. The boiled DNA lysate preparations were stored at -20°C and used as a template for subsequent PCR reactions.

#### 4.2.3 16S rRNA V3-V4 sequencing

#### 4.2.3.1 DNA extraction and sequencing

Faecal samples for DNA extraction were removed from the -80°C freezer, defrosted on ice and transferred to a bead-beating tube for DNA extraction. Genomic DNA was extracted using the Presto Stool DNA Extraction Kit (Geneaid Biotech Ltd, New Taipei City, Taiwan) according to the manufacturers instructions with minor modifications. Briefly, 0.2 g of faeces was used for the DNA extraction. For sample lysis, 800 µL of ST1 buffer was added, mixed by vortexing and incubated at 70°C for 5 min. The sample was vortexed at maximum speed for 7 min (Vortex Mixer, Labnet International, New Jersey, USA) and then centrifuged at 8,000 x q for 2 min. Next, 500  $\mu$ L of supernatant was transferred to a new tube and 5 µL RNAse A (100 mg/mL, QIAGEN, Hilden, Germany) was added and incubated for 10 min at 37°C. Subsequently, 150 µL of ST2 buffer was added and mixed briefly by vortexing. The sample was incubated at 0 - 4°C for 5 min and centrifuged at  $16,000 \ge q$  for 3 min to remove PCR inhibitors. Next, 500 µL of clear supernatant was transferred to a PCR inhibitor removal column, centrifuged at  $16,000 \ge q$  for 1 min and the column discarded. For DNA binding, 800 µL of ST3 buffer was added to the flow through, mixed vigorously by hand for 5 sec and 700 µL transferred to a GD column. The sample was centrifuged at  $16,000 \ge q$  for 1 min, the flow through was discarded and the remaining sample mixture transferred to the GD column. The GD column and sample was again centrifuged at 16,000 x q for 1 min and the flow through discarded. To wash the DNA, 400  $\mu$ L of ST3 buffer was added to the GD column and centrifuged at 16,000 x q for 30 sec. The flow through was discarded,  $600 \ \mu L$  of wash buffer was added and the sample centrifuged at 16,000 x q for 30 sec. The flow through was discarded and the wash step was repeated. After the final wash step, the GD column was centrifuged at 16,000 x q for 3 min to dry the column.

To elute the DNA, the GD column was transferred to a new tube and 30 µL of elution buffer pre-heated to 60°C was added to the centre of the column and left to stand for 2 min. The sample was then centrifuged at 16,000 x g for 2 min and the elution step repeated using the elution buffer containing the DNA. The DNA was transferred to a new microcentrifuge tube and stored at -20°C. All centrifuge steps were carried out at room temperature. The DNA concentration was quantified using a Qubit 4.0 fluorometer (ThermoFisher Scientific Inc., USA) and the  $A_{260/280}$  and  $A_{260/230}$  ratios were determined using a Nanodrop microvolume spectrophotometer (Nanodrop 2000c, ThermoFisher Scientific Inc., USA). The DNA integrity and size was visualised on a 0.8% [w/v] agarose gel which was prepared using 0.8g agarose (Invitrogen, Auckland, New Zealand) and 100 mL of 0.5x Tris-borate-EDTA buffer. The gel was stained with RedSafe (iNtRON Biotechnology, South Korea) and run at 100V for 4 hours. A high molecular weight *Hind* III/ $\lambda$  ladder (ThermoFisher Scientific Inc., USA) was used for size comparison and the gel images were captured and stored using the GelDoc XR<sup>+</sup> system (BioRad, New Zealand).

Next-generation sequencing was conducted using an Illumina MiSeq v2 platform with 2 x 250 base paired-end sequencing reads (Massey Genome Service, Massey University, Palmerston North, New Zealand). The total bacterial community was targeted using dual index primers which flank the V3-V4 hypervariable region of the 16S rRNA gene (16SF V3 and 16SR V4 primers, as detailed in [393]) and the 161 samples were run on a single Illumina MiSeq v2 lane.

#### 4.2.3.2 Bioinformatics

Adapter sequences and PhiX control library reads were removed from the demultiplexed fastq sequencing reads using FASTQ-MCF [394] by the Massey Genome Service (Massey University, Palmerston North, New Zealand) and the reads were quality filtered to a Phred score of Q20 (Massey Genome Service, Palmerston North, Massey University, New Zealand). The 16S rRNA data was analysed using the dada2 package v1.16.0 [395] in R v4.0.2 [396]. Briefly, the demultiplexed reads were quality filtered and 10 bp and 40 bp trimmed from forward and reverse reads, respectively. The denoised reads were merged, the chimeras removed and the reads remaining post-processing were used to construct an amplicon sequence variant (ASV) table. Taxonomy was assigned to the ASV table using the Silva v138.1 database [397, 398, 399]. The ASV table, sample metadata (Appendix P) and taxonomy table were combined to construct a phyloseq object using the R package Phyloseq v1.32.0 [400]. Diversity metrics were calculated using the vegan [401] and microbiome [402] R packages. The data was normalised to the median sequencing depth prior to calculating the diversity metrics. Data visualisation was conducted in R using a range of packages including ggplot2 [403], tidyverse [404], dplyr [405], Manu [406], RColorBrewer [407] and the website Colorgorical [408].

#### 4.2.4 Statistical tests

The  $\alpha$ -diversity of the faecal samples was compared using a generalised linear mixed effect model in R using the lme4 package [409]. Animal health status (treated or untreated control) and sample collection order (S1, S2 or S3) were included as fixed effects in the model as well as the interaction of both health status and sample order. The individual animal was included as a random effect. Pairwise comparison of health status and sample order was tested using the emmeans package and p-values were adjusted using the Tukey method [410].

#### 4.2.5 Microbiological methods and molecular characterisation

Faecal enrichments were removed from storage at -80°C and kept on ice. Culture-based methods (detailed in Chapter 3, sections 3.2.3 and 3.2.4) were used to enrich and isolate third-generation cephalosporin resistant *Enterobacteriaceae* from the faecal enrichments. Briefly, the faecal enrichments from the ceftiofur treated cows (n=3) and associated control cows (n=3) were plated on MC agar as a positive control to confirm E. coli detection as well as MC agar supplemented with cefotaxime (1  $\mu g/mL$ ), MC agar supplemented with ceftazidime (1 µg/mL) and CHROMagar<sup>TM</sup> ESBL plates to select for third-generation cephalosporin resistant Enterobacteriaceae, particularly E. coli or Klebsiella spp. Presumptive Enterobacteriaceae were identified using MALDI-TOF MS. Kirby-Bauer disc diffusion ASTs were undertaken on *E. coli* isolates for six antimicrobials (Table 3.3) and AmpC and ESBL phenotypes were confirmed using double-disc and three-disc comparison assays respectively, as outlined in section 3.2.4. For genotypic confirmation, E. coli with an AmpC phenotype were tested for pAmpC gene families using a multiplex PCR [177]. A PCR targeting the  $bla_{\rm CMY}$  gene family [313] was performed on isolates positive for the CITM primer set (Table 3.4), indicative of CMY-positive E. coli (using boiled DNA lysate preparations as DNA templates). The presence of pAmpC gene families in boiled DNA lysate preparations from enriched faecal samples was also analysed using the multiplex PCR. The PCR reaction and gel electrophoresis conditions are detailed in Appendix A.

#### 4.2.6 Whole genome sequencing

#### 4.2.6.1 DNA extraction

Bacterial isolates from glycerol broths stored at -80°C (section 3.2.3) were inoculated on Columbia Sheep Blood agar (5% blood) (Fort Richard Laboratories, Auckland, New Zealand) and incubated for 18 hours at 35°C. An individual colony was subsequently subcultured onto a fresh Columbia Sheep Blood agar plate (5% blood) and incubated at 35°C for 18 hours. Pure cultures were sent on LB-agar slopes to The Institute of Environmental Science and Research (Christchurch, New Zealand) where the DNA was extracted with the Qiagen DNeasy Blood and Tissue kit (QIAGEN, Hilden, Germany) according to the manufacturers instructions, using a QIAcube (QIAGEN, Hilden, Germany). The DNA libraries were prepared using the Nextera XT v2 Library Preparation Kit (Illumina, San Diego, USA) and sequenced on a NextSeq 550 sequencer (Illumina, San Diego, USA) by The Institute of Environmental Science and Research (Christchurch, New Zealand).

#### 4.2.6.2 Genome analysis

The Illumina MiSeq sequencing reads were processed using the Nullarbor v2.0 [320] pipeline with default parameters. In summary, the adapters were removed from raw reads using Trimmomatic v0.39 [321], the species identification by k-mer analysis was performed using the Kraken v1.1.1 database [322], the genomes were assembled using SKESA v2.4.0 [323] and annotated with Prokka v1.14.6 [324]. The sequence type was determined using mlst v2.19.0 [325] with information downloaded from PubMLST [326], the resistome profiles were assessed with ABRicate v1.0.1 [327] using the ResFinder database (downloaded 19/04/2020) [328] and the Centre for Genomic Epidemiology website [329] was used to detect the virulence genes and serotype in assembled genomes using the VirulenceFinder 2.0.3 database (v2020-05-29) [73] and the SerotypeFinder 2.0.1 database (v1.0.0), respectively. The SNP variation was assessed using Snippy v4.4.3 [331] with isolate DF0183e (ST56) as the reference genome (Table 3.5).

#### 4.3 Results

A total of 30 treated and 24 healthy (untreated) cows were recruited into the study (Appendix P). A further 15 treated cows were initially recruited in the study but were not included in the final analysis due to missing untreated control cow samples (n=6) or incomplete sampling (i.e. three faecal samples not collected, n=9) due to various reasons including animal death or welfare. Sample collection was at the farm manager's discretion. For one control cow, only two faecal samples were collected. Therefore, 161 faecal samples were analysed and from these, 15,263 unique ASVs were observed. Of these, 118 were singletons and 1,687 were doubletons, highlighting the high diversity and variation of ASVs within the faecal samples. The minimum, maximum and mean number of reads across the faecal samples was 16,100, 88,841 and 48,002 reads, respectively. Metadata for the faecal samples is detailed in Appendix P.

#### 4.3.1 Microbiome diversity

Comparison of the Shannon diversity from the faecal microbiomes of treated and control cows suggested that the microbiota was diverse and species-rich, as the diversity index was relatively high (4.09 - 6.59) which is consistent with the large number of unique ASVs

detected across the samples (Figure 4.2). For the treated cows, the Shannon diversity decreased during treatment (S2) compared to pre- (S1) and post-treatment (S3) samples (p = 0.0029), which suggests that the microbiome composition of these samples was less diverse with fewer abundant ASVs. Interestingly, the Shannon diversity from the treated cow faecal samples collected post-treatment (S3) was higher compared to S2 (during) and was more comparable with pre-treatment (S1) samples. No statistically significant changes in  $\alpha$ -diversity for control cows were observed. The Chao1 diversity index indicated that the community richness was relatively even across the samples (Figure 4.2), suggesting a similar number of ASVs in the faecal microbiomes, except from S2 for treated cows (p = 0.0124) which had a decreased Chao1 diversity index.

The microbiome profiles did not cluster according to health status or sample order (pre-(S1), during (S2) and post-treatment (S3)), with a low amount of variance explained in the Principal Component Analysis (Figure 4.3, PC1: 5.36%, PC2: 3.50%). Although the S2 faecal microbiomes from treated cows did not cluster together, they were generally more dispersed compared to the S1 and S3 samples. The dispersion of the faecal microbiome samples in the PCA plot and the low variance explained may be attributed to the variety of factors which can influence the microbiome composition such as variance at the individual animal level as well as treatment with specific antimicrobials and illnesses. Therefore, the faecal microbiomes from individual cows were further analysed as case studies differentiated by antimicrobial treatment type.



Figure 4.2: Diversity between treated (n=30) and control (n=24) cows for A: Shannon diversity and B: Chao1 diversity



Figure 4.3: Principal Component Analysis for the faecal microbiome in treated animals (red) and control animals (blue) over time (shapes). The percentage of variation explained in the Principal Component Analysis is indicated on the axis labels.

#### 4.3.2 Case studies

The animals were grouped and analysed by antimicrobial treatment. The systemic antimicrobial treatments included in this study were procaine penicillin G (n=23 cows), ceftiofur (n=3 cows) and penethamate hydriodide (n=3 cows) which are all  $\beta$ -lactam antibiotics, and a mixture of marbofloxacin/penethamate hydriodide (n=1 cow) which is a quinolone and  $\beta$ -lactam antibiotic, respectively. The untreated control cows associated with each treated cow were analysed in the respective case study.

#### 4.3.2.1 Procaine penicillin G treated cows

Procaine penicillin G was the active ingredient in a number of products used on the farms during the study including Intracillin<sup>®</sup> 300 and Phoenix Pharmacillin 300 (Appendix M). These products were used to treat numerous conditions including between claw/footrot, a penetration on the leg, left displacement of the abomasum, a swollen vulva and a retained membrane. Procaine penicillin G is a  $\beta$ -lactam antibiotic and is classified as green tier by the NZVA [388]. Penicillin G has *in vitro* activity against Gram-positive and

Gram-negative bacteria, however it is usually used to treat infections caused by susceptible Gram-positive bacteria such as *Streptococcus* spp [387]. Therefore, it was hypothesised that systemic antimicrobial therapy with procaine penicillin G would reduce the relative abundance of Gram-positive bacteria in the faecal microbiome. There was a high level of variation in the microbiome composition of the 23 cows treated with procaine penicillin G (Appendix Q) as well as variation at the individual animal level. Therefore, four of these cows and their associated controls were analysed in more detail as a case study. The following cows were included in the case study and represented a range of treatments: (i) cow 9: treated for between claw/footrot, (ii) cow 10: treated for left displacement of the abomasum, (iii) cow 16: treated for a swollen vulva and (iv) cow 24: treated for a retained membrane post-calving.

Compared to the post-treatment faecal sample (S3), the Shannon and Chao1  $\alpha$ -diversity was lower for samples S1 and S2 for cow 10 (Figure 4.4A-B). Compared to the other treated and control cows, cow 24 had a lower Shannon and Chao1  $\alpha$ -diversity across all faecal samples collected (S1, S2, S3), except for S2 collected from cow 10. The faecal samples from treated cows 9 and 16 had a similar Shannon  $\alpha$ -diversity compared to the control cows, however, the Chao1  $\alpha$ -diversity was lower for the S2 samples for all four treated cows, suggesting that there were fewer unique ASVs from these faecal samples. The Chao1  $\alpha$ -diversity of S3 samples from cows 9, 10 and 16 was similar compared to the control cows (S1, S2 and S3), whereas the S3 Chao1  $\alpha$ -diversity for cow 24 was the lowest across all faecal microbiomes in this case study. The faecal microbiomes from each control cow clustered together in the PCA plot (Figure 4.4D) compared to the cows treated with procaine penicillin G (n=4). The faecal microbiomes from individual animals (S1, S2 and S3) generally clustered together, suggesting that the faecal microbiomes from the same animal were more similar (demonstrating that animal is a major driver of variation).

The faecal microbiome composition across samples S1, S2 and S3 from cow 16 was relatively stable (Figure 4.4C). Faecal sample S1 for cow 9 had a higher abundance of an ASV belonging to *Romboutsia* compared to samples S2 and S3, whereas the remaining microbiome composition at the genus level was relatively consistent. In comparison, perturbations were observed in the faecal microbiome of treated cows 10 and 24. For cow 10, an ASV classified as *Escherichia-Shigella* was present in S1, with a higher abundance in S2 and was absent in S3. Sample S2 also had a higher abundance of an ASV belonging to the order *Bacteroidales*, which is labelled as "NA" in the plot as it could not be classified at the genus level. The microbiome composition of faecal samples S1, S2 and S3 for treated cow 24 varied, with *Prevotellaceae* UCG-003 present in a higher abundance in samples S1 (pre-) compared to S2 (during) and S3 (post-treatment). Interestingly, six ASVs belonging to *Prevotellaceae* UCG-003 were present in S1 compared to two in S2 and only one in S3. An ASV classified as *Bacteroides* was more abundant in S2 compared to S1 or S3 from cow 24. For the control cows in this case study (n=3), the faecal microbiome composition was relatively consistent for each animal during the study period (S1, S2 and S3) compared to the treated cows, although there was some perturbations in the relative abundance of specific ASVs (Figure 4.4C).





#### 4.3.2.2 Penethamate hydriodide treated cows

One cow from Dairy 1 and two cows from Dairy 4 farm were treated for mastitis with penethamate hydriodide (Penethaject) during the study period. These mastitis cases were diagnosed by positive culture and/or had culture susceptibility testing undertaken. Penethamate hydriodide is a benzyl penicillin prodrug [65] and is classed as green tier by the NZVA [388]. The spectrum of activity for penethamate hydriodide includes a number of Gram-positive mastitis causing pathogens such as *Staphylococcus aureus*, *Streptococcus uberis* and *Streptococcus dysgalactiae* subsp. *dysgalactiae* [65, 411]. One cow was used as the untreated control for two treated cows as they started antimicrobial treatment on the same day (control cow 18C/19C).

The Shannon diversity was comparatively lower from the pre-treatment and during treatment samples (S1 and S2) from cow 19 and the S2 sample from cow 18 (Figure 4.5A). The  $\alpha$ -diversity of post-treatment samples (S3) from cows 18 and 19 were similar to control cows. In comparison, both the Shannon and Chao1 diversity of all three faecal samples from cow 6 (S1, S2, S3) was consistently similar to the diversity observed in control cows (Figure 4.5B), whereas sample S3 from control cow 6C showed a decrease in both the Shannon and Chao1  $\alpha$ -diversity. The faecal microbiome samples clustered at the individual animal level for control cows 6C and 18C/19C and treated cow 6 (Figure 4.5D). In comparison, pre- (S1) and post-treatment (S3) samples for cows 18 and 19 were clustered together, however the during treatment samples (S2) were distinct (Figure 4.5D).

As suggested by the diversity metrics, the bacterial microbiome composition of samples S1, S2 and S3 from cow 6 remained relatively stable during the study (Figure 4.5C), with small perturbations in the abundance of key ASVs. The microbiome composition for control cow 6C was also stable during the study period, despite a reduction of  $\alpha$ -diversity from sample S3, which suggests there was a reduction in ASVs present at a lower relative abundance post-treatment. At the individual animal level, the faecal samples collected during treatment (S2) from cows 18 and 19 differed from samples S1 and S3, with ASVs belonging to *Escherichia-Shigella* being more abundant in S2 for both cows. ASVs classified as *Alloprevotella* were more abundant in S2 from cow 18 and S1 from cow 19. In addition, ASVs belonging to *Paeniclostridium* were in low abundance in S2 for both cows and ASVs classified as *Romboutsia* were reduced in S2 compared to S1 and S3 for cow 19. The most abundant ASVs from faecal samples of control cow 18C/19C were relatively stable during the sampling period (S1, S2 and S3).





Page 104

#### 4.3.2.3 Marbofloxacin/penethamate hydriodide treated cow

One cow from Dairy 4 (n=1) was treated for mastitis with Marbocyl<sup>®</sup> 10% and Penethaject, which is a combination of marbofloxacin and penethamate hydriodide belonging to the quinolone and  $\beta$ -lactam antibiotic classes, respectively. In addition to penethamate hydriodide which is classified as green tier by the NZVA, marbofloxacin is classified as red tier [388]. Cow 14 was treated with the intramammary antibiotic Mastalone<sup>®</sup> (oxytetracycline, oleandomycin and neomycin) 24 hours before the S1 sample was collected.

The Shannon and Chao1  $\alpha$ -diversity of the faecal microbiome of the untreated control cow in this case study (14C) was relatively high and consistent across all sample periods (S1, S2 and S3) which indicates a high level of bacterial diversity. In comparison, pre (S1) and during treatment (S2) samples collected from treated cow 14 had a comparatively lower Shannon and Chao1 diversity compared to the post-treatment sample (S3) and this is likely due treatment with penethamate hydriodide and marbofloxacin. This combination of antibiotics would likely have a significant effect on the faecal microbiome diversity and richness by reducing the abundance of susceptible populations. The  $\alpha$ -diversity of sample S3 appeared to have recovered post-treatment and was more similar to the diversity observed in the control cow faecal microbiomes (S1, S2 and S3) than to S1 or S2 from treated cow 14.

For treated cow 14, the microbiome composition of S1 was unique, with a number of ASVs belonging to genera which were absent or in low abundance in S2 and S3 such as *Burkholderia/Caballeronia/Paraburkholderia* (could not be differentiated at a higher resolution), *Clostridium sensu stricto* 1, *Succiniclasticum* and *Terrisporobacter*. In cow 14, the microbiome composition was altered during treatment (S2), with a higher abundance of ASVs belonging to *Lachnoclostridium*, *Prevotellaceae* UCG-001 and UCG-005. ASVs belonging to *Escherichia-Shigella* were more abundant in S1 and S2 of treated cow 14, but were absent in S3 from cow 14 as well as from the control cow faecal microbiomes (14C: S1, S2 and S3). The post-treatment sample (S3) for cow 14 was more comparable to the faecal microbiome of the control cow (14C) compared to the pre- and during treatment samples for cow 14. ASVs belonging to *Paeniclostridium* and *Romboutsia* were more abundant in S3 samples from both the treated and control cows.



Figure 4.6: Shannon (A) and Chao1 (B)  $\alpha$ -diversity of faecal samples from a cow treated with marbofloxacin/penethamate hydriodide (n=1) and the respective control cow (n=1).



Figure 4.7: Abundance of the top 30 ASVs at the genus level of faecal samples from a cow treated with marbofloxacin/penethamate hydriodide (n=1) and the respective control cow (n=1).

#### 4.3.2.4 Ceftiofur treated cows

Ceftiofur is a third-generation cephalosporin which has a broad spectrum of activity against a variety of Gram-negative pathogens [386]. Ceftiofur is the active ingredient in a number of products used to treat infections on dairy farms, particularly infections other than mastitis such as metritis [387]. Ceftiofur is classified as red tier by the NZVA [388] and therefore its use should be limited to specific organisms or resistant infections. Three cows were treated with ceftiofur for caesarean calf deliveries during the study period (Excede LA; Appendix P).

Both the Shannon and Chao1  $\alpha$ -diversity was similar across samples S1 for both treated and control cows (Figures 4.8A-B) and the  $\alpha$ -diversity remained relatively consistent across the treated and control cows, except for sample S2 from control cow 11C which increased and samples S2 and S3 from treated cow 12 which was reduced during and post-treatment compared to pre-treatment (S1) levels. Perturbations in the microbiome composition were observed at the individual animal level during treatment (Figure 4.8C). For cow 11, ASVs belonging to the genera Paeniclostridium, Romboutsia and Prevotellaceae UCG-001 were in low abundance from sample S2 and Prevotellaceae UCG-004 associated ASVs were reduced during treatment (S2). In comparison, the microbiome composition in the associated control cow (11C) was relatively stable, with some perturbations in ASV abundance. For example, ASVs belonging to the genus Alloprevotella were more abundant in S3 compared to S1 and S2. For treated cow 12, Paeniclostridium classified ASVs were in low abundance during treatment (S2) and a higher abundance of an ASV belonging to the family Bacteroidales RF16 group was detected during treatment (this ASV could not be classified at the genus level and is listed as "NA" in Figure 4.8C). The microbiome composition of the associated control cow 12C differed in S1, with ASVs belonging to Alloprevotella and *Paeniclostridium* either present in low abundance or were absent. The microbiome composition for treated cow 13 varied during the study period. During treatment (S2), ASVs belonging to *Prevotellaceae* UCG-001 were in low abundance and there was a higher abundance of two ASVs listed as "NA" which corresponded to the family Bacteroidales RF16 group and p-251-05 groups. The microbiome composition of key genera in control cow 13C was stable during the study period. According to PCA analysis, the three faecal microbiomes (S1, S2 and S3) from control cows were clustered together (cows 11C, 12C and 13C) whereas the S2 (during) samples for treated cows were distinct compared to S1 and S3, except for cow 12 in which samples S2 and S3 clustered together (Figure 4.8D).



with ceftiofur (n=3) and respective control cows (n=3). Principal Component Analysis  $(\mathbf{D})$  for the faecal microbiome of treated and control cows at the individual animal level (colours) over the sampling period (shapes). The percentage of variation explained in the Principal Component Analysis is indicated on the axis labels.

#### 4.3.3 AMR in action

The use of third- and fourth-generation cephalosporins is a known risk factor for AmpC and/or ESBL-E being detected on a dairy farm [7]. Three cows were treated with ceftiofur during the study period (cows 11, 12 and 13) and despite no obvious change in abundance of ASVs belonging to *Escherichia-Shigella* during or post-treatment (Appendix R), culture enrichments from faecal samples obtained from treated and control animals (S1, S2 and S3) were investigated further to determine the presence and emergence of third-generation cephalosporin resistant *Enterobacteriaceae*. *Enterobacteriaceae* were identified from all enrichments on the positive control plates (MC agar). Presumptive third-generation cephalosporin resistant *E. coli* were isolated on selective agar plates from samples S2 and S3 collected from the treated cow 12 (Table 4.1). No *E. coli* were isolated from the selective agar plates for sample S1 from cow 12 nor from the other seven treated or nine control cow enrichments.

Boiled DNA lysates from the faecal enrichments were screened using a multiplex PCR to identify pAmpC genes (Table 3.4). PCR of boiled lysate preparations from enrichments DG079 (S2) and DG082 (S3) from cow 12 demonstrated the presence of the  $bla_{CMY-2}$  type gene, whereas enrichment S1 did not yield a PCR product. Faecal sample enrichments from control cow 12C (S1, S2 and S3) had a faint band which did not match any expected amplicon PCR product sizes (380 bp) and may have been due to non-specific priming. Faecal enrichment boiled lysates from the remaining ceftiofur treated cows (n=2) and associated control cows (n=2) did not yield any PCR products from the multiplex PCR.

Animal	Lab ID	Farm	$\mathbf{Type}$	Sample	No. isolates	MALDI-
				$\operatorname{order}^{\mathbf{a}}$		TOF ID
11	DG067	Dairy 4	Treated	S1	0	-
11	DG073	Dairy 4	Treated	S2	0	-
11	DG087	Dairy 4	Treated	S3	0	-
11C	DG069	Dairy 4	Control	S1	0	-
11C	DG074	Dairy 4	Control	S2	0	-
11C	DG088	Dairy 4	Control	S3	0	-
12	DG070	Dairy 1	Treated	S1	0	-
12	DG079	Dairy 1	Treated	S2	6	E. coli
12	DG082	Dairy 1	Treated	S3	4	E. coli
12C	DG071	Dairy 1	Control	S1	4	Morganella
						morganii
12C	DG080	Dairy 1	Control	S2	0	-
12C	DG083	Dairy 1	Control	S3	0	-
13	DG075	Dairy 4	Treated	S1	0	-
13	DG077	Dairy 4	Treated	S2	0	-
13	DG084	Dairy 4	Treated	S3	0	-
13C	DG076	Dairy 4	Control	S1	0	-
13C	DG078	Dairy 4	Control	S2	0	-
13C	DG085	Dairy 4	Control	S3	0	-

Table 4.1: Isolates identified from faecal enrichments of ceftiofur treated cows (n=3) and associated control cows (n=3)

<sup>a</sup> Faecal samples were collected pre- (S1), during (S2) and post-treatment (S3) from sick dairy cows. Healthy control (untreated) cow samples were collected during the same sampling period.

Species identification of presumptive *Enterobacteriaceae* was determined using MALDI-TOF MS. Isolates from the MC agar plates, which were used as a positive control, were not analysed further nor were any *Morganella morganii* isolates. *E. coli* which were isolated from selective agar plates (n=10) were subjected to Kirby-Bauer disc diffusion ASTs for six antimicrobials (Table 3.3). All isolates were resistant to cefotaxime, cefpodoxime and cefoxitin and susceptible to tetracycline, streptomycin and ciprofloxacin; except for isolate DG079e which was intermediate for cefoxitin (Table 4.2). All isolates were confirmed as AmpC positive and ESBL negative using the three-disc or double-disc comparison assays, respectively (Figure 3.1). The AmpC-producing *E. coli* were investigated using a multiplex PCR (Table 3.4) to identify pAmpC genes. All *E. coli* isolates (n=10) were positive for the CITM primer set (Table 3.4), indicative of a CMY-2 type  $\beta$ -lactamase. The majority of isolates had the same resistance profile, however the AST zone sizes varied between isolates. Comparing sample metadata, resistance profiles and AST zone sizes suggested that three groups of *E. coli* isolated from the S2 and S3 faecal samples, indicated by the bold lines in Table 4.2. Group 1 isolates (DG079c-f) were isolated from the faecal sample collected during treatment (S2) on either MC agar supplemented with cefotaxime (1 µg/mL) or ceftazidime (1 µg/mL). Group 2 isolates (DG079g-h) were isolated on CHROMagar<sup>TM</sup> ESBL plates from sample S2 and in general had smaller AST zone sizes for both cefotaxime and cefoxitin compared to the other isolates (Table 4.2). Group 3 consisted of isolates DG082c-f isolated on either MC agar supplemented with cefotaxime (1 µg/mL) or ceftazidime (1 µg/mL) from the post-treatment faecal sample (S3). *E. coli* isolates from group 1 and group 3 had similar AST zone sizes. One isolate from each group was further analysed using WGS.

Isolate	Phenotype	ampC gene	CTX <sup>a</sup>	CPD	FOX	TET	STR	CIP
DG079c	AmpC	$bla_{\rm CMY-2}$ type	20.3	6.6	14.0	23.7	15.3	33.5
DG079d	AmpC	$bla_{\rm CMY-2}$ type	19.3	6.6	13.9	21.0	15.1	32.5
DG079e	AmpC	$bla_{\rm CMY-2}$ type	19.3	6.6	15.0	23.7	16.7	33.7
DG079f	AmpC	$bla_{\rm CMY-2}$ type	18.8	6.6	14.0	23.5	15.3	33.3
DG079g	AmpC	$bla_{\rm CMY-2}$ type	9.6	6.6	6.6	22.4	16.0	30.6
DG079h	AmpC	$bla_{\rm CMY-2}$ type	12.0	6.6	6.6	23.3	15.9	31.2
DG082c	AmpC	$bla_{\rm CMY-2}$ type	19.3	6.6	14.0	22.4	15.0	34.5
DG082d	AmpC	$bla_{\rm CMY-2}$ type	19.0	6.6	13.5	23.1	15.7	34.9
DG082e	AmpC	$bla_{\rm CMY-2}$ type	20.2	6.6	13.7	22.2	15.1	31.2

Table 4.2: Antimicrobial susceptibility testing and molecular characterisation of the  $E.\ coli$  isolates

<sup>a</sup> Zone sizes in millimetres. CTX, cefotaxime (30μg); CPD, cefpodoxime (10μg); FOX, cefoxitin (30μg); TET, tetracycline (30μg); STR, streptomycin (10μg); CIP, ciprofloxacin (5μg).

18.8

 $bla_{\rm CMY-2}$  type

6.6

13.9

22.6

15.5

33.1

#### 4.3.3.1 Whole genome sequencing

AmpC

DG082f

Whole genome sequence reads were obtained from DNA extracts of three *E. coli* to understand the molecular epidemiology and genetic context of ARGs from these strains. The three *E. coli* were all isolated from cow 12, two during treatment from MC agar supplemented with ceftazidime (DG079c) and CHROMagar<sup>TM</sup> ESBL (DG079h), and one *E. coli* isolated from the post-treatment sample on MC agar supplemented with ceftazime (DG082f). The three *E. coli* isolates were the same serotype (O160:H32) and ST (ST5514) and were genetically similar, differing by 53 - 170 SNPs (Table 4.4). Interestingly, according to the core SNP genome, isolates DG079c and DG082f (isolated from S2 and S3, respectively) were the most similar differing by 53 SNPs. Isolate DF0049.2e, which is a ST57 CMY-2 producing *E. coli* isolated from Dairy 1 during the longitudinal study (Table 3.5), was included as the reference genome as four isolates are required to carry out the core genome SNP analysis. DF0049e differed from the other isolates by 59,783 - 62,032 SNPs.

Isolate	Contigs	Genome	GC	Depth	N50	$\mathbf{ST}$	Serotype	Plasmids
		size (bp)	(%)					
DG079c	112	4,880,186	50.3	120	110,495	5514	O160:H32	IncFIB_1,
								IncI1
DG079h	308	4,789,972	50.3	82	37,402	5514	O160:H32	IncFIB_1,
								IncI1
DG082f	128	4,875,060	50.8	89	91,090	5514	O160:H32	IncFIB_1,
								IncI1

Table 4.3: Genome composition of three *E. coli* isolates

The ARGs  $bla_{CMY-2}$ , chromosomal ampC, aph(6)-Id, dfrA14 and sul2 were detected in the three *E. coli* genomes, which potentially confer resistance to the  $\beta$ -lactam, aminoglycoside, trimethoprim and sulfonamide classes, respectively. The detection of the  $bla_{CMY-2}$  gene was consistent with the AST data and resistance to the third-generation cephalosporins. In contrast, all three isolates carried the aph(6)-Id gene which hypothetically confers resistance to streptomycin [412], yet the *E. coli* were susceptible in the AST data highlighting the necessity to confirm a resistance phenotype. Phenotypic testing for trimethoprim and sulfonamide resistance was not undertaken. All three *E. coli* carried the two plasmid incompatibility factors IncFIB\_1 and IncI1\_1 $\alpha$ . According to VirulenceFinder [73], isolates DG079c and DG082f harboured seven and DG079h harboured eight virulence factors (Appendix S). Compared to the other two isolates, DG079h also carried the gad gene (glutamate decarboxylase). All of the *E. coli* harboured the f17A and f17G fimbriae genes. No STEC or EPEC associated virulence genes were identified among the three *E. coli* isolates.

Table 4.4: Core genome single nucleotide polymorphism analysis of three AmpC-producing  $E.\ coli$  from in this study

Isolate	DF0049.2e	DG079c	DG079h	DG082f
DF0049.2e (reference)	0	61,753	59,783	62,032
DG079c	61,753	0	153	53
DG079h	59,783	153	0	170
DG082f	62,032	53	170	0

#### 4.4 Discussion

It was hypothesised that systemic antimicrobial treatment would impact the bovine faecal microbiome by decreasing species diversity and reducing the abundance of susceptible bacterial populations which could allow non-susceptible bacteria to flourish. The impact of treatment with various antimicrobials in diseased animals on working dairy farms in NZ is unknown. Therefore, faecal samples were collected pre- (S1), during (S2) and posttreatment (S3) from cows receiving systemic antimicrobial therapy across two working dairy farms. Faecal samples were collected at the same time from healthy control/untreated cows on the same farm to account for temporal variation in farm management factors such as feed type and breed as well as any farm-specific variation throughout the duration of the experiment. The results of this study demonstrate that systemic antimicrobial treatment with either procaine penicillin G, ceftiofur, penethamate or marbofloxacin/penethamate reduced the bacterial diversity and richness during treatment (S2) in the bovine faecal microbiome, which was likely a result of the antimicrobials targeting susceptible bacteria. Generally the  $\alpha$ -diversity increased post-antimicrobial treatment (S3) when the cow reentered the milking herd and perturbations in the bovine faecal microbiome were more pronounced in cows with severe disease (for example, cows with left displacement of the abomasum compared to footrot).

Other studies have also demonstrated a reduction in the bovine faecal microbiome richness post antimicrobial treatment. The bovine faecal microbiome OTU richness was reduced two days post-treatment in beef cattle that received a single oxytetracycline injection (n=12)two days after being transported to a feedlot [382]. The rumen microbiota was also reduced after treatment with doxycycline (n=3) in beef cattle [413] and rumen, colon and caecum digesta samples from steers fed a diet supplemented with monesin and tylosin showed decreased bacterial diversity and richness compared to control steers [219]. In comparison, another study found the difference in  $\alpha$ -diversity between cattle fed a diet supplemented with tylosin compared to control cattle at feedlots in the USA was not statistically significant [243], although treated cattle had a lower  $\alpha$ -diversity compared to that of controls. No significant difference in microbiome diversity or richness was also observed between beef cattle treated with tulathromycin (n=15) or controls (n=15) at a commercial feedlot [414]. Such findings highlight that numerous factors can affect the microbiome richness and diversity, such as the spectrum of antimicrobial activity as well as host-dependent factors. In many cases in this study, the bacterial diversity and richness recovered post-treatment (median 6 days) when the cow re-entered the milking herd (Figures 4.4, 4.5, 4.6, 4.8). However, the microbiome was analysed at the ASV level and the abundance of ASVs were grouped by genus for the top 30 ASVs in the case studies, therefore further work is required to determine whether the same ASVs are present in the pre- and post-treatment samples, or whether different ASVs belonging to the same genus are flourishing post-treatment.

The shared changes in bacterial microbiome composition between treated and control animals could not be investigated due to the study being set in a natural farm environment, with variation in animal health status, prescribed treatment product and duration in individual animals as well as temporal changes in seasonality and feed at the farm level. Therefore, alterations in the bovine faecal microbiome during systemic antimicrobial treatment (S1, S2 and S3) was analysed at the individual animal level. In general, perturbations in the bacterial microbiome composition were observed during treatment (S2) compared to pre- (S1) and post-treatment (S3) samples for treated cows.

Changes in the relative abundance of specific ASVs during treatment differed at the individual animal level. For example, cow 10 treated for a left displaced abomasum with procaine penicillin G displayed major perturbations in bacterial diversity, composition and richness during treatment (Figure 4.4), whereas in comparison the bacterial diversity and richness was lower in cow 12 treated with ceftiofur during and post-treatment, however changes in the microbiome composition of the top 30 ASVs was less apparent (Figure 4.8). This may be due to only the top 30 ASVs across the animals in the case study being analysed and therefore there may have been more apparent changes in ASVs which were present in a lower abundance. Perturbations in the microbiome for cows with the same illness and treatment regime was also animal specific. For example, the diversity, richness and taxonomic composition was relatively stable in the faecal microbiome from cow 6 treated for mastitis with penethamate hydriodide, whereas the  $\alpha$ -diversity metrics and bacterial community composition changed for the other two cows treated with penethamate hydriodide (cows 18 and 19; Figure 4.5). These findings highlight that a multitude of factors other than antimicrobial treatment and disease can drive the diversity and composition of the bovine faecal microbiome, such as animal-level variation, diet [286, 415], housing and geographical location [414]. A study of oxytetracycline (tetracycline class) or tulathromycin (macrolide) single injection treatment in feedlot cattle found that the treatments altered the faecal microbiome at days two and five post-treatment, however the microbiome community composition was more affected by transport to the feedlot and time (day 2 to 34 in the study) compared to antimicrobial treatment [382]. Another study analysing pooled faecal samples from pen floors of control cattle or cattle fed a diet supplemented with tylosin did not detect any changes in microbiome composition among treated or control cattle, however geographic location of the feedlot had a significant association with microbiome composition [416]. Both studies demonstrate that environmental and farm management factors can also have a significant effect on the bovine faecal microbiome.

Procaine penicillin G is the active ingredient in a number of products used to treat infections susceptible to penicillin on NZ dairy farms [387]. Penicillin G has *in vitro* activity against Gram-positive and Gram-negative bacteria, however it is usually used to treat infections caused by susceptible Gram-positive bacteria such as *Streptococcus* spp. Therefore, it was hypothesised that systemic antimicrobial therapy with procaine penicillin G would reduce the relative abundance of Gram-positive bacteria in the faecal microbiome. There was higher variation of microbiome composition at the individual animal level across cows treated with procaine penicillin G (n=23) compared to other treatment groups, with some treated cows showing large perturbations in community composition (e.g. cows 10 and 24) and other cows in which microbiota composition was comparatively stable during treatment (e.g. cows 9 and 16). This may be reflective of the larger sample size for animals treated with procaine penicillin G, which is classified as green tier by the NZVA [388] and is therefore is routinely used to treat infections caused by bacteria susceptible to penicillin. The relative abundance of ASVs classified as *Paeniclostridium* and *Romboutsia*, which are both Gram-positive bacteria, was reduced during treatment in cows 9, 16 and 24 and may be a result the activity of procaine penicillin G against Gram-positive bacteria. The impact of procaine penicillin G on *Lactobacillales*, which includes *Streptococcus* spp., could not be assessed as they were present in low abundance and were not included in the top 30 ASVs detected from faeces in this case study.

In addition, ASVs belonging to the *Bacteroidales* order, which are Gram-negative bacteria, were more abundant in faecal samples collected during treatment (S2) for cows 10 and 24 (Figure 4.5C). Treated cow 10 had an increased abundance of *Escherichia-Shiqella* pre- and during treatment and an increase of a ASV belonging to the *Bacteroidales*. The increase in Gram-negative bacteria during treatment may be due to reduced competition with Grampositive bacteria, which are generally susceptible to procaine penicillin G. In contrast, Escherichia-Shigella were higher in abundance post-treatment for cow 24 (Figure 4.4). A decrease in Gram-positive bacteria at the genus level was also observed in cattle fed feed supplemented with monesin and tylosin, which have activity against Gram-positive bacteria [219], yet no changes in Gram-positive bacterial population structure was observed at the phylum level in another study analysing tylosin as a feed additive in beef feedlot cattle [416]. The family *Prevotellaceae* was identified as a core member of the faecal microbiota in cattle [385, 415] and are metabolically diverse [417], with some members of *Prevotella* being identified as more abundant in animals fed concentrates [415]. Interestingly, six ASVs classified as *Prevotellaceae* UCG-003, a genus which was detected a higher relative abundance pre- and during treatment for cow 24, were detected in sample S1 compared to two in S2 and only one in S3 (Figure 4.4), suggesting some *Prevotellaceae* UCG-003 ASVs were more persistent after treatment with procaine penicillin G.

A higher number of animals were treated with procaine penicillin G in this study (23 of 30, 76.7%), which may account for the higher animal level variation seen in this treatment group. Disease severity can also impact on the faecal microbiome, for example more severe disease such as left displacement of the abomasum (cow 10) compared to between claw/-footrot (cow 9) are likely to have an increased impact on overall animal health. Treated animals within other case studies were all treated for the same illness (e.g. ceftiofur for a caesarean section and penethamate hydriodide for mastitis), therefore the impact of illness on the faecal microbiome could not be assessed. It should be noted their perturbations in bacterial community composition were less common in control (untreated) cow faecal samples where the control cows were sampled at the same time as treated cows. The treated

and control cows had access to the same feed sources on farm. Therefore, one hypothesis is that antimicrobial therapy, illness and behavioural changes such as reduced feed intake in ill cattle may have a significant impact on the microbiome.

Penethamate hydriodide (Penethaject) is recommended for the treatment of mastitis caused by Gram-positive bacteria as well as for metritis, foot-rot and respiratory infections [387]. As such, the spectrum of activity of penethamate hydriodide includes a number of Grampositive mastitis causing pathogens such as S. aureus, Strep. uberis and Strep. dysgalactiae subsp. dysqalactiae [411, 65]. Therefore, it was hypothesised that the relative abundance of Gram-positive bacteria will be reduced post-treatment with penethamate hydriodide. A small number of cows were treated for mastitis with penethamate hydriodide during the study period (n=3) and all cows were diagnosed by positive culture and/or susceptibility testing undertaken. The majority of mastitis cases on farm were treated with procaine penicillin G and both of these antimicrobials are classified as green tier by the NZVA, which reflects the prudent antimicrobial stewardship on Dairy 1 and Dairy 4. The bacterial microbiome composition was altered in two of three cows treated with penethamate hydriodide (Figure 4.5). The relative abundance of Escherichia-Shiqella and Alloprevotella was higher in faecal samples collected during treatment (S1 and S2) for cows 18 and 19. This increase in *Escherichia-Shiqella* and *Alloprevotella*, which are both Gram-negative bacteria, may be due to a reduction in Gram-positive bacteria which are susceptible to penethamate hydriodide allowing for Gram-negative bacteria to persist and increase in overall abundance. ASVs classified as *Paeniclostridium* and *Romboutsia* were less abundant in some during treatment faecal samples (S2), which may be due to the susceptibility of Gram-positive bacteria to penethamate hydriodide.

Marbofloxacin is a third-generation fluoroquinolone developed for veterinary treatment and has a broad spectrum of activity against most Gram-negative and some Gram-positive bacteria [418]. The label use of Marbocyl<sup>®</sup> 10%, containing marbofloxacin as the active ingredient, is for the treatment of respiratory or Gram-negative infections [387], with limits on its use for severe or resistant infections as it is classified as red tier by the NZVA [388]. Penethamate hydriodide is active against a number of Gram-positive mastitis causing pathogens such as *S. aureus*, *Strep. uberis* and *Strep. dysgalactiae* subsp. *dysgalactiae* [411, 65], as discussed above. Therefore, it is hypothesised that simultaneous treatment with both marbofloxacin and penethamate hydriodide would alter the faecal microbiome during and post-treatment by reducing the abundance of susceptible bacteria, due to the combined wide spectrum of activity of these two antimicrobials.

The sample S1 faecal microbiome for cow 14 was unique (Figure 4.7), as this cow was also treated 24 hours prior to initial sample collection with the intramammary treatment Mastalone, which is used for severe cases of mastitis and consists of a combination of the three antibiotics oxytetracycline, oleandomycin and neomycin [387]. A number of ASVs were abundant in S1 and either absent or low in abundance from samples S2 and S3

from cow 14 as well as from the top 30 ASVs detected in the other case studies, in particular ASVs classified as *Burkholderia/Caballeronia/Paraburkholderia, Clostridium sensu stricto* 1, *Succiniclasticum* and *Terrisporobacter*, which includes a mixture of Gram-positive and Gram-negative bacteria. Some bacterial species within these genera, for example *Burkholderia* spp. [419], have high levels of acquired or intrinsic resistance which may account for the increased abundance of ASVs in this sample post-treatment with oxytetracycline, oleandomycin and neomycin compared to other faecal samples in this study. Interestingly, network analysis of swine manure inoculated with mature compost (high temperature compost containing mesophilic and thermophilic microorganisms) identified *Romboutsia, Clostridium sensu stricto* 1 and *Terrisporobacter* as harbouring a number of ARGs and mobile genetic elements [420]. These ASVs were not highly abundant in faecal samples collected during (S2) and post antimicrobial treatment (S3) for cow 14.

The relative abundance of *Escherichia-Shigella* was higher in S1 and S2 for treated cow 14. An increased ratio of Proteobacteria compared to Firmicutes and Bacteroidetes has been associated with rumen dysbiosis identified in cattle mainly fed a concentrate-based diet [286], which could potentially increase the prevalence of ARGs, which are often found in Proteobacteria [390]. Proteobacteria were present at a higher abundance in cattle treated with doxycycline (n=3) compared to healthy controls (n=3) [413], however other external factors such as feed may also affect their abundance as Proteobacteria were more abundant in animals fed a concentrate diet, compared to those with a forage-based diet [286]. Therefore, the higher relative abundance of *Escherichia-Shigella* in samples S1 and S2 may have been due to illness or external factors such as feed. ASVs classified as *Paeniclostridium* and *Romboutsia* were more abundant in samples S3 from both the treated and control cows (14 and 14C, respectively). The relative abundance of such ASVs is variable in some animals from the case studies described, although the relative abundance was higher in both the treated and control (untreated) cow within this case study and therefore may be due to factors other than antimicrobial treatment.

Ceftiofur is a third-generation cephalosporin which has a broad spectrum of activity against a variety of Gram-negative pathogens [386], including susceptible *E. coli.* Across the ceftiofur treated cows (n=3), ASVs belonging to *Alloprevotella* and *Bacteroidales* (Gramnegative bacteria) were enriched during treatment. Interestingly, alterations in the faecal microbiome community structure in cattle treated with ceftiofur (n=3) compared to controls (n=3) was largely due to an increase in the relative abundance of *Bacteroidia* and a decrease in *Actinobacteria* in treated cows [292]. In the same study, in the faeces of ceftiofur treated cows (n=3) the most abundant genes within the  $\beta$ -lactam class were cfxA2and cfxA3 [292], which encode class A  $\beta$ -lactamases that are often found in *Prevotella* spp. [391] and *Capnocytophaga* spp. [392]. The cfxA gene has also been detected in *Alloprevotella* spp. isolated from patients with periodontitis in Spain [421]. As seen in other case studies (Figures 4.4 and 4.5), ASVs classified as *Paeniclostridium* and *Romboutsia* (Gram-positive) were absent or in low abundance from the S2 sample from one and two cows, respectively. ASVs belonging to these two genera varied across treatment groups within this study.

It was hypothesised that treatment with a third-generation cephalosporin (ceftiofur) would increase the prevalence of AmpC and/or ESBL-producing *E. coli* in bovine faeces. Therefore, faecal enrichments from cows treated with ceftiofur (n=3) and associated control cows (n=3) were plated on agar selective for cephalosporin-resistant *Enterobacteriaceae*. *M. morganii* were isolated from the pre-treatment sample for control cow 12C (Table 4.1). Inducible AmpC- $\beta$ -lactamases are frequently found in *M. morganii* and confer resistance to third-generation cephalosporins [134]. Interestingly, emergent AmpC-producing *E. coli* were isolated during treatment (n=6, 48 hours after ceftiofur treatment) and posttreatment (n=4, 4 days after treatment) for cow 12. Molecular characterisation confirmed that the AmpC-producing *E. coli* harboured the plasmid-mediated *bla*<sub>CMY-2</sub> gene, which is the most prevalent pAmpC gene globally [131] and were isolated from seven samples during the longitudinal study (Chapter 3, Table 3.9).

The genomes of three E. coli were sequenced and all were identified as the same serotype and ST (O160:H32 and ST5514). No AmpC-producing  $E. \ coli$  of this serotype or ST were isolated from the longitudinal study (Chapter 2). To the best of my knowledge this is the first report of an ST5514 AmpC-producing E. coli and only one E. coli with this sequence type has been reported on Enterobase (https://enterobase.warwick.ac.uk/species/index/ ecoli; accessed 06 December 2021). The plasmid incompatibility factors IncFIB and IncI1 were detected in the three AmpC-producing E. coli. IncFIB was found in two chromosomal AmpC-producing  $E. \ coli$  isolated from Dairy 1 in the longitudinal study and all plasmidmediated AmpC-producing E. coli had the IncI1 plasmid incompatibility factor (Chapter 3, Table 3.5). The  $bla_{\rm CMY-2}$  gene has been identified on IncI1 plasmids in E. coli isolated from humans, pigs, turkey and various meats (chicken, turkey and pork) [360]. The two E. *coli* identified as the most genetically similar according to the core genome SNP analysis (Table 4.4; isolates DG079c and DG082f) were isolated from samples S2 and S3, and were more similar compared to the two E. coli isolated from sample S2. This finding was consistent with the AST zone sizes (Table 4.2) and suggests that genetically similar AmpC-producing *E. coli* could be isolated from bovine faeces two and four days posttreatment with ceftiofur. Additional samples were not collected post-treatment, therefore it is not known how long the emergent AmpC-producing E. coli persisted and were shed in the faeces of cow 12. There was an increased risk of detecting AmpC-producing E. coli in the facees of dogs treated with cefalexin (first-generation cephalosporin) or cefovecin (third-generation cephalosporin) in a clinical setting immediately after treatment, however one-month post-treatment the risk of detection was not significant [422].

An increase in third-generation cephalosporin resistant *E. coli* in cattle faeces has been identified during and post-treatment with ceftiofur in previous culture-based studies. A pair-matched longitudinal study (n=124 pairs) of cows treated with two-doses of ceftiofur crystalline-free acid across USA dairy farms (n=3) found that third-generation

cephalosporin resistant *E. coli* were higher in the treated group compared to to the untreated group (1.5  $\log_{10}$  colony forming units [CFU]) by day six of the study, and remained higher at day 16 (0.5  $\log_{10}$  CFU) [423]. By days 28 and 56 the population of third-generation cephalosporin resistant *E. coli* in the treated group had decreased and was comparable with the untreated group [423]. The molecular mechanisms for the thirdgeneration cephalosporin resistance was not investigated. Interestingly, a higher abundance of third-generation cephalosporin resistant *E. coli* was identified in cows with a lactation number >3 (compared to first lactation) [423], which may be due to pre-exposure to antimicrobials. *E. coli* with reduced susceptibility to ceftriaxone (third-generation cephalosporin) were more likely to be isolated from farms that reported using ceftiofur in the USA (n=18 farms), however no associations were identified at the individual animal level in cows treated with ceftiofur within six months of sampling [424]. In comparison, no association was identified between ceftiofur use and cephalosporin resistance in *E. coli* across 42 farms across the USA [425]. However, the aforementioned study only examined ceftiofur use on farm [425], and the total AMU and other antimicrobials were not included in the analysis.

Similar to this study, faecal samples from ceftiofur treated (n=5) and untreated (n=5) cattle on a dairy farm in the USA were analysed for ceftiofur resistant *E. coli* [426]. Cows were housed in a barn and treated for infertility due to *Leptospira borgpetersenii* serovar Hardjo-bovis infection. No ceftiofur resistant *E. coli* were isolated from faecal samples collected from untreated cows (0 of 265 *E. coli*) or before treatment for treated cows (days -1 and 0), and were only isolated from faecal samples collected during and directly post-treatment from treated cows (12 of 203 *E. coli*) using MC agar without antibiotics [426]. The  $bla_{\rm CMY-2}$  gene was detected in all 12 ceftiofur resistant *E. coli* and was not detected after day 6 (cows sampled post-treatment at days 5 - 11, 14, 18, 25 and 32). In addition, total *E. coli* counts decreased during treatment in treated cows, however the number of resistant *E. coli* did not increase in treated cows and was not persistent post-treatment [426].

Genome sequencing revealed the *E. coli* (n=3) carried the aph(6)-*Id* gene, which encodes a streptomycin phosphotransferase that has been shown to confer resistance to streptomycin [412], yet all ten of the *E. coli* were susceptible to streptomycin in phenotypic testing (Table 4.2). Studies have shown that APH(6) enzymes are less efficient at inactivating streptomycin [412] and the aph(6)-*Id* gene is often found in combination with other aminoglycoside resistance genes such as aph(6)-*Id*, aph(3"), and ant(3"), suggesting that additional 3'-phosphotransferase or 3'-adenylyltransferase activity may be required for sufficient streptomycin inactivation in the presence of aph(6)-*Id* [412]. This finding highlights the necessity to confirm a resistance phenotype and caution should be used when reporting a resistance phenotype predicted from genotypic information.

Isolation of plasmid-mediated AmpC-producing  $E. \ coli$  post-treatment with ceftiofur is a public and animal health concern due to the health impacts of resistant infections and the potential dissemination of pathogenic bacteria harbouring clinically relevant ARGs. The

lack of detection of AmpC-producing *E. coli* pre-treatment, and culture-based isolation and molecular detection from enrichment broths during and post-treatment potentially represents an interesting case of "AMR in action" and corroborates the recommendation to reduce the use of critically important antimicrobials such as third- and fourth-generation cephalosporins [155].

## 4.5 Conclusion

This research assessed the impact of systemic antimicrobial treatment on the bovine faecal microbiome in a natural environment by studying diseased animals treated with a range of antimicrobials, providing important information in a natural or "uncontrolled" setting. A decrease in bacterial microbiome diversity and richness was observed (measured as Shannon and Chao1  $\alpha$ -diversity, respectively) during systemic antimicrobial treatment, but the  $\alpha$ -diversity had often recovered post-treatment (S3) when the cow re-entered the milking herd. In general, the faecal microbiome of control cows was more stable compared to those which were treated with systemic antimicrobials, with more perturbations observed during treatment (S2) compared to samples S1 and S3 for treated cows. Although perturbations in the microbiome composition and the ability of microbiomes to recover was specific on a case by case basis, highlighting that the individual animal is the main driver of variation, additional factors such as disease severity, the prescribed antimicrobial treatment and duration and changes in farm management factors such as feed may also impact the bovine faecal microbiome.

Emergent AmpC-producing E. coli were isolated from faecal enrichments collected during (S2) and post (S3) treatment with ceftiofur for one cow (1 of 3 cows; 33.3%) and no third-generation cephalosporin resistant E. coli were isolated from pre-treatment nor from any untreated control cow samples. This finding represents an interesting case of "AMR in action" and indicated that either a resistant E. coli population was present in low abundance pre-treatment at undetectable levels using culture-methods and was able to flourish post ceftiofur treatment or the antimicrobial resistant E. coli strain was able to successfully colonise the bovine gut in the 48 hours between the S1 (pre-treatment) and S2 (during treatment) sample occasions due to reduced competing bacterial microbiota. The isolation of plasmid-mediated AmpC-producing E. coli which were genetically similar during and post-treatment has implications for the development and dissemination of clinically relevant ARGs as well as supporting the need to limit the use of critically important antimicrobials to reduce the development and dissemination of AMR. Future research should focus on any changes in the abundance of ARGs or antimicrobial resistant bacteria during (S2) and post-antimicrobial treatment (S3) by analysing the abundance of specific ARGs in boiled DNA lysate preparations from enriched faecal samples S1, S2 and S3 from treated and control cows using real-time PCR and plating faecal enrichments on various selective agars, respectively. In addition, this work and the example of "AMR in action"

should be communicated through a publication and by discussion with veterinarians and farmers' to encourage prudent antimicrobial stewardship in NZ dairy farming, particularly reducing the use of critically important antimicrobials.

## Chapter 5

# A longitudinal study of the resistome on two New Zealand dairy farms

#### 5.1 Introduction

The development and transmission of antimicrobial resistance (AMR) is a serious global public and animal health concern. Antimicrobial resistance genes (ARGs) have been identified in varying abundance across numerous environments, including urban sewage [427, 428], hospital wastewater [429], agricultural environments such as dairy farms [430, 431] and soil samples with low anthropogenic impact [432]. Traditional methods for AMR surveillance have focused on culture-based screening of specific bacterial pathogens such as Escherichia coli [194], or polymerase chain reaction (PCR) of a limited number of target ARGs [433]. Advances in next-generation sequencing technologies and a reduction in their costs [30] has led to an increase in the number of studies utilising shotgun metagenomic sequencing to study AMR. This has enabled a deeper understanding of the bacterial taxa and ARGs present in various ecosystems. An advantage of shotgun metagenomic sequencing methods is the depth of sequencing data which can be achieved, allowing for detection of genes of interest that may be present in low abundances [236]. If the sequencing depth is sufficient, it can allow for the genomic context of ARGs to be determined. Acquired ARGs can be transferred via horizontal gene transfer (HGT) and are of particular concern due to their ability to disseminate within bacterial populations. This contextual inference is crucial as it is important to assess (i) whether the ARGs are transferable and (ii) which bacterial species harbour the resistance gene(s) to assess the public and animal risk associated with each specific ARG [246].

Globally, ARGs have been detected in the dairy farm environment including in faeces, farm dairy effluent (FDE) and soil [430, 431, 218, 244] as well as raw bulk tank milk [434]. Although the misuse and overuse of antimicrobials in human and animal health have been suggested as the main drivers of AMR [1], the human, animal and environment AMR nexus is complex and numerous factors are involved. For example, heavy metal [280, 279, 301, 302] and biocide use [281] may co-select for AMR, and farm management factors such as buying in cattle or bulls [284] and feeding waste milk to calves [435, 436] may influence the prevalence and shedding of antimicrobial resistant bacteria in the dairy farm environment. The factors which may influence AMR are not constant, such as seasonality changes, farm management practices and antimicrobial use. For example, the majority of antimicrobial usage (AMU) on New Zealand (NZ) dairy farms is for mastitis treatment and prevention [21], and therefore AMU is likely to be higher post-lactation. A recent study suggested that single time point sampling or collecting samples from a limited number of locations on a farm may be insufficient to accurately estimate the prevalence of antibiotic resistant *E. coli* [195]. Therefore, longitudinal study designs may be more suitable for AMR surveillance in agricultural environments, as they account for changes in AMR prevalence, seasonality, AMU and farm management practices.

NZ dairy farm management practices differ compared to international systems in that they are largely pasture-based [437], have a larger average herd size (431 cows) [438] and a low prevalence of diseases such as coliform mastitis [19]. AMU for growth promotion is banned in NZ [41]. The New Zealand Veterinary Association (NZVA) has aspired that by 2030 antimicrobials will not be required for the maintenance of animal health and welfare [439] and instead their use will be reserved for the treatment of disease. As such, the NZVA have categorised antimicrobials used in animal medicine with a tier traffic light system [388], with the aim of reducing the use of antimicrobials which are critical in treating refractory conditions in human and animal health. Compared to many international systems, NZ uses a low amount of antimicrobials in food-producing animals [16, 17].

It can therefore be hypothesised that the prevalence of ARGs in the NZ dairy farm environment will be low. The aim of this study was to assess the abundance and diversity of ARGs in two NZ dairy farm environments, taking into account changes in seasonality, AMU and farm management practices. Few longitudinal studies have investigated ARGs in NZ dairy farm environments, and to the best of my knowledge this is the first study to utilise shotgun metagenomic sequencing technology for ARG surveillance in NZ dairy farm systems.

### 5.2 Methods

#### 5.2.1 Study population, sample collection and processing

The study population and sample collection methods are outlined in section 3.2.1. Environmental samples for DNA extraction were promptly stored at -80°C to prevent DNA degradation and changes to the bacterial community composition. For faeces, a pea-size amount of each faecal sample collected per sampling visit (n=16) was transferred and homogenised in a 15 mL centrifuge tube (Greiner, Sigma-Aldrich, St. Louis, Missouri, USA). Composite faecal material was weighed and 0.25 g stored in triplicate at -80°C. Soil samples were homogenised by hand in the Whirl-Pak<sup>®</sup> bag (Nasco, Fort Atkinson, Wisconsin, USA), and 0.25 g was stored in triplicate at -80°C. In addition, 1 g of both faecal and soil samples were stored in a cryovial at -80°C. To concentrate the microbial DNA and reduce

the protein and fat content, approximately 400 mL of milk was centrifuged at 10,000 x g (Sorvall LYNX 4000 Superspeed Centrifuge, Thermo Fisher Scientific, Waltham, Massachusetts, United States) for 45 min at 4°C. The fat was scooped off and 95% of the supernatant was decanted and the pellet was re-suspended in the remainder of the supernatant. The re-suspended pellet mix was transferred to a 50 mL centrifuge tube (Greiner, Sigma-Aldrich, St. Louis, Missouri, USA) and centrifuged at 10,000 x g for 10 min at 4°C. The supernatant was discarded, and the pellet washed twice in 5 mL PBS (pH 7.4) (Thermo Fisher Scientific, Waltham, Massachusetts, United States) at 10,000 x g for 10 min at 4°C. The re-suspended pellet was transferred and stored in aliquots at -80°C. To concentrate the FDE (based on the method described in [244]), approximately 400 mL of FDE was centrifuged at 10,000 x g (Sorvall LYNX 4000 Superspeed Centrifuge, Thermo Fisher Scientific, Waltham, Massachusetts, United States) for 20 min at 4°C. After, 95% of the supernatant was decanted and the pellet was re-suspended in the remainder of the supernatant and stored at -80°C.

#### 5.2.2 DNA extraction

Samples for DNA extraction were removed from the -80°C freezer, defrosted on ice and transferred to a bead-beating tube for DNA extraction. Genomic DNA was extracted using the Presto Stool DNA Extraction Kit (Geneaid Biotech Ltd, New Taipei City, Taiwan) according to the manufacturer's instructions with minor modifications. Briefly, 0.25 g faeces and soil and 200 µL of milk and FDE was used for the DNA extraction. For sample lysis, 800 µL of ST1 buffer was added, mixed by vortexing and incubated at 70°C for 5 min. The sample was vortexed at maximum speed for 7 min (Vortex Mixer, Labret International, New Jersey, USA) and then centrifuged at  $8,000 \ge g$  for 2 min. Next, 500 µL of supernatant was transferred to a new tube and 5 µL RNAse A (100 mg/mL; QIAGEN, Hilden, Germany) added and incubated for 10 min at 37°C. To remove PCR inhibitors, 150  $\mu$ L of ST2 buffer was added and mixed briefly by vortex. The sample was incubated at 0 -4°C for 5 min and subsequently centrifuged at 16,000 x g for 3 min. Next, 500  $\mu$ L of clear supernatant was transferred to a PCR inhibitor removal column, centrifuged at  $16,000 \ge q$ for 1 min and the column discarded. For DNA binding, 800 µL of ST3 buffer was added to the flow through, mixed vigorously by hand for 5 sec and 700  $\mu$ L transferred to a GD column. The sample was centrifuged at 16,000 x g for 1 min, the flow through discarded and the remaining sample mixture transferred to the GD column. The GD column and sample was again centrifuged at  $16,000 \ge g$  for 1 min and flow through discarded. To wash the DNA, 400  $\mu$ L of ST3 buffer was added to the GD column and centrifuged at 16,000 x g for 30 sec. The flow through was discarded, 600  $\mu$ L of wash buffer added and centrifuged at  $16,000 \ge q$  for 30 sec. The flow through was discarded and the wash step was repeated. After the final wash step, the GD column was centrifuged at  $16,000 \ge g$  for 3 min to dry the column.

To elute the DNA, the GD column was transferred to a new tube, 30  $\mu L$  of pre-heated (60°C)

elution buffer was added to the centre of the column and left to stand for 2 min. The sample was then centrifuged at 16,000 x g for 2 min and the elution step repeated using the elution buffer containing the eluted DNA. The DNA was transferred to a new microcentrifuge tube and stored at -20°C. All centrifuge steps were carried out at room temperature. The DNA concentration was quantified using a Qubit 4.0 fluorometer (ThermoFisher Scientific Inc., USA) and  $A_{260/280}$  and  $A_{260/230}$  ratios determined using the Nanodrop microvolume spectrophotometer (Nanodrop 2000c, ThermoFisher Scientific Inc., USA). DNA integrity and size was visualised on a 0.8% [w/v] agarose gel which was prepared using 0.8 g agarose (Invitrogen, Auckland, New Zealand) and 100 mL of 0.5x Tris-borate-EDTA buffer. The gel was stained with RedSafe (Invitrogen, Auckland, New Zealand) and run at 100V for 4 hours. A high molecular weight *Hind* III/ $\lambda$  ladder was used for size comparison (ThermoFisher Scientific Inc., USA) and gel images were captured and stored using the GelDoc XR<sup>+</sup> system (BioRad, New Zealand).

#### 5.2.3 Shotgun metagenomic sequencing

The metagenomic sequencing run (faeces n=30; soil n=30; FDE n=28; milk n=25 and waste milk n=1) was performed on an Illumina Novaseq S4 platform with 2 x 150 pairedend reads ( $\geq$ 40 million paired-end reads or  $\approx$ 13 Gb per sample). The libraries were prepared using the NEBNext<sup>®</sup> DNA Library Preparation Kit (New England Biolabs, Inc, Ipswich, Massachusetts, USA) and sequenced by Novogene, Singapore. The following controls (n=10) were included in this study: (i) blank reagent control for each DNA extraction kit batch (n=2), (ii) phosphate buffered saline (PBS) (n=2) used for preparation of the milk samples and (iii) ZymoBIOMICS<sup>TM</sup> Microbial Community DNA Standard (D6305; n=3) [440] and ZymoBIOMICS<sup>TM</sup> Microbial Community DNA Standard II (log distribution) (D6311; n=3) [441].

#### 5.2.4 Bioinformatic analysis

The shotgun metagenomic sequencing analysis was based on a previously described workflow by Liu *et al.*, [11] and the analysis was run on the AgResearch Ltd. High Performance Computing servers. The key steps of the workflow are outlined in Figure 5.1 and the bioinformatic pipeline is detailed here. Briefly, TrimGalore v0.6.6 [442] was used for raw read trimming and quality assessment, using Cutadapt v1.18 [443] and FastQC v0.11.9 [444], respectively. A Phred quality score threshold of 20 was used. Host contamination (*Bos taurus*) was removed by aligning the reads to the bovine genome UMD3.1.1 (accession number: AAFC00000000.3) using BMTagger in bmtools v3.101 [445]. For the ARG normalisation calculations, the number of 16S rRNA genes per sample were identified using METAXA2 v2.2 [446]. The average genome size was estimated using MicrobeCensus v1.1.1 [447], with the number of reads set to 100,000,000 to utilise the maximum reads in each sample. The processed sequencing reads were taxonomically classified using Kraken2 v2.1.1 [448] with the Kraken2 standard database v20200919. The relative abundance of taxa within the samples was estimated at various classification levels using Bracken v2.6.0 [449] and phyla present in  $\leq 5\%$  total abundance were grouped together as "Other" for subsequent analyses. The processed reads were assembled into contigs using MEGAHIT v1.2.9 [450] with default parameters.

#### 5.2.5 Resistance gene analysis

The resistome (collection of all ARGs) in the processed sequencing reads was analysed following the AMR++ pipeline (https://megares.meglab.org/amrplusplus/latest/html/index. html) with minor modifications. Briefly, the antimicrobial, biocide, and heavy metal resistance genes were identified by mapping the processed sequencing reads to the ARG MEGARes database v2.0 [451] using BWA v0.7.17 with default settings [452]. The SAM formatted alignment files were analysed using ResistomeAnalyzer v2018.09.06 (https:// github.com/cdeanj/resistomeanalyzer) to generate the sample resistome for each level of the database hierarchy (gene, group, mechanism and class). For gene-level analysis, a gene fraction threshold of 80% was used to reduce false positive hits. Rarefaction analysis was performed per sample to determine whether the sequencing depth used in this study was sufficient to detect the antimicrobial, biocide and heavy metal resistance genes present. The SAM formatted alignment file was used as input for RarefactionAnalyzer v2018.09.06 (https://github.com/cdeanj/rarefactionanalyzer), with sub-sampling of sequencing reads at 5% increments and a gene fraction threshold of 80%.

To allow for more accurate comparisons, ARG gene-level data was normalised to avoid bias associated with ARG size and the microbial load per sample. ARG abundance was expressed as "copy of ARG per copy of 16S rRNA gene" and normalisation calculations performed as previously described by Li *et al.*, [453] using the formula described below. In equation 5.1, <sup>*n*</sup> represents each ARG, N<sub>ARG-like sequence</sub> indicates the number of reads annotated as a specific ARG; L<sub>ARG reference sequence</sub> is the sequence length (bp) of the corresponding ARG sequence in the MEGARes database v2.0; N<sub>16S rRNA sequence</sub> is the number of reads mapping to the 16S rRNA gene in the sample as identified by METAXA2; L<sub>16S rRNA sequence</sub> was set to be 1,432 bp which corresponds to the average length of 16S rRNA gene in the Greengenes database [454] and L<sub>reads</sub> is the length of the sequencing reads used in this study (150 bp).

Abundance = 
$$\sum_{1}^{n} \frac{N_{ARG-like \ sequence} \times L_{reads} \ / \ L_{ARG \ reference \ sequence}}{N_{16S \ rRNA \ sequence} \times L_{reads} / L_{16S \ rRNA \ sequence}}$$
(5.1)

Acquired ARGs were identified with Abricate v1.0.1 [https://github.com/tseemann/abricate] [327] using the MEGARes database v2.0 [451]. Contigs containing acquired ARGs were taxonomically classified with the Contig Annotation Tool (CAT) v5.2.1 [455] using NCBI taxonomy files and the NCBI nr database (generated 2021-01-07) with default settings.


#### 5.2.6 Statistical analysis and data visualisation

Statistical tests and data visualisation was conducted in R Core v4.0.2 [396] and RStudio v1.3.959 [456]. Permutational multivariate analysis of variance test (PERMANOVA) was implemented in the vegan package in R [401] and was used to test the effects of farm (Dairy 1 or Dairy 4) and AMU (mg/PCU per month) and the interactions of these variables on normalised ARGs (copies of ARG per 16S rRNA gene) in each sample using a generalised mixed effect linear model. Sample type (faeces, FDE, milk and soil) was included in the model as a random effect. Only genes in which the normalised ARG value was >0 for at least one sample was included in the analysis. Post-hoc analysis was calculated using the R package emmeans [410] and pairwise comparisons adjusted using the Tukev method. Statistical tests for the resistome analysis were performed in Minitab<sup> $\mathbb{R}$ </sup> 19.1.1 [347] using a generalised linear model with post-hoc analysis using the Tukey method and a 95% confidence interval. Differences in microbiome profiles based on Bray-Curtis distance measures were analysed using non-metric multidimensional scaling (NMDS) in the vegan package [401]. Data visualisation was conducted in R using a range of packages including ggplot2 [403], tidyverse [404], dplyr [405], Manu [406], Colorgorical [408] and RColorBrewer [407].

## 5.3 Results

In this study, 123 metagenomic libraries were sequenced consisting of faeces (n=30), soil (n=30), FDE (n=28), milk (n=25) and waste milk (n=1) samples collected over a 15 month period (sample metadata here). The following controls were also included: a mock community DNA standard (n=3), mock community log distribution standard (n=3), PBS (n=2) and negative blank reagent (n=2) controls. The milk sample collected in October 2019 from Dairy 1 was waste milk (DF0167), rather than bulk tank milk. The sample collection issue was discovered post-sequencing, when anomalies were detected in this sample such as a high abundance of ARGs as well as multiple assembled contigs belonging to mastitis-associated pathogens including *Streptococcus uberis*. Therefore, 25 bulk tank milk (referred to hereafter as milk) and one waste milk sample were included in the study. The shotgun metagenomic libraries were analysed to determine their microbiome composition, abundance and distribution of ARGs within the farm environment as well as identifying the bacterial hosts carrying acquired ARGs. A comparison of farm management practices and AMU between the two farms included in this study is detailed in sections 3.2.1 and 3.10.

#### 5.3.1 Microbial community composition

At the phylum level, the microbiome composition was similar across the faecal samples (n=30) from both farms, with 23 phyla identified, excluding phyla present in  $\leq 5\%$  total abundance which were grouped together as "Other" (Appendix T). The phylum Firmicutes (45.2%; range 42.5% - 48.5%), Bacteroidetes (20.4%; range 18.4% - 23.9%), Proteobacteria (17.7%; range 15.6% - 19.7%) and Actinobacteria (8.6%; range 6.9% - 11.8%) were dominant in total abundance across the samples. The same four phyla were also the most abundant in the FDE samples, however the total abundances varied (Proteobacteria [45.1%; range 23.7% - 87.57%], Actinobacteria [17.4%; range 2.7% - 47.4%], Firmicutes [16.6%; range 2.4% - 31.2%] and Bacteroidetes [11.9%; range 2.9% - 18.9%]) and the taxonomic composition was more diverse across the different FDE samples with 31 distinct phyla identified (Appendix T). In particular, the total abundance of Proteobacteria in FDE was the most variable over the 15 month sampling period (23.7 - 87.5%).

A total of 18 different phyla were identified across the soil samples (n=30) (excluding the phyla which were present in  $\leq 5\%$  total abundance grouped together as "Other" (Appendix T)). The microbial community composition was relatively similar across the soil samples at the phylum level, with the two dominant phyla Proteobacteria and Actinobacteria accounting for on average 57.7% (52.3% - 61.0%) and 27.8% (24.3% - 33.8%) of the total abundance, respectively. The predominant bacterial phyla in the milk samples were Proteobacteria and Firmicutes (Appendix T) and the total abundance of these phyla varied across the milk samples collected during this study. In general, either Proteobacteria (54.2%; range 0.6% - 89.5%) or Firmicutes (21.7%; range 0.7% - 68.6%) was dominant in each sample, which reflects the wide range in total abundance of these two phyla across the samples. The waste milk sample had a comparatively high total abundance of both Proteobacteria (59.8\%) and Firmicutes (33.9\%). The milk samples also had a high total abundance of the phylum Chordata (21.0\%; range 6.2\% - 32.5\%), which is likely remnants of bovine host contamination still present after read processing.

The microbiome profiles clustered by sample type, irrespective of farm (Figure 5.2). The faecal and soil microbiomes were homogeneous between the farms. According to the microbiome composition, the milk samples clustered in two groups, which was independent of farm or collection date. The two clusters were separated by the most abundant phyla in the samples (Appendix T), with Firmicutes being the predominant phyla in cluster one (n=7) and Proteobacteria in cluster two (n=18). The waste milk microbiome profile clustered separately from the other sample types. The FDE microbiome profiles were more diverse across both farms, reflecting the complex composition of FDE. On Dairy 1, the clustering of FDE microbiome profiles was independent of collection site (Table 3.1). Surprisingly, the microbiome profile of one FDE sample collected from Dairy 1 was more similar to the soil microbiomes compared to the other FDE samples. This was the last sample from the first collection site on Dairy 1, which was collected when there was issues with the FDE filtration system pump. The FDE in the sump during this collection period was visibly dryer than usual and therefore the FDE was likely more stagnant. Hence, a different microbiome composition was observed compared to the other FDE samples.



Figure 5.2: NMDS of microbiome profiles at the phylum level of farm samples based on Bray-Curtis dissimilarity matrix (k=3; stress=0.023). Samples are coloured by type and shapes represent farm, as indicated in the figure key.

#### 5.3.2 Resistome analysis

Shotgun metagenomic sequencing of faeces, FDE, soil, milk and waste milk samples from Dairy 1 and Dairy 4 identified 372 unique ARGs, representing 37 resistance classes belonging to the drugs, biocide, metal and multi-compound classes, as classified by the MEGARes2.0 database. Focusing on ARGs belonging to the drug and multi-compound classes (excluding biocide and heavy metal classes), the copies of ARG per 16S rRNA gene per sample ranged from 0.03 - 0.37 in effluent, 0.08 - 0.17 in faeces and the lowest and highest abundance in milk and soil, respectively (0.0 - 0.12; 0.20 - 0.63) (Figure 5.3). ARG abundance in soil was higher compared to FDE and faeces, despite harbouring fewer unique ARGs. Three FDE samples were identified as outliers in the boxplots, with a higher ARG abundance, two collected from Dairy 1 in in October 2018 (DF0009) and December 2019 (DF0188) and one from Dairy 4 in July 2019 (DF0115). The copies of ARG per 16S rRNA gene was not statistically significant between Dairy 1 and Dairy 4 over the 15 month period (p=0.321). ARG abundance between the FDE samples on Dairy 1 and Dairy 4 was statistically significant (p = 0.010) but the ARG abundance between soil (p = 0.271), faces (p=0.145) and milk (p=0.872) samples from Dairy 1 and Dairy 4 was not statistically significant. Between all sample types, ARG abundance was statistically significant  $(p = \langle 0.001 \rangle)$ , except for the FDE and faces resistomes (p = 0.954). Compared to all other sample types, the waste milk sample (n=1) had a higher abundance of antimicrobial, heavy metal and biocide resistance genes, with 0.0 - 3.05 copies of resistance genes per 16S rRNA gene. The antimicrobial, heavy metal and biocide resistance gene abundance (copies of ARG per 16S rRNA gene) at the class level is shown in Figures 5.5 and 5.6.



Figure 5.3: Normalised antimicrobial resistance gene abundance (copies of ARG per 16S rRNA gene) in farm dairy effluent, faeces, soil and milk samples collected over a 15 month period on Dairy 1 and Dairy 4. The boxes show the median and upper and lower quartiles. The whiskers show the minimum and maximum values within the interquartile range and the outliers are indicated by a black dot.



Figure 5.4: Relative abundance of antimicrobial resistance genes in farm dairy effluent, faeces, soil and milk samples on Dairy 1 and Dairy 4. ARGs are classified by antibiotic class (including any ARG conferring resistance across multiple classes but excluding genes only involved in biocide and heavy metal resistance) as indicated in the figure legend.





The number of unique ARGs identified in the five sample types varied (Figure 5.4); FDE samples harboured the most diverse range of ARGs, with 164 unique ARGs representing 19 resistance classes identified, followed by faeces and soil with 51 and 30 ARGs, representing 10 and 11 resistance classes, respectively. The fewest unique ARGs (n=3) were identified in the milk, representing only three resistance classes. In the waste milk sample, 81 unique ARGs belonging to 14 resistance classes were identified (Figure 5.9).

The ARGs detected in faeces on both farms predominantly belonged to the  $\beta$ -lactam or macrolide, lincosamide and streptogramin (MLS) antibiotic classes. The predominance of ARGs belonging to the  $\beta$ -lactam or MLS antibiotic classes is likely due to the presence of the *cfxA* and *lnuC* genes respectively, which were identified in all faecal samples (n=30; Appendix U). The *cfxA* gene encodes a class A  $\beta$ -lactamase found on a mobilisable transposon detected in *Bacteroides* species [457] and the abundance of the *cfxA* gene ranged from 0.07 - 0.14 copies of ARG per 16S rRNA gene. The *lnuC* gene encodes a lincosamide nucleotidyltransferase, which was first characterised in a clinical *Streptococcus agalactiae* on a transposon-like element [458]. In comparison to the *cfxA* gene, the *lnuC* gene was detected at a lower abundance ranging from 0.01 - 0.03 copies of ARG per 16S rRNA gene.

The FDE samples contained the most diverse range of ARGs and the predominant antibiotic classes included the  $\beta$ -lactams, MLS and aminoglycoside classes. Resistance genes potentially conferring resistance to multi-compounds were also identified (drug/biocide resistance and biocide/metal resistance). Although the ARGs identified in FDE were more diverse, they were generally detected at a relatively low abundance (compared to the soil resistome which had the highest ARG abundance in this study). The *cfxA* gene group was identified in 27 of 28 of FDE samples (0.0 - 0.10 copies of ARG per 16S rRNA gene), and the *lnuC* and *sodB* genes belonging to MLS and peroxide resistance classes, were identified in 27 of 28 and 25 of 28 of FDE samples respectively, albeit at a low abundance (0.0 - 0.01 and 0.0 - 0.06 copies of ARG per 16S rRNA gene, respectively; Appendix U).

In soil samples, the resistance classes identified were comparatively similar between Dairy 1 and Dairy 4, with ARGs predominantly belonging to multi-drug resistance, drug/biocide resistance, glycopeptides, aminocoumarins, MLS and rifampin resistance classes. ARGs were detected in a limited number of milk samples (1 of 11 on Dairy 1; 2 of 13 on Dairy 4) and they belonged to trimethoprim, aminoglycoside and drug/biocide resistance classes and the abundance of these genes was very low (0.00 - 0.12, 0.00 - 0.05, 0.00 - 0.01 copies of ARG per 16S rRNA gene, respectively). The gene dfrB potentially conferring trimethoprim resistance, was identified from a single milk sample from both Dairy 1 (DF0012) and Dairy 4 (DF0001). The same milk sample from Dairy 4 also harboured the *ant6* and  $qac \Delta 1$  genes, belonging to aminoglycosides and drug/biocide resistance classes, respectively. The *ant6* gene was also identified from another single milk sample from Dairy 4 (DF0184).

The abundance of heavy metal and biocide resistance genes between Dairy 1 and Dairy 4 (Figure 5.7) was not statistically significant (p=0.712). However, heavy metal and biocide resistance gene abundance between the four sample types was statistically significant

(faeces and effluent (p= 0.028), soil and milk (p= 0.005) and all other combinations (p= <0.001)). Heavy metal and biocide resistance gene abundance in FDE was statistically significant between the farms (p= 0.003), with a higher abundance detected in Dairy 1 (0.01 - 0.33 copies of ARG per 16S rRNA gene) compared to Dairy 4 (0.00 - 0.08 copies of ARG per 16S rRNA gene). Across the FDE samples, 149 unique heavy metal and biocide resistance genes were detected representing 14 resistance classes (Figure 5.8). The *qac* gene encoding resistance to quaternary ammonium compounds which are commonly used in disinfectants, were identified in FDE sporadically on both farms (Appendix U).

In the milk samples, a higher abundance of heavy metal and biocide resistance genes were identified on Dairy 4 (0.09 - 0.59 copies of ARG per 16S rRNA gene) compared to Dairy 1 (0.00 - 0.19), which was statistically significant (p= 0.012). Interestingly, one peroxide resistance gene was detected (*sodB*) on Dairy 4 and the remaining resistance genes all conferred mercury resistance (*merABCDEFPRT*, Appendix U). The abundance of heavy metal and biocide resistance genes in faeces was very low on both farms (0.00 -0.01 copies of ARG per 16S rRNA gene; p= 0.972). Heavy metal and biocide resistance gene abundance was highest in soil samples compared to the other sample types and the abundance between soil on Dairy 1 (0.18 - 0.43 copies of ARG per 16S rRNA gene) and Dairy 4 (0.00 - 0.37 copies of ARG per 16S rRNA gene) was not statistically significant (p=0.093). Genes representing a diverse range of resistance classes were identified from the soil samples, with the most resistance genes belonging including copper, peroxide and iron resistance classes.



Figure 5.7: Normalised heavy metal and biocide resistance gene abundance (copies of ARG per 16S rRNA gene) in farm dairy effluent, faeces, soil and milk samples collected over a 15 month period on Dairy 1 and Dairy 4. The boxes show the median and upper and lower quartiles. The whiskers show the minimum and maximum values within the interquartile range and the outliers are indicated by a black dot.



Figure 5.8: Relative abundance of heavy metal and biocide resistance genes in farm dairy effluent, faeces, soil and milk samples on Dairy 1 and Dairy 4. Resistance genes are classified by resistance class as indicated in the figure legend.



Figure 5.9: Antimicrobial, heavy metal and biocide resistance gene abundance (copies of ARG per 16S rRNA gene) identified in the waste milk sample. Resistance genes are classified at the class level.

From the waste milk sample, 103 unique antimicrobial, heavy metal or biocide resistance genes were identified representing 14 resistance classes including copper, acid, nickel, zinc, arsenic and peroxide (Figure 5.9). Interestingly, the most abundant resistant classes in the waste milk were often multi-compound such as multi-metal (3.05 copies of ARG per 16S rRNA gene), drug/biocide (2.97 copies of ARG per 16S rRNA gene), biocide/metal (1.18 copies of ARG per 16S rRNA gene) and multi-biocide (0.78 copies of ARG per 16S rRNA gene), drug/biocide/metal (0.70 copies of ARG per 16S rRNA gene) and multi-drug (0.27 copies of ARG per 16S rRNA gene) and multi-drug (0.27 copies of ARG per 16S rRNA gene) classes were also relatively abundant.

No significant difference in antimicrobial, heavy metal or biocide resistance gene abundance was observed between farms and differences in AMU (p=0.6264), taking into account the random effect of sample type (faeces, FDE, soil or milk). PERMANOVA analysis suggested that AMU did not account for the total variation in ARG abundance between the samples.

#### 5.3.3 Bacterial host range harbouring contigs containing acquired ARGs

The detection of acquired ARGs is a concern due to their ability to spread via HGT and thus the potential for the development of multi-drug resistant phenotypes to emerge. Across the sequenced farm samples (n=113), 147,822,721 contigs were assembled. Of these, 1,014 contigs (0.0007%) harboured at least one acquired ARG and were assembled from 88 metagenomic samples (88 of 113; 78.8%). A total of 200 contigs could be classified at the phylum level (200 of 1,014; 19.7%). The number of contigs which were taxonomically assigned reduced as the classification levels decreased down to the class (140 of 200; 70.0%), order (67 of 200; 6.6%), family (58 of 200; 29.0%), genus (42 of 200; 21.0%) and species (12 of 200; 6.0%) levels. The predicted taxonomic ranks of contigs harbouring acquired ARGs at the class, order, genus and species level is shown in Appendix V. The remaining contigs could not be taxonomically classified at a lower taxonomic rank with the classification parameters used (section 5.2.5).

Within the 200 contigs originating from 70 metagenomic samples (Figure 5.10), 57 unique ARGs were identified, representing 14 antibiotic classes. Seven contigs (7 of 200; 3.5%) coharboured multiple ARGs (Table 5.1). Interestingly, contigs co-harbouring the aph(3") and aph6 gene groups belonging to Proteobacteria were assembled from FDE samples on both Dairy 1 and Dairy 4 in November 2018 and March 2019, respectively. Three contigs coharbouring the  $bla_{\rm PER}$  and fosA genes belonging to *Gammaproteobacteria* were assembled from FDE samples (n=2 samples) from November and December 2019. The  $bla_{\rm ACC}$  and crp genes were identified on a contig assembled from the waste milk sample and belonged to the bacterial order *Enterobacterales*. The  $bla_{\rm ACC}$  gene encodes a plasmid-mediated Class C  $\beta$ -lactamase.



TMP, Trimethoprim; WM, Waste milk.

Most of the acquired ARGs were on contigs that belonged to the Proteobacteria (30 genes) and Firmicutes (19 genes) phylum, followed by the Actinobacteria (6 genes), Bacteroidetes (3 genes) and Fibrobacteres (1 gene) phylum. At the phylum level, the majority of the contigs harbouring acquired ARGs were assembled from FDE samples (n=102 contigs), with a higher number of contigs assembled from Dairy 1 (n=57) compared to Dairy 4 (n=46). ARGs identified in contigs assembled from FDE samples potentially confer resistance across ten antimicrobial classes, with the highest number of gene groups belonging to drug and biocide resistance (n=9),  $\beta$ -lactams (n=7), MLS (n=5) and aminoglycosides (n=4). These genes were predominantly found in Proteobacteria but were detected in contigs belonging to Actinobacteria, Bacteroidetes and Firmicutes. In addition, a number of *qac* genes were found in contigs belonging to Proteobacteria from both farms. The second highest number of contigs harbouring acquired ARGs were assembled from soil samples (n=52 contigs) and the number of contigs from each farm was similar (Dairy 1) n=27; Dairy 4 n=25). Relatively fewer contigs with ARGs were assembled from faces samples (n=26), with a higher number from Dairy 4 (n=17) compared to Dairy 1 (n=9). No containing acquired ARGs were assembled from milk samples. From the waste milk sample, 19 contigs containing ARGs were assembled, with one contig co-harbouring two ARGs. The ARGs identified in the waste milk sample potentially confer resistance to the  $\beta$ -lactam, drug and biocide resistance and MLS classes from contigs belonging to the phylum Proteobacteria, as well as  $\beta$ -lactam, fosfomycin, glycopeptide, multi-drug, MLS and nucleoside classes detected in contigs belonging to the Firmicutes phylum.

Sample	Farm <sup>a</sup>	$Type^{b}$	Gene groups	Class	Classification <sup>c</sup>
DF0025	D1	FDE	aph(3"), aph6	Aminoglycosides	Proteobacteria
DF0068	D4	FDE	aph(3"), aph6	Aminoglycosides	Proteobacteria
DF0097	D4	Soil	emrAsm, emrBsm emrCsm, emrRsm	Drug and biocide	Stenotrophomonas
DF0167	D1	WM	$bla_{ACC}, crp$	$\beta$ -lactams Drug and biocide	Enterobacterales
DF0176	D1	FDE	$bla_{\rm PER}, fos A$	$\beta$ -lactams Fosfomycin	Gamma proteo bacteria
DF0188	D1	FDE	$bla_{\rm PER}, fos A$	$\beta$ -lactams Fosfomycin	Gamma proteo bacteria
DF0188	D1	FDE	$bla_{\rm PER}, fos A$	$\beta$ -lactams Fosfomycin	Gamma proteo bacteria

Table 5.1: Taxonomic classification of contigs co-harbouring two acquired resistance genes

<sup>a</sup> D1, Dairy 1; D4, Dairy 4.

<sup>b</sup> FDE, Farm dairy effluent; WM, Waste milk.

<sup>c</sup> Classification recorded for the lowest taxonomic rank identified.

The phylum Proteobacteria, which includes clinically relevant pathogens such as *E. coli*, *Klebsiella pneumoniae* and *Acinetobacter baumannii*, harbours the most diverse range of ARGs across the sample types. Proteobacteria was one of the predominant phyla found in the FDE samples (average: 45.1%, Appendix T), and the number of contigs containing

ARGs classified as Proteobacteria across FDE samples from both farms was high (Dairy 1 n=43; Dairy 4 n=26). Acquired ARGs from contigs belonging to the Proteobacteria phylum were found in the waste milk sample (n=4 contigs) and soil samples (n=4 contigs), although the number of contigs was lower from these sources. Samples which had a high ARG abundance at the sequencing read level (Figure 5.12) also had a higher number of contigs with acquired ARGs belonging to Proteobacteria (FDE: DF0176 and DF0188; waste milk: DF0167). Despite being a predominant phylum detected in the faecal microbiome composition analysis, no Proteobacteria contigs containing ARGs were detected in faecal samples from either farm during the study. Acquired ARGs were detected in contigs assembled from faecal samples belonged to Firmicutes (n=13 contigs), Actinobacteria (n=11 contigs), Bacteroidetes (n=1 contig) and Fibrobacteres (n=1 contig) phyla.

The phylum Firmicutes also harboured a diverse range of ARGs (n=19) and contigs were assembled from FDE (n=14 contigs), faces (n=13 contigs) and waste milk (n=14 contigs) samples. The lnuC gene, which encodes a lincosamide nucleotidyltransferase and was identified from all faecal samples (n=30) and most FDE samples (n=27; 96.4%), was identified in 14 contigs across FDE (n=4) and faces (n=10) samples. Four of these contigs could be classified at the family level, and belonged to the Lachnospiraceae, Ruminococcaceae and Sporomusaceae families (Figure 5.11). The identification of lnuC across multiple bacteria families may explain the high detection of this gene from faeces and FDE. Acquired ARGs were identified in contigs belonging to Bacteroidetes (n=17 contigs) from FDE (n=13 sam-)ples) and faeces samples (n=1 sample). Of these, 14 contigs harboured the  $bla_{OXA}$  gene and five contigs could be classified at the family level, belonging to *Flavobacteriaceae*. One contig harboured the tetX gene and belonged to the Sphingobacteriaceae family. Two contigs assembled from faces and FDE (n=1 each) harboured the  $bla_{CFX}$  gene and the cfxAgene was identified in all faecal samples (n=30) and most FDE samples (n=27; 96.4%) at the sequencing read level. These two contigs could not be classified at a lower taxonomic rank.

The phylum Actinobacteria, which was prevalent in the soil and FDE microbiomes (Appendix T), had a high number of contigs containing acquired ARGs across the soil samples on both farms (Dairy 1: 13 of 15 samples; 27 contigs; Dairy 4: 11 of 15, 21 contigs), as well as in faeces (Dairy 1: 2 of 15, 2 contigs; Dairy 4: 9 of 15, 9 contigs) and only FDE on Dairy 4 (4 of 15, 4 contigs). A range of acquired ARGs (six genes) were carried by contigs belonging to Actinobacteria, with *ileS* (encoding an isoleucyl-tRNA synthetase which confers resistance to mupirocin) found in faeces and FDE and *rbpA* (encoding an RNA-polymerase binding protein which confers resistance to rifampin) found in soil and FDE being the most prevalent. FDE samples from both farms harboured the most diverse host range of contigs with contigs belonging to *Gammaproteobacteria*, Actinobacteria, *Bacilli* and *Clostridia* which were also identified in other sample types, as well as *Betaproteobacteria*, *Flavobacteria*.





2

Effluent

D1

Zoogloeaceae Sporomusaceae Pseudomonadaceae

Sphingobacteriaceae

Flavobacteriaceae Comamonadaceae

Moraxellaceae

Chromatiaceae Cellvibrionaceae

Clostridiaceae

## 5.3.4 Examination of sequencing depth for the detection of antimicrobial, heavy metal and biocide resistance genes

To determine whether the sequencing depth used in this study was sufficient to detect the resistome in each sample, including resistance genes present in low abundance, rarefaction analysis was used which identifies the number of unique genes while by sub-sampling the sequencing reads at 5% increments. If the depth was sufficient to detect the total resistome, the number of unique genes should plateau after 100% of the reads have been sampled. Previous research has shown that the relative proportion of reads assigned to specific ARG classes was relatively consistent across various sampling depths in bovine faeces (average read counts used in the study: 26, 59 and 117 million), however increasing the sequencing depth resulted in higher numbers of ARG assigned reads [236]. An average sample read count of 59 million was suggested as suitable to sufficiently describe the resistome in bovine faeces [236]. Given that the abundance of ARGs will vary depending on sample type as well as population level factors, such as a hypothesised low abundance of ARGs in NZ dairy farm environments, the sequencing depth required to analyse the resistome is study-specific.

This study used a sequencing depth of  $\geq 40$  million read pairs per sample and rarefaction analysis suggested that this depth was sufficient to study the resistome in the FDE, faeces, milk and soil samples sequenced in this study. However the relative abundance of ARGs was higher in three FDE samples (DF0145, DF0176, DF0188) and the waste milk sample (DF0167) and the number of unique genes was still increasing after all of the reads had been sub-sampled (Figure 5.12). Compared to the other samples analysed in this study, the number of resistance genes was much higher (>200 genes) in these four samples. The three FDE samples identified as outliers compared to the other FDE samples were collected from Dairy 1 in September, November and December 2019 from the second collection point in the study which was from a grate in the cowshed compared to the effluent sump (see section 3.1). Interestingly, the samples collected in July, August and October 2019 from this location had lower antimicrobial, heavy metal and biocide resistance genes identified (<200 genes) and the number of unique genes appeared to plateau off, suggesting the sequencing depth was sufficient to detect the resistome within these samples. The waste milk sample contained a high number of ARGs, even when a low proportion of reads were sub-sampled.



Figure 5.12: Rarefaction analysis of antimicrobial, heavy metal and biocide resistance genes detected in sequencing reads. Samples with >200 genes detected are highlighted in colour as defined in the legend.

#### 5.3.5 Sequencing controls

Controls were included in the shotgun metagenomic sequencing runs in this project including a mock community DNA control to identify any biases introduced during the library preparation process and quantification of any sequencing errors and a mock community log distribution control which was included to assess the detection limits of the bioinformatics workflow. These two controls were run in triplicate across the two sequencing lanes. Blank reagent controls (n=2) of the two batches of DNA extraction kits and PBS controls (n=2) were included to identify reagent contamination associated with the DNA extraction kits and the PBS used to wash the milk pellet, respectively (section 5.2.1).

Sequencing of the microbial community mock DNA standard identified all eight bacteria (five Gram-positive, three Gram-negative) included in the control and the proportion of reads assigned to each bacterial species was similar to the expected values (Table 5.2). Reads classified as *Pseudomonas aeruginosa* and *S. enterica* (both Gram negative) were over-represented in the reads from the microbial community mock DNA standard compared to the expected proportion and the reads classified as *Bacillus subtilis* were much lower than expected (6.6% compared to 12%). However, 4.7% of reads in the mock community

control were classified as *Bacillus intestinalis* and these two species combined account for 11.3% of reads in the sample which is similar to the overall expected value for *B. subtilis*. The proportion of reads belonging to the remaining five bacteria were relatively similar (9.1 - 11.2%) to the expected proportion (12%). Fungi were not included in the database used to taxonomically classify reads in this study, therefore the expected and actual proportion of reads classified as *Saccharomyces cerevisiae* and *Cryptococcus neoformans* was not assessed. *Lactobacillus fermentum* reads were taxonomically classified as *Limosilactobacillus fermentum* in the database, as the *Lactobacillus* taxonomy has recently undergone changes [459]. These results suggest a slight bias towards *P. aeruginosa* and *S. enterica* during the library preparation process and that the Gram-positives were detected at a lower proportion than expected. There were classification issues within the database used for *B. subtilis* and instead a proportion of the reads were likely incorrectly classified as *B. intestinalis*.

Table 5.2: The expected and actual proportion of reads taxonomically classified in the DNA mock community sequencing control

Species	Expected (%)	Reads <sup>a</sup> (%)
Pseudomonas aeruginosa	12	17.1
Escherichia coli	12	11.2
$Salmonella\ enterica$	12	17.7
Bacillus subtilis	12	6.6
Lactobacillus fermentum	12	9.1
Enterococcus faecalis	12	10.0
$Listeria\ monocytogenes$	12	10.2
$Staphylococcus \ aureus$	12	9.8
Saccharomyces cerevisiae	2	NA
Cryptococcus neoformans	2	NA

<sup>a</sup> NA, Not assessed.

Generally, the proportion of reads taxonomically classified from the microbial community log distribution standard was similar expected distributions (Table 5.3). As with the microbial community DNA standard, *P. aeruginosa* and *S. enterica* were detected at a higher proportion than expected. In addition, the proportion of reads taxonomically classified as *B. subtilis* was lower than expected, however some reads from this sample were classified as *B. intestinalis* (0.3%) and *Bacillus cereus* (0.2%) which may account for the under-representation of reads classified as *B. subtilis*. These findings suggest that there was taxonomic classification issues within the *Bacillus* genus. Both *L. fermentum* and *S. aureus*, which were present at low concentrations, could not be detected in the microbial community log distribution standard using the bioinformatics pipeline used in this study.

At the genus level, the nine most abundant contaminants in both the PBS (n=2) and blank reagent controls (n=2) were similar (Table 5.4), suggesting that these bacteria are common contaminants in the laboratory and/or in the DNA extraction kits used in this study. Contaminating human DNA sequences were infrequently identified (<1%) in the PBS and negative blank reagent controls.

Species	Genomic DNA	Reads <sup>a</sup> (%)
Listeria monocytogenes	89.1	85.4
Pseudomonas aeruginosa	8.9	12.6
Bacillus subtilis	0.89	0.4
Saccharomyces cerevisiae	0.89	NA
Escherichia coli	0.089	0.1
$Salmonella\ enterica$	0.089	0.1
Lactobacillus fermentum	0.0089	0
Enterococcus faecalis	0.00089	0.04
Cryptococcus neoformans	0.00089	NA
Staphylococcus aureus	0.000089	0

Table 5.3: Genomic DNA content and the proportion of reads taxonomically classified in the microbial community log distribution standard

<sup>a</sup> NA, Not assessed.

Table 5.4: The most abundant contaminants in the PBS and blank reagent controls

Genus	Reads $(\%)$		$\operatorname{Reads}(\%)$	
	PBS		Blank reagent	
	C01	C02	C03	C04
Rhizobium	10.8	11.4	11.3	10.8
A quabacterium	9.3	9.8	9.4	10.8
Brevundimonas	11.5	11.3	11.6	10.7
Azospira	8.7	8.0	7.7	8.9
Acidovorax	7.7	7.4	7.7	6.7
Pseudomonas	4.2	4.1	4.2	4.4
Diaphorobacter	3.4	3.2	3.4	2.9
Sphingomonas	3.0	2.9	3.0	2.8
Agrobacterium	2.5	2.6	2.6	2.5

## 5.4 Discussion

The role of the NZ dairy farm environment in the development and spread of AMR is not fully understood. Due to the comparatively low use of antimicrobials in food-producing animals in NZ and the largely pasture-based dairy farm system, it was hypothesised that the abundance of ARGs in the dairy farm environment would be low. To test this hypothesis, shotgun metagenomic sequencing was utilised to examine and compare the resistome of environmental samples (faeces, FDE, soil and milk) collected over a 15 month period from two NZ dairy farms with contrasting farm management practices, taking into consideration seasonal variation and AMU. ARG abundance from FDE, faeces and milk collected over a 15 month period on Dairy 1 and Dairy 4 was relatively low compared to levels found in more intensive farming systems overseas, and was similar between the two farms (Figure 5.3). Interestingly, a higher number of unique ARGs were detected in FDE compared to soil, however they were detected at a lower abundance. FDE harbours a diverse range of ARGs [433, 460] and applying FDE to pasture has been associated with higher detection of samples positive for specific antimicrobial resistant bacteria, for example, AmpC-producing E. coli on beef farms in the UK [284]. Similarly, a study of a USA dairy farm using pasture-based grazing (eight months of the year and an indoor barn in winter) used qPCR targeting 113 ARGs to assess the dairy farm resistome. The most diverse ARGs and highest ARG abundance was identified in the stagnant lagoon, compared to cow faeces, the agitated lagoon, compost, soil, corn silage, animal drinking water or environmental water [433]. Congruent with this study, these findings suggest that FDE is likely a reservoir of ARGs and applying FDE to pasture may be a transmission pathway for the dissemination of ARGs within the dairy farm environment. The effluent management strategy differed between the two farms: Dairy 1 had two effluent management strategies during the study period including storage in a small sump which may allow for concentration of bacteria and/or ARGs prior to being filtered and the treated FDE being applied to paddocks, or being discharged into the sewage system (Table 3.1), therefore Dairy 1 does not apply raw FDE to pasture. On Dairy 4, the FDE is stored in a large pond prior to being applied to pasture as a source of nitrogen. Therefore, the on-farm FDE strategy is a farm management practice which may influence the levels on AMR within a farm. The ARG abundance in FDE from Dairy 4 was relatively low, and due to the large FDE pond size, any ARGs present would be diluted prior to being applied to pasture. It is also recommended that paddocks are rested for 10 - 14 days between FDE application and grazing [461]. The soil samples in this study were collected from recently grazed paddocks, therefore FDE would not have recently been applied to these paddocks prior to the sample collection periods (>10 days). There may also be regional specific compliance requirements for on-farm FDE management.

The diverse number of ARGs detected in FDE may reflect the complex composition and microbiota of this sample type, consisting of faeces and run-off from the cowshed. These results suggest that factors other than on-farm FDE management are likely to have a role in the variation in ARG abundance in FDE between Dairy 1 and Dairy 4. Factors which may influence the diversity or abundance of ARGs in effluent include the bacterial community composition [462], temperature [462, 463] or storage conditions [464]. Variation in ARG abundance has also been observed for FDE collected at different sampling depths [464], which may be due to the different nutrient and pH levels as well as the microbiome composition at different depths.

The phyla Firmicutes, Bacteroidetes, Proteobacteria and Actinobacteria were predominant in the faecal samples from both farms in this study. Consistent with this finding, the most abundant bacterial phyla from faecal samples from Canadian feedlots were Firmicutes, Bacteroidetes, Proteobacteria, Actinobacteria, Spirochaetes and Euryarchaeota [12]. ARG abundance in faeces ranged from 0.08 - 0.17 copies of ARG per 16S rRNA gene, which was lower compared to the abundance of ARGs in 145 animal-associated metagenomes including from faeces, manure, agricultural soil and water [465] and the total ARG abundance in young calf faeces (0.77 - 5.14 copies of ARG per 16S rRNA gene) [11]. A study comparing the faecal resistome of preweaned calves compared to lactating dairy cows on 17 commercial farms in the USA found a significantly higher ARG abundance in calves (0.43 - 2.9 copies of ARG per 16S rRNA gene) compared to dairy cows (0.11 - 0.6 copies of ARG per 16S rRNA gene) [13]. ARG abundance in the lactating dairy cow faeces was higher compared to this study, but was more comparable than ARG abundances found in calf faeces [13].

ARGs belonging to the  $\beta$ -lactam resistance class were the most abundant in faces which was due to the presence of the *cfxA* gene. The *cfxA* gene encodes a class A  $\beta$ -lactamase, which has been associated with the non-autonomous conjugative transposon  $Tn_{4555}$  [457, 466] and it is proposed that this transposon is involved in the horizontal transfer of the cfxA gene between Bacteroides and Prevotella species [466, 467, 468]. Therefore the high abundance of this gene in bovine faces may be due to Bacteroidetes being a predominant phylum detected in faecal microbiome of lactating dairy cows [385]. Similar to this study, a high abundance of  $\beta$ -lactam resistance genes were identified in a dried faecal sample (n=1) from a USA dairy farm, which was largely driven by the presence of the *cfxA* gene [433]. The aforementioned study used targeted qPCR for 113 ARGs, of which only 17 belonged to the  $\beta$ -lactam resistance class. The average number of ARGs was also lower in faeces compared to stagnant/agitated lagoon and compost samples in a targeted qPCR based study of a USA dairy farm [433], although admittedly only one sample of each type was analysed. The abundance of heavy metal and biocide resistance genes in faeces was extremely low on both farms (0.00 - 0.01 copies of resistance genes per 16S rRNA gene), suggesting these genes are not prevalent in the faecal microbes from healthy dairy cattle.

The abundance of ARGs in milk was very low (0.00 - 0.12 copies of ARG per 16S rRNA)gene) on both farms during the study period. Normalised ARGs were only detected in three milk samples which were collected in October 2018 (n=2) and December 2019 (n=1). Raw retail milk had a higher abundance of ARGs compared to pasteurised milk and using the HiSeq 4000 sequencing platform, ARGs were not detected from raw retail milk at the first time point (0 hours) but were detected after 24 hour room temperature incubation [210]. In the same study, additional raw retail milk samples collected across states in the USA were analysed using the NovaSeq S4 sequencing platform and ARGs were detected across most samples at an abundance of 0.0 - 1.0 copies of ARG per 16S rRNA gene [210]. The aforementioned study used a sequencing depth of approximately 40 million reads (150 bp paired-end) on the NovaSeq S4 platform, which is similar to the depth used in this study. These findings suggest that the abundance of ARGs in raw milk is low and that a high sequencing depth is required to detect any ARGs if they are present. In contrast, the milk from both Dairy 1 and Dairy 4 had a higher abundance of heavy metal resistance genes (0.00 - 0.59 copies of resistance genes per 16S rRNA). All of the metal resistance genes detected in milk potentially conferred mercury resistance and only one biocide resistance gene (sodB) was detected in one milk sample (0.10 copies of resistance genes per 16S rRNA gene). Other studies utilising shotgun metagenomics to study the resistome of retail raw milk have not investigated the abundance of heavy metal and biocide resistance genes [210, 434]. The abundance of mercury resistance genes in soil  $(0.003 - 0.007 \text{ copies of ARG per 16S rRNA gene across three samples), waste milk <math>(0.001)$  and faeces (0.00) were comparatively lower compared to milk samples. Mercury resistance genes were detected in FDE in four samples at a similar abundance compared to milk (0.03 - 0.06 copies) of resistance genes per 16S rRNA gene). However, the abundance was lower across the majority of FDE samples (average: 0.01 copies of resistance genes per 16S rRNA gene). Genes conferring mercury resistance (*mer*ABCDEFPRT) have been identified in numerous bacteria, including among members of Proteobacteria and Firmicutes [469], which were the two predominant phyla across the milk samples. The milk samples collected in this study were raw bulk tank milk, and therefore would be pasteurised prior to human consumption.

The milk samples clustered into two distinct groups according to milk microbiome composition (Figure 5.2 and Appendix T). The predominant bacterial phyla differed between the two clusters. Cluster one contained two samples from Dairy 1 and five from Dairy 4, which had a high abundance of Firmicutes. Except for samples collected in December 2018 (n=2), the remaining five samples did not overlap in collection date (Dairy 1: December 2018 and February 2019; Dairy 4: October - December 2018, August - September 2019). The variation in microbiome compositional changes may be due to changes in supplementary feed type on farm (Appendix W). No single feed type was only fed during the months where the milk microbiome composition was dominated by Firmicutes, although the amount of each feed type fed may have fluctuated per month and this data was not recorded. The remaining 18 milk samples clustered together and the predominant phylum was Proteobacteria. Two retail raw milk samples collected from markets in two cities in Hungary showed a large amount of diversity in microbiome composition, with *Bacilli* the predominant class in the first sample and *Gammaproteobacteria* being predominant in the second sample, followed by Actinobacteria and *Bacilli* classes [434]. Although the sequencing depth of used in the aforementioned study was much lower compared to this study (17,773,004 and 8,425,326 paired-end reads from the two samples). The microbiome composition of retail raw milk samples across states in the USA also varied [210]. To the best of my knowledge, this is the first longitudinal study to utilise shotgun metagenomic sequencing to analyse the microbiome and resistome of raw bulk tank milk (prior to pasteurisation).

Waste milk (sometimes referred to as discard milk) is any form of unsaleable milk produced on farm. The composition of waste milk varies and may consist of (i) milk from dairy cows receiving antimicrobial treatment (either systemic or intramammary) that has a withholding period, (ii) milk from cows receiving non-antimicrobial drugs, (iii) colostrum from cows shortly after calving and (iv) milk from ill cows or milk with a somatic cell count exceeding the saleable limit [470, 436]. The waste milk disposal strategy on farm depends on the volume of waste milk as well as farm management practices. Disposal of waste milk generally includes either disposal into the effluent pond (or similar storage area) or drains or may be fed to calves [470]. Both disposal strategies may facilitate the dissemination of antimicrobial, heavy metal and biocide resistance genes.

Feeding waste milk, which may contain low concentrations of antimicrobials [471], to young calves has been suggested as a risk factor for shedding and transmission of antimicrobial resistant bacteria [472]. This is also a concern if waste milk contains pathogenic bacteria. Antimicrobial resistant bacteria, including multi-drug resistant E. coli have been isolated from waste milk samples on USA dairy farms (2 of 10 E. coli isolated from waste milk samples) [471]. A higher prevalence of E. coli resistant to antimicrobials including cefotaxime, nalidixic acid, trimethoprim/sulfamethoxazole and cephalothin were isolated from the faeces of calves that were fed waste milk (pasteurised and unpasteurised) compared to calves fed bulk tank milk [436] on a commercial dairy farm in Germany. Other factors may also be associated with a higher prevalence of antimicrobial resistant bacteria detected in calf faces, such as the use of specific antimicrobials on farm and calf housing [472]. The aforementioned studies utilised culture-based methods and focused on specific AMR phenotypes particularly in E. coli. The abundance of specific ARGs was compared using quantitative PCR between calves pre- and post-weaning which were fed milk replacer with or without antibiotics (oxytetracycline and neomycin). Generally, feeding milk replacer with antibiotics was not associated with an increased absolute abundance of ARGs, however the relative abundance of tetO was higher in calves fed with waste milk containing a higher dose of the antibiotics (therapeutic compared to sub-therapeutic levels) [473]. These findings also suggested feeding waste milk to calves may increase ARG abundance in faeces.

The inclusion of waste milk in this study was unplanned, yet provided useful insights into the ARG abundance in waste milk in NZ. The waste milk sample was collected from Dairy 1 in October 2019. According to individual antimicrobial animal treatments recorded on farm, five cows were receiving antimicrobial treatment within six days prior to the October sampling date. The illnesses treated were mastitis and/or between claw/footrot (Appendix M). The active antimicrobial compounds in the products used were  $\beta$ -lactams, namely procaine penicillin G and penethamate. Therefore, milk from these cows may have been present in the waste milk sample (DF0167). Smaller volumes of waste milk on Dairy 1 and Dairy 4 are disposed of in accordance with the effluent management strategy on farm (sewage system and FDE pond, respectively). The waste milk sample (n=1) had a higher abundance of ARGs as well as heavy metal and biocide resistance genes (0.0 - 3.05 copies)of resistance genes per 16S rRNA gene) compared to soil, FDE, faeces and milk samples. Interestingly, the most abundant resistance mechanisms identified were ARGs belonging to multi-compound classes, particularly multi-metal, drug/biocide and biocide/metal classes including a number of ARGs encoding efflux pumps or regulators, highlighting that waste milk is not only a source of ARGs but also genes conferring resistance to other compounds such as heavy metals and biocides. Acquired ARGs were identified in 46 contigs assembled from the waste milk sample, however, only 19 contigs could be taxonomically classified at the phylum level. The detection of clinically relevant ARGs, such as  $bla_{\text{CTX-M}}$  which encodes an ESBL enzyme in *Enterobacterales* from waste milk is a public and animal health concern.

Soil, including those with low anthropogenic impact, has been shown to harbour a diverse range and abundance of ARGs [474, 475] and many antimicrobial compounds are naturally produced by soil microorganisms [475]. ARG abundance as well as heavy metal and biocide resistance gene abundance was highest in the soil samples throughout the 15 month study period (0.20 - 0.63 copies of ARG per 16S rRNA gene and 0.00 - 0.43 copies of resistance genes per 16S rRNA gene) compared to the FDE, faeces and milk sample types. Oddly, one soil sample collected in October 2018 on Dairy 4 (DF0006) did not contain any heavy metal or biocide resistance genes and this sample also had the lowest ARG abundance among soil samples from Dairy 4 (0.20 copies of ARG per 16S rRNA gene). At the time of sampling, it was noted that this paddock had poor drainage, although no changes were observed in the microbiome composition of this soil sample compared to the other soil samples (Appendix T). Further soil samples could be analysed from this paddock to determine whether the observed ARG abundance was consistently low or whether the previous ARG abundance may have been affected by poor drainage at the time of sampling. The high abundance of antimicrobial, heavy metal and biocide resistance genes identified in soil compared to FDE, faeces and milk and highlights that the soil resistome is abundant on both farms throughout the 15 month study period.

Soil microbiome composition can be highly complex [476]. Both ARG and bacterial diversity were significantly correlated in a study comparing soil from three distinct ecosystems (tundra, temperate and tropical) [474], which may suggest that microbiome sample variation drives ARG diversity. The microbiome from soil samples collected on Dairy 1 and Dairy 4 were relatively homogeneous, with the two predominant phyla being Proteobacteria and Actinobacteria, which is congruent with previous studies of soil from feedlots (n=4)in Canada [12] and soil samples collected from five USA dairy farms [218], although the soil samples in the aforementioned study were collected at a depth of 5 cm (compared to 10 cm in this study). However, soil communities were shown to be similar in composition and diversity at various sampling depths (samples compared at depths of 15.24 and 30.48 cm) [464]. There was no statistical difference (p = 0.271) in ARG abundance in the soil samples between the two dairy farms, which is not unexpected given the close geographical proximity of the farms (<5km) and the similarity in microbiome between the soil samples. The soil composition on Dairy 1 consists of a mixture of Rangitikei or Karapoti Brown sandy loam with some Manawatu sandy loam/gravelly phase and Manawatu mottled silt loam. On Dairy 1, the soil is free-draining alluvial soil. The soil on Dairy 4 is predominantly Tokomaru silt loam, with some compact clay loam and Ohakea silt loam. The soil on Dairy 4 has poor natural drainage and is artificially drained. ARG abundance in soil samples collected in this study (0.20 - 0.63 copies of ARG per 16S rRNA gene) was comparatively low compared to soil microcosms with/without compost manure collected from cows treated with specific antibiotics [477] and ARG abundance in soil collected from pens housing untreated (n=3) and florfenicol treated calves (n=3) [478] which ranged from 0.62 - 4.53 copies of ARG per 16S rRNA gene. ARG abundance in soil defined as pristine with little anthropogenic impact had an ARG abundance of 0.05 - 0.28 copies of ARG per 16S rRNA gene [432], which is less than the highly curated soils in this study.

The soil samples had the highest ARG abundance and was the third most diverse sample type (after FDE and faeces), with 30 ARG groups belonging to 11 resistance classes (Figures 5.3 and 5.4). The most abundant resistance classes for ARGs were glycopeptides, multi-drug resistance and drug/biocide resistance followed by aminocoumarins and MLS resistance classes and copper, iron and peroxide resistance classes were the most abundant for heavy metal and biocide resistance. Actinomycetes belong to the phylum Actinobacteria, which was the second most abundant phylum detected in soil samples in this study. Interestingly, soil samples (n=4) from feedlots in Canada had a smaller number of unique ARG groups identified compared to this study, with only nine ARG groups belonging to six classes detected [12], although the sample size was considerably smaller compared to this study. The ARGs from soil belonging to the multi-drug resistance class encoded multi-drug efflux pumps (*muxB* and *tap*) or regulators (*mtrad*). Multi-drug efflux pumps are also important for functions other than AMR such as detoxification of intracellular metabolites, cell homeostasis and bacterial virulence within plant and animal hosts [52]. A number of drug/biocide resistance genes were identified in soil (Appendix T), all of which encode efflux pumps or regulators. Despite ARGs detected at a higher abundance in soil, some ARGs identified in soil have been shown to share relatively low sequence similarity to the corresponding ARG in clinical pathogens [474], although a limited number of clinical ARGs were analysed and the phenotypic consequence of this finding is unknown. A threshold of >80% similarity at the nucleotide level was used to identify ARGs in this study to reduce false positive results. Individual ARGs identified in soil samples, particularly those of clinical significance, could be examined in closer detail and the genetic similarity compared to the corresponding ARG from clinical pathogens.

The bacterial host and genetic context of ARGs is crucial to assess the health risk posed. In addition, some ARGs such as efflux pumps, can have alternative functions and be involved in physiological processes unrelated to AMR. Mobile ARGs which were associated with humans were identified as the highest risk in a recent risk framework study [246]. Analysis of the ARGs used in the risk framework found that 70% of ARGs analysed (1,816 of 2,579) were not human-associated (enriched in environments with anthropogenic impact) and were classified as the lowest risk. The majority of these ARGs were not found on mobile genetic elements [246]. A diverse range of ARGs (n=372) belonging to the drugs, biocide, metal and multi-compound classes were identified in this study, many of which were identified in FDE. Despite a large number of unique ARGs being identified, the majority of these were not classified as the highest risk gene families [246]. At the sequencing read level, the high risk ARGs aac6' were identified in faces, dfrB from one milk sample (dfrB1 highest risk ARG) and  $bla_{OXA}$ ,  $bla_{CTX}$  and lnuA from FDE (Appendix T). The soil samples had the highest ARG abundance in this study, yet no high risk ARGs were identified in soil, suggesting that although the overall ARG abundance in these samples was high compared to faeces, FDE and milk, the public and animal health risk associated with these ARGs is low.

ARGs potentially conferring resistance to critically important antimicrobials in human

medicine [98] including plasmid-mediated mcr genes conferring resistance to colistin [46], carbapenem resistance genes [479] or the resistance gene mecA from methicillin-resistant S. aureus [480] were not identified in this study. Of the acquired ARGs detected in contigs assembled from the metagenomic samples, some high risk ARGs were identified. Of particular concern, the  $bla_{OXA}$  gene was detected in a contig belonging to the Campylobacteraceae family from FDE, and blaZ and the extended-spectrum  $\beta$ -lactamase resistance gene bla<sub>CTX-M</sub> were identified in Bacillaceae and Enterobacterales from waste milk, respectively. The tet(W) gene, which is classified as high risk rank II gene (which includes resistance emerging in non-pathogens) was detected in *Fibrobacteraceae* and *Lachnospiraceae* from facees and FDE samples on Dairy 4, respectively. ARGs were analysed at the gene family level compared to gene sequence, as was used in the risk framework [246], which is potentially a limitation of this study. These findings suggest that although a diverse range of ARGs were detected across the faeces, FDE, soil and milk samples, albeit at a comparatively low abundance, the majority of the ARGs detected do not pose a high public and animal health risk. These results highlight the importance of determining the bacterial host and mobility of ARGs to assess the relevant public and animal health risk posed.

Comparison of ARG abundance between shotgun metagenomic studies can be difficult due to the various bioinformatic methods and databases to identify ARGs, as well as the normalisation used for reporting ARG abundance. For example, some ARG databases contain only acquired resistance genes (e.g. ResFinder [216]), whereas other such as MEGARes2.0 [451] contain both acquired genes as well as chromosomal gene mutations. Therefore the ARG abundances reported in studies focusing on acquired resistance may be lower when compared to similar study populations in which resistance databases containing both acquired genes as well as chromosomal gene mutations were used. The normalisation methods may also differ between studies. For example, Zaheer et al., uses the Cumulative Sum Scaling normalisation method for microbiome and resistome analyses [12], compared to the method by Li et al., [453] using ARG sequencing hits and gene length, read length as well as the number of 16S rRNA genes identified per sample which was utilised in this study. Normalisation by ARG gene length was shown to have an important effect on the distribution of ARG allelic variants from pig caeca and human treated effluent samples [239]. ARG abundance in this study was normalised and expressed as "ARG per copy of 16S rRNA gene" as this method adjusts for bias relating to sequencing length, the microbial load per sample which is important for low biomass samples such as milk and this method is well-utilised in this field and thus allows for a comparison of the results from this study with the current literature.

The sequencing depth across studies can vary and the depth required to analyse the resistome is study-specific. Previous research has shown that the relative proportion of reads assigned to specific ARG classes was relatively consistent across various sampling depths in bovine faeces (average read counts used in the study: 26, 59 and 117 million), however, increasing the sequencing depth resulted in higher numbers of ARG assigned reads [236]. Studies with low sequencing depth may not be able to analyse the total number of unique ARGs in a resistome, or examine the genetic context of specific ARGs. The sequencing depth in this study (Figure 5.12) was sufficient to analyse the resistome structure for the majority of samples. Although the rarefaction curves for three FDE and the waste milk sample did not plateau after 100% of reads were sub-sampled, suggesting that an increased sequencing depth may have identified additional unique ARGs. The sequencing depth can be specific to the sample matrix and dependent on the proportion of ARGs within a sample. For example, deep sequencing (at least 80 million reads per sample) of pooled pig caeca, treated human effluent and river sediment upstream from sewage treatment found that a low proportion of sequencing reads belonged to ARGs (<0.05%) and that the sequencing depth was not sufficient to detect the full allelic diversity in the treated effluent sample [239]. Even at a high sequencing depth, some ARGs could not be assigned to specific allelic variants, although a stringent threshold of 100% identity was used in the aforementioned study. However, the sequencing depth was sufficient for the treated effluent and pig caeca when analysing ARG families [239].

Analysis of the mock community DNA standard highlighted that taxonomic classification at the species level between related bacteria may lead to classification errors, such as those identified with *B. subtilis* and *B. intestinalis* (Table 5.2). Sequencing of the negative blank reagent (n=2) and PBS controls (n=2) identified that the nine most abundant genera were contaminants of both the negative blank reagent and PBS controls, suggesting that these bacteria are common contaminants in the laboratory and/or in the DNA extraction kits. The majority of these genera, except for *Agrobacterium*, have previously been reported as contaminants in sequenced negative blank controls [226, 481].

The high proportion of host reads (*Bos taurus*) in milk samples can complicate shotgun metagenomic sequencing analysis [242] and a higher sequencing depth is required to obtain sufficient coverage of the microbial DNA [482]. A high proportion of the sequencing reads across the milk samples was host DNA and was removed from subsequent analyses (metagenomic sequence read numbers can be accessed here). High proportions of host DNA has been shown to reduce microbiome profile sensitivity [482, 242] and this finding most likely can be extrapolated to include ARG detection sensitivity. The high amount of host DNA in milk samples drastically reduced the sequencing depth in these samples and may reduce the taxonomic classification sensitivity. To overcome this issue in future studies, host-depletion methods could be utilised prior to shotgun metagenomic sequencing, such as methods which utilise differential lysis of microbial and mammalian cells or postextraction methods that selectively bind and remove CpG-methylated host DNA [242]. Alternatively, sequencing of milk filters have been proposed as a representative sample to analyse ARGs in bulk tank milk [242]. However, the milk is filtered through milk filters prior to storage in the bulk tank vat and although the pore size is large enough for bacteria to pass through, any ARGs or bacteria detected in the milk filter may not necessarily be present in the bulk tank milk.

## 5.5 Conclusion

ARG abundance in faeces, FDE, soil and milk samples across two NZ dairy farms over 15 months was comparatively low compared to overseas studies. No statistically significant difference in overall antimicrobial, heavy metal and biocide resistance genes was observed between Dairy 1 and Dairy 4. However, ARG abundance between FDE from the two farms was statistically significant, with a higher ARG abundance in FDE from Dairy 1. Dairy 1 also had greater AMU (17.09 mg/PCU) between October 2018 and December 2019 compared to Dairy 4 (5.36 mg/PCU). FDE samples harboured the most diverse range of ARGs, some of which are classified as a high risk to public health. Compared to FDE, faeces samples had lower diversity of ARGs and the faecal resistome on both farms was dominated by genes potentially conferring  $\beta$ -lactam resistance, in particular the cfxA gene. Milk samples had a low abundance of ARGs, but had a higher abundance of heavy metal, particularly mercury resistance genes. Soil samples had the highest ARG abundance (excluding the waste milk sample), however the ARGs in soil were not classified as high risk to public and animal health, highlighting that abundance as well as genetic context and risk (human-associated and mobility) should be considered when analysing resistomes using shotgun metagenomic sequencing. The waste milk sample had a high abundance of antimicrobial, heavy metal and biocide resistance genes. The detection of clinically significant ARGs, as well as contigs from mastitis-associated pathogens in the waste milk sample suggests that waste milk may be a source of pathogens and ARGs in the dairy farm environment. Due to the numerous ARG identification and normalisation methods, as well as variation in sequencing depth, caution should be used when comparing between studies using shotgun metagenomic sequencing methods. No association was identified between normalised ARG abundance and AMU or seasonality (assessed as sample collection period). Although the low ARG abundances and AMU on the farms during the study period may have made any associations difficult to detect. Future research on dairy farms with contrasting AMU are needed to address these research hypotheses.

## Chapter 6

# General discussion

#### 6.1 Overview

Antimicrobial resistance (AMR) poses a significant public and animal health concern worldwide. The development and transmission of AMR between humans, animals and the environment is multi-factorial [28] and requires a holistic "One Health" approach to address. New Zealand (NZ) is a comparatively low user of antimicrobials in animal production [16, 17] and dairy farming uses less antimicrobials compared to both the poultry and swine industries [17]. In comparison to international standards, a study found NZ dairy herds (n=477) were generally low users of antimicrobials, however, some high-use herds were also identified [43]. The majority of antibiotic use on NZ dairy farms is for the treatment and prevention of mastitis [21].

Studies investigating AMR in NZ dairy farm environments have generally taken a culturebased approach, focusing on specific target organisms and resistant phenotypes [22] while utilising a cross-sectional study design [25, 26]. The research in this thesis using a longitudinal approach addresses the hypothesis that antimicrobial use on NZ dairy farms influences the prevalence of AMR in dairy farm environments, taking into consideration seasonality and contrasting farm management practices and utilises a combination of phenotypic, molecular and next-generation sequencing techniques to study AMR in NZ dairy farm production systems (Figure 6.1). Therefore, a three-fold approach was used to assess AMR in the NZ dairy farm environment, (i) targeting specific antibiotic resistant E. coli using culture-based methods (Chapter 3), (ii) investigating the impact of systemic antimicrobial therapy on the bovine faecal microbiome and the emergence of antimicrobial resistant bacteria at the individual animal level (Chapter 4) and (iii) an extensive approach utilising shotgun metagenomic sequencing to estimate the abundance of antimicrobial, heavy metal and biocide resistance genes in farm environmental samples collected over a 15 month period and to identify the bacterial host and genomic context of acquired antimicrobial resistance genes (ARGs) (Chapter 5).



## 6.2 Findings, implications and general discussion

Overall, the research presented in this thesis has shown a low sample level prevalence of ESBL-producing E. coli from two NZ dairy farms with AmpC-producing E. coli being more frequently isolated across both farms (Chapter 3). AmpC- and ESBL-producing E. *coli* were isolated in spring and summer, during months with varying levels of antimicrobial usage (AMU). Analysis at the individual animal level showed a decrease in bacterial diversity, richness and composition during systemic antimicrobial treatment, and in many cases the microbiome structure recovered post-treatment when the cow re-entered the milking herd (median 6 days; Chapter 4). Changes to the faecal microbiome varied at the individual animal level which was likely due to a multitude of factors such as illness, disease severity, prescribed antimicrobial and treatment duration as well as feed type. Shotgun metagenomic sequencing of farm environmental samples provided a broader approach to study AMR in the dairy farm environment, examining a diverse range of antimicrobial, heavy metal and biocide resistance genes. Compared to overseas data in a similar context [478, 477, 465, 13, 210], NZ dairy farm environments had a low abundance of ARGs (normalised to per copy of the 16S rRNA gene), with the highest abundance detected in soil (Chapter 5). However, many of the ARGs identified in soil were not classified as high risk [246]. FDE samples had a lower abundance of ARGs but these were more diverse, suggesting that FDE may be a reservoir for ARGs in the dairy farm environment. Acquired resistance genes were predominantly found in Firmicutes and Proteobacteria. Proteobacteria contains a number of human bacterial pathogens, however, only a small number of contigs could be classified at the species level.

## 6.2.1 Antimicrobial resistance in the New Zealand dairy farm environment

ESBL-producing E. coli are hypothesised to be at a low prevalence in NZ dairy herds due to NZ's predominantly pasture-based dairy farm production system, coupled with the low AMU in animal production. The sample level prevalence of ESBL-producing *E. coli* in this study was low on Dairy 4 (1 of 60 (1.7%)) from pooled faeces; 1 of 15 (6.7%) from FDE) compared to overseas studies [7, 6, 199] and no ESBL-producing *E. coli* were isolated from Dairy 1. A low prevalence of ESBL-producing E. coli was also detected in a regional crosssectional study (1 of 15, 6.7%) [25] and a nationwide cross-sectional study (n=26 farms) in NZ did not detect any ESBL-producing E. coli [26]. These findings are consistent with overseas studies of pasture-based farming systems, for example, a cross-sectional study of ruminant farms in Spain found beef cattle herds and sheep flocks, which have a semiintensive pasture-based farming system, had significantly lower herd level prevalence of cefotaxime-resistant E. coli compared to dairy cattle, which are generally housed inside pens [9]. A study across 40 cattle farms in Israel found that pasture-based farms had the lowest number of ESBL-producing Enterobacteriaceae compared to quarantine stations (housing calves only) and fattening farms (intensive farming without grazing) [483], but pasture-based farms did not include calves <4 months of age [483], which may confound the results as studies have shown a higher prevalence of AmpC- and ESBL-producing E. *coli* in calves compared to adult dairy cows [483, 194]. Few studies have investigated the prevalence of AmpC- and ESBL-producing E. coli on pasture-based dairy farms and further longitudinal studies across a larger number of farms are required to identify the risk factors for AmpC- and ESBL-producing E. coli in these farm production systems.

In contrast to ESBL-producing *E. coli*, the sample level prevalence of AmpC-producing *E. coli* was higher in this study and they were isolated from pooled faecal and FDE enrichments on both the Dairy 1 and Dairy 4 farms (Table 3.9). A nationwide cross-sectional study in NZ (n=26 dairy farms) detected chromosomal AmpC-producing *E. coli* from 14% (11 of 78) of pooled faecal enrichments originating from seven farms [26]. In comparison, a cross-sectional study in the Manawatū region (the same region as this study) found 7.8% (9 of 116) of pooled faecal samples were positive for chromosomal AmpC-producing *E. coli* with pAmpC genes were identified [26]. In this present study, the AmpC-positive sample level prevalence from pooled faecal enrichments was comparable to other cross-sectional studies undertaken in NZ, however the sample level prevalence was higher in FDE (which was not assessed in the other studies). This suggests that FDE may be a reservoir of AmpC-producing *E. coli* and would therefore represent a good proxy to assess antimicrobial resistant bacteria in the dairy farm environment.

In NZ, veterinarians and farmers are being encouraged to reduce the use of antimicrobials classified as orange and red tier by the NZVA, which will help facilitate improved antibiotic stewardship in agricultural production systems. The aforementioned study [26] across NZ dairy farms (n=26) found that a larger herd size and greater total AMU was associated

with AmpC positive farms. Dairy 4 has a larger herd size compared to Dairy 1, with 584 and 261 cows in each herd respectively, yet Dairy 1 had a higher AMU during the study period (D1, 17.09 mg/PCU; D4, 5.36 mg/PCU). Further research is required to elucidate these differences in AMU between the two farms. For example, in depth interviews could be conducted with the farm managers and consulting veterinarians from the two dairy farms to identify factors which contribute to the AMU on both Dairy 1 and Dairy 4. The AMU on Dairy 1 and Dairy 4 was within the range reported in a cross-sectional study of 26 dairy farms across NZ (4.39 - 20.92 mg/PCU) [26], albeit at the higher and lower end of the scale, respectively. However, the AMU in this study was calculated using individual antimicrobial treatments recorded on farm which were provided by the farmer, in contrast to antimicrobial sales data in the study by Burgess *et al.*, [26]. Antimicrobial sales data from Dairy 1 and Dairy 4 could be collected and compared with individual farm records as well as published AMU data from dairy farms throughout NZ.

Despite differences in herd size and AMU between the farms, the sample level prevalence of AmpC-producing *E. coli* in pooled faeces was comparable (Dairy 1: 3.3%; Dairy 4: 8.3%). The sample level prevalence was higher in FDE from Dairy 1, which may be due to contrasting effluent management strategies between the farms with Dairy 1 having a smaller storage sump, potentially allowing for an increased bacterial concentration. Despite the higher sample level prevalence of AmpC-producing *E. coli* in FDE, the resistance was mediated through chromosomal point mutations in the promoter region of the ampC gene. This mechanism of resistance is of concern for the relevant bacterium that harbours it and for vertical transmission, but is less likely to spread horizontally within bacterial populations.

The use of injectable amoxicillin [156, 26] and third- and fourth-generation cephalosporins [7] have been associated with the detection of AmpC- and/or ESBL-producing  $E. \ coli$  on dairy farms and *in vitro* studies have shown an association between amoxicillin use and AmpC-producing E. coli arising from mutations in the promoter region of the ampC gene [277, 355]. Both farms in this study did not report any injectable amoxicillin-clavulanic acid use during the study period (Appendix M). However, a low level of third-generation cephalosporins were used on both farms (Appendices K - L) and this research has demonstrated that plasmid-mediated AmpC-producing E. coli were isolated from an individual cow during and post-treatment with ceftiofur (Chapter 4, section 4.3.3 and Appendix P), suggesting that treatment with third-generation cephalosporins may be associated with the enrichment of AmpC-producing E. coli. Interestingly the plasmid-mediated AmpCproducing E. coli isolated during the microbiome study (Chapter 4) was from an individual animal on Dairy 1, yet during the longitudinal study in which only pooled faeces samples were examined, only one plasmid-mediated AmpC-producing E. coli was isolated from this farm (Chapter 3, Table 3.9). The virulence factors found in the E. coli (n=3) isolated from an individual animal (Appendix S) were also found in AmpC-producing E. coli isolated during the longitudinal study (Figure 3.4). This finding reflects the high resolution analysis which can be undertaken when sampling at the individual animal level, although such study designs are not always feasible. Additional studies with a larger sample size utilising both culture-based and molecular methods (e.g. WGS of antimicrobial resistant isolates and resistome analysis of faecal samples using shotgun metagenomic sequencing or targeted qPCR) are required to assess any potential associations. The total amount of amoxicillin and ceftiofur used on Dairy 1 and Dairy 4 during the study period was low (Appendices K and L), and this may have contributed to the low sample level prevalence of ESBL-producing  $E. \ coli$  on these two dairy farms.

ARGs have been found in a wide range of environments including human sewage [427], agricultural settings [13, 244, 484] and pristine soil with low anthropogenic impact [432]. Metagenomic studies assessing AMR often focus on the abundance of ARGs, however there is growing evidence to suggest that the risk associated with ARGs in the resistome should also be considered [53]. A framework for ARG risk ranking has been proposed and the key considerations include: (i) whether the ARG is part of the intrinsic resistome, (ii) fitness cost of the resistance and (iii) the environment in which the ARG was detected, for example high risk genes found in humans pose a higher risk than if the gene was detected in a remote environment [53].

Intrinsic resistance or resistance caused by mutations in chromosomal genes is relevant for the bacterium that harbours it and for vertical transmission, however ARGs which are transferable are of higher concern as they may result in ARG dissemination horizontally within a bacterial population. For example, the cfxA gene, encoding a class A  $\beta$ -lactamase often found on a mobilisable transposon detected in *Bacteroides* species [457], was found in all faecal and most FDE samples in this study (Chapter 5). Thus, the identification of this gene in faeces and FDE may therefore be indicative of the predominance of Bacteroidetes in the faecal microbiome of lactating dairy cows [385] rather than suggesting the detection of the cfxA gene in other bacterial species. ARGs classified as the highest risk (defined as resistance readiness condition 1 as described in [53]) are those which can be identified on mobile genetic elements, are known to contribute to the failure of antibiotic treatment and have previously been found in human bacterial pathogens [53]. An example of a high risk ARG is the *bla*<sub>CTX-M</sub> gene, which was identified in a contig belonging to the *Enterobac*terales order that was assembled from the waste milk sample (Chapter 5), suggesting that clinically relevant antimicrobial resistant bacteria may be present in waste milk. Therefore, to estimate the burden of AMR in an environment or ecosystem, metagenomic studies should evaluate not only the abundance of ARGs but they should also assess the risk each ARG poses. A similar risk framework approach has been applied to assess ARGs from human gut microbiome genomes and only a small proportion of ARGs were ranked as the highest risk group in the study (3.6%) [246]. A limitation of the current proposed ARG risk frameworks is that they are often anthropocentric [53, 247, 246] with the risk described in relation to human health. Future modifications to risk frameworks should take a "One Health" approach and include risk assessments which collectively consider human, animal and environmental health.

ARG abundance measured using shotgun metagenomic sequencing was highest in soil samples during the study period (Chapter 5, Figure 5.3), however, many of the ARGs identified were not classified as high risk [246] suggesting that soil poses a lower public and animal health risk. Instead, the ARGs present in soil may be due to the natural production of antimicrobials by soil microorganisms [475], which acts as a selection pressure within the soil environment. Therefore, the high abundance of ARGs in the soil from the two dairy farm environments (collected as soil cores from the paddock) is more likely due to the natural environment and microbial populations rather than the anthropogenic impact of dairy farming. Further work analysing soil samples collected from sites with various levels of anthropogenic impact, such as higher impacted sites on dairy farms including around calf pens or water troughs and low impact sites such as native forest, is required to address this hypothesis. Despite being identified in a natural environment such as soil, some ARGs may pose a health risk if they are identified in bacterial pathogens, or if they are encoded on mobile genetic elements that could be transferred to pathogenic bacteria [51]. Thus, it is important to understand the genomic context of ARGs to assess risk (e.g. bacterial host, mobility), rather than only focusing on abundance.

The ARG abundance was low and no cephalosporin resistant *Enterobacteriaceae* were isolated from bulk tank milk samples in this study (Chapter 5, Figure 5.3). FDE harboured the most diverse range of ARGs, albeit at a lower abundance (Chapter 5, Figure 5.3). Consistent with this finding, multiple AmpC and/or ESBL-producing E. coli clonal types were isolated from FDE samples (Chapter 3, Figure 3.7). The diverse range of ARGs along with the presence of AmpC- and/or ESBL-producing E. coli provides further support that FDE is a reservoir for a diverse range of antimicrobial resistant bacteria and ARGs in the dairy farm environment. The composition of FDE and effluent management strategy will differ between farms, however FDE is a complex matrix of faeces, urine, waste milk (if present on farm and/or not fed to calves) and wastewater from the cow shed environment. As such, FDE/manure is a nutrient rich matrix with a high abundance and diversity of microbial populations which may contain antibiotic and biocide/disinfectant residues. Therefore, FDE provides a suitable environment for HGT and the dissemination of antimicrobial resistant bacteria and ARGs [460, 485]. Sampling FDE for the detection of antimicrobial resistant bacteria and ARGs could be used as a proxy for an entire farm. Urban sewage has previously been used to monitor ARGs globally [427] as well as for other pathogens such as testing wastewater to monitor SARS-CoV-2 in the NZ community [486]. Composite samples, compared to a grab sample, should be used for surveillance studies. Future work should also assess the limit of detection of both ARGs and AmpC- and ESBL-E from FDE, as well as investigating how representative a sample of FDE would be of the entire dairy herd.

Seasonality and months following increased AMU were not associated with an increased abundance of ARGs (Chapter 3, Figure 3.7). However, any correlations may have been difficult to identify due to the low ARG abundance across farm environmental samples in this study. Three FDE samples were identified as outliers and had a higher ARG abundance
(Chapter 5, Figure 5.3) but no AmpC- or ESBL-producing *E. coli* were isolated from these samples. This was not unexpected as the relative abundance of the genes associated with these resistant phenotypes were infrequently identified in the metagenomic sequence data (Appendix U). Interestingly, AmpC- and/or ESBL-producing E. coli were only identified during spring and summer, which correlates with calving/the start of lactation and the dryoff period, respectively which is a time period associated with increased AMU and mastitis cases. The third-generation cephalosporin resistant E. coli (DF0102.4e-h; AmpC and ESBL negative) was isolated in autumn (Chapter 3, Figure 3.7). No AmpC- and ESBL-producing E. coli were isolated during winter. It is important to note however that the longitudinal sample collection was carried out over 15 months and there was uneven sampling during some seasons which may exaggerate the number of AmpC or ESBL positive samples (Figure 3.7; summer: 4 visits; autumn: 3 visits; winter: 3 and spring: 5 visits). There was also a low sample level prevalence for some resistance phenotypes (e.g. ESBL producers) which makes any associations between AMU, seasonality and sample level prevalence difficult to determine. Further longitudinal studies with a larger sample size and number of farms are required to assess any potential associations between seasonality and greater AMU with ARG abundance and the prevalence of of antimicrobial resistant bacteria.

ESBL-producing E. coli were only isolated from one farm (1 of 15) during spring in NZ [25] which was consistent with this study. Detection of ESBL-positive faecal samples on dairy farms (n=3) in France varied across sampling months and was highest during spring and summer on one farm which was positive for ESBL-producing E. coli [487]. Increased temperatures have also been associated with a higher likelihood of detecting bla<sub>CTX-M</sub> positive E. coli samples in the UK [195]. Season significantly affected E. coli persistence in cow pat faeces, with higher counts detected in summer and autumn and the most rapid decline in E. coli during winter [488], suggesting that seasonal trends are also associated with other faecal bacteria and not limited to AmpC- and ESBL-producing E. coli. These findings as well as the isolation of AmpC- and ESBL-producing E. coli from dairy farms in spring and summer only in this study, suggests that prevalence studies in NZ should have more intensive sampling during spring and summer. Further studies with a larger sample size are required to assess seasonal trends in the prevalence of AmpCand ESBL-producing E. coli in NZ dairy farm environments. Climate change may have a negative impact on dairy farm production systems and animal welfare; for example, increased frequency of heavy rainfall or extreme weather events, heat stress in animals, changes in annual pasture yield and increased incidence of mastitis could lead to increased AMU and persistence of pathogenic and antimicrobial resistant bacteria in the dairy farm envrionment [271, 272, 273].

## 6.2.2 Selection of antimicrobial resistant bacteria

Analysis of AMR at the individual animal level and targeted culture-based methods allowed for high resolution analysis at the isolate level compared to the broader culture-based and culture-independent techniques used for the farm environmental samples. This analysis provided an example of the emergence of antimicrobial resistant bacteria, whereby plasmid-mediated AmpC-producing *E. coli* were isolated during and post ceftiofur treatment in an individual animal (Chapter 4, Table 4.1). No cephalosporin resistant *E. coli* were isolated from the pre-treatment sample for this cow nor from any of the associated control cow samples (S1, S2 and S3), which suggests that treatment with ceftiofur may have enriched and selected for AmpC-producing *E. coli*. Interestingly, third-generation cephalosporin resistant *E. coli* were only isolated from one of the three cows treated with ceftiofur (Chapter 4, Table 4.1), which indicates that antimicrobial treatment does not always enrich for resistant bacteria, or they were enriched but at numbers not detectable using culture-based methods. However, a PCR targeting the  $bla_{CMY-2}$  gene from the crude boiled lysates from the faecal enrichments of ceftiofur treated cows was concordant with the culture-based methods. Further work is required to assess the abundance of additional ARGs from the faecal samples from individual animals in this study to see whether other antimicrobial resistant phenotypes are also enriched.

# 6.3 Potential limitations and future work

#### 6.3.1 Farm choice

The rationale for recruiting the Massey University dairy farms (Dairy 1 and Dairy 4) in this study was (i) contrasting farm management practices, (ii) they are research farms and therefore have good farm records and farm staff who are able to assist with research projects and (iii) are in close proximity to the Hopkirk Research Institute (the latter two were required for the research outlined in Chapter 4, Figure 4.1). Some on farm processes may differ between research and conventional dairy farms which is a potential limitation of this study. For example, the Massey University dairy farms were more likely to consult with veterinarians in the diagnosis and treatment of sick dairy cows compared to conventional farms (personal communication with farm managers). In addition, access to veterinary consultation was not a barrier on the research farms and therefore even dairy cows with severe illness may have been treated rather than being culled.

The Massey University dairy farms are also used for veterinary teaching purposes which may result in scenarios that would not typically occur on conventional farms e.g. the caesarean cows that were treated with ceftiofur (Appendix P). A high level of veterinarian involvement on the two Massey University dairy farms may have contributed to the low use of antimicrobials classified as red tier by the NZVA and the good antimicrobial stewardship on these farms, such as the low use of third- and fourth-generation cephalosporins and marbofloxacin (Appendices K - L). Despite these differences between the Massey University research farms and conventional dairy farms, the inclusion of these farms was appropriate to answer the research aims and meet the requirements of this study design. Across NZ dairy farms there is also regional variation in the topography, soil type, rainfall and climate as well as differences in farm management practices such as feed type, housing and waste milk disposal. In addition to between farm variation, there is variation within farm in regards to seasonality, farming practices (e.g. calving), feed sources and antimicrobial use. To account for within farm variation, a longitudinal study design was utilised. Although the Massey University research farms are not representative of all NZ dairy farms, the conclusions drawn from this research are still applicable as the core farm production system and management is typical of NZ dairy farms to ensure that they adhere to the food quality, health and safety, animal welfare and environmental regulations set out by the governing organisations and/or local government. Thus, a representative sample is not always a requirement for scientific research and is instead dependent on the research aims [489].

#### 6.3.2 AMR risk factors

This research has assessed AMR on two NZ dairy farm environments, taking into consideration seasonality and farm management practices. Additional research is required to assess whether any AmpC-producing  $E. \ coli$  isolated in this study are resistant to fourthgeneration cephalosporins, and subsequently have ESAC  $\beta$ -lactamase activity (Chapter 3). Additionally the resistance mechanism of isolates DF0102.4e-h to third-generation cephalosporins should be further elucidated (Chapter 3). In silico analysis suggested that all plasmids characterised in this study were conjugative, except for plasmid pDF0059.2e 2 which belonged to the IncY group (Chapter 3, section 3.3.6). Conjugation experiments are required to determine whether these plasmids can be transferred under laboratory conditions. Further studies with a larger sample size are also required across NZ dairy farms to understand the risk factors for AmpC- and ESBL-producing Enterobacteriaceae and higher ARG abundance across NZ dairy farms. Samples should be collected at multiple time points to account for within farm variation and potentially during spring and summer when an increased number of AmpC or ESBL positive samples have previously been detected (Chapter 3, Figure 3.9). Future work should also focus on understanding ARG abundance and the prevalence of AmpC- and ESBL-producing Enterobacteriaceae in higher risk animals (e.g. calves, including a comparison between calves fed waste milk compared to milk replacer) and higher-risk farms (e.g. those with higher total AMU and/or use of antimicrobials which are risk factors for AmpC- and ESBL-producing E. coli such as amoxicillin/clavulanic acid and third- and fourth-generation cephalosporins). Calves were not included in this study as more intensive sampling at shorter intervals would be required (compared to once a month sampling), which was outside the scope of this research. However, on dairy farms calves have been associated with a high ARG abundance [13] and prevalence of AmpC- and ESBL-producing E. coli [194]. Therefore, it is hypothesised that NZ calf faeces would have a higher prevalence of AmpC- and ESBL-producing E. coli compared to other farm environmental samples analysed in this study. In addition, further studies using WGS are required to determine the transmission pathways of antimicrobial resistant bacteria at the dairy farm, human and environmental interface, utilising

a broader "One Health" approach encompassing human, animal, environmental (e.g. farm drains and nearby waterways) and wildlife (e.g. birds and mammalian pests within the farm environment) sources.

## 6.3.3 Sample variables

The faecal samples collected from sick dairy cows receiving systemic antimicrobial therapy (Chapter 4) were collected from working dairy farms (Dairy 1 and Dairy 4). As such, the study variables were not all controlled and there was variation in illness type and severity at the individual animal level as well as temporal variation in farm management factors. Although the illness for each cow was described in the sampling questionnaire (Appendix O), no measurement of disease severity was included as this is subjective and multiple vet/farm staff were involved in the faecal sample collection across the two farms, making a consistent comparison difficult. To overcome the latter study limitation, healthy untreated cows were included as controls to account for temporal variation in season as well as differences in management factors. A farm level comparison of the impact of antimicrobial therapy on the faecal microbiome could not be undertaken due to the difference in sample number across the two farms, with 28 and 2 treated cows recruited in the study from Dairy 4 and Dairy 1, respectively (Appendix P). Both farms were recruited in the study for the same period of time and data collected from the longitudinal study indicated that AMU was higher on Dairy 1 (Chapter 3), however the smaller sample size on Dairy 1 may have been due to fewer farm staff working and available for sample collection or differences between the number of cows treated with systemic antimicrobials during the microbiome study (only six months of the longitudinal (Chapters 3 and 5) and microbiome (Chapter 4) study periods overlapped). Despite these limitations, the results of this research shed light on the impact on the bovine faecal microbiome of a range of systemic antimicrobial treatments in diseased cows from working NZ dairy farms, providing "real life" scenarios.

## 6.3.4 Sequencing methodologies

This research assessed the impact of antimicrobial treatment on the faecal microbiome using 16S rRNA amplicon sequencing which is a cost-effective, high throughput method to study the microbiome [221]. However, limitations of this method include potential primer biases [221] and in comparison, shotgun metagenomic methods allow deeper sequencing and have been shown to identify taxa present at a lower abundance [490]. Shotgun metagenomic sequencing also allows for functional analysis and detection of the resistome, including antimicrobial, heavy metal and biocide resistance genes [30]. However, 16S rRNA amplicon sequencing was suitable to answer the specific research objectives in this study, allowing a high resolution comparison of the bovine faecal microbiome between sampling periods at the ASV level (Chapter 4) [491]. This study did not focus on archaea or investigate changes in fungal populations in the faecal microbiome pre-, during and post-treatment, the latter

of which could be additionally analysed using internal transcribed spacer marker gene analysis [221]. Due to the high amount of variation at the individual animal level, treated animals within a case study could not be grouped at the treatment level. Therefore, robust statistical tests to compare changes in the relative abundance of specific ASVs between samples S1, S2 and S3 could not be used. Future studies should collect replicate faecal samples from individual cows for such analyses. The data also suggested that generally the bacterial diversity, richness and composition had recovered post-treatment when the cow re-entered the milking herd (median 6 days). Therefore, additional studies should focus on a smaller number of cows and collect samples more frequently between treatment and when the cow re-enters the milking herd to examine the stages of microbiome recovery in more detail e.g. the sequence which ASVs return to pre-treatment levels.

#### 6.3.5 Computational limitations

The bioinformatic tools and databases used for AMR research are constantly evolving. As such, additional modifications to improve the pipeline used for the analysis of shotgun metagenomic sequencing reads (Chapter 5, Figure 5.1) should be routinely undertaken. Specific modifications to include are secondary functional validation of ARGs which were identified as requiring SNP confirmation using the Resistance Gene Identifier [492]. This secondary validation is used to reduce any false positive results in ARGs which confer resistance due to SNPs, such as mutations in the gyrA gene which may lead to reduced susceptibility to fluoroquinolones. Additionally, the sequencing reads and contigs were taxonomically classified using Kraken and CAT, respectively (Chapter 5, sections 5.2.4 and 5.2.5). The databases used were not customised to include additional metagenomeassembled genomes (MAGs). Inclusion of such sequences, for example a collection of 4,941 rumen MAGs [493], could be added to create a customised database which may improve the number of reads and contigs taxonomically classified in this study.

# 6.3.6 Analysis of faecal samples enriched in media containing specific antibiotics

High resolution analysis of the impact of systemic antimicrobial treatment on the faecal microbiome and resistome is crucial to understand the potential selection pressures driving AMR in the dairy farm environment with an increased focus at the individual animal level. Further research in this area could include enriching the faecal samples in media containing specific antibiotics to select for functional resistant phenotypes and analysing the samples using shotgun metagenomic sequencing. These enriched faecal samples could be plated on selective media to isolate antimicrobial resistant bacteria of interest. Antimicrobial susceptibility testing and WGS, particularly long-read sequencing (e.g. PacBio or MinION sequencing platforms), of resistant isolates would enable high resolution analysis of the molecular mechanisms for resistance and mobility, including the genomic context of ARGs.

In addition, shotgun metagenomic sequencing of the enriched faecal sample would allow an in depth analysis of antimicrobial, heavy metal and biocide resistance genes enriched during antimicrobial treatment and assembly of MAGs would enable host identification, genomic context and gene linkage for resistance genes to be assessed. Analysing pre-enriched faecal samples from cows receiving systemic antimicrobial treatment would provide a comparison of the selective pressures and impact of treatment in the "natural" farm environment as well as *in vitro* enrichment under laboratory conditions.

# 6.3.7 Waste milk

Waste milk has been linked to the development of antimicrobial resistant bacteria [436, 494] and it is not recommended to feed waste milk to calves in NZ (https://www.dairynz. co.nz/news/waste-milk-is-it-calf-feed/), although some calves are fed waste milk in NZ. Shotgun metagenomic sequencing analysis of one waste milk sample revealed a number of contigs belonging to mastitis-associated pathogens such as S. uberis as well as a plethora of antimicrobial, heavy metal and biocide resistance genes (Chapter 5, Figure 5.9), many of which potentially confer resistance across a number of compounds. A small proportion of contigs assembled from the waste milk harboured acquired ARGs (Chapter 5, Figure 5.10), however MAGs were not assembled from the samples in this study. Future work should focus on assembling MAGs from the farm environmental samples and identifying MAGs harbouring antimicrobial, heavy metal or biocide resistance genes to elucidate the genomic context and the bacterial hosts harbouring such genes. If the ARGs are present on plasmids, additional Hi-C analysis may be required to link the plasmid with the bacterial host [495]. These findings suggests that waste milk may be a source of pathogenic bacteria and ARGs within the dairy farm environment. Further research is required to understand how many farms in NZ feed waste milk to calves, any risks this practice poses as well as providing clear industry guidelines on the use of waste milk as calf feed in NZ.

## 6.3.8 Environmental vectors

AmpC- and ESBL-producing *E. coli* have been identified from cloacal and surface swabs from migratory and wild birds near Ohio dairy farms [368] as well as from wild birds in the Netherlands [496] and in Spain [497]. AmpC- and ESBL-producing *E. coli* have also been isolated from other non-avian vectors including wildlife such as rats (*Rattus* spp.), mice (*Mus musculus*) and hedgehogs (for example, *Erinaceus europaeus*) [498, 499], all of which are pests that can be found on NZ dairy farms. White stork (*Ciconia ciconia*) colonies, which are migratory birds, were closer in distance to landfills with human waste were more likely to be positive for third-generation cephalosporin resistant *E. coli* [500] and the isolation rate of AmpC- and ESBL-producing *E. coli* was significantly higher in birds associated with aquatic environments [496]. The environment and feed may be a source of antimicrobial resistant bacteria carried by wild birds and therefore birds may be a potential vector for the dissemination of antimicrobial resistant bacteria.

The Dairy 1 and Dairy 4 farms are located near the city of Palmerston North, as well as in close proximity to other dairy farms. Therefore, birds and wildlife could potentially mix between urban and agricultural environments. If present, bird faecal samples were collected from the dairy farm environment at each sampling visit in this study. The samples were processed in accordance with the methods detailed in Chapter 3, section 3.2.3 and the enrichments stored with glycerol (30% [v/v]) at - $80^{\circ}$ C. In addition, supplementary feed was collected during sampling visits if it was being used (Appendix W) and processed as described above. Enrichment samples from bird faeces and supplementary feed could be plated on selective agar plates to select for AmpC- and ESBL-producing *Enterobacteriaceae* to identify whether birds or supplementary feed may be a vector for these resistant bacteria in the NZ dairy farm environment. To the best of the author's knowledge, no studies have investigated the prevalence of AmpC- and ESBL-producing *Enterobacteriaceae* from birds or wildlife in the NZ dairy farm environment.

# 6.4 Conclusion

This research used multiple approaches to study AMR in the NZ dairy farm environment; (i) targeted culture-based methods to examine the sample level prevalence of AmpC- and ESBL-producing *E. coli*, (ii) 16S rRNA sequencing to examine the impact of systemic antimicrobial therapy on the bovine faecal microbiome coupled with culture-based isolation of specific antimicrobial resistant bacteria at the individual animal level, and (iii) using shotgun metagenomics to assess the abundance of antimicrobial, heavy metal and biocide resistance genes as well as identifying the bacterial host and genomic context of ARGs where possible.

The sample level prevalence of ESBL-producing *E. coli*, which are listed as "critical" on the WHO Priority Pathogens List [54], was low in the two dairy farms included in this study. This low prevalence may be attributed to NZ's predominantly pasture-based dairy farm production system and the comparatively low AMU in animal production, particularly on dairy farms. The recommendation to reduce the use of antimicrobials classified as red tier by the NZVA may also influence AMU and the low sample level prevalence of ESBL-producing E. coli. The AMU on both farms was comparable to other dairy farms throughout NZ and interestingly Dairy 1, which has a smaller herd size, had a higher AMU during the study period, although both farms used a low amount of antimicrobials classified as red tier by the NZVA. Compared to ESBL-producing E. coli, the sample level prevalence was higher for AmpC-producing E. coli. All AmpC-producers on the larger farm (Dairy 4) were plasmid-mediated, compared to Dairy 1 in which most of the AmpCproducers were chromosomally mediated (14 of 16 E. coli). Further studies are required to assess the risk factors between plasmid and chromosomally mediated AmpC-producing E. coli and to explore any seasonal effects on the prevalence of AmpC- and ESBL-producing E. coli as they were only isolated during spring and summer in this study.

This research demonstrated that systemic antimicrobial therapy reduced the diversity and richness of the bovine faecal microbiome during treatment and that this perturbation in bacterial community structure generally recovered post-treatment when the cow re-entered the milking herd. However, the animal level variation was high and cows with more serious conditions showed greater perturbations in the faecal microbiome composition (e.g. cows with left displacement of the abomasum compared to footrot). Treatment with ceftiofur resulted in the emergence of plasmid-mediated AmpC-producing *E. coli* in one of the three animals treated, which highlights the need for prudent antimicrobial stewardship in both animal and human health to reduce the development and spread of AMR. The abundance of ARGs across the NZ dairy farm environment was comparatively low compared to overseas data [478, 477, 465, 13, 210]. There was no statistical difference in the ARG abundance between either farm. Overall, the highest ARG abundance was in soil, yet the most diverse range of ARGs was in FDE. There was no association between the normalised ARG abundance and AMU or seasonality, although the low ARG abundances in farm

environmental samples may have made any associations difficult to detect.

At current rates and left unchecked it is estimated that by 2050, 10 million human deaths per year will be attributable to AMR worldwide [5]. In 2019 it was estimated that globally 4.95 million deaths and 1.27 millions deaths were associated with or attributable to bacterial AMR, respectively [36]. Therefore, AMR is a global burden for human, animal and environmental health and requires a holistic "One Health" approach to address [501]. The "New Zealand Antimicrobial Resistance Action Plan" was published in 2017 and priority areas for action were identified including (i) increasing awareness and understanding of AMR, (ii) strengthening AMR surveillance and research, (iii) improving infection prevention and control measures to prevent the transmission of antibiotic reistant bacteria, (iv) improving antimicrobial stewardship in human health, animal health and agriculture and (v) establishing clear governance, collaboration and investment for approaches to counter AMR [502]. This PhD research contributes to objective two of NZ's AMR action plan. To encourage prudent antimicrobial stewardship in animal husbandry, the NZVA has aspired that by 2030 NZ will not require antimicrobials for the maintenance of animal health and wellness [42], instead their use will be reserved for the treatment of disease. The SARS-CoV-2 pandemic has demonstrated how globally connected the world is, thus efforts to combat AMR need to include both national and global initiatives which encompass a "One Health" approach. Prevention is better than a cure and urgent action is required to slow the development and dissemination of AMR worldwide as well as encouraging prudent antimicrobial stewardship and continuing the research and development of new antimicrobial treatments.

# Bibliography

- Laxminarayan, R., Duse, A., Wattal, C., Zaidi, A. K. M., Wertheim, H. F. L., Sumpradit, N., Vlieghe, E., Hara, G. L., Gould, I. M., Goossens, H., Greko, C., So, A. D., Bigdeli, M., Tomson, G., Woodhouse, W., Ombaka, E., Peralta, A. Q., Qamar, F. N., Mir, F., Kariuki, S., Bhutta, Z. A., Coates, A., Bergstrom, R., Wright, G. D., Brown, E. D., and Cars, O. Antibiotic resistance-the need for global solutions. *Lancet Infectious Diseases*, 13(12):1057– 1098, 2013, DOI: https://doi.org/10.1016/s1473-3099(13)70318-9.
- [2] Holmes, A., Moore, L., Sundsfjord, A., Steinbakk, M., Regmi, S., Karkey, A., Guerin, P., and Piddock, L. Understanding the mechanisms and drivers of antimicrobial resistance. *Lancet*, 387(10014):176–187, 2016, DOI: https://doi.org/10.1016/s0140-6736(15)00473-0.
- [3] Seiler, C. and Berendonk, T. U. Heavy metal driven co-selection of antibiotic resistance in soil and water bodies impacted by agriculture and aquaculture. *Frontiers in Microbiology*, 3:10, 2012, DOI: https://doi.org/10.3389/fmicb.2012.00399.
- [4] Partridge, S. R., Kwong, S. M., Firth, N., and Jensen, S. O. Mobile genetic elements associated with antimicrobial resistance. *Clinical Microbiology Reviews*, 31(4):00088–17, 8 2018, DOI: https://doi.org/10.1128/CMR.00088-17.
- [5] O'Neill, J. Antimicrobial Resistance: Tackling a crisis for the health and wealth of nations. Report, 2014, https://amr-review.org/sites/default/files/ AMRReviewPaper-Tacklingacrisisforthehealthandwealthofnations\_1.pdf.
- [6] Dahms, C., Hubner, N. O., Kossow, A., Mellmann, A., Dittmann, K., and Kramer, A. Occurrence of ESBL-producing *Escherichia coli* in livestock and farm workers in Mecklenburg-Western Pomerania, Germany. *PLOS One*, 10(11):13, 2015, DOI: https://doi.org/10.1371/ journal.pone.0143326.
- [7] Gonggrijp, M., Santman-Berends, I., Heuvelink, A., Buter, G., van Schaik, G., Hage, J., and Lam, T. Prevalence and risk factors for extended-spectrum beta-lactamase- and AmpCproducing *Escherichia coli* in dairy farms. *Journal of Dairy Science*, 99(11):9001–9013, 3 2016, DOI: https://doi.org/10.3168/jds.2016-11134.
- [8] Heuvelink, A. E., Gonggrijp, M. A., Buter, R. G. J., ter Bogt-Kappert, C. C., van Schaik, G., Velthuis, A. G. J., and Lam, T. Prevalence of extended-spectrum and AmpC beta-lactamaseproducing *Escherichia coli* in Dutch dairy herds. *Veterinary Microbiology*, 232:58–64, 2019, DOI: https://doi.org/10.1016/j.vetmic.2019.04.005.
- [9] Tello, M., Ocejo, M., Oporto, B., and Hurtado, A. Prevalence of cefotaxime-resistant Escherichia coli isolates from healthy cattle and sheep in Northern Spain: Phenotypic and genome-based characterization of antimicrobial susceptibility. Applied and Environmental Microbiology, 86(15):00742–20, 2020, DOI: https://doi.org/10.1128/aem.00742-20.

- [10] Manga, I., Hasman, H., Smidkova, J., Medvecky, M., Dolejska, M., and Cizek, A. Fecal carriage and whole-genome sequencing-assisted characterization of CMY-2 beta-lactamaseproducing *Escherichia coli* in calves at Czech dairy cow farm. *Foodborne Pathogens and Disease*, 16(1):42–53, 1 2019, DOI: https://doi.org/10.1089/fpd.2018.2531.
- [11] Liu, J., Taft, D. H., Maldonado-Gomez, M. X., Johnson, D., Treiber, M. L., Lemay, D. G., DePeters, E. J., and Mills, D. A. The fecal resistome of dairy cattle is associated with diet during nursing. *Nature Communications*, 10(1):4406, 2019, DOI: https://doi.org/10.1038/ s41467-019-12111-x.
- [12] Zaheer, R., Lakin, S. M., Polo, R. O., Cook, S. R., Larney, F. J., Morley, P. S., Booker, C. W., Hannon, S. J., Van Domselaar, G., Read, R. R., and McAllister, T. A. Comparative diversity of microbiomes and resistomes in beef feedlots, downstream environments and urban sewage influent. *BMC Microbiology*, 19(1):197, 2019, DOI: https://doi.org/10.1186/ s12866-019-1548-x.
- [13] Haley, B. J., Kim, S.-W., Salaheen, S., Hovingh, E., and Van Kessel, J. A. S. Differences in the microbial community and resistome structures of feces from preweaned calves and lactating dairy cows in commercial dairy herds. *Foodborne Pathogens and Disease*, 17(8):494–503, 3 2020, DOI: https://doi.org/10.1089/fpd.2019.2768.
- [14] George, A. Antimicrobial resistance, trade, food safety and security. One Health, 5:6–8, 2018, DOI: https://doi.org/10.1016/j.onehlt.2017.11.004.
- [15] Wemette, M., Greiner Safi, A., Wolverton, A. K., Beauvais, W., Shapiro, M., Moroni, P., Welcome, F. L., and Ivanek, R. Public perceptions of antibiotic use on dairy farms in the United States. *Journal of Dairy Science*, 104(3):2807–2821, 2021, DOI: https://doi.org/ https://doi.org/10.3168/jds.2019-17673.
- [16] Hillerton, J., Irvine, C., Bryan, M., Scott, D., and Merchant, S. Use of antimicrobials for animals in New Zealand, and in comparison with other countries. *New Zealand Veterinary Journal*, 65(2):71–77, 2017, DOI: https://doi.org/10.1080/00480169.2016.1171736.
- [17] Hillerton, J., Bryan, M., Beattie, B., Scott, D., Millar, A., and French, N. Use of antimicrobials for food animals in New Zealand: updated estimates to identify a baseline to measure targeted reductions. *New Zealand Veterinary Journal*, 69(3):180–185, 3 2021, DOI: https://doi.org/10.1080/00480169.2021.1890648.
- [18] Holmes, C. W., Brookes, I. M., Garrick, D. J., Mackenzie, D. D. S., Parkinson, T. J., and Wilson, G. F. Pastoral dairy farming in New Zealand. In Swain, D., editor, *Milk production from pasture: Principles and practices*. Massey University, 2002.
- [19] Lacy-Hulbert, J., Williamson, J., Kolver, E., Doohan, H., and Shelgren, J. Is coliform mastitis an emerging issue? In *Proceedings of the New Zealand Milk Quality Conference*. New Zealand Veterinary Association, 2012.
- [20] Ministry for Primary Industries. Antibiotic sales analysis 2018. Report, 2020, https://www.mpi.govt.nz/dmsdocument/42075/direct.

- [21] Compton, C. and McDougall, S. Patterns of antibiotic sales to Waikato dairy farms. Vetscript, 27:22–24, 2014.
- [22] McDougall, S., Hussein, H., and Petrovski, K. Antimicrobial resistance in Staphylococcus aureus, Streptococcus uberis and Streptococcus dysgalactiae from dairy cows with mastitis. New Zealand Veterinary Journal, 62(2):68–76, 2014, DOI: https://doi.org/10.1080/00480169. 2013.843135.
- [23] Greening, S. S., Zhang, J., Midwinter, A. C., Wilkinson, D. A., McDougall, S., Gates, M. C., and French, N. P. The genetic relatedness and antimicrobial resistance patterns of mastitiscausing *Staphylococcus aureus* strains isolated from New Zealand dairy cattle. *Veterinary Sciences*, 8(11), 2021, DOI: https://doi.org/10.3390/vetsci8110287.
- [24] McDougall, S., Clausen, L., Ha, H.-J., Gibson, I., Bryan, M., Hadjirin, N., Lay, E., Raisen, C., Ba, X., Restif, O., Parkhill, J., and Holmes, M. A. Mechanisms of β-lactam resistance of *Streptococcus uberis* isolated from bovine mastitis cases. *Veterinary Microbiology*, 242:108592, 2020, DOI: https://doi.org/https://doi.org/10.1016/j.vetmic.2020.108592.
- [25] Burgess, S. A., Aplin, J., Biggs, P. J., Breckell, G., Benschop, J., Fayaz, A., Toombs-Ruane, L. J., and Midwinter, A. C. Characterisation of AmpC and extended-spectrum beta-lactamase producing *E. coli* from New Zealand dairy farms. *International Dairy Journal*, 117:104998, 2021, DOI: https://doi.org/https://doi.org/10.1016/j.idairyj.2021.104998.
- [26] Burgess, S. A., Cookson, A. L., Brousse, L., Ortolani, E., Benschop, J., Akhter, R., Brightwell, G., and McDougall, S. The epidemiology of AmpC-producing *Escherichia coli* isolated from dairy cattle faeces on pasture-fed farms. *Journal of Medical Microbiology*, 70(10), 10 2021, DOI: https://doi.org/10.1099/jmm.0.001447.
- [27] Mwenifumbo, M. Extended-spectrum β-lactamase (ESBL) and AmpC β-lactamase (AmpC) producing Escherichia coli in dairy calves from the Canterbury region. Thesis, Massey University, 2020.
- [28] Collis, R. M., Burgess, S. A., Biggs, P. J., Midwinter, A. C., French, N. P., Toombs-Ruane, L., and Cookson, A. L. Extended-spectrum beta-lactamase-producing *Enterobacteriaceae* in dairy farm environments: A New Zealand perspective. *Foodborne Pathogens and Disease*, 16(1):5–22, 2019, DOI: https://doi.org/10.1089/fpd.2018.2524.
- [29] Pal, C., Bengtsson-Palme, J., Kristiansson, E., and Larsson, D. G. J. Co-occurrence of resistance genes to antibiotics, biocides and metals reveals novel insights into their co-selection potential. BMC Genomics, 16(1):964, 2015, DOI: https://doi.org/10.1186/s12864-015-2153-5.
- [30] Boolchandani, M., D'Souza, A. W., and Dantas, G. Sequencing-based methods and resources to study antimicrobial resistance. *Nature Reviews Genetics*, 2019, DOI: https://doi.org/10. 1038/s41576-019-0108-4.
- [31] Pankey, G. A. and Sabath, L. D. Clinical relevance of bacteriostatic versus bactericidal mechanisms of action in the treatment of Gram-positive bacterial infections. *Clinical Infectious Diseases*, 38(6):864–870, 3 2004, DOI: https://doi.org/10.1086/381972.

- [32] Christaki, E., Marcou, M., and Tofarides, A. Antimicrobial resistance in bacteria: mechanisms, evolution, and persistence. *Journal of Molecular Evolution*, 88(1):26–40, 1 2020, DOI: https://doi.org/10.1007/s00239-019-09914-3.
- [33] Pitout, J. D. D. and Laupland, K. B. Extended-spectrum beta-lactamase-producing Enterobacteriaceae: An emerging public-health concern. Lancet Infectious Diseases, 8(3):159–166, 2008, DOI: https://doi.org/10.1016/s1473-3099(08)70041-0.
- [34] Bloom, D. E. and Cadarette, D. Infectious disease threats in the Twenty-First Century: strengthening the global response. *Frontiers in Immunology*, 10:549, 2019, DOI: https://doi. org/10.3389/fimmu.2019.00549.
- [35] Gardiner, S. J., Duffy, E. J., Chambers, S. T., et al. Antimicrobial stewardship in human healthcare in Aotearoa New Zealand: urgent call for national leadership and coordinated efforts to preserve antimicrobial effectiveness. *The New Zealand Medical Journal*, 134(1544):113–128, 10 2021.
- [36] Murray, C. J. L., Ikuta, K. S., Sharara, F., et al. Global burden of bacterial antimicrobial resistance in 2019: a systematic analysis. *The Lancet*, 1 2022, DOI: https://doi.org/10.1016/ S0140-6736(21)02724-0.
- [37] Horner, C., Mawer, D., and Wilcox, M. Reduced susceptibility to chlorhexidine in staphylococci: Is it increasing and does it matter? Journal of Antimicrobial Chemotherapy, 67(11):2547–2559, 2012, DOI: https://doi.org/10.1093/jac/dks284.
- [38] Jaglic, Z. and Cervinkova, D. Genetic basis of resistance to quaternary ammonium compounds - the qac genes and their role: A review. Veterinarni Medicina, 57(6):275–281, 2012.
- [39] Martinez, J. L. The role of natural environments in the evolution of resistance traits in pathogenic bacteria. *Proceedings of the Royal Society B: Biological Sciences*, 276(1667):2521– 2530, 7 2009, DOI: https://doi.org/10.1098/rspb.2009.0320.
- [40] Martinez, J. L. General principles of antibiotic resistance in bacteria. Drug Discovery Today: Technologies, 11:33–39, 2014, DOI: https://doi.org/https://doi.org/10.1016/j.ddtec.2014.02. 001.
- [41] Anonymous. Agricultural compounds and veterinary medicines act 1997, 1997, http://www. legislation.govt.nz/act/public/1997/0087/latest/DLM414577.html.
- [42] Anonymous. Antibiotic judicious use guidelines for the New Zealand veterinary profession: Dairy. Report, Wellington, New Zealand, 2016, https://www.amrvetcollective.com/assets/ guidelines/guide\_dairy.pdf.
- [43] Compton, C. and McDougall, S. Anti-microbial usage on Waikato dairy farms. Vetscript, 26:34–35, 2013.
- [44] Harrison, E. M., Paterson, G. K., Holden, M. T. G., Larsen, J., Stegger, M., Larsen, A. R., Petersen, A., Skov, R. L., Christensen, J. M., Bak Zeuthen, A., Heltberg, O., Harris, S. R., Zadoks, R. N., Parkhill, J., Peacock, S. J., and Holmes, M. A. Whole genome sequencing identifies zoonotic transmission of MRSA isolates with the novel mecA homologue mecC. EMBO Molecular Medicine, 5(4):509–515, 2013, DOI: https://doi.org/10.1002/emmm.201202413.

- [45] Wu, G. H., Day, M. J., Mafura, M. T., Nunez-Garcia, J., Fenner, J. J., Sharma, M., van Essen-Zandbergen, A., Rodriguez, I., Dierikx, C., Kadlec, K., Schink, A. K., Wain, J., Helmuth, R., Guerra, B., Schwarz, S., Threlfall, J., Woodward, M. J., Woodford, N., Coldham, N., and Mevius, D. Comparative analysis of ESBL-positive *Escherichia coli* isolates from animals and humans from the UK, the Netherlands and Germany. *PLOS One*, 8(9):10, 2013, DOI: https://doi.org/10.1371/journal.pone.0075392.
- [46] Liu, Y., Wang, Y., Walsh, T., Yi, L., Zhang, R., Spencer, J., Doi, Y., Tian, G., Dong, B., Huang, X., Yu, L., Gu, D., Ren, H., Chen, X., Lv, L., He, D., Zhou, H., Liang, Z., Liu, J., and Shen, J. Emergence of plasmid-mediated colistin resistance mechanism MCR-1 in animals and human beings in China: A microbiological and molecular biological study. *Lancet Infectious Diseases*, 16(2):161–168, 2016, DOI: https://doi.org/10.1016/s1473-3099(15)00424-7.
- [47] Rozwandowicz, M., Brouwer, M. S. M., Fischer, J., Wagenaar, J. A., Gonzalez-Zorn, B., Guerra, B., Mevius, D. J., and Hordijk, J. Plasmids carrying antimicrobial resistance genes in *Enterobacteriaceae*. Journal of Antimicrobial Chemotherapy, 73(5):1121–1137, 5 2018, DOI: https://doi.org/10.1093/jac/dkx488.
- [48] McGann, P., Snesrud, E., Maybank, R., Corey, B., Ong, A., Clifford, R., Hinkle, M., Whitman, T., Lesho, E., and Schaecher, K. *Escherichia coli* harboring *mcr-1* and *bla*CTX-M on a novel IncF plasmid: First report of *mcr-1* in the United States. *Antimicrobial Agents and Chemotherapy*, 60(7):4420–4421, 2016, DOI: https://doi.org/10.1128/aac.01103-16.
- [49] Poirel, L., Bonnin, R. A., and Nordmann, P. Analysis of the resistome of a multidrugresistant NDM-1-producing *Escherichia coli* strain by high-throughput genome sequencing. *Antimicrobial Agents and Chemotherapy*, 55(9):4224–4229, 2011, DOI: https://doi.org/10. 1128/aac.00165-11.
- [50] Ho, P., Chan, J., Lo, W., Law, P., Li, Z., Lai, E., and Chow, K. Dissemination of plasmid-mediated fosfomycin resistance *fosA3* among multidrug-resistant *Escherichia coli* from livestock and other animals. *Journal of Applied Microbiology*, 114(3):695–702, 2013, DOI: https://doi.org/10.1111/jam.12099.
- [51] Martínez, J. L. Antibiotics and antibiotic resistance genes in natural environments. Science, 321(5887):365–367, 7 2008, DOI: https://doi.org/10.1126/science.1159483.
- [52] Martinez, J. L., Sánchez, M. B., Martínez-Solano, L., Hernandez, A., Garmendia, L., Fajardo, A., and Alvarez-Ortega, C. Functional role of bacterial multidrug efflux pumps in microbial natural ecosystems. *FEMS Microbiology Reviews*, 33(2):430–449, 3 2009, DOI: https://doi. org/10.1111/j.1574-6976.2008.00157.x.
- [53] Martínez, J. L., Coque, T. M., and Baquero, F. What is a resistance gene? Ranking risk in resistomes. *Nature Reviews Microbiology*, 13(2):116–123, 2015, DOI: https://doi.org/10. 1038/nrmicro3399.
- [54] Tacconelli, E., Carrara, E., Savoldi, A., Harbarth, S., Mendelson, M., Monnet, D. L., Pulcini, C., Kahlmeter, G., Kluytmans, J., Carmeli, Y., Ouellette, M., Outterson, K., Patel, J., Cavaleri, M., Cox, E. M., Houchens, C. R., Grayson, M. L., Hansen, P., Singh, N.,

Theuretzbacher, U., Magrini, N., and WHO Pathogens Priority List Working Group. Discovery, research, and development of new antibiotics: The WHO priority list of antibiotic-resistant bacteria and tuberculosis. *Lancet Infectious Diseases*, 18(3):318–327, 2018, DOI: https://doi.org/10.1016/s1473-3099(17)30753-3.

- [55] Ali, T., Rahman, S. U., Zhang, L. M., Shahid, M., Zhang, S. Y., Liu, G., Gao, J., and Han, B. ESBL-producing *Escherichia coli* from cows suffering mastitis in China contain clinical class 1 integrons with CTX-M linked to ISCR1. *Frontiers in Microbiology*, 7:11, 2016, DOI: https://doi.org/10.3389/fmicb.2016.01931.
- [56] Rodriguez-Rivera, L. D., Cummings, K. J., Loneragan, G. H., Rankin, S. C., Hanson, D. L., Leone, W. M., and Edrington, T. S. *Salmonella* prevalence and antimicrobial susceptibility among dairy farm environmental samples collected in Texas. *Foodborne Pathogens and Disease*, 13(4):205–211, 2016, DOI: https://doi.org/10.1089/fpd.2015.2037.
- [57] Vandendriessche, S., Vanderhaeghen, W., Soares, F. V., Hallin, M., Catry, B., Hermans, K., Butaye, P., Haesebrouck, F., Struelens, M. J., and Denis, O. Prevalence, risk factors and genetic diversity of methicillin-resistant *Staphylococcus aureus* carried by humans and animals across livestock production sectors. *Journal of Antimicrobial Chemotherapy*, 68(7):1510– 1516, 2013, DOI: https://doi.org/10.1093/jac/dkt047.
- [58] Gomes, F. and Henriques, M. Control of bovine mastitis: Old and recent therapeutic approaches. *Current Microbiology*, 72(4):377–382, 2016, DOI: https://doi.org/10.1007/ s00284-015-0958-8.
- [59] Oliveira, L., Hulland, C., and Ruegg, P. L. Characterization of clinical mastitis occurring in cows on 50 large dairy herds in Wisconsin. *Journal of Dairy Science*, 96(12):7538–7549, 2013, DOI: https://doi.org/10.3168/jds.2012-6078.
- [60] Bal, A. M., Coombs, G. W., Holden, M. T. G., Lindsay, J. A., Nimmo, G. R., Tattevin, P., and Skov, R. L. Genomic insights into the emergence and spread of international clones of healthcare-, community- and livestock-associated meticillin-resistant *Staphylococcus aureus*: Blurring of the traditional definitions. *Journal of Global Antimicrobial Resistance*, 6:95–101, 2016, DOI: https://doi.org/10.1016/j.jgar.2016.04.004.
- [61] De Vliegher, S., Fox, L. K., Piepers, S., McDougall, S., and Barkema, H. W. Mastitis in dairy heifers: Nature of the disease, potential impact, prevention, and control. *Journal of Dairy Science*, 95(3):1025–1040, 2012, DOI: https://doi.org/10.3168/jds.2010-4074.
- [62] Zadoks, R. N., Middleton, J. R., McDougall, S., Katholm, J., and Schukken, Y. H. Molecular epidemiology of mastitis pathogens of dairy cattle and comparative relevance to humans. *Journal of Mammary Gland Biology and Neoplasia*, 16(4):357–372, 2011, DOI: https://doi. org/10.1007/s10911-011-9236-y.
- [63] Gao, J., Barkema, H. W., Zhang, L. M., Liu, G., Deng, Z. J., Cai, L. J., Shan, R. X., Zhang, S. Y., Zou, J. Q., Kastelic, J. P., and Han, B. Incidence of clinical mastitis and distribution of pathogens on large Chinese dairy farms. *Journal of Dairy Science*, 100(6):4797–4806, 2017, DOI: https://doi.org/10.3168/jds.2016-12334.

- [64] Sztachanska, M., Baranski, W., Janowski, T., Pogorzelska, J., and Zdunczyk, S. Prevalence and etiological agents of subclinical mastitis at the end of lactation in nine dairy herds in North-East Poland. *Polish Journal of Veterinary Sciences*, 19(1):119–124, 2016, DOI: https://doi.org/10.1515/pjvs-2016-0015.
- [65] McDougall, S., Arthur, D. G., Bryan, M. A., Vermunt, J. J., and Weir, A. M. Clinical and bacteriological response to treatment of clinical mastitis with one of three intramammary antibiotics. *New Zealand Veterinary Journal*, 55(4):161–170, 2007, DOI: https://doi.org/10. 1080/00480169.2007.36762.
- [66] Petrovski, K. R., Laven, R. A., and Lopez-Villalobos, N. A descriptive analysis of the antimicrobial susceptibility of mastitis-causing bacteria isolated from samples submitted to commercial diagnostic laboratories in New Zealand (2003-2006). New Zealand Veterinary Journal, 59(2):59–66, 2011, DOI: https://doi.org/10.1080/00480169.2011.552853.
- [67] Dyson, R., Charman, N., Hodge, A., Rowe, S. M., and Taylor, L. F. A survey of mastitis pathogens including antimicrobial susceptibility in southeastern Australian dairy herds. *Jour*nal of Dairy Science, 2021, DOI: https://doi.org/https://doi.org/10.3168/jds.2021-20955.
- [68] McDougall, S., Niethammer, J., and Graham, E. M. Antimicrobial usage and risk of retreatment for mild to moderate clinical mastitis cases on dairy farms following on-farm bacterial culture and selective therapy. NZ Veterinary Journal, 66(2):98–107, 2018, DOI: https://doi.org/10.1080/00480169.2017.1416692.
- [69] Nobrega, D. B., Naqvi, S. A., Dufour, S., Deardon, R., Kastelic, J. P., De Buck, J., and Barkema, H. W. Critically important antimicrobials are generally not needed to treat nonsevere clinical mastitis in lactating dairy cows: Results from a network metaanalysis. *Journal of Dairy Science*, 103(11):10585–10603, 2020, DOI: https://doi.org/https: //doi.org/10.3168/jds.2020-18365.
- [70] Kaper, J. B., Nataro, J. P., and Mobley, H. L. T. Pathogenic Escherichia coli. Nature Reviews Microbiology, 2(2):123–140, 2004, DOI: https://doi.org/10.1038/nrmicro818.
- [71] Orskov, I., Orskov, F., Jann, B., and Jann, K. Serology, chemistry and genetics of O and K antigens of *Escherichia coli*. *Bacteriological Reviews*, 41(3):667–710, 1977.
- [72] Stenutz, R., Weintraub, A., and Widmalm, G. The structures of *Escherichia coli* O-polysaccharide antigens. *FEMS Microbiology Reviews*, 30(3):382–403, 2006, DOI: https://doi.org/10.1111/j.1574-6976.2006.00016.x.
- [73] Joensen, K. G., Tetzschner, A. M. M., Iguchi, A., Aarestrup, F. M., and Scheutz, F. Rapid and easy in silico serotyping of *Escherichia coli* isolates by use of whole genome sequencing data. *Journal of Clinical Microbiology*, 53(8):2410–2426, 2015, DOI: https://doi.org/10.1128/ jcm.00008-15.
- [74] Larsen, M. V., Cosentino, S., Rasmussen, S., Friis, C., Hasman, H., Marvig, R. L., Jelsbak, L., Sicheritz-Ponten, T., Ussery, D. W., Aarestrup, F. M., and Lund, O. Multilocus sequence typing of total genome sequenced bacteria. *Journal of Clinical Microbiology*, 50(4):1355–1361, 2012, DOI: https://doi.org/10.1128/jcm.06094-11.

- [75] Clermont, O., Bonacorsi, S., and Bingen, E. Rapid and simple determination of the Escherichia coli phylogenetic group. Applied and Environmental Microbiology, 66(10):4555– 4558, 10 2000, DOI: https://doi.org/10.1128/AEM.66.10.4555-4558.2000.
- [76] Clermont, O., Christenson, J., Denamur, E., and Gordon, D. The Clermont Escherichia coli phylo-typing method revisited: improvement of specificity and detection of new phylogroups. Environmental Microbiology Reports, 5(1):58–65, 2013, DOI: https://doi.org/10. 1111/1758-2229.12019.
- [77] Beghain, J., Bridier-Nahmias, A., Le Nagard, H., Denamur, E., and Clermont, O. Clermon-Typing: an easy-to-use and accurate *in silico* method for *Escherichia* genus strain phylotyping. *Microbial genomics*, 4(7):e000192, 7 2018, DOI: https://doi.org/10.1099/mgen.0.000192.
- [78] Clermont, O., Dixit, O. V. A., Vangchhia, B., Condamine, B., Dion, S., Bridier-Nahmias, A., Denamur, E., and Gordon, D. Characterization and rapid identification of phylogroup G in *Escherichia coli*, a lineage with high virulence and antibiotic resistance potential. *Environmental Microbiology*, 21(8):3107–3117, 8 2019, DOI: https://doi.org/10.1111/1462-2920. 14713.
- [79] Gordon, D. M. and Cowling, A. The distribution and genetic structure of *Escherichia coli* in Australian vertebrates: host and geographic effects. *Microbiology*, 149(12), 2003, DOI: https://doi.org/https://doi.org/10.1099/mic.0.26486-0.
- [80] Escobar-Páramo, P., Le Menac'h, A., Le Gall, T., Amorin, C., Gouriou, S., Picard, B., Skurnik, D., and Denamur, E. Identification of forces shaping the commensal *Escherichia coli* genetic structure by comparing animal and human isolates. *Environmental Microbiology*, 8(11):1975–1984, 11 2006, DOI: https://doi.org/10.1111/j.1462-2920.2006.01077.x.
- [81] Robins-Browne, R. M., Holt, K. E., Ingle, D. J., Hocking, D. M., Yang, J., and Tauschek, M. Are *Escherichia coli* pathotypes still relevant in the era of whole-genome sequencing? *Frontiers in Cellular and Infection Microbiology*, 6:141, 2016, DOI: https://doi.org/10.3389/ tcimb.2016.00141.
- [82] Fluit, A. C., Florijn, A., Verhoef, J., and Milatovic, D. Presence of tetracycline resistance determinants and susceptibility to tigecycline and minocycline. *Antimicrobial Agents and Chemotherapy*, 49(4):1636–1638, 2005, DOI: https://doi.org/10.1128/aac.49.4.1636-1638. 2005.
- [83] Liang, W.-j., Liu, H.-y., Duan, G.-C., Zhao, Y.-x., Chen, S.-y., Yang, H.-Y., and Xi, Y.-L. Emergence and mechanism of carbapenem-resistant *Escherichia coli* in Henan, China, 2014. *Journal of Infection and Public Health*, 11(3):347–351, 2018, DOI: https://doi.org/https: //doi.org/10.1016/j.jiph.2017.09.020.
- [84] Ambler, R. P., Baddiley, J., and Abraham, E. P. The structure of β-lactamases. *Philosophical Transactions of the Royal Society of London. B, Biological Sciences*, 289(1036):321–331, 5 1980, DOI: https://doi.org/10.1098/rstb.1980.0049.
- [85] Bush, K. Past and Present Perspectives on beta-Lactamases. Antimicrobial Agents and Chemotherapy, 62(10):20, 2018, DOI: https://doi.org/10.1128/aac.01076-18.

- [86] Bush, K. and Jacoby, G. A. Updated functional classification of beta-lactamases. Antimicrobial Agents and Chemotherapy, 54(3):969–976, 3 2010, DOI: https://doi.org/10.1128/AAC. 01009-09.
- [87] Liebana, E., Carattoli, A., Coque, T. M., Hasman, H., Magiorakos, A.-P., Mevius, D., Peixe, L., Poirel, L., Schuepbach-Regula, G., Torneke, K., Torren-Edo, J., Torres, C., and Threlfall, J. Public health risks of enterobacterial isolates producing extended-spectrum beta-lactamases or AmpC beta-lactamases in food and food-producing animals: An EU perspective of epidemiology, analytical methods, risk factors, and control options. *Clinical Infectious Diseases*, 56(7):1030–1037, 2013, DOI: https://doi.org/10.1093/cid/cis1043.
- [88] Majiduddin, F. K., Materon, I. C., and Palzkill, T. G. Molecular analysis of beta-lactamase structure and function. *International Journal of Medical Microbiology*, 292(2):127–137, 2002, DOI: https://doi.org/https://doi.org/10.1078/1438-4221-00198.
- [89] Tooke, C. L., Hinchliffe, P., Bragginton, E. C., Colenso, C. K., Hirvonen, V. H. A., Takebayashi, Y., and Spencer, J. β-lactamases and β-lactamase inhibitors in the 21st century. Journal of Molecular Biology, 431(18):3472–3500, 2019, DOI: https://doi.org/https: //doi.org/10.1016/j.jmb.2019.04.002.
- [90] Bush, K., Jacoby, G. A., and Medeiros, A. A. A functional classification scheme for beta-lactamases and its correlation with molecular structure. *Antimicrobial Agents and Chemotherapy*, 39(6):1211–1233, 6 1995, DOI: https://doi.org/10.1128/AAC.39.6.1211.
- [91] Pitout, J. D. D. Extraintestinal pathogenic Escherichia coli: An update on antimicrobial resistance, laboratory diagnosis and treatment. Expert Review of Anti-Infective Therapy, 10(10):1165–1176, 2012, DOI: https://doi.org/10.1586/eri.12.110.
- [92] Rubin, J. E. and Pitout, J. D. D. Extended-spectrum beta-lactamase, carbapenemase and AmpC producing *Enterobacteriaceae* in companion animals. *Veterinary Microbiology*, 170(1-2):10–18, 2014, DOI: https://doi.org/10.1016/j.vetmic.2014.01.017.
- [93] Jacoby, G. A. and Munoz-Price, L. S. Mechanisms of disease: The new beta-lactamases. New England Journal of Medicine, 352(4):380–391, 2005, DOI: https://doi.org/10.1056/ NEJMra041359.
- [94] Hirvonen, V. H. A., Spencer, J., and van der Kamp, M. W. Antimicrobial resistance conferred by OXA-48 β-lactamases: Towards a detailed mechanistic understanding. *Antimicrobial Agents and Chemotherapy*, 65(6):00184–21, 1 2022, DOI: https://doi.org/10.1128/AAC. 00184-21.
- [95] Iredell, J., Brown, J., and Tagg, K. Antibiotic resistance in *Enterobacteriaceae*: Mechanisms and clinical implications. *Bmj-British Medical Journal*, 352:19, 2016, DOI: https://doi.org/ 10.1136/bmj.h6420.
- [96] Bonnet, R. Growing group of extended-spectrum beta-lactamases: the CTX-M enzymes. Antimicrobial Agents and Chemotherapy, 48(1):1–14, 1 2004, DOI: https://doi.org/10.1128/ AAC.48.1.1-14.2004.

- [97] Collignon, P., Powers, J. H., Chiller, T. M., Aidara-Kane, A., and Aarestrup, F. M. World Health Organization ranking of antimicrobials according to their importance in human medicine: A critical step for developing risk management strategies for the use of antimicrobials in food production animals. *Clinical Infectious Diseases*, 49(1):132–141, 2009, DOI: https://doi.org/10.1086/599374.
- [98] Collignon, P. C., Conly, J. M., Andremont, A., McEwen, S. A., Aidara-Kane, A., and Group, W. H. O. A. World Health Organization ranking of antimicrobials according to their importance in human medicine: A critical step for developing risk management strategies to control antimicrobial resistance from food animal production. *Clinical Infectious Diseases*, 63(8):1087–1093, 2016, DOI: https://doi.org/10.1093/cid/ciw475.
- [99] McEwen, S. A. and Collignon, P. J. Antimicrobial resistance: a One Health perspective. *Microbiology spectrum*, 6(2), 3 2018, DOI: https://doi.org/10.1128/microbiolspec. ARBA-0009-2017.
- [100] Naas, T., Poirel, L., and Nordmann, P. Minor extended-spectrum beta-lactamases. Clinical Microbiology and Infection, 14:42–52, 2008, DOI: https://doi.org/10.1111/j.1469-0691.2007. 01861.x.
- [101] Zhao, W. H. and Hu, Z. Q. Epidemiology and genetics of CTX-M extended-spectrum βlactamases in Gram-negative bacteria. Crit Rev Microbiol, 39(1):79–101, 2013, DOI: https: //doi.org/10.3109/1040841x.2012.691460.
- [102] Canton, R., Gonzalez-Alba, J. M., and Galán, J. C. CTX-M enzymes: origin and diffusion. Frontiers in Microbiology, 3:110, 2012, DOI: https://doi.org/10.3389/fmicb.2012.00110.
- [103] Ramadan, A. A., Abdelaziz, N. A., Amin, M. A., and Aziz, R. K. Novel blaCTX-M variants and genotype-phenotype correlations among clinical isolates of extended spectrum beta lactamase-producing *Escherichia coli*. Scientific Reports, 9(1):4224, 3 2019, DOI: https://doi.org/10.1038/s41598-019-39730-0.
- [104] Rossolini, G. M., D'Andrea, M. M., and Mugnaioli, C. The spread of CTX-M-type extendedspectrum β-lactamases. *Clinical Microbiology and Infection*, 14:33–41, 2008, DOI: https: //doi.org/https://doi.org/10.1111/j.1469-0691.2007.01867.x.
- [105] Gniadkowski, M. Evolution of extended-spectrum β-lactamases by mutation. Clinical Microbiology and Infection, 14(s1):11–32, 1 2008, DOI: https://doi.org/https://doi.org/10.1111/j. 1469-0691.2007.01854.x.
- [106] Evans, B. A. and Amyes, S. G. B. OXA β-lactamases. Clinical Microbiology Reviews, 27(2):241–263, 4 2014, DOI: https://doi.org/10.1128/CMR.00117-13.
- [107] Brolund, A. and Sandegren, L. Characterization of ESBL disseminating plasmids. Infectious Diseases, 48(1):18–25, 1 2016, DOI: https://doi.org/10.3109/23744235.2015.1062536.
- [108] Nicolas-Chanoine, M.-H., Bertrand, X., and Madec, J.-Y. Escherichia coli ST131, an intriguing clonal group. Clinical Microbiology Reviews, 27(3):543–574, 7 2014, DOI: https: //doi.org/10.1128/CMR.00125-13.

- [109] Banerjee, R. and Johnson, J. R. A new clone sweeps clean: the enigmatic emergence of Escherichia coli sequence type 131. Antimicrobial Agents and Chemotherapy, 58(9):4997– 5004, 9 2014, DOI: https://doi.org/10.1128/AAC.02824-14.
- [110] Bevan, E. R., Jones, A. M., and Hawkey, P. M. Global epidemiology of CTX-M β-lactamases: Temporal and geographical shifts in genotype. Journal of Antimicrobial Chemotherapy, 72(8):2145–2155, 10 2017, DOI: https://doi.org/10.1093/jac/dkx146.
- [111] Canton, R., Novais, A., Valverde, A., Machado, E., Peixe, L., Baquero, F., and Coque, T. M. Prevalence and spread of extended-spectrum beta-lactamase-producing *Enterobacteriaceae* in Europe. *Clinical Microbiology and Infection*, 14:144–153, 2008, DOI: https://doi.org/10. 1111/j.1469-0691.2007.01850.x.
- [112] Livermore, D. M., Canton, R., Gniadkowski, M., Nordmann, P., Rossolini, G. M., Arlet, G., Ayala, J., Coque, T. M., Kern-Zdanowicz, I., Luzzaro, F., Poirel, L., and Woodford, N. CTX-M: Changing the face of ESBLs in Europe. *Journal of Antimicrobial Chemotherapy*, 59(2):165–174, 10 2006, DOI: https://doi.org/10.1093/jac/dkl483.
- [113] Day, M. J., Rodríguez, I., van Essen-Zandbergen, A., Dierikx, C., Kadlec, K., Schink, A.-K., Wu, G., Chattaway, M. A., DoNascimento, V., Wain, J., Helmuth, R., Guerra, B., Schwarz, S., Threlfall, J., Woodward, M. J., Coldham, N., Mevius, D., and Woodford, N. Diversity of STs, plasmids and ESBL genes among *Escherichia coli* from humans, animals and food in Germany, the Netherlands and the UK. *The Journal of Antimicrobial Chemotherapy*, 71(5):1178–1182, 5 2016, DOI: https://doi.org/10.1093/jac/dkv485.
- [114] Cormier, A., Zhang, P. L. C., Chalmers, G., Weese, J. S., Deckert, A., Mulvey, M., McAllister, T., and Boerlin, P. Diversity of CTX-M-positive *Escherichia coli* recovered from animals in Canada. *Veterinary Microbiology*, 231:71–75, 2019, DOI: https://doi.org/https://doi.org/ 10.1016/j.vetmic.2019.02.031.
- [115] Afema, J. A., Ahmed, S., Besser, T. E., Jones, L. P., Sischo, W. M., and Davis, M. A. Molecular epidemiology of dairy cattle-associated *Escherichia coli* carrying *bla*(CTX-M) genes in Washington state. *Applied and Environmental Microbiology*, 84(6):02430–17, 3 2018, DOI: https://doi.org/10.1128/AEM.02430-17.
- [116] Ewers, C., Bethe, A., Semmler, T., Guenther, S., and Wieler, L. H. Extended-spectrum βlactamase-producing and AmpC-producing *Escherichia coli* from livestock and companion animals, and their putative impact on public health: a global perspective. *Clinical Microbiology and Infection*, 18(7):646–655, 2012, DOI: https://doi.org/https://doi.org/10.1111/j. 1469-0691.2012.03850.x.
- [117] Ewers, C., de Jong, A., Prenger-Berninghoff, E., El Garch, F., Leidner, U., Tiwari, S. K., and Semmler, T. Genomic diversity and virulence potential of ESBL- and AmpC-β-lactamaseproducing *Escherichia coli* strains from healthy food animals across Europe. *Frontiers in Microbiology*, 12:626774, 2021, DOI: https://doi.org/10.3389/fmicb.2021.626774.
- [118] Kaye, K. and Pogue, J. Infections caused by resistant gram-negative bacteria: Epidemiology and management. *Pharmacotherapy*, 35(10):949–962, 2015, DOI: https://doi.org/10.1002/ phar.1636.

- [119] Laupland, K., Church, D., Vidakovich, J., Mucenski, M., and Pitout, J. Community-onset extended-spectrum beta-lactamase (ESBL) producing *Escherichia coli*: Importance of international travel. *Journal of Infection*, 57(6):441–448, 4 2008, DOI: https://doi.org/10.1016/ j.jinf.2008.09.034.
- [120] Toubiana, J., Timsit, S., Ferroni, A., Grasseau, M., Nassif, X., Lortholary, O., Zahar, J. R., and Chalumeau, M. Community-onset extended-spectrum beta-lactamase-producing *Enter*obacteriaceae invasive infections in children in a university hospital in France. Medicine, 95(12):7, 2016, DOI: https://doi.org/10.1097/md.00000000003163.
- [121] Heffernan, H., Woodhouse, R., Draper, J., and Ren, X. 2016 survey of extended-spectrum beta-lactamase-producing *Enterobacteriaceae*. Report, Institute of Environmental Science and Research Limited, 2018, https://surv.esr.cri.nz/PDF\_surveillance/Antimicrobial/ ESBL/ESBL\_2016.pdf.
- [122] Le, Q. P., Ueda, S., Nguyen, T. N., Dao, T. V., Van Hoang, T. A., Tran, T. T., Hirai, I., Nakayawa, H., Kawahara, R., Do, T. H., Vien, Q. M., and Yamamoto, Y. Characteristics of extended-spectrum beta-lactamase-producing *Escherichia coli* in retail meats and shrimp at a local market in Vietnam. *Foodborne Pathogens and Disease*, 12(8):719–725, 2015, DOI: https://doi.org/10.1089/fpd.2015.1954.
- [123] Zhang, H. N., Zhou, Y. F., Guo, S. Y., and Chang, W. S. Multidrug resistance found in extended-spectrum beta-lactamase-producing *Enterobacteriaceae* from rural water reservoirs in Guantao, China. *Frontiers in Microbiology*, 6:4, 2015, DOI: https://doi.org/10.3389/ fmicb.2015.00267.
- [124] Filioussis, G., Kachrimanidou, M., Christodoulopoulos, G., Kyritsi, M., Hadjichristodoulou, C., Adamopoulou, M., Tzivara, A., Kritas, S. K., and Grinberg, A. Short communication: Bovine mastitis caused by a multidrug-resistant, mcr-1-positive (colistin-resistant), extended-spectrum β-lactamase-producing Escherichia coli clone on a Greek dairy farm. Journal of Dairy Science, 103(1):852–857, 2020, DOI: https://doi.org/https://doi.org/10. 3168/jds.2019-17320.
- [125] World Health Organization. Critically important antimicrobials for human medicine: 6th revision. Report, 2018, https://www.who.int/publications/i/item/9789241515528.
- [126] Zheng, B., Feng, C., Xu, H., Yu, X., Guo, L., Jiang, X., and Song, X. Detection and characterization of ESBL-producing *Escherichia coli* expressing *mcr-1* from dairy cows in China. *Journal of Antimicrobial Chemotherapy*, 74(2):321–325, 2 2019, DOI: https://doi. org/10.1093/jac/dky446.
- [127] Dahmen, S., Metayer, V., Gay, E., Madec, J. Y., and Haenni, M. Characterization of extended-spectrum beta-lactamase (ESBL)-carrying plasmids and clones of *Enterobacteri*aceae causing cattle mastitis in France. Veterinary Microbiology, 162(2-4):793-799, 2013, DOI: https://doi.org/10.1016/j.vetmic.2012.10.015.
- [128] Pehlivanoglu, F., Turutoglu, H., and Ozturk, D. CTX-M-15-type extended-spectrum betalactamase-producing *Escherichia coli* as causative agent of bovine mastitis. *Foodborne Pathogens and Disease*, 13(9):477–482, 2016, DOI: https://doi.org/10.1089/fpd.2015.2114.

- [129] Tsuka, T., Ozaki, H., Saito, D., Murase, T., Okamoto, Y., Azuma, K., Osaki, T., Ito, N., Murahata, Y., and Imagawa, T. Genetic characterization of CTX-M-2-producing *Klebsiella* pneumoniae and *Klebsiella oxytoca* associated with bovine mastitis in Japan. Frontiers in Veterinary Science, 8:412, 2021, DOI: https://doi.org/10.3389/fvets.2021.659222.
- [130] McDougall, S. Bovine mastitis: epidemiology, treatment and control. New Zealand Veterinary Journal, 50(3):81–84, 2002, DOI: https://doi.org/10.1080/00480169.2002.36274.
- [131] Jacoby, G. A. AmpC β-Lactamases. Clinical Microbiology Reviews, 22(1):161–182, 2009, DOI: https://doi.org/10.1128/cmr.00036-08.
- [132] Mammeri, H., Poirel, L., Fortineau, N., and Nordmann, P. Naturally occurring extendedspectrum cephalosporinases in *Escherichia coli*. Antimicrobial Agents and Chemotherapy, 50(7):2573–2576, 7 2006, DOI: https://doi.org/10.1128/AAC.01633-05.
- [133] Walther-Rasmussen, J. and Høiby, N. Plasmid-borne AmpC β-lactamases. Canadian Journal of Microbiology, 48(6):479–493, 8 2002, DOI: https://doi.org/10.1139/w02-039.
- [134] Jones, R. N., Baquero, F., Privitera, G., Inoue, M., and Wiedemann, B. Inducible βlactamase-mediated resistance to third-generation cephalosporins. *Clinical Microbiology and Infection*, 3(s1):s7–s20, 4 1997, DOI: https://doi.org/https://doi.org/10.1111/j.1469-0691. 1997.tb00643.x.
- [135] Honoré, N., Nicolas, M. H., and Cole, S. T. Inducible cephalosporinase production in clinical isolates of *Enterobacter cloacae* is controlled by a regulatory gene that has been deleted from *Escherichia coli*. The EMBO journal, 5(13):3709–3714, 12 1986.
- [136] Jaurin, B., Grundström, T., Edlund, T., and Normark, S. The E. coli β-lactamase attenuator mediates growth rate-dependent regulation. Nature, 290(5803):221–225, 1981, DOI: https: //doi.org/10.1038/290221a0.
- [137] Peter-Getzlaff, S., Polsfuss, S., Poledica, M., Hombach, M., Giger, J., Böttger, E. C., Zbinden, R., and Bloemberg, G. V. Detection of AmpC beta-lactamase in *Escherichia coli*: Comparison of three phenotypic confirmation assays and genetic analysis. *Journal of Clinical Microbiology*, 49(8):2924–2932, 2011, DOI: https://doi.org/10.1128/JCM.00091-11.
- [138] Caroff, N., Espaze, E., Bérard, I., Richet, H., and Reynaud, A. Mutations in the ampC promoter of Escherichia coli isolates resistant to oxyiminocephalosporins without extended spectrum beta-lactamase production. FEMS Microbiology Letters, 173(2):459–465, 1999, DOI: https://doi.org/10.1111/j.1574-6968.1999.tb13539.x.
- [139] Mulvey, M. R., Bryce, E., Boyd, D. A., Ofner-Agostini, M., Land, A. M., Simor, A. E., and Paton, S. Molecular characterization of cefoxitin-resistant *Escherichia coli* from Canadian hospitals. *Antimicrobial Agents and Chemotherapy*, 49(1):358–365, 2005, DOI: https://doi. org/10.1128/aac.49.1.358-365.2005.
- [140] Tracz, D. M., Boyd, D. A., Hizon, R., Bryce, E., McGeer, A., Ofner-Agostini, M., Simor, A. E., Paton, S., and Mulvey, M. R. ampC gene expression in promoter mutants of cefoxitinresistant *Escherichia coli* clinical isolates. *FEMS Microbiology Letters*, 270(2):265–271, 2007, DOI: https://doi.org/10.1111/j.1574-6968.2007.00672.x.

- [141] Nakano, R., Okamoto, R., Nakano, Y., Kaneko, K., Okitsu, N., Hosaka, Y., and Inoue, M. CFE-1, a novel plasmid-encoded AmpC β-lactamase with an ampR gene originating from Citrobacter freundii. Antimicrobial Agents and Chemotherapy, 48(4):1151–1158, 4 2004, DOI: https://doi.org/10.1128/AAC.48.4.1151-1158.2004.
- [142] Chen, C.-M., Huang, M., Wu, H.-J., Guo, M.-K., and Wu, L.-T. Identification of CFE-2, a new plasmid-encoded AmpC β-lactamase from a clinical isolate of *Citrobacter freundii*. *International Journal of Antimicrobial Agents*, 52(3):421–424, 2018, DOI: https://doi.org/ https://doi.org/10.1016/j.ijantimicag.2018.06.013.
- [143] Philippon, A., Arlet, G., and Jacoby, G. A. Plasmid-determined AmpC-type beta-lactamases. Antimicrobial Agents and Chemotherapy, 46(1):1–11, 2002, DOI: https://doi.org/10.1128/ aac.46.1.1-11.2002.
- [144] Barnaud, G., Arlet, G., Verdet, C., Gaillot, O., Lagrange, P. H., and Philippon, A. Salmonella enteritidis: AmpC plasmid-mediated inducible beta-lactamase (DHA-1) with an ampR gene from Morganella morganii. Antimicrobial Agents and Chemotherapy, 42(9):2352– 2358, 9 1998, DOI: https://doi.org/10.1128/AAC.42.9.2352.
- [145] Reisbig, M. D. and Hanson, N. D. The ACT-1 plasmid-encoded AmpC beta-lactamase is inducible: detection in a complex beta-lactamase background. *The Journal of Antimicrobial Chemotherapy*, 49(3):557–560, 3 2002, DOI: https://doi.org/10.1093/jac/49.3.557.
- [146] Nadjar, D., Rouveau, M., Verdet, C., Donay, J.-L., Herrmann, J.-L., Lagrange, P. H., Philippon, A., and Arlet, G. Outbreak of *Klebsiella pneumoniae* producing transferable AmpC-type β-lactamase (ACC-1) originating from *Hafnia alvei*. *FEMS Microbiology Letters*, 187(1):35–40, 6 2000, DOI: https://doi.org/10.1111/j.1574-6968.2000.tb09133.x.
- [147] Bauernfeind, A., Stemplinger, I., Jungwirth, R., and Giamarellou, H. Characterization of the plasmidic beta-lactamase CMY-2, which is responsible for cephamycin resistance. *Antimicrobial Agents and Chemotherapy*, 40(1):221–224, 1 1996, DOI: https://doi.org/10. 1128/AAC.40.1.221.
- [148] Jacoby, G. A. and Tran, J. Sequence of the MIR-1 beta-lactamase gene. Antimicrobial Agents and Chemotherapy, 43(7):1759–1760, 7 1999, DOI: https://doi.org/10.1128/AAC.43.7.1759.
- [149] Rottman, M., Benzerara, Y., Hanau-Berçot, B., Bizet, C., Philippon, A., and Arlet, G. Chromosomal ampC genes in Enterobacter species other than Enterobacter cloacae, and ancestral association of the ACT-1 plasmid-encoded cephalosporinase to Enterobacter asburiae. FEMS Microbiology Letters, 210(1):87–92, 4 2002, DOI: https://doi.org/10.1111/j.1574-6968.2002. tb11164.x.
- [150] Poirel, L., Guibert, M., Girlich, D., Naas, T., and Nordmann, P. Cloning, sequence analyses, expression, and distribution of *ampC-ampR* from *Morganella morganii* clinical isolates. *Antimicrobial Agents and Chemotherapy*, 43(4):769–776, 4 1999, DOI: https: //doi.org/10.1128/AAC.43.4.769.
- [151] Ebmeyer, S., Kristiansson, E., and Larsson, D. G. J. CMY-1/MOX-family AmpC βlactamases MOX-1, MOX-2 and MOX-9 were mobilized independently from three Aeromonas

species. Journal of Antimicrobial Chemotherapy, 74(5):1202–1206, 5 2019, DOI: https://doi.org/10.1093/jac/dkz025.

- [152] Ebmeyer, S., Kristiansson, E., and Larsson, D. G. J. The mobile FOX AmpC beta-lactamases originated in Aeromonas allosaccharophila. International Journal of Antimicrobial Agents, 54(6):798–802, 2019, DOI: https://doi.org/https://doi.org/10.1016/j.ijantimicag.2019.09.017.
- [153] Nordmann, P. and Mammeri, H. Extended-spectrum cephalosporinases: structure, detection and epidemiology. *Future Microbiology*, 2(3):297–307, 5 2007, DOI: https://doi.org/10.2217/ 17460913.2.3.297.
- [154] Santiago, G. S., Coelho, I. S., Bronzato, G. F., Moreira, A. B., Gonçalves, D., Alencar, T. A., Ferreira, H. N., Castro, B. G., Souza, M. M. S., and Coelho, S. M. O. Short communication: Extended-spectrum AmpC-producing *Escherichia coli* from milk and feces in dairy farms in Brazil. *Journal of Dairy Science*, 101(9):7808–7811, 2018, DOI: https://doi.org/https: //doi.org/10.3168/jds.2017-13658.
- [155] World Health Organization. Critically important antimicrobials for human medicine. Report, Geneva, Switzerland, 2011, https://apps.who.int/iris/bitstream/handle/10665/ 312266/9789241515528-eng.pdf.
- [156] Alzayn, M., Findlay, J., Schubert, H., Mounsey, O., Gould, V., Heesom, K., Turner, K., Barrett, D., Reyher, K., and Avison, M. Characterization of AmpC-hyperproducing *Escherichia coli* from humans and dairy farms collected in parallel in the same geographical region. *Journal of Antimicrobial Chemotherapy*, 75(9):2471–2479, 9 2020, DOI: https://doi.org/10.1093/jac/dkaa207.
- [157] Haenni, M., Châtre, P., and MADEC, J.-Y. Emergence of Escherichia coli producing extended-spectrum AmpC β-lactamases (ESAC) in animals. *Frontiers in Microbiology*, 5:53, 2014, DOI: https://doi.org/10.3389/fmicb.2014.00053.
- [158] Barnaud, G., Labia, R., Raskine, L., Sanson-Le Pors, M. J., Philippon, A., and Arlet, G. Extension of resistance to cefepime and cefpirome associated to a six amino acid deletion in the H-10 helix of the cephalosporinase of an *Enterobacter cloacae* clinical isolate. *FEMS Microbiology Letters*, 195(2):185–190, 2 2001, DOI: https://doi.org/10.1111/j.1574-6968.2001. tb10519.x.
- [159] Barnaud, G., Benzerara, Y., Gravisse, J., Raskine, L., Sanson-Le Pors, M. J., Labia, R., and Arlet, G. Selection during cefepime treatment of a new cephalosporinase variant with extended-spectrum resistance to cefepime in an *Enterobacter aerogenes* clinical isolate. *Antimicrobial Agents and Chemotherapy*, 48(3):1040–1042, 3 2004, DOI: https: //doi.org/10.1128/AAC.48.3.1040-1042.2004.
- [160] Mammeri, H., Poirel, L., Bemer, P., Drugeon, H., and Nordmann, P. Resistance to cefepime and cefpirome due to a 4-amino-acid deletion in the chromosome-encoded AmpC β-Lactamase of a Serratia marcescens clinical isolate. Antimicrobial Agents and Chemotherapy, 48(3):716–720, 3 2004, DOI: https://doi.org/10.1128/AAC.48.3.716-720.2004.

- [161] Turnidge, J. and Paterson, D. L. Setting and revising antibacterial susceptibility breakpoints. *Clinical Microbiology Reviews*, 20(3):391–408, 7 2007, DOI: https://doi.org/10.1128/CMR. 00047-06.
- [162] Kahlmeter, G. and European Committee on Antimicrobial Susceptibility Testing Steering Commitee. Redefining susceptibility testing categories S, I and R, 2019, https://www.eucast. org/newsiandr/.
- [163] CLSI. CLSI M100-ED31:2021 Performance Standards for Antimicrobial Susceptibility Testing, 31st Edition. Report, Clinical and Laboratory Standards Institute, 2021, https: //clsi.org/.
- [164] European Committee on Antimicrobial Susceptibility Testing. The European Committee on Antimicrobial Susceptibility Testing. Breakpoint tables for interpretation of MICs and zone diameters, version 10.0, 2020, 2020, http://www.eucast.org/clinical\_breakpoints/.
- [165] Matuschek, E., Brown, D., and Kahlmeter, G. Development of the EUCAST disk diffusion antimicrobial susceptibility testing method and its implementation in routine microbiology laboratories. *Clinical Mcrobiology and Infection*, 20(4):255–66, 4 2014, DOI: https://doi.org/ 10.1111/1469-0691.12373.
- [166] European Committee on Antimicrobial Susceptibility Testing. MIC and zone diameter distributions and ECOFFs, 2021, https://www.eucast.org/mic\_distributions\_and\_ecoffs/.
- [167] Berendonk, T. U., Manaia, C. M., Merlin, C., Fatta-Kassinos, D., Cytryn, E., Walsh, F., Bürgmann, H., Sørum, H., Norström, M., Pons, M.-N., Kreuzinger, N., Huovinen, P., Stefani, S., Schwartz, T., Kisand, V., Baquero, F., and Martinez, J. L. Tackling antibiotic resistance: the environmental framework. *Nature Reviews Microbiology*, 13(5):310–317, 2015, DOI: https://doi.org/10.1038/nrmicro3439.
- [168] Kronvall, G. and Smith, P. Normalized resistance interpretation, the NRI method. APMIS, 124(12):1023–1030, 12 2016, DOI: https://doi.org/https://doi.org/10.1111/apm.12624.
- [169] Dias, D., Torres, R. T., Kronvall, G., Fonseca, C., Mendo, S., and Caetano, T. Assessment of antibiotic resistance of *Escherichia coli* isolates and screening of *Salmonella* spp. in wild ungulates from Portugal. *Research in Microbiology*, 166(7):584–593, 2015, DOI: https://doi. org/https://doi.org/10.1016/j.resmic.2015.03.006.
- [170] European Committee on Antimicrobial Susceptibility Testing. EUCAST guideline for the detection of resistance mechanisms and specific resistances of clinical and/or epidemiological importance. Report, 2017, http://www.eucast.org/resistance mechanisms/.
- [171] Garrec, H., Drieux-Rouzet, L., Golmard, J.-L., Jarlier, V., and Robert, J. Comparison of nine phenotypic methods for detection of extended-spectrum beta-lactamase production by *Enterobacteriaceae. Journal of Clinical Microbiology*, 49(3):1048–1057, 3 2011, DOI: https: //doi.org/10.1128/JCM.02130-10.
- [172] Stürenburg, E. and Mack, D. Extended-spectrum β-lactamases: implications for the clinical microbiology laboratory, therapy, and infection control. *Journal of Infection*, 47(4):273–295, 2003, DOI: https://doi.org/https://doi.org/10.1016/S0163-4453(03)00096-3.

- [173] Dallenne, C., Da Costa, A., Decré, D., Favier, C., and Arlet, G. Development of a set of multiplex PCR assays for the detection of genes encoding important β-lactamases in *Enterobacteriaceae*. Journal of Antimicrobial Chemotherapy, 65(3):490–495, 11 2010, DOI: https://doi.org/10.1093/jac/dkp498.
- [174] Lee, S., Teng, L., DiLorenzo, N., Weppelmann, T. A., and Jeong, K. C. Prevalence and molecular characteristics of extended-spectrum and AmpC β-Lactamase producing *Escherichia coli* in grazing beef cattle. *Frontiers in Microbiology*, 10(3076), 2020, DOI: https://doi.org/10.3389/fmicb.2019.03076.
- [175] Correa-Martínez, C. L., Idelevich, E. A., Sparbier, K., Kostrzewa, M., and Becker, K. Rapid Detection of Extended-Spectrum β-Lactamases (ESBL) and AmpC β-Lactamases in *Enterobacterales*: Development of a Screening Panel Using the MALDI-TOF MS-Based Direct-on-Target Microdroplet Growth Assay. *Frontiers in Microbiology*, 10:13, 2019, DOI: https://doi.org/10.3389/fmicb.2019.00013.
- [176] Bajaj, P., Singh, N. S., and Virdi, J. S. Escherichia coli β-Lactamases: What Really Matters. Frontiers in Microbiology, 7:417, 3 2016, DOI: https://doi.org/10.3389/fmicb.2016.00417.
- [177] Pérez-Pérez, F. and Hanson, N. Detection of plasmid-mediated AmpC β-lactamase genes in clinical isolates by using multiplex PCR. Journal of Clinical Microbiology, 40(6):2153–2162, 2002, DOI: https://doi.org/10.1128/jcm.40.6.2153-2162.2002.
- [178] Dierikx, C., van Essen-Zandbergen, A., Veldman, K., Smith, H., and Mevius, D. Increased detection of extended spectrum beta-lactamase producing *Salmonella enterica* and *Escherichia coli* isolates from poultry. *Vet Microbiology*, 145(3-4):273–278, 2010, DOI: https://doi.org/10.1016/j.vetmic.2010.03.019.
- [179] Martínez-Martínez, L., Hernández-Allés, S., Albertí, S., Tomás, J. M., Benedi, V. J., and Jacoby, G. A. In vivo selection of porin-deficient mutants of *Klebsiella pneumoniae* with increased resistance to cefoxitin and expanded-spectrum-cephalosporins. Antimicrobial Agents and Chemotherapy, 40(2):342–348, 2 1996, DOI: https://doi.org/10.1128/AAC.40.2.342.
- [180] Correa-Martínez, C. L., Idelevich, E. A., Sparbier, K., Kuczius, T., Kostrzewa, M., and Becker, K. Development of a MALDI-TOF MS-based screening panel for accelerated differential detection of carbapenemases in *Enterobacterales* using the direct-on-target microdroplet growth assay. *Scientific Reports*, 10(1):4988, 2020, DOI: https://doi.org/10.1038/ s41598-020-61890-7.
- [181] Waade, J., Seibt, U., Honscha, W., Rachidi, F., Starke, A., Speck, S., and Truyen, U. Multidrug-resistant enterobacteria in newborn dairy calves in Germany. *PLOS One*, 16(3):e0248291, 3 2021, DOI: https://doi.org/10.1371/journal.pone.0248291.
- [182] Hordijk, J., Wagenaar, J. A., Kant, A., van Essen-Zandbergen, A., Dierikx, C., Veldman, K., Wit, B., and Mevius, D. Cross-sectional study on prevalence and molecular characteristics of plasmid mediated ESBL/AmpC-producing *Escherichia coli* isolated from veal calves at slaughter. *PLOS One*, 8(5):7, 2013, DOI: https://doi.org/10.1371/journal.pone.0065681.

- [183] Hordijk, J., Wagenaar, J. A., van de Giessen, A., Dierikx, C., van Essen-Zandbergen, A., Veldman, K., Kant, A., and Mevius, D. Increasing prevalence and diversity of ESBL/AmpCtype beta-lactamase genes in *Escherichia coli* isolated from veal calves from 1997 to 2010. *Journal of Antimicrobial Chemotherapy*, 68(9):1970–1973, 2013, DOI: https://doi.org/10. 1093/jac/dkt132.
- [184] Hutchinson, H., Finney, S., Munoz-Vargas, L., Feicht, S., Masterson, M., and Habing, G. Prevalence and transmission of antimicrobial resistance in a vertically integrated veal calf production system. *Foodborne Pathogens and Disease*, 14(12):711–718, 2017, DOI: https: //doi.org/10.1089/fpd.2017.2310.
- [185] Duggett, N., AbuOun, M., Randall, L., Horton, R., Lemma, F., Rogers, J., Crook, D., Teale, C., and Anjum, M. F. The importance of using whole genome sequencing and extended spectrum beta-lactamase selective media when monitoring antimicrobial resistance. *Scientific Reports*, 10(1):19880, 2020, DOI: https://doi.org/10.1038/s41598-020-76877-7.
- [186] Smith, S., Meade, J., McGill, K., Gibbons, J., Bolton, D., and Whyte, P. Restoring the selectivity of modified charcoal cefoperazone deoxycholate agar for the isolation of *Campy-lobacter* species using tazobactam, a beta-lactamase inhibitor. *International Journal of Food Microbiology*, 210:131–135, 2015, DOI: https://doi.org/10.1016/j.ijfoodmicro.2015.06.014.
- [187] Moran, L., Kelly, C., Cormican, M., McGettrick, S., and Madden, R. H. Restoring the selectivity of Bolton broth during enrichment for *Campylobacter* spp. from raw chicken. *Letters in Applied Microbiology*, 52(6):614–618, 6 2011, DOI: https://doi.org/https://doi. org/10.1111/j.1472-765X.2011.03046.x.
- [188] Prendergast, D. M., O'Doherty, Burgess, C. M., Howe, N., McMahon, F., Murphy, D., Leonard, F., Morris, D., Harrington, C., Carty, A., Moriarty, J., and Gutierrez, M. Critically important antimicrobial resistant *Enterobacteriaceae* in Irish farm effluent and their removal in integrated constructed wetlands. *Science of The Total Environment*, 806:151269, 2022, DOI: https://doi.org/https://doi.org/10.1016/j.scitotenv.2021.151269.
- [189] Terentjeva, M., Streikiša, M., Avsejenko, J., Trofimova, J., Kovalenko, K., Elferts, D., and Bērziņš, A. Prevalence and antimicrobial resistance of *Escherichia coli*, *Enterococcus* spp. and the major foodborne pathogens in calves in Latvia. *Foodborne Pathogens and Disease*, 16(1):35–41, 11 2018, DOI: https://doi.org/10.1089/fpd.2018.2523.
- [190] Hille, K., Ruddat, I., Schmid, A., Hering, J., Hartmann, M., von Münchhausen, C., Schneider, B., Messelhäusser, U., Friese, A., Mansfeld, R., Käsbohrer, A., Hörmansdorfer, S., Roesler, U., and Kreienbrock, L. Cefotaxime-resistant *E. coli* in dairy and beef cattle farms- Joint analyses of two cross-sectional investigations in Germany. *Preventive Veterinary Medicine*, 142:39–45, 7 2017, DOI: https://doi.org/10.1016/j.prevetmed.2017.05.003.
- [191] Kürekci, C., Osek, J., Aydın, M., Tekeli, O., Kurpas, M., Wieczorek, K., and Sakin, F. Evaluation of bulk tank raw milk and raw chicken meat samples as source of ESBL producing *Escherichia coli* in Turkey: Recent insights. *Journal of Food Safety*, 39(2):e12605, 4 2019, DOI: https://doi.org/https://doi.org/10.1111/jfs.12605.

- [192] Santman-Berends, I., Gonggrijp, M., Hage, J., Heuvelink, A., Velthuis, A., Lam, T., and van Schaik, G. Prevalence and risk factors for extended-spectrum beta-lactamase or AmpCproducing *Escherichia coli* in organic dairy herds in the Netherlands. *Journal of Dairy Science*, 100(1):562–571, 2017, DOI: https://doi.org/10.3168/jds.2016-11839.
- [193] Khachatryan, A. R., Hancock, D. D., Besser, T. E., and Call, D. R. Role of calf-adapted Escherichia coli in maintenance of antimicrobial drug resistance in dairy calves. Applied and Environmental Microbiology, 70(2):752–757, 2 2004, DOI: https://doi.org/10.1128/AEM.70. 2.752-757.2004.
- [194] Massé, J., Lardé, H., Fairbrother, J., Roy, J., Francoz, D., Dufour, S., and Archambault, M. Prevalence of antimicrobial resistance and characteristics of *Escherichia coli* isolates from fecal and manure pit samples on dairy farms in the province of Québec, Canada. *Frontiers* in Veterinary Science, 8:438, 2021, DOI: https://doi.org/10.3389/fvets.2021.654125.
- [195] Schubert, H., Morley, K., Puddy, E. F., Arbon, R., Findlay, J., Mounsey, O., Gould, V. C., Vass, L., Evans, M., Rees, G. M., Barrett, D. C., Turner, K. M., Cogan, T. A., Avison, M. B., and Reyher, K. K. Reduced antibacterial drug resistance and *bla*CTX-M β-lactamase gene carriage in cattle-associated *Escherichia coli* at low temperatures, at sites dominated by older animals, and on pastureland: Implications for surveillance. *Applied and Environmental Microbiology*, 87(6):01468–20, 7 2021, DOI: https://doi.org/10.1128/AEM.01468-20.
- [196] Heffernan, H., Wong, T., Lindsay, J., Bowen, B., and Woodhouse, R. A baseline survey of antimicrobial resistance in bacteria from selected New Zealand foods, 2009-2010. Report, Christchurch, 2011, https://www.mpi.govt.nz/dmsdocument/ 21464-A-baseline-survey-of-antimicrobial-resistance-in-bacteria-from-selected-New-Zealand-foods-2009-2010.
- [197] Browne, A. S., Midwinter, A. C., Withers, H., Cookson, A. L., Biggs, P. J., Marshall, J. C., Benschop, J., Hathaway, S., Haack, N. A., Akhter, R. N., and French, N. P. Molecular epidemiology of Shiga toxin-producing *Escherichia coli* (STEC) on New Zealand dairy farms: Application of a culture-independent assay and whole-genome sequencing. *Applied and Environmental Microbiology*, 84(14):16, 2018, DOI: https://doi.org/10.1128/aem.00481-18.
- [198] Watson, E., Jeckel, S., Snow, L., Stubbs, R., Teale, C., Wearing, H., Horton, R., Toszeghy, M., Tearne, O., Ellis-Iversen, J., and Coldham, N. Epidemiology of extended spectrum beta-lactamase *E. coli* (CTX-M-15) on a commercial dairy farm. *Vet Microbiology*, 154(3-4):339–346, 2012, DOI: https://doi.org/10.1016/j.vetmic.2011.07.020.
- [199] Schmid, A., Hormansdorfer, S., Messelhausser, U., Kasbohrer, A., Sauter-Louis, C., and Mansfeld, R. Prevalence of extended-spectrum beta-lactamase-producing *Escherichia coli* on Bavarian dairy and beef cattle farms. *Applied and Environmental Microbiology*, 79(9):3027– 3032, 2013, DOI: https://doi.org/10.1128/aem.00204-13.
- [200] Ohnishi, M., Okatani, A. T., Esaki, H., Harada, K., Sawada, T., Murakami, M., Marumo, K., Kato, Y., Sato, R., Shimura, K., Hatanaka, N., and Takahashi, T. Herd prevalence of *Enterobacteriaceae* producing CTX-M-type and CMY-2-lactamases among Japanese dairy farms. *Journal of Applied Microbiology*, 115(1):282–289, 2013, DOI: https://doi.org/10.1111/ jam.12211.

- [201] Braun, S. D., Ahmed, M. F. E., El-Adawy, H., Hotzel, H., Engelmann, I., Wei, D., Monecke, S., and Ehricht, R. Surveillance of extended-spectrum beta-lactamase-producing *Escherichia coli* in dairy cattle farms in the Nile Delta, Egypt. *Frontiers in Microbiology*, 7:1020, 2016, DOI: https://doi.org/10.3389/fmicb.2016.01020.
- [202] Hartmann, A., Locatelli, A., Amoureux, L., Depret, G., Jolivet, C., Gueneau, E., and Neuwirth, C. Occurrence of CTX-M producing *Escherichia coli* in soils, cattle, and farm environment in France (Burgundy region). *Frontiers in Microbiology*, 3:7, 2012, DOI: https://doi.org/10.3389/fmicb.2012.00083.
- [203] Haenni, M., Chatre, P., Metayer, V., Bour, M., Signol, E., Madec, J. Y., and Gay, E. Comparative prevalence and characterization of ESBL-producing *Enterobacteriaceae* in dominant versus subdominant enteric flora in veal calves at slaughterhouse, France. *Veterinary Microbiology*, 171(3-4):321–327, 2014, DOI: https://doi.org/10.1016/j.vetmic.2014.02.023.
- [204] Hordijk, J., Mevius, D. J., Kant, A., Bos, M. E. H., Graveland, H., Bosman, A. B., Hartskeerl, C. M., Heederik, D. J. J., and Wagenaar, J. A. Within-farm dynamics of ESBL/AmpCproducing *Escherichia coli* in veal calves: A longitudinal approach. *Journal of Antimicrobial Chemotherapy*, 68(11):2468–2476, 2013, DOI: https://doi.org/10.1093/jac/dkt219.
- [205] Awosile, B., McClure, J., Sanchez, J., Rodriguez-Lecompte, J. C., Keefe, G., and Heider, L. C. Salmonella enterica and extended-spectrum cephalosporin-resistant Escherichia coli recovered from Holstein dairy calves from 8 farms in New Brunswick, Canada. Journal of Dairy Science, 101(4):3271–3284, 2018, DOI: https://doi.org/https://doi.org/10.3168/jds. 2017-13277.
- [206] Day, M. J., Hopkins, K. L., Wareham, D. W., Toleman, M. A., Elviss, N., Randall, L., Teale, C., Cleary, P., Wiuff, C., Doumith, M., Ellington, M. J., Woodford, N., and Livermore, D. M. Extended-spectrum beta-lactamase-producing *Escherichia coli* in human-derived and foodchain-derived samples from England, Wales, and Scotland: an epidemiological surveillance and typing study. *The Lancet Infectious Diseases*, 19(12):1325–1335, 12 2019, DOI: https://doi.org/10.1016/S1473-3099(19)30273-7.
- [207] Geser, N., Stephan, R., and Hachler, H. Occurrence and characteristics of extended-spectrum beta-lactamase (ESBL) producing *Enterobacteriaceae* in food producing animals, minced meat and raw milk. *BMC Veterinary Research*, 8:9, 2012, DOI: https://doi.org/10.1186/ 1746-6148-8-21.
- [208] Kamaruzzaman, E. A., Aziz, S. A., Bitrus, A. A., Zakaria, Z., and Hassan, L. Occurrence and characteristics of extended-spectrum beta-lactamase-producing *Escherichia coli* from dairy cattle, milk, and farm environments in peninsular Malaysia. *Pathogens*, 9(12), 12 2020, DOI: https://doi.org/10.3390/pathogens9121007.
- [209] Odenthal, S., Akineden, O., and Usleber, E. Extended-spectrum beta-lactamase producing Enterobacteriaceae in bulk tank milk from German dairy farms. International Journal of Food Microbiology, 238:72–78, 2016, DOI: https://doi.org/10.1016/j.ijfoodmicro.2016.08.036.
- [210] Liu, J., Zhu, Y., Jay-Russell, M., Lemay, D. G., and Mills, D. A. Reservoirs of antimicrobial resistance genes in retail raw milk. *Microbiome*, 8(1):99, 2020, DOI: https://doi.org/10.1186/ s40168-020-00861-6.

- [211] Liu, H., Meng, L., Dong, L., Zhang, Y., Wang, J., and Zheng, N. Prevalence, antimicrobial susceptibility, and molecular characterization of *Escherichia coli* isolated from raw milk in dairy herds in Northern China. *Frontiers in Microbiology*, 12:2720, 2021, DOI: https: //doi.org/10.3389/fmicb.2021.730656.
- [212] Skockova, A., Bogdanovicova, K., Kolackova, I., and Karpiskova, R. Antimicrobial-resistant and extended-spectrum beta-lactamase-producing *Escherichia coli* in raw cow's milk. *Journal of Food Protection*, 78(1):72–77, 2015, DOI: https://doi.org/10.4315/0362-028x. jfp-14-250.
- [213] Sudarwanto, M., Akineden, O., Odenthal, S., Gross, M., and Usleber, W. Extended spectrum beta-lactamase (ESBL)-producing *Klebsiella pneumoniae* in bulk tank milk from dairy farms in Indonesia. *Foodborne Pathogens and Disease*, 12(7), 2015, DOI: https://doi.org/10.1089/ fpd.2014.18950.
- [214] Crofts, T. S., Gasparrini, A. J., and Dantas, G. Next-generation approaches to understand and combat the antibiotic resistome. *Nature Reviews Microbiology*, 15(7):422–434, 2017, DOI: https://doi.org/10.1038/nrmicro.2017.28.
- [215] Jia, B., Raphenya, A. R., Alcock, B., Waglechner, N., Guo, P., Tsang, K. K., Lago, B. A., Dave, B. M., Pereira, S., Sharma, A. N., Doshi, S., Courtot, M., Lo, R., Williams, L. E., Frye, J. G., Elsayegh, T., Sardar, D., Westman, E. L., Pawlowski, A. C., Johnson, T. A., Brinkman, F. S., Wright, G. D., and McArthur, A. G. CARD 2017: Expansion and modelcentric curation of the comprehensive antibiotic resistance database. *Nucleic Acids Research*, 45(D1):D566–d573, 2017, DOI: https://doi.org/10.1093/nar/gkw1004.
- [216] Zankari, E., Hasman, H., Cosentino, S., Vestergaard, M., Rasmussen, S., Lund, O., Aarestrup, F. M., and Larsen, M. V. Identification of acquired antimicrobial resistance genes. *Journal of Antimicrobial Chemotherapy*, 67(11):2640–2644, 2012, DOI: https://doi.org/10. 1093/jac/dks261.
- [217] Zankari, E., Allesøe, R., Joensen, K. G., Cavaco, L. M., Lund, O., and Aarestrup, F. M. PointFinder: a novel web tool for WGS-based detection of antimicrobial resistance associated with chromosomal point mutations in bacterial pathogens. *Journal of Antimicrobial Chemotherapy*, 72(10):2764–2768, 10 2017, DOI: https://doi.org/10.1093/jac/dkx217.
- [218] Pitta, D. W., Dou, Z. X., Kumar, S., Indugu, N., Toth, J. D., Vecchiarelli, B., and Bhukya, B. Metagenomic evidence of the prevalence and distribution patterns of antimicrobial resistance genes in dairy agroecosystems. *Foodborne Pathogens and Disease*, 13(6):296–302, 2016, DOI: https://doi.org/10.1089/fpd.2015.2092.
- [219] Thomas, M., Webb, M., Ghimire, S., Blair, A., Olson, K., Fenske, G. J., Fonder, A. T., Christopher-Hennings, J., Brake, D., and Scaria, J. Metagenomic characterization of the effect of feed additives on the gut microbiome and antibiotic resistome of feedlot cattle. *Scientific Reports*, 7(1):12257, 2017, DOI: https://doi.org/10.1038/s41598-017-12481-6.
- [220] Walsh, T. R. A one-health approach to antimicrobial resistance. Nature Microbiology, 3(8):854–855, 2018, DOI: https://doi.org/10.1038/s41564-018-0208-5.

- [221] Knight, R., Vrbanac, A., Taylor, B. C., Aksenov, A., Callewaert, C., Debelius, J., Gonzalez, A., Kosciolek, T., McCall, L. I., McDonald, D., Melnik, A. V., Morton, J. T., Navas, J., Quinn, R. A., Sanders, J. G., Swafford, A. D., Thompson, L. R., Tripathi, A., Xu, Z. J. Z., Zaneveld, J. R., Zhu, Q. Y., Caporaso, J. G., and Dorrestein, P. C. Best practices for analysing microbiomes. *Nature Reviews Microbiology*, 16(7):410–422, 2018, DOI: https: //doi.org/10.1038/s41579-018-0029-9.
- [222] Quince, C., Walker, A. W., Simpson, J. T., Loman, N. J., and Segata, N. Shotgun metagenomics, from sampling to analysis. *Nature Biotechnology*, 35:833, 2017, DOI: https://doi.org/10.1038/nbt.3935.
- [223] Cardona, S., Eck, A., Cassellas, M., Gallart, M., Alastrue, C., Dore, J., Azpiroz, F., Roca, J., Guarner, F., and Manichanh, C. Storage conditions of intestinal microbiota matter in metagenomic analysis. *BMC Microbiology*, 12(1):158, 2012, DOI: https://doi.org/10.1186/ 1471-2180-12-158.
- [224] Lauber, C. L., Zhou, N., Gordon, J. I., Knight, R., and Fierer, N. Effect of storage conditions on the assessment of bacterial community structure in soil and human-associated samples. *FEMS Microbiology Letters*, 307(1):80–86, 2010, DOI: https://doi.org/10.1111/j.1574-6968. 2010.01965.x.
- [225] Rubin, B. E. R., Gibbons, S. M., Kennedy, S., Hampton-Marcell, J., Owens, S., and Gilbert, J. A. Investigating the impact of storage conditions on microbial community composition in soil samples. *PLOS One*, 8(7):e70460–e70460, 2013, DOI: https://doi.org/10.1371/journal. pone.0070460.
- [226] Salter, S. J., Cox, M. J., Turek, E. M., Calus, S. T., Cookson, W. O., Moffatt, M. F., Turner, P., Parkhill, J., Loman, N. J., and Walker, A. W. Reagent and laboratory contamination can critically impact sequence-based microbiome analyses. *BMC Biology*, 12:12, 2014, DOI: https://doi.org/10.1186/s12915-014-0087-z.
- [227] Costea, P. I., Zeller, G., Sunagawa, S., et al. Towards standards for human fecal sample processing in metagenomic studies. *Nature Biotechnology*, 35:1069, 2017, DOI: https://doi. org/10.1038/nbt.3960.
- [228] Hart, M. L., Meyer, A., Johnson, P. J., and Ericsson, A. C. Comparative evaluation of DNA extraction methods from feces of multiple host species for downstream next-generation sequencing. *Plos One*, 10(11):16, 2015, DOI: https://doi.org/10.1371/journal.pone.0143334.
- [229] Desneux, J. and Pourcher, A. M. Comparison of DNA extraction kits and modification of DNA elution procedure for the quantitation of subdominant bacteria from piggery effluents with real-time PCR. *Microbiologyopen*, 3(4):437–445, 2014, DOI: https://doi.org/10.1002/ mbo3.178.
- [230] Guo, F. and Zhang, T. Biases during DNA extraction of activated sludge samples revealed by high throughput sequencing. *Applied Microbiology and Biotechnology*, 97(10):4607–4616, 2013, DOI: https://doi.org/10.1007/s00253-012-4244-4.
- [231] Knudsen, B. E., Bergmark, L., Munk, P., Lukjancenko, O., Prieme, A., Aarestrup, F. M., and Pamp, S. J. Impact of sample type and DNA isolation procedure on genomic inference

of microbiome composition. mSystems, 1(5), 2016, DOI: https://doi.org/10.1128/mSystems. 00095-16.

- [232] Quigley, L., O'Sullivan, O., Beresford, T. P., Ross, R. P., Fitzgerald, G. F., and Cotter, P. D. A comparison of methods used to extract bacterial DNA from raw milk and raw milk cheese. *Journal of Applied Microbiology*, 113(1):96–105, 2012, DOI: https://doi.org/10.1111/ j.1365-2672.2012.05294.x.
- [233] Sagar, K., Singh, S. P., Goutam, K. K., and Konwar, B. K. Assessment of five soil DNA extraction methods and a rapid laboratory-developed method for quality soil DNA extraction for 16S rDNA-based amplification and library construction. *Journal of Microbiological Methods*, 97:68–73, 2014, DOI: https://doi.org/10.1016/j.mimet.2013.11.008.
- [234] Dahlberg, J., Sun, L., Persson Waller, K., Östensson, K., McGuire, M., Agenäs, S., and Dicksved, J. Microbiota data from low biomass milk samples is markedly affected by laboratory and reagent contamination. *PLOS One*, 14(6):e0218257–e0218257, 2019, DOI: https://doi.org/10.1371/journal.pone.0218257.
- [235] Glassing, A., Dowd, S. E., Galandiuk, S., Davis, B., and Chiodini, R. J. Inherent bacterial DNA contamination of extraction and sequencing reagents may affect interpretation of microbiota in low bacterial biomass samples. *Gut Pathogens*, 8(1):24, 2016, DOI: https://doi.org/10.1186/s13099-016-0103-7.
- [236] Zaheer, R., Noyes, N., Polo, R. O., Cook, S. R., Marinier, E., Van Domselaar, G., Belk, K. E., Morley, P. S., and McAllister, T. A. Impact of sequencing depth on the characterization of the microbiome and resistome. *Scientific Reports*, 8:11, 2018, DOI: https://doi.org/10.1038/ s41598-018-24280-8.
- [237] Huttenhower, C., Gevers, D., Knight, R., et al. Structure, function and diversity of the healthy human microbiome. *Nature*, 486(7402):207–214, 2012, DOI: https://doi.org/10.1038/ nature11234.
- [238] Hillmann, B., Al-Ghalith, G. A., Shields-Cutler, R. R., Zhu, Q. Y., Gohl, D. M., Beckman, K. B., Knight, R., and Knights, D. Evaluating the information content of shallow shotgun metagenomics. *mSystems*, 3(6):12, 2018, DOI: https://doi.org/10.1128/mSystems.00069-18.
- [239] Gweon, H. S., Shaw, L. P., Swann, J., et al. The impact of sequencing depth on the inferred taxonomic composition and AMR gene content of metagenomic samples. *Environmental Microbiome*, 14(1):7, 2019, DOI: https://doi.org/10.1186/s40793-019-0347-1.
- [240] Marotz, C. A., Sanders, J. G., Zuniga, C., Zaramela, L. S., Knight, R., and Zengler, K. Improving saliva shotgun metagenomics by chemical host DNA depletion. *Microbiome*, 6(1):42, 2018, DOI: https://doi.org/10.1186/s40168-018-0426-3.
- [241] Li, Y. F., Chen, P. H., and Yu, Z. T. Spatial and temporal variations of microbial community in a mixed plug-flow loop reactor fed with dairy manure. *Microbial Biotechnology*, 7(4):332– 346, 2014, DOI: https://doi.org/10.1111/1751-7915.12125.
- [242] Rubiola, S., Chiesa, F., Dalmasso, A., Di Ciccio, P., and Civera, T. Detection of antimicrobial resistance genes in the milk production environment: Impact of host DNA and sequencing

depth. *Frontiers in Microbiology*, 11:1983, 2020, DOI: https://doi.org/10.3389/fmicb.2020. 01983.

- [243] Weinroth, M. D., Scott, H. M., Norby, B., Loneragan, G. H., Noyes, N. R., Rovira, P., Doster, E., Yang, X., Woerner, D. R., Morley, P. S., and Belk, K. E. Effects of ceftiofur and chlortetracycline on the resistomes of feedlot cattle. *Applied and Environmental Microbiology*, 84(13):00610–18, 2018, DOI: https://doi.org/10.1128/aem.00610-18.
- [244] Noyes, N. R., Yang, X., Linke, L. M., Magnuson, R. J., Cook, S. R., Zaheer, R., Yang, H., Woerner, D. R., Geornaras, I., McArt, J. A., Gow, S. P., Ruiz, J., Jones, K. L., Boucher, C. A., McAllister, T. A., Belk, K. E., and Morley, P. S. Characterization of the resistome in manure, soil and wastewater from dairy and beef production systems. *Scientific Reports*, 6:12, 2016, DOI: https://doi.org/10.1038/srep24645.
- [245] Carini, P., Marsden, P. J., Leff, J. W., Morgan, E. E., Strickland, M. S., and Fierer, N. Relic DNA is abundant in soil and obscures estimates of soil microbial diversity. *Nature Microbiology*, 2:16242, 2016, DOI: https://doi.org/10.1038/nmicrobiol.2016.242.
- [246] Zhang, A.-N., Gaston, J. M., Dai, C. L., Zhao, S., Poyet, M., Groussin, M., Yin, X., Li, L.-G., van Loosdrecht, M. C. M., Topp, E., Gillings, M. R., Hanage, W. P., Tiedje, J. M., Moniz, K., Alm, E. J., and Zhang, T. An omics-based framework for assessing the health risk of antimicrobial resistance genes. *Nature Communications*, 12(1):4765, 2021, DOI: https: //doi.org/10.1038/s41467-021-25096-3.
- [247] Ashbolt, N. J., Amézquita, A., Backhaus, T., Borriello, P., Brandt, K. K., Collignon, P., Coors, A., Finley, R., Gaze, W. H., Heberer, T., Lawrence, J. R., Larsson, D. G. J., McEwen, S. A., Ryan, J. J., Schönfeld, J., Silley, P., Snape, J. R., Van den Eede, C., and Topp, E. Human Health Risk Assessment (HHRA) for environmental development and transfer of antibiotic resistance. *Environmental Health Perspectives*, 121(9):993–1001, 9 2013, DOI: https://doi.org/10.1289/ehp.1206316.
- [248] Lhermie, G., Grohn, Y. T., and Raboisson, D. Addressing antimicrobial resistance: An overview of priority actions to prevent suboptimal antimicrobial use in food-animal production. Frontiers in Microbiology, 7:11, 2017, DOI: https://doi.org/10.3389/fmicb.2016.02114.
- [249] McDougall, S., Gohary, K., Bates, A., and Compton, C. Antimicrobial usage and resistance. In Proceedings of the Society if Dairy Cattle Veterinarians of the NZVA Annual Conference, 2016.
- [250] Schar, D., Sommanustweechai, A., Laxminarayan, R., and Tangcharoensathien, V. Surveillance of antimicrobial consumption in animal production sectors of low- and middle-income countries: Optimizing use and addressing antimicrobial resistance. *PLoS Medicine*, 15(3), 2018, DOI: https://doi.org/10.1371/journal.pmed.1002521.
- [251] Barkema, H. W., von Keyserlingk, M. A. G., Kastelic, J. P., Lam, T., Luby, C., Roy, J. P., LeBlanc, S. J., Keefe, G. P., and Kelton, D. F. Invited review: Changes in the dairy industry affecting dairy cattle health and welfare. *Journal of Dairy Science*, 98(11):7426–7445, 2015, DOI: https://doi.org/10.3168/jds.2015-9377.

- [252] Maron, D. F., Smith, T. J. S., and Nachman, K. E. Restrictions on antimicrobial use in food animal production: An international regulatory and economic survey. *Globalization* and Health, 9:11, 2013, DOI: https://doi.org/10.1186/1744-8603-9-48.
- [253] Van Boeckel, T. P., Brower, C., Gilbert, M., Grenfell, B. T., Levin, S. A., Robinson, T. P., Teillant, A., and Laxminarayan, R. Global trends in antimicrobial use in food animals. *Pro*ceedings of the National Academy of Sciences of the United States of America, 112(18):5649– 5654, 2015, DOI: https://doi.org/10.1073/pnas.1503141112.
- [254] Sorge, U. S., Moon, R., Wolff, L. J., Michels, L., Schroth, S., Kelton, D. F., and Heins, B. Management practices on organic and conventional dairy herds in Minnesota. *Journal of Dairy Science*, 99(4):3183–3192, 2016, DOI: https://doi.org/10.3168/jds.2015-10193.
- [255] AsureQuality. AsureQuality organic standard. Report, Auckland, New Zealand, 2018, https: //www.asurequality.com/assets/Organic-Files/AQ-Organics-Standard-2018-v7.pdf.
- [256] BioGro NZ. BioGro organic standards, 2009, https://static1.squarespace.com/static/ 5783012e1b631b1a87b5f0de/t/583cb0c459cc68a8c3c94bda/1480372422096/Module\_5\_ Livestock.pdf.
- [257] Kuipers, A., Koops, W. J., and Wemmenhove, H. Antibiotic use in dairy herds in the Netherlands from 2005 to 2012. *Journal of Dairy Science*, 99(2):1632–1648, 2016, DOI: https://doi.org/10.3168/jds.2014-8428.
- [258] Lacy-Hulbert, J., Blackwell, M., and McDougall, S. SmartSAMM The smart approach to minimising mastitis. In Proceedings of the Society of Dairy Cattle Veterinarians of the NZVA Annual Conference, (VetLearn Foundation), 2011.
- [259] Woolford, M. and Lacy-Hulbert, S. J. Mastitis research in New Zealand. In Volume Proceedings of the 2nd Pan Pacific Veterinary Conference: Cattle Sessions - incorporating the 13th Annual Seminar of the Society of Dairy Cattle Veterinarians of the New Zealand Veterinary Association, (VetLearn Foundation), 1996.
- [260] Bryan, M. and Hea, S. Y. A survey of antimicrobial use in dairy cows from farms in four regions of New Zealand. New Zealand Veterinary Journal, 65(2):93–98, 2017, DOI: https: //doi.org/10.1080/00480169.2016.1256794.
- [261] Hyde, R. M., Remnant, J. G., Bradley, A. J., Breen, J. E., Hudson, C. D., Davies, P. L., Clarke, T., Critchell, Y., Hylands, M., Linton, E., Wood, E., and Green, M. J. Quantitative analysis of antimicrobial use on British dairy farms. *The Veterinary Record*, 181(25):683, 12 2017, DOI: https://doi.org/10.1136/vr.104614.
- [262] Merle, R. and Meyer-Kühling, B. Sales data as a measure of antibiotics usage: Concepts, examples and discussion of influencing factors. *Veterinary Medicine and Science*, 6(1):154– 163, 2 2020, DOI: https://doi.org/10.1002/vms3.205.
- [263] Nobrega, D. B., Tang, K. L., Caffrey, N. P., De Buck, J., Cork, S. C., Ronksley, P. E., Polachek, A. J., Ganshorn, H., Sharma, N., Kastelic, J. P., Kellner, J. D., Ghali, W. A., and Barkema, H. W. Prevalence of antimicrobial resistance genes and its association with restricted antimicrobial use in food-producing animals: a systematic review

and meta-analysis. Journal of Antimicrobial Chemotherapy, 76(3):561–575, 3 2021, DOI: https://doi.org/10.1093/jac/dkaa443.

- [264] McDougall, S., Compton, C. W. R., and Botha, N. Factors influencing antimicrobial prescribing by veterinarians and usage by dairy farmers in New Zealand. New Zealand Veterinary Journal, 65(2):84–92, 2017, DOI: https://doi.org/10.1080/00480169.2016.1246214.
- [265] Fischer, K., Sjöström, K., Stiernström, A., and Emanuelson, U. Dairy farmers' perspectives on antibiotic use: A qualitative study. *Journal of Dairy Science*, 102(3):2724–2737, 2019, DOI: https://doi.org/https://doi.org/10.3168/jds.2018-15015.
- [266] Rabiee, A. R. and Lean, I. J. The effect of internal teat sealant products (Teatseal and Orbeseal) on intramammary infection, clinical mastitis, and somatic cell counts in lactating dairy cows: A meta-analysis. *Journal of Dairy Science*, 96(11):6915–6931, 6 2013, DOI: https://doi.org/10.3168/jds.2013-6544.
- [267] Singer, A. C., Shaw, H., Rhodes, V., and Hart, A. Review of antimicrobial resistance in the environment and its relevance to environmental regulators. *Frontiers in Microbiology*, 7, 2016, DOI: https://doi.org/10.3389/fmicb.2016.01728.
- [268] Riekerink, R., Barkema, H. W., Kelton, D. F., and Scholl, D. T. Incidence rate of clinical mastitis on Canadian dairy farms. *Journal of Dairy Science*, 91(4):1366–1377, 2008, DOI: https://doi.org/10.3168/jds.2007-0757.
- [269] Lacy-Hulbert, S. J., Kolver, E. S., Williamson, J. H., and Napper, A. R. Incidence of mastitis among cows of different genotypes in differing nutritional environments. In *Proceedings of New Zealand Society of Animal Production*, volume 62, pages 24–29, Palmerston North, 2002. New Zealand Society of Animal Production.
- [270] Saini, V., McClure, J., Léger, D., Keefe, G., Scholl, D., Morck, D., and Barkema, H. Antimicrobial resistance profiles of common mastitis pathogens on Canadian dairy farms. *Journal* of Dairy Science, 95(8):4319–4332, 2012, DOI: https://doi.org/https://doi.org/10.3168/jds. 2012-5373.
- [271] Kalaugher, E., Beukes, P., Bornman, J. F., Clark, A., and Campbell, D. I. Modelling farmlevel adaptation of temperate, pasture-based dairy farms to climate change. Agricultural Systems, 153:53–68, 2017, DOI: https://doi.org/https://doi.org/10.1016/j.agsy.2017.01.008.
- [272] Hellberg, R. S. and Chu, E. Effects of climate change on the persistence and dispersal of foodborne bacterial pathogens in the outdoor environment: A review. *Critical Reviews in Microbiology*, 42(4):548–572, 7 2016, DOI: https://doi.org/10.3109/1040841X.2014.972335.
- [273] MacFadden, D. R., McGough, S. F., Fisman, D., Santillana, M., and Brownstein, J. S. Antibiotic resistance increases with local temperature. *Nature Climate Change*, 8(6):510– 514, 2018, DOI: https://doi.org/10.1038/s41558-018-0161-6.
- [274] Knapp, J. R., Laur, G. L., Vadas, P. A., Weiss, W. P., and Tricarico, J. M. Invited review: Enteric methane in dairy cattle production: Quantifying the opportunities and impact of reducing emissions. *Journal of Dairy Science*, 97(6):3231–3261, 2014, DOI: https://doi.org/ https://doi.org/10.3168/jds.2013-7234.

- [275] Markland, S., Weppelmann, T. A., Ma, Z., Lee, S., Mir, R. A., Teng, L., Ginn, A., Lee, C., Ukhanova, M., Galindo, S., Carr, C., DiLorenzo, N., Ahn, S., Mah, J.-H., Kim, H.-Y., Mai, V., Mobley, R., Morris, J. G., and Jeong, K. C. High prevalence of cefotaxime resistant bacteria in grazing beef cattle: a cross sectional study. *Frontiers in Microbiology*, 10:176, 2019, DOI: https://doi.org/10.3389/fmicb.2019.00176.
- [276] Saini, V., McClure, J. T., Scholl, D. T., DeVries, T. J., and Barkema, H. W. Herd-level relationship between antimicrobial use and presence or absence of antimicrobial resistance in gram-negative bovine mastitis pathogens on Canadian dairy farms. *Journal of Dairy Science*, 96(8):4965–4976, 2013, DOI: https://doi.org/10.3168/jds.2012.5713.
- [277] Stohr, J. J. M., Kluytmans-van den Bergh, M. F. Q., Verhulst, C. J. M. M., Rossen, J. W. A., and Kluytmans, J. A. J. W. Development of amoxicillin resistance in *Escherichia coli* after exposure to remnants of a non-related phagemid-containing *E. coli*: an exploratory study. *Antimicrobial Resistance & Infection Control*, 9(1):48, 2020, DOI: https://doi.org/10.1186/s13756-020-00708-7.
- [278] Williamson, J. H. and Lacy-Hulbert, S. J. Effect of disinfecting teats post-milking or pre- and post-milking on intramammary infection and somatic cell count. New Zealand Veterinary Journal, 61(5):262–268, 2013, DOI: https://doi.org/10.1080/00480169.2012.751576.
- [279] Ji, X., Shen, Q., Liu, F., Ma, J., Xu, G., Wang, Y., and Wu, M. Antibiotic resistance gene abundances associated with antibiotics and heavy metals in animal manures and agricultural soils adjacent to feedlots in Shanghai; China. *Journal of Hazardous Materials*, 235:178–185, 2012, DOI: https://doi.org/10.1016/j.jhazmat.2012.07.040.
- [280] Knapp, C., McCluskey, S., Singh, B., Campbell, C., Hudson, G., and Graham, D. Antibiotic resistance gene abundances correlate with metal and geochemical conditions in archived Scottish soils. *PLOS One*, 6(11):6, 2011, DOI: https://doi.org/10.1371/journal.pone.0027300.
- [281] Romero, J., Burgos, M., Perez-Pulido, R., Galvez, A., and Lucas, R. Resistance to antibiotics, biocides, preservatives and metals in bacteria isolated from seafoods: Co-selection of strains resistant or tolerant to different classes of compounds. *Frontiers in Microbiology*, 8:16, 2017, DOI: https://doi.org/10.3389/fmicb.2017.01650.
- [282] Wales, A. D. and Davies, R. H. Co-Selection of resistance to antibiotics, biocides and heavy metals, and its relevance to foodborne pathogens. *Antibiotics-Basel*, 4(4):567–604, 2015, DOI: https://doi.org/10.3390/antibiotics4040567.
- [283] Snow, L. C., Warner, R. G., Cheney, T., Wearing, H., Stokes, M., Harris, K., Teale, C. J., and Coldham, N. G. Risk factors associated with extended spectrum beta-lactamase *Escherichia* coli (CTX-M) on dairy farms in North West England and North Wales. *Preventive Veterinary Medicine*, 106(3-4):225–234, 2012, DOI: https://doi.org/10.1016/j.prevetmed.2012.03.009.
- [284] Velasova, M., Smith, R. P., Lemma, F., Horton, R. A., Duggett, N. A., Evans, J., Tongue, S. C., Anjum, M. F., and Randall, L. P. Detection of extended-spectrum beta-lactam, AmpC and carbapenem resistance in *Enterobacteriaceae* in beef cattle in Great Britain in 2015. *Journal of Applied Microbiology*, 126(4):1081–1095, 2019, DOI: https://doi.org/10. 1111/jam.14211.
- [285] Maynou, G., Migura-Garcia, L., Chester-Jones, H., Ziegler, D., Bach, A., and Terre, M. Effects of feeding pasteurized waste milk to dairy calves on phenotypes and genotypes of antimicrobial resistance in fecal *Escherichia coli* isolates before and after weaning. *Journal* of Dairy Science, 100(10):7967–7979, 2017, DOI: https://doi.org/10.3168/jds.2017-13040.
- [286] Auffret, M. D., Dewhurst, R. J., Duthie, C. A., Rooke, J. A., Wallace, R. J., Freeman, T. C., Stewart, R., Watson, M., and Roehe, R. The rumen microbiome as a reservoir of antimicrobial resistance and pathogenicity genes is directly affected by diet in beef cattle. *Microbiome*, 5:11, 2017, DOI: https://doi.org/10.1186/s40168-017-0378-z.
- [287] One Health High Level Expert Panel. Tripartite and UNEP support OHH-LEP's definition of "One Health", 2021, https://www.who.int/news/item/ 01-12-2021-tripartite-and-unep-support-ohhlep-s-definition-of-one-health?
- [288] Ludden, C., Raven, K. E., Jamrozy, D., Gouliouris, T., Blane, B., Coll, F., de Goffau, M., Naydenova, P., Horner, C., Hernandez-Garcia, J., Wood, P., Hadjirin, N., Radakovic, M., Brown, N. M., Holmes, M., Parkhill, J., and Peacock, S. J. One Health genomic surveillance of *Escherichia coli* demonstrates distinct lineages and mobile genetic elements in isolates from humans versus livestock. *mBio*, 10(1), 1 2019, DOI: https://doi.org/10.1128/mBio.02693-18.
- [289] Duarte, A., Röder, T., Van Gompel, L., Petersen, T., Hansen, R., Hansen, I., Bossers, A., Aarestrup, F., Wagenaar, J., and Hald, T. Metagenomics-based approach to sourceattribution of antimicrobial resistance determinants – identification of reservoir resistome signatures. *Frontiers in Microbiology*, 11:3447, 2021, DOI: https://doi.org/10.3389/fmicb. 2020.601407.
- [290] Boerlin, P. and Reid-Smith, R. J. Antimicrobial resistance: its emergence and transmission. Animal Health Research Reviews, 9(2):115–126, 2008, DOI: https://doi.org/10.1017/ S146625230800159X.
- [291] Hitch, T. C. A., Thomas, B. J., Friedersdorff, J. C. A., Ougham, H., and Creevey, C. J. Deep sequence analysis reveals the ovine rumen as a reservoir of antibiotic resistance genes. *Environmental Pollution*, 235:571–575, 2018, DOI: https://doi.org/10.1016/j.envpol.2017.12. 067.
- [292] Chambers, L., Yang, Y., Littier, H., Ray, P., Zhang, T., Pruden, A., Strickland, M., and Knowlton, K. Metagenomic analysis of antibiotic resistance genes in dairy cow feces following therapeutic administration of third generation cephalosporin. *PLOS One*, 10(8):18, 2015, DOI: https://doi.org/10.1371/journal.pone.0133764.
- [293] Wichmann, F., Udikovic-Kolic, N., Andrew, S., and Handelsman, J. Diverse antibiotic resistance genes in dairy cow manure. *mBio*, 5(2):9, 2014, DOI: https://doi.org/10.1128/ mBio.01017-13.
- [294] McKinney, C. W., Dungan, R. S., Moore, A., and Leytem, A. B. Occurrence and abundance of antibiotic resistance genes in agricultural soil receiving dairy manure. *Fems Microbiology Ecology*, 94(3):10, 2018, DOI: https://doi.org/10.1093/femsec/fiy010.
- [295] Lazarus, B., Paterson, D. L., Mollinger, J. L., and Rogers, B. A. Do human extraintestinal Escherichia coli infections resistant to expanded-spectrum cephalosporins originate from

food-producing animals? A systematic review. *Clinical Infectious Diseases*, 60(3):439–452, 2015, DOI: https://doi.org/10.1093/cid/ciu785.

- [296] Sparham, S. J., Kwong, J. C., Valcanis, M., Easton, M., Trott, D. J., Seemann, T., Stinear, T. P., and Howden, B. P. Emergence of multidrug resistance in locally-acquired human infections with *Salmonella* Typhimurium in Australia owing to a new clade harbouring *bla*(CTX-M-9). *International Journal of Antimicrobial Agents*, 50(1):101–105, 6 2017, DOI: https://doi.org/10.1016/j.ijantimicag.2017.02.014.
- [297] Börjesson, S., Ny, S., Egervärn, M., Bergström, J., Rosengren, , Englund, S., Löfmark, S., and Byfors, S. Limited dissemination of extended-spectrum β-lactamase- and plasmidencoded AmpC-producing *Escherichia coli* from food and farm animals, Sweden. *Emerging infectious diseases*, 22(4):634–640, 4 2016, DOI: https://doi.org/10.3201/eid2204.151142.
- [298] Tepeli, S. O. and Zorba, N. N. D. Frequency of extended-spectrum beta-lactamase (ESBL)- and AmpC beta-lactamase-producing *Enterobacteriaceae* in a cheese production process. *Journal of Dairy Science*, 101(4):2906–2914, 2018, DOI: https://doi.org/10.3168/ jds.2017-13878.
- [299] Mughini-Gras, L., Dorado-García, A., van Duijkeren, E., van den Bunt, G., Dierikx, C. M., Bonten, M. J. M., Bootsma, M. C. J., Schmitt, H., Hald, T., Evers, E. G., de Koeijer, A., van Pelt, W., Franz, E., Mevius, D. J., and Heederik, D. J. J. Attributable sources of community-acquired carriage of *Escherichia coli* containing beta-lactam antibiotic resistance genes: a population-based modelling study. *The Lancet Planetary Health*, 3(8):e357–e369, 8 2019, DOI: https://doi.org/10.1016/S2542-5196(19)30130-5.
- [300] Queenan, K., Häsler, B., and Rushton, J. A One Health approach to antimicrobial resistance surveillance: is there a business case for it? *International Journal of Antimicrobial Agents*, 48(4):422–427, 2016, DOI: https://doi.org/https://doi.org/10.1016/j.ijantimicag. 2016.06.014.
- [301] Thomas IV, J., Oladeinde, A., Kieran, T., Finger, J. J., Bayona-Vásquez, N., Cartee, J., Beasley, J., Seaman, J., McArthur, J., Rhodes, O. J., and Glenn, T. Co-occurrence of antibiotic, biocide, and heavy metal resistance genes in bacteria from metal and radionuclide contaminated soils at the Savannah River site. *Microbial Biotechnology*, 13(4):1179–1200, 7 2020, DOI: https://doi.org/https://doi.org/10.1111/1751-7915.13578.
- [302] Chen, J., Li, J., Zhang, H., Shi, W., and Liu, Y. Bacterial heavy-metal and antibiotic resistance genes in a copper tailing dam area in Northern China. *Frontiers in Microbiology*, 10:1916, 8 2019, DOI: https://doi.org/10.3389/fmicb.2019.01916.
- [303] von Tippelskirch, P., Golz, G., Projahn, M., Daehre, K., Friese, A., Roesler, U., Alter, T., and Orquera, S. Prevalence and quantitative analysis of ESBL and AmpC beta-lactamase producing *Enterobacteriaceae* in broiler chicken during slaughter in Germany. *International Journal of Food Microbiology*, 281:82–89, 2018, DOI: https://doi.org/10.1016/j.ijfoodmicro. 2018.05.022.

- [304] Daehre, K., Projahn, M., Semmler, T., Roesler, U., and Friese, A. Extended-spectrum betalactamase-/AmpC beta-lactamase-producing *Enterobacteriaceae* in broiler farms: Transmission dynamics at farm level. *Microbial Drug Resistance*, 24(4):511–518, 2018, DOI: https://doi.org/10.1089/mdr.2017.0150.
- [305] Gazal, L., Medeiros, L., Dibo, M., Nishio, E., Koga, V., Gonçalves, B., Grassotti, T., de Camargo, T., Pinheiro, J., Vespero, E., de Brito, K., de Brito, B., Nakazato, G., and Kobayashi, R. Detection of ESBL/AmpC-producing and fosfomycin-resistant *Escherichia coli* from different sources in poultry production in Southern Brazil. *Frontiers in Microbiology*, 11:3387, 2021, DOI: https://doi.org/10.3389/fmicb.2020.604544.
- [306] von Salviati, C., Laube, H., Guerra, B., Roesler, U., and Friese, A. Emission of ESBL/AmpCproducing *Escherichia coli* from pig fattening farms to surrounding areas. *Veterinary Microbiology*, 175(1):77–84, 2015, DOI: https://doi.org/https://doi.org/10.1016/j.vetmic.2014. 10.010.
- [307] Dohmen, W., Van Gompel, L., Schmitt, H., Liakopoulos, A., Heres, L., Urlings, B., Mevius, D., Bonten, M., and Heederik, D. ESBL carriage in pig slaughterhouse workers is associated with occupational exposure. *Epidemiology and Infection*, 145(10):2003–2010, 2017, DOI: https://doi.org/10.1017/s0950268817000784.
- [308] Hamza, D., Dorgham, S., Ismael, E., El-Moez, S., Elhariri, M., Elhelw, R., and Hamza, E. Emergence of β-lactamase- and carbapenemase- producing *Enterobacteriaceae* at integrated fish farms. *Antimicrobial Resistance & Infection Control*, 9(1):67, 2020, DOI: https://doi. org/10.1186/s13756-020-00736-3.
- [309] Hordijk, J., Fischer, E., van Werven, T., Sietsma, S., Van Gompel, L., Timmerman, A., Spaninks, M., Heederik, D., Nielen, M., Wagenaar, J., and Stegeman, A. Dynamics of faecal shedding of ESBL- or AmpC-producing *Escherichia coli* on dairy farms. *Journal* of Antimicrobial Chemotherapy, 74(6):1531–1538, 2019, DOI: https://doi.org/10.1093/jac/ dkz035.
- [310] Forsi Innovations. Dairy effluent recycling system, 2015, https://www.forsi.co.nz/wp-content/uploads/Forsi\_Effluent\_Recycling\_System\_2015.pdf.
- [311] Lévesque, S., Dufresne, P., Soualhine, H., Domingo, M., Bekal, S., Lefebvre, B., and Tremblay, C. A side by side comparison of Bruker Biotyper and VITEK MS: Utility of MALDI-TOF MS technology for microorganism identification in a public health reference laboratory. *PlOS One*, 10(12):e0144878, 12 2015, DOI: https://doi.org/10.1371/journal.pone.0144878.
- [312] Weinstein, M., Patel, J., Bobenchik, A., Campeau, S., Cullen, S., Galas, M., Gold, H., Humphries, R., Kirn, T., Lewis, J., Limbago, B., Mathers, A., Mazzulli, T., Richter, S., Satlin, M., Schuetz, A., Swenson, J., and Tamma, P. CLSI M100-ED29:2019 Performance standards for antimicrobial susceptibility testing. Report, 2019, https://clsi.org/standards/ products/microbiology/documents/m100/.
- [313] Dierikx, C., van Duijkeren, E., Schoormans, A., van Essen-Zandbergen, A., Veldman, K., Kant, A., Huijsdens, X., van der Zwaluw, K., Wagenaar, J., and Mevius, D. Occurrence

and characteristics of extended-spectrum-β-lactamase- and AmpC-producing clinical isolates derived from companion animals and horses. *Journal of Antimicrobial Chemotherapy*, 67(6):1368–1374, 6 2012, DOI: https://doi.org/10.1093/jac/dks049.

- [314] Kearse, M., Moir, R., Wilson, A., Stones-Havas, S., Cheung, M., Sturrock, S., Buxton, S., Cooper, A., Markowitz, S., Duran, C., Thierer, T., Ashton, B., Meintjes, P., and Drummond, A. Geneious basic: An integrated and extendable desktop software platform for the organization and analysis of sequence data. *Bioinformatics*, 28(12):1647–1649, 2012, DOI: https://doi.org/10.1093/bioinformatics/bts199.
- [315] Biomatters. Genious Prime, https://www.geneious.com/.
- [316] Altschul, S., Gish, W., Miller, W., Myers, E., and Lipman, D. Basic local alignment search tool. Journal of Molecular Biology, 215(3):403–410, 1990, DOI: https://doi.org/10.1016/ s0022-2836(05)80360-2.
- [317] Clermont, O., Bonacorsi, S., and Bingen, E. Characterization of an anonymous molecular marker strongly linked to *Escherichia coli* strains causing neonatal meningitis. *Journal* of *Clinical Microbiology*, 42(4):1770–1772, 4 2004, DOI: https://doi.org/10.1128/JCM.42.4. 1770-1772.2004.
- [318] Lescat, M., Clermont, O., Woerther, P. L., Glodt, J., Dion, S., Skurnik, D., Djossou, F., Dupont, C., Perroz, G., Picard, B., Catzeflis, F., Andremont, A., and Denamur, E. Commensal *Escherichia coli* strains in Guiana reveal a high genetic diversity with hostdependant population structure. *Environmental Microbiology Reports*, 5(1):49–57, 2 2013, DOI: https://doi.org/10.1111/j.1758-2229.2012.00374.x.
- [319] Hall, M. B. Rasusa: Randomly subsample sequencing reads to a specified coverage. Journal of Open Source Software, 7(69):3941, 2022, DOI: https://doi.org/10.21105/joss.03941.
- [320] Seemann, T., Goncalves, d. S. A., Bulach, D., Schultz, M., Kwong, J., and Howden, B. Nullarbor Github, https://github.com/tseemann/nullarbor.
- [321] Bolger, A. M., Lohse, M., and Usadel, B. Trimmomatic: a flexible trimmer for Illumina sequence data. *Bioinformatics*, 30(15):2114–2120, 11 2014, DOI: https://doi.org/10.1093/ bioinformatics/btu170.
- [322] Wood, D. E. and Salzberg, S. L. Kraken: Ultrafast metagenomic sequence classification using exact alignments. *Genome Biology*, 15(3):R46–R46, 2014, DOI: https://doi.org/10. 1186/gb-2014-15-3-r46.
- [323] Souvorov, A., Agarwala, R., and Lipman, D. J. SKESA: strategic k-mer extension for scrupulous assemblies. *Genome Biology*, 19(1):153, 2018, DOI: https://doi.org/10.1186/ s13059-018-1540-z.
- [324] Seemann, T. Prokka: Rapid prokaryotic genome annotation. *Bioinformatics*, 30(14):2068–2069, 2014, DOI: https://doi.org/10.1093/bioinformatics/btu153.
- [325] Seemann, T. mlst, Github, https://github.com/tseemann/mlst.

- [326] Jolley, K. A. and Maiden, M. C. J. BIGSdb: Scalable analysis of bacterial genome variation at the population level. *BMC Bioinformatics*, 11(1):595, 2010, DOI: https://doi.org/10. 1186/1471-2105-11-595.
- [327] Seemann, T. Abricate, Github, https://github.com/tseemann/abricate.
- [328] Bortolaia, V., Kaas, R., Ruppe, E., et al. ResFinder 4.0 for predictions of phenotypes from genotypes. *Journal of Antimicrobial Chemotherapy*, 75(12):3491–3500, 12 2020, DOI: https://doi.org/10.1093/jac/dkaa345.
- [329] Center for Genomic Epidemiology. Overview of services, 2011, http://www.genomicepidemiology.org/.
- [330] Letunic, I. and Bork, P. Interactive tree of life (iTOL) v3: an online tool for the display and annotation of phylogenetic and other trees. *Nucleic Acids Research*, 44(W1):242–245, 2016, DOI: https://doi.org/10.1093/nar/gkw290.
- [331] Seemann, T. Snippy, 2016, https://github.com/tseemann/snippy/blob/master/README. md.
- [332] Kozlov, A. M., Darriba, D., Flouri, T., Morel, B., and Stamatakis, A. RAxML-NG: a fast, scalable and user-friendly tool for maximum likelihood phylogenetic inference. *Bioinformatics*, 35(21):4453–4455, 19 2019, DOI: https://doi.org/10.1093/bioinformatics/btz305.
- [333] Zhou, Z., Alikhan, N.-F., Sergeant, M. J., Luhmann, N., Vaz, C., Francisco, A. P., Carriço, J. A., and Achtman, M. GrapeTree: visualization of core genomic relationships among 100,000 bacterial pathogens. *Genome Research*, 28(9):1395–1404, 9 2018, DOI: https://doi. org/10.1101/gr.232397.117.
- [334] Oxford Nanopore Technologies Ltd. qcat, Github, 2019, https://github.com/nanoporetech/ qcat.
- [335] Wick, R. Porechop, Github, 2018, https://github.com/rrwick/Porechop.
- [336] Wick, R. Filtlong, Github, 2018, https://github.com/rrwick/Filtlong.
- [337] Wick, R. R., Judd, L. M., Gorrie, C. L., and Holt, K. E. Unicycler: Resolving bacterial genome assemblies from short and long sequencing reads. *PLOS Computational Biology*, 13(6):e1005595, 6 2017, DOI: https://doi.org/10.1371/journal.pcbi.1005595.
- [338] Robertson, J. and Nash, J. H. E. MOB-suite: software tools for clustering, reconstruction and typing of plasmids from draft assemblies. *Microbial Genomics*, 4(8):e000206, 8 2018, DOI: https://doi.org/10.1099/mgen.0.000206.
- [339] Robertson, J., Bessonov, K., Schonfeld, J., and Nash, J. Universal whole-sequence-based plasmid typing and its utility to prediction of host range and epidemiological surveillance. *Microbial Genomics*, 6(10), 2020.
- [340] Carattoli, A., Zankari, E., Garcia-Fernandez, A., Larsen, M. V., Lund, O., Villa, L., Aarestrup, F. M., and Hasman, H. In silico detection and typing of plasmids using PlasmidFinder and plasmid multilocus sequence typing. Antimicrobial Agents and Chemotherapy, 58(7):3895–3903, 2014, DOI: https://doi.org/10.1128/aac.02412-14.

- [341] Huerta-Cepas, J., Forslund, K., Coelho, L. P., Szklarczyk, D., Jensen, L. J., von Mering, C., and Bork, P. Fast genome-wide functional annotation through orthology assignment by eggNOG-mapper. *Molecular Biology and Evolution*, 34(8):2115–2122, 8 2017, DOI: https: //doi.org/10.1093/molbev/msx148.
- [342] Cantalapiedra, C. P., Hernández-Plaza, A., Letunic, I., Bork, P., and Huerta-Cepas, J. eggNOG-mapper v2: Functional annotation, orthology assignments, and domain prediction at the metagenomic scale. *Molecular Biology and Evolution*, 9(38):5825–5829, 1 2021, DOI: https://doi.org/10.1093/molbev/msab293.
- [343] Huerta-Cepas, J., Szklarczyk, D., Heller, D., Hernández-Plaza, A., Forslund, S. K., Cook, H., Mende, D. R., Letunic, I., Rattei, T., Jensen, L. J., von Mering, C., and Bork, P. eggNOG 5.0: a hierarchical, functionally and phylogenetically annotated orthology resource based on 5090 organisms and 2502 viruses. *Nucleic Acids Research*, 47(D1):D309–D314, 1 2019, DOI: https://doi.org/10.1093/nar/gky1085.
- [344] Darling, A. C. E., Mau, B., Blattner, F. R., and Perna, N. T. Mauve: multiple alignment of conserved genomic sequence with rearrangements. *Genome Research*, 14(7):1394–1403, 7 2004, DOI: https://doi.org/10.1101/gr.2289704.
- [345] Luo, H. and Gao, F. DoriC 10.0: an updated database of replication origins in prokaryotic genomes including chromosomes and plasmids. *Nucleic Acids Research*, 47(D1):D74–D77, 1 2019, DOI: https://doi.org/10.1093/nar/gky1014.
- [346] INKSCAPE 0.92 Draw Freely, https://inkscape.org/.
- [347] Minitab Inc. Minitab 19 Statistical Software, 2019, https://www.minitab.com/en-us/.
- [348] Olsson, O., Bergström, S., and Normark, S. Identification of a novel ampC beta-lactamase promoter in a clinical isolate of *Escherichia coli*. The EMBO Journal, 1(11):1411–1416, 1982.
- [349] Taylor, D. E. Bacterial tellurite resistance. Trends in Microbiology, 7(3):111–115, 1999, DOI: https://doi.org/https://doi.org/10.1016/S0966-842X(99)01454-7.
- [350] Johnson, T., Wannemuehler, Y., Doetkott, C., Johnson, S., Rosenberger, S., and Nolan, L. Identification of minimal predictors of avian pathogenic *Escherichia coli* virulence for use as a rapid diagnostic tool. *Journal of Clinical Microbiology*, 46(12):3987–3996, 12 2008, DOI: https://doi.org/10.1128/JCM.00816-08.
- [351] Brouwer, M. S. M., Jurburg, S. D., Harders, F., Kant, A., Mevius, D. J., Roberts, A. P., and Bossers, A. The shufflon of IncI1 plasmids is rearranged constantly during different growth conditions. *Plasmid*, 102:51–55, 2019, DOI: https://doi.org/https://doi.org/10.1016/ j.plasmid.2019.03.003.
- [352] Yau, S., Liu, X., Djordjevic, S. P., and Hall, R. M. RSF1010-like plasmids in Australian Salmonella enterica Serovar Typhimurium and origin of their sul2-strA-strB antibiotic resistance gene cluster. Microbial Drug Resistance, 16(4):249–252, 7 2010, DOI: https://doi.org/10.1089/mdr.2010.0033.

- [353] Sukupolvi, S. and O'Connor, C. D. TraT lipoprotein, a plasmid-specified mediator of interactions between gram-negative bacteria and their environment. *Microbiological Reviews*, 54(4):331–341, 12 1990, DOI: https://doi.org/10.1128/mr.54.4.331-341.1990.
- [354] Ceccarelli, D., Kant, A., van Essen-Zandbergen, A., Dierikx, C., Hordijk, J., Wit, B., Mevius, D. J., and Veldman, K. T. Diversity of plasmids and genes encoding resistance to extended spectrum cephalosporins in commensal *Escherichia coli* from Dutch livestock in 2007–2017. *Frontiers in Microbiology*, 10:76, 2019, DOI: https://doi.org/10.3389/fmicb.2019.00076.
- [355] Händel, N., Schuurmans, J. M., Feng, Y., Brul, S., and ter Kuile, B. H. Interaction between mutations and regulation of gene expression during development of *de novo* antibiotic resistance. *Antimicrobial Agents and Chemotherapy*, 58(8):4371–4379, 8 2014, DOI: https://doi.org/10.1128/AAC.02892-14.
- [356] Meini, S., Tascini, C., Cei, M., Sozio, E., and Rossolini, G. AmpC β-lactamase-producing Enterobacterales: what a clinician should know. Infection, 47(3):363–375, 2019, DOI: https: //doi.org/10.1007/s15010-019-01291-9.
- [357] Rizi, K., Mosavat, A., Youssefi, M., Jamehdar, S., Ghazvini, K., Safdari, H., Amini, Y., and Farsiani, H. High prevalence of *bla*CMY AmpC beta-lactamase in ESBL co-producing *Escherichia coli* and *Klebsiella* spp. clinical isolates in the northeast of Iran. *Journal of Global Antimicrobial Resistance*, 22:477–482, 2020, DOI: https://doi.org/https://doi.org/10. 1016/j.jgar.2020.03.011.
- [358] Fonterra Co-operative Group Ltd. Fonterra farmers' handbook 2018/2019, 2018, https: //www.fonterra.com/nz/en/our-co-operative/the-way-we-farm/sustainability-on-farm/ evaluating-and-supporting-farmers.html.
- [359] Petrovski, K. R., Heuer, C., Parkinson, T. J., and Williamson, N. B. The incidence and aetiology of clinical bovine mastitis on 14 farms in Northland, New Zealand. New Zealand Veterinary Journal, 57(2):109–115, 2009, DOI: https://doi.org/10.1080/00480169.2009.36887.
- [360] Pietsch, M., Irrgang, A., Roschanski, N., et al. Whole genome analyses of CMY-2-producing *Escherichia coli* isolates from humans, animals and food in Germany. *BMC Genomics*, 19(1):601, 2018, DOI: https://doi.org/10.1186/s12864-018-4976-3.
- [361] Ritu, B. and R., J. J. A new clone sweeps clean: the enigmatic emergence of *Escherichia coli* sequence type 131. Antimicrobial Agents and Chemotherapy, 58(9):4997–5004, 9 2014, DOI: https://doi.org/10.1128/AAC.02824-14.
- [362] Jafari, A., Falahatkar, S., Delpasand, K., Sabati, H., and Sedigh Ebrahim-Saraie, H. Emergence of *Escherichia coli* ST131 causing urinary tract infection in Western Asia: A systematic review and meta-analysis. *Microbial Drug Resistance*, 26(11):1357–1364, 5 2020, DOI: https://doi.org/10.1089/mdr.2019.0312.
- [363] Guillouzouic, A., Caroff, N., Dauvergne, S., Lepelletier, D., Perrin Guyomard, A., Kempf, I., Reynaud, A., and Corvec, S. MLST typing of *Escherichia coli* isolates overproducing AmpC β-lactamase. *Journal of Antimicrobial Chemotherapy*, 63(6):1290–1292, 6 2009, DOI: https://doi.org/10.1093/jac/dkp099.

- [364] Coolen, J. P. M., den Drijver, E. P. M., Verweij, J. J., Schildkraut, J. A., Neveling, K., Melchers, W. J. G., Kolwijck, E., Wertheim, H. F. L., Kluytmans, J. A. J. W., and Huynen, M. A. Genome-wide analysis in *Escherichia coli* unravels a high level of genetic homoplasy associated with cefotaxime resistance. *Microbial Genomics*, 7(4), 2021, DOI: https://doi. org/https://doi.org/10.1099/mgen.0.000556.
- [365] Toombs-Ruane, L. J., Benschop, J., French, N. P., Biggs, P. J., Midwinter, A. C., Marshall, J. C., Chan, M., Drinković, D., Fayaz, A., Baker, M. G., Douwes, J., Roberts, M. G., and Burgess, S. A. Carriage of extended-spectrum-beta-lactamase- and AmpC beta-lactamase-producing Escherichia coli strains from humans and pets in the same households. *Applied and Environmental Microbiology*, 86(24):01613–20, 7 2021, DOI: https://doi.org/10.1128/AEM.01613-20.
- [366] Burgess, S., Francois, M., Midwinter, A., Biggs, P., and Rasko, D. Draft genome sequences of seven extended-spectrum β-lactamase-producing Escherichia coli strains isolated from New Zealand waterways. Microbiology Resource Announcements, 10(11):01445–20, 9 2021, DOI: https://doi.org/10.1128/MRA.01445-20.
- [367] Maluta, R., Logue, C., Casas, M., Meng, T., Guastalli, E., Rojas, T., Montelli, A., Sadatsune, T., de Carvalho Ramos, M., Nolan, L., and da Silveira, W. Overlapped sequence types (STs) and serogroups of avian pathogenic (APEC) and human extra-intestinal pathogenic (ExPEC) *Escherichia coli* isolated in Brazil. *PLOS One*, 9(8):e105016, 8 2014, DOI: https: //doi.org/10.1371/journal.pone.0105016.
- [368] Mathys, D., Mathys, B., Mollenkopf, D., Daniels, J., and Wittum, T. Enterobacteriaceae harboring AmpC (blaCMY) and ESBL (blaCTX-M) in migratory and monmigratory wild songbird populations on Ohio dairies. Vector-Borne and Zoonotic Diseases, 17(4):254–259, 2 2017, DOI: https://doi.org/10.1089/vbz.2016.2038.
- [369] Torres, E., López-Cerero, L., Rodríguez-Martínez, J. M., and Pascual, Reduced susceptibility to cefepime in clinical isolates of *Enterobacteriaceae* producing OXA-1 beta-lactamase. *Microbial Drug Resistance*, 22(2):141–146, 8 2015, DOI: https://doi.org/10.1089/mdr.2015. 0122.
- [370] Beceiro, A., Maharjan, S., Gaulton, T., Doumith, M., Soares, N. C., Dhanji, H., Warner, M., Doyle, M., Hickey, M., Downie, G., Bou, G., Livermore, D. M., and Woodford, N. False extended-spectrum β-lactamase phenotype in clinical isolates of *Escherichia coli* associated with increased expression of OXA-1 or TEM-1 penicillinases and loss of porins. *Journal of Antimicrobial Chemotherapy*, 66(9):2006–2010, 9 2011, DOI: https://doi.org/10.1093/jac/ dkr265.
- [371] Boyle, F., Healy, G., Hale, J., Kariuki, S., Cormican, M., and Morris, D. Characterization of a novel extended-spectrum β-lactamase phenotype from OXA-1 expression in Salmonella Typhimurium strains from Africa and Ireland. Diagnostic Microbiology and Infectious Disease, 70(4):549–553, 8 2011, DOI: https://doi.org/10.1016/j.diagmicrobio.2011.04.007.
- [372] Choi, U. and Lee, C. Distinct roles of outer membrane porins in antibiotic resistance and membrane integrity in *Escherichia coli*. Frontiers in Microbiology, 10:953, 2019, DOI: https: //doi.org/10.3389/fmicb.2019.00953.

- [373] Fernández, L. and Hancock, R. Adaptive and mutational resistance: role of porins and efflux pumps in drug resistance. *Clinical Microbiology Reviews*, 25(4):661–681, 10 2012, DOI: https://doi.org/10.1128/CMR.00043-12.
- [374] Orlek, A., Phan, H., Sheppard, A., Doumith, M., Ellington, M., Peto, T., Crook, D., Walker, A., Woodford, N., Anjum, M., and Stoesser, N. Ordering the mob: Insights into replicon and MOB typing schemes from analysis of a curated dataset of publicly available plasmids. *Plasmid*, 91:42–52, 2017, DOI: https://doi.org/https://doi.org/10.1016/j.plasmid.2017.03.002.
- [375] Ribeiro, T. G., Novais, Peixe, L., and Machado, E. Atypical epidemiology of CTX-M-15 among *Enterobacteriaceae* from a high diversity of non-clinical niches in Angola. *Journal of Antimicrobial Chemotherapy*, 71(5):1169–1173, 5 2016, DOI: https://doi.org/10.1093/jac/ dkv489.
- [376] Chunping, Z., Yuqing, F., Fei, L., Hui, J., Zhina, Q., Meng, L., Jianfeng, W., Bing, Z., Yongfei, H., Jiabo, D., and Baoli, Z. A phage-like IncY plasmid carrying the mcr-1 gene in Escherichia coli from a pig farm in China. Antimicrobial Agents and Chemotherapy, 61(3):02035–16, 7 2021, DOI: https://doi.org/10.1128/AAC.02035-16.
- [377] Choi, K., Li, Y., Sarnovsky, R., and Craig, N. Direct interaction between the ThsA and ThsB subunits controls the heteromeric Th7 transposase. *Proceedings of the National Academy of Sciences*, 110(22):E2038 LP – E2045, 5 2013, DOI: https://doi.org/10.1073/pnas.1305716110.
- [378] He, J., Li, C., Cui, P., and Wang, H. Detection of Tn7-like transposons and antibiotic resistance in *Enterobacterales* from animals used for food production with identification of three novel transposons Tn6813, Tn6814, and Tn6765. *Frontiers in Microbiology*, 11:2049, 2020, DOI: https://doi.org/10.3389/fmicb.2020.02049.
- [379] Looft, T., Allen, H. K., Cantarel, B. L., Levine, U. Y., Bayles, D. O., Alt, D. P., Henrissat, B., and Stanton, T. B. Bacteria, phages and pigs: The effects of in-feed antibiotics on the microbiome at different gut locations. *ISME Journal*, 8(8):1566–1576, 2014, DOI: https: //doi.org/10.1038/ismej.2014.12.
- [380] Xiong, W. G., Wang, Y. L., Sun, Y. X., Ma, L. P., Zeng, Q. L., Jiang, X. T., Li, A. D., Zeng, Z. L., and Zhang, T. Antibiotic-mediated changes in the fecal microbiome of broiler chickens define the incidence of antibiotic resistance genes. *Microbiome*, 6:11, 2018, DOI: https://doi.org/10.1186/s40168-018-0419-2.
- [381] Modi, S. R., Collins, J. J., and Relman, D. A. Antibiotics and the gut microbiota. The Journal of Clinical Investigation, 124(10):4212–4218, 10 2014, DOI: https://doi.org/10.1172/ JCI72333.
- [382] Holman, D. B., Yang, W., and Alexander, T. W. Antibiotic treatment in feedlot cattle: a longitudinal study of the effect of oxytetracycline and tulathromycin on the fecal and nasopharyngeal microbiota. *Microbiome*, 7(1):86, 2019, DOI: https://doi.org/10.1186/ s40168-019-0696-4.
- [383] Dethlefsen, L., Huse, S., Sogin, M. L., and Relman, D. A. The pervasive effects of an antibiotic on the human gut microbiota, as revealed by deep 16S rRNA sequencing. *PLOS Biology*, 6(11):e280, 11 2008, DOI: https://doi.org/10.1371/journal.pbio.0060280.

- [384] Zeineldin, M., Aldridge, B., and Lowe, J. Dysbiosis of the fecal microbiota in feedlot cattle with hemorrhagic diarrhea. *Microbial Pathogenesis*, 115:123–130, 2 2018, DOI: https://doi. org/10.1016/j.micpath.2017.12.059.
- [385] Hagey, J. V., Bhatnagar, S., Heguy, J. M., Karle, B. M., Price, P. L., Meyer, D., and Maga, E. A. Fecal microbial communities in a large representative cohort of California dairy cows. *Frontiers in Microbiology*, 10:1093, 2019, DOI: https://doi.org/10.3389/fmicb.2019.01093.
- [386] Deshpande, L., Pfaller, M. A., and Jones, R. N. In vitro activity of ceftiofur tested against clinical isolates of Escherichia coli and Klebsiella pneumoniae including extended spectrum β-lactamase producing strains. International Journal of Antimicrobial Agents, 15(4):271– 275, 2000, DOI: https://doi.org/https://doi.org/10.1016/S0924-8579(00)00184-9.
- [387] Burgess, S. and French, N. Antimicrobial resistant bacteria in dairy cattle: A review. Report, Palmerston North, New Zealand, 2017, https://www.nzfssrc.org.nz/node/79.
- [388] New Zealand Veterinary Association. Judicious use of antimicrobials, 2015, https://www.nzva.org.nz/page/policyantimicrobials.
- [389] Allen, H. K., Looft, T., Bayles, D. O., Humphrey, S., Levine, U. Y., Alt, D., and Stanton, T. B. Antibiotics in feed induce prophages in swine fecal microbiomes. *mBio*, 2(6):9, 2011, DOI: https://doi.org/10.1128/mBio.00260-11.
- [390] Durso, L. M., Miller, D. N., and Wienhold, B. J. Distribution and quantification of antibiotic resistant genes and bacteria across agricultural and non-agricultural metagenomes. *PLOS One*, 7(11):e48325, 11 2012, DOI: https://doi.org/10.1371/journal.pone.0048325.
- [391] Madinier, I., Fosse, T., Giudicelli, J., and Labia, R. Cloning and biochemical characterization of a class A beta-lactamase from *Prevotella intermedia*. Antimicrobial Agents and Chemotherapy, 45(8):2386–2389, 2001, DOI: https://doi.org/10.1128/AAC.45.8.2386-2389. 2001.
- [392] Jolivet-Gougeon, A., Tamanai-Shacoori, Z., Desbordes, L., Burggraeve, N., Cormier, M., and Bonnaure-Mallet, M. Genetic analysis of an ambler class A extended-spectrum betalactamase from *Capnocytophaga ochracea*. Journal of Clinical Microbiology, 42(2):888–890, 2004, DOI: https://doi.org/10.1128/jcm.42.2.888-890.2004.
- [393] Kozich, J. J., Westcott, S. L., Baxter, N. T., Highlander, S. K., and Schloss, P. D. Development of a dual-index sequencing strategy and curation pipeline for analyzing amplicon sequence data on the MiSeq Illumina sequencing platform. *Applied and Environmental Mi*crobiology, 79(17):5112–5120, 9 2013, DOI: https://doi.org/10.1128/AEM.01043-13.
- [394] Aronesty, E. ea-utils: Command-line tools for processing biological sequencing data, 2011, https://expressionanalysis.github.io/ea-utils/.
- [395] Callahan, B. J., McMurdie, P. J., Rosen, M. J., Han, A. W., Johnson, A. J. A., and Holmes, S. P. DADA2: High-resolution sample inference from Illumina amplicon data. *Nature Meth*ods, 13(7):581–583, 7 2016, DOI: https://doi.org/10.1038/nmeth.3869.
- [396] R Core Team. A language and environment for statistical computing., 2020, https://www. r-project.org/.

- [397] Quast, C., Pruesse, E., Yilmaz, P., Gerken, J., Schweer, T., Yarza, P., Peplies, J., and Glöckner, F. O. The SILVA ribosomal RNA gene database project: improved data processing and web-based tools. *Nucleic Acids Research*, 41(Database issue):D590–D596, 1 2013, DOI: https://doi.org/10.1093/nar/gks1219.
- [398] Yilmaz, P., Parfrey, L. W., Yarza, P., Gerken, J., Pruesse, E., Quast, C., Schweer, T., Peplies, J., Ludwig, W., and Glöckner, F. O. The SILVA and "All-species Living Tree Project (LTP)" taxonomic frameworks. *Nucleic Acids Research*, 42(Database issue):D643–D648, 1 2014, DOI: https://doi.org/10.1093/nar/gkt1209.
- [399] Glöckner, F. O., Yilmaz, P., Quast, C., Gerken, J., Beccati, A., Ciuprina, A., Bruns, G., Yarza, P., Peplies, J., Westram, R., and Ludwig, W. 25 years of serving the community with ribosomal RNA gene reference databases and tools. *Journal of Biotechnology*, 261:169–176, 11 2017, DOI: https://doi.org/10.1016/j.jbiotec.2017.06.1198.
- [400] McMurdie, P. J. and Holmes, S. phyloseq: An R Package for Reproducible Interactive Analysis and Graphics of Microbiome Census Data. *PLOS One*, 8(4):e61217, 4 2013, DOI: https://doi.org/10.1371/journal.pone.0061217.
- [401] Oksanen, J., Guillaume Blanchet, F., Friendly, M., Kindt, R., Legendre, P., McGlinn, D., Minchin, P. R., O'Hara, R. B., Simpson, G. L., Solymos, P., Stevens, M. H. H., Szoecs, E., and Wagner, H. vegan: Community Ecology Package, 2020, https://cran.r-project.org/ package=vegan.
- [402] Lahti, L. and Sudarshan, S. microbiome R package, 2012, https://microbiome.github.io/ tutorials/.
- [403] Wickham, H. ggplot2: Elegant Graphics for Data Analysis, 2016, https://ggplot2.tidyverse. org.
- [404] Wickham, H., Averick, M., Bryan, J., Chang, W., D'Agostino McGowan, L., François, R., Grolemund, G., Hayes, A., Henry, L., Hester, J., Kuhn, M., Pederson, T., Miller, E., Bache, S., Müller, K., Ooms, J., Robinson, D., Seidel, D., Spinu, V., Takahashi, K., Vaughan, D., Wilke, C., Woo, K., and Yutani, H. Welcome to the Tidyverse. *Journal of Open Source Software*, 4(43):1686, 2019, DOI: https://doi.org/10.21105/joss.01686.
- [405] Wickham, H., Romain, F., Lionel, H., and Müller, K. dplyr: A grammar of data manipulation, 2021, https://cran.r-project.org/package=dplyr.
- [406] Thomson, G. Manu: NZ Bird Colour Palettes, 2020, https://github.com/G-Thomson/Manu.
- [407] Neuwirth, E. RColorBrewer: ColorBrewer palettes, 2014, https://cran.r-project.org/package=RColorBrewer.
- [408] Gramazio, C. C., Laidlaw, D. H., and Schloss, K. B. Colorgorical: Creating discriminable and preferable color palettes for information visualization. *IEEE Transactions on Visualization* and Computer Graphics, 23(1):521–530, 2017, DOI: https://doi.org/10.1109/TVCG.2016. 2598918.

- [409] Bates, D., Mächler, M., Bolker, B., and Walker, S. Fitting linear mixed-effects models using lme4. Journal of Statistical Software, 67(1 SE - Articles):1–48, 10 2015, DOI: https: //doi.org/10.18637/jss.v067.i01.
- [410] Lenth, R. emmeans: estimated marginal means, aka least-squares means, 2021, https://cran. r-project.org/package=emmeans.
- [411] Almeida, R. A., Patel, D., Friton, G. M., and Oliver, S. P. Intracellular killing of mastitis pathogens by penethamate hydriodide following internalization into mammary epithelial cells. Journal of Veterinary Pharmacology and Therapeutics, 30(2):151–156, 4 2007, DOI: https://doi.org/https://doi.org/10.1111/j.1365-2885.2007.00830.x.
- [412] Ashenafi, M., Ammosova, T., Nekhai, S., and Byrnes, W. M. Purification and characterization of aminoglycoside phosphotransferase APH(6)-Id, a streptomycin-inactivating enzyme. *Molecular and Cellular Biochemistry*, 387(1-2):207–216, 2 2014, DOI: https: //doi.org/10.1007/s11010-013-1886-1.
- [413] Chen, F., Cheng, G., Xu, Y., Wang, Y., Xia, Q., and Hu, S. Rumen microbiota distribution analyzed by high-throughput sequencing after oral doxycycline administration in beef cattle. *Frontiers in Veterinary Science*, 7:251, 2020, DOI: https://doi.org/10.3389/fvets.2020.00251.
- [414] Doster, E., Rovira, P., Noyes, N. R., Burgess, B. A., Yang, X., Weinroth, M. D., Lakin, S. M., Dean, C. J., Linke, L., Magnuson, R., Jones, K. I., Boucher, C., Ruiz, J., Belk, K. E., and Morley, P. S. Investigating effects of tulathromycin metaphylaxis on the fecal resistome and microbiome of commercial feedlot cattle early in the feeding period. *Frontiers* in Microbiology, 9:1715, 2018, DOI: https://doi.org/10.3389/fmicb.2018.01715.
- [415] Henderson, G., Cox, F., Ganesh, S., et al. Rumen microbial community composition varies with diet and host, but a core microbiome is found across a wide geographical range. *Scientific Reports*, 5(1):14567, 2015, DOI: https://doi.org/10.1038/srep14567.
- [416] Weinroth, M. D., Martin, J. N., Doster, E., Geornaras, I., Parker, J. K., Carlson, C. R., Metcalf, J. L., Morley, P. S., and Belk, K. E. Investigation of tylosin in feed of feedlot cattle and effects on liver abscess prevalence, and fecal and soil microbiomes and resistomes. *Journal* of Animal Science, 97(11):4567–4578, 11 2019, DOI: https://doi.org/10.1093/jas/skz306.
- [417] Deusch, S., Camarinha-Silva, A., Conrad, J., Beifuss, U., Rodehutscord, M., and Seifert, J. A structural and functional elucidation of the rumen microbiome influenced by various diets and microenvironments. *Frontiers in Microbiology*, 8:1605, 8 2017, DOI: https://doi.org/10. 3389/fmicb.2017.01605.
- [418] Aliabadi, F. S. and Lees, P. Pharmacokinetics and pharmacokinetic/pharmacodynamic integration of marbofloxacin in calf serum, exudate and transudate. *Journal of Veteri*nary Pharmacology and Therapeutics, 25(3):161–174, 6 2002, DOI: https://doi.org/https: //doi.org/10.1046/j.1365-2885.2002.00399.x.
- [419] Rhodes, K. A. and Schweizer, H. P. Antibiotic resistance in Burkholderia species. Drug resistance updates : reviews and commentaries in antimicrobial and anticancer chemotherapy, 28:82–90, 9 2016, DOI: https://doi.org/10.1016/j.drup.2016.07.003.

- [420] Wang, J., Gu, J., Wang, X., Song, Z., Dai, X., Guo, H., Yu, J., Zhao, W., and Lei, L. Enhanced removal of antibiotic resistance genes and mobile genetic elements during swine manure composting inoculated with mature compost. *Journal of Hazardous Materi*als, 411:125135, 2021, DOI: https://doi.org/https://doi.org/10.1016/j.jhazmat.2021.125135.
- [421] Arredondo, A., Blanc, V., Mor, C., Nart, J., and León, R. Resistance to β-lactams and distribution of β-lactam resistance genes in subgingival microbiota from Spanish patients with periodontitis. *Clinical Oral Investigations*, 24(12):4639–4648, 2020, DOI: https://doi. org/10.1007/s00784-020-03333-1.
- [422] Schmidt, V. M., Pinchbeck, G., McIntyre, K. M., Nuttall, T., McEwan, N., Dawson, S., and Williams, N. J. Routine antibiotic therapy in dogs increases the detection of antimicrobialresistant faecal *Escherichia coli*. Journal of Antimicrobial Chemotherapy, 73(12):3305–3316, 12 2018, DOI: https://doi.org/10.1093/jac/dky352.
- [423] Taylor, E. A., Jordan, E. R., Garcia, J. A., Hagevoort, G. R., Norman, K. N., Lawhon, S. D., Piñeiro, J. M., and Scott, H. M. Effects of two-dose ceftiofur treatment for metritis on the temporal dynamics of antimicrobial resistance among fecal *Escherichia coli* in Holstein-Friesian dairy cows. *PLOS One*, 14(7):e0220068, 7 2019, DOI: https://doi.org/10.1371/ journal.pone.0220068.
- [424] Tragesser, L. A., Wittum, T. E., Funk, J. A., Winokur, P. L., and Rajala-Schultz, P. J. Association between ceftiofur use and isolation of *Escherichia coli* with reduced susceptibility to ceftriaxone from fecal samples of dairy cows. *American Journal of Veterinary Research*, 67(10):1696–1700, 2006, DOI: https://doi.org/10.2460/ajvr.67.10.1696.
- [425] Daniels, J. B., Call, D. R., Hancock, D., Sischo, W. M., Baker, K., and Besser, T. E. Role of ceftiofur in selection and dissemination of *bla*CMY-2-mediated cephalosporin resistance in *Salmonella enterica* and commensal *Escherichia coli* isolates from cattle. *Applied and Environmental Microbiology*, 75(11):3648–3655, 6 2009, DOI: https://doi.org/10.1128/AEM. 02435-08.
- [426] Singer, R. S., Patterson, S. K., and Wallace, R. I. Effects of therapeutic ceftiofur administration to dairy cattle on *Escherichia coli* dynamics in the intestinal tract. *Applied and En*vironmental Microbiology, 74(22):6956–6962, 11 2008, DOI: https://doi.org/10.1128/AEM. 01241-08.
- [427] Hendriksen, R. S., Munk, P., Njage, P., et al. Global monitoring of antimicrobial resistance based on metagenomics analyses of urban sewage. *Nature Communications*, 10:12, 2019, DOI: https://doi.org/10.1038/s41467-019-08853-3.
- [428] Rodríguez, E., Ramirez, D., Balcázar, J., and Jiménez, J. Metagenomic analysis of urban wastewater resistome and mobilome: A support for antimicrobial resistance surveillance in an endemic country. *Environmental Pollution*, 276:116736, 2021, DOI: https://doi.org/https: //doi.org/10.1016/j.envpol.2021.116736.
- [429] Marathe, N., Berglund, F., Razavi, M., Pal, C., Dröge, J., Samant, S., Kristiansson, E., and Larsson, D. Sewage effluent from an Indian hospital harbors novel carbapenemases and integron-borne antibiotic resistance genes. *Microbiome*, 7(1):97, 2019, DOI: https://doi.org/ 10.1186/s40168-019-0710-x.

- [430] Pitta, D. W., Indugu, N., Toth, J. D., Bender, J. S., Baker, L. D., Hennessy, M. L., Vecchiarelli, B., Aceto, H., and Dou, Z. The distribution of microbiomes and resistomes across farm environments in conventional and organic dairy herds in Pennsylvania. *Environmental Microbiome*, 15(1):21, 2020, DOI: https://doi.org/10.1186/s40793-020-00368-5.
- [431] Rovira, P., McAllister, T., Lakin, S., Cook, S., Doster, E., Noyes, N., Weinroth, M., Yang, X., Parker, J., Boucher, C., Booker, C., Woerner, D., Belk, K., and Morley, P. Characterization of the microbial resistome in conventional and "raised without antibiotics" beef and dairy production systems. *Frontiers in Microbiology*, 10:12, 2019, DOI: https://doi.org/10.3389/ fmicb.2019.01980.
- [432] Li, B., Chen, Z., Zhang, F., Liu, Y., and Yan, T. Abundance, diversity and mobility potential of antibiotic resistance genes in pristine Tibetan Plateau soil as revealed by soil metagenomics. *FEMS Microbiology Ecology*, 96(10), 10 2020, DOI: https://doi.org/10.1093/ femsec/fiaa172.
- [433] Guo, X., Akram, S., Stedtfeld, R., Johnson, M., Chabrelie, A., Yin, D., and Mitchell, J. Distribution of antimicrobial resistance across the overall environment of dairy farms – A case study. *Science of the Total Environment*, 788:147489, 2021, DOI: https://doi.org/https: //doi.org/10.1016/j.scitotenv.2021.147489.
- [434] Tóth, A., Csabai, I., Krikó, E., Tőzsér, D., Maróti, G., Patai, , Makrai, L., Szita, G., and Solymosi, N. Antimicrobial resistance genes in raw milk for human consumption. *Scientific Reports*, 10(1):7464, 2020, DOI: https://doi.org/10.1038/s41598-020-63675-4.
- [435] Brunton, L., Reeves, H., Snow, L., and Jones, J. A longitudinal field trial assessing the impact of feeding waste milk containing antibiotic residues on the prevalence of ESBL-producing *Escherichia coli* in calves. *Preventive Veterinary Medicine*, 117(2):403–412, 2014, DOI: https: //doi.org/https://doi.org/10.1016/j.prevetmed.2014.08.005.
- [436] Aust, V., Knappstein, K., Kunz, H.-J., Kaspar, H., Wallmann, J., and Kaske, M. Feeding untreated and pasteurized waste milk and bulk milk to calves: effects on calf performance, health status and antibiotic resistance of faecal bacteria. *Journal of Animal Physiology and Animal Nutrition*, 97(6):1091–1103, 12 2013, DOI: https://doi.org/https://doi.org/10.1111/ jpn.12019.
- [437] Holmes, C. W. Milk production from pasture. Palmerston North, N.Z. : Massey University, 2002, 2002.
- [438] Dairy NZ. QuickStats about dairying New Zealand, 2019, https://www.dairynz.co.nz/ media/5791052/quickstats-about-dairying-new-zealand-2019.pdf.
- [439] Anonymous. 2016 New Zealand Organic Market Report. Report, Auckland, New Zealand, 2016.
- [440] ZYMO Research Corp. ZymoBIOMICS<sup>™</sup> Microbial Community DNA Standard, https://files.zymoresearch.com/protocols/\_d6305\_d6306\_zymobiomics\_microbial\_ community\_dna\_standard.pdf.

- [441] ZYMO Research Corp. ZymoBIOMICS<sup>™</sup> Microbial Community DNA Standard II (Log Distribution), https://files.zymoresearch.com/protocols/\_d6311\_zymobiomics\_microbial\_ community\_dna\_standard\_ii\_(log\_distribution).pdf.
- [442] Krueger, F. TrimGalore, 2019, https://github.com/FelixKrueger/TrimGalore.
- [443] Martin, M. Cutadapt removes adapter sequences from high-throughput sequencing reads. EMBnet.journal, 17(1):10–12, 2011, DOI: https://doi.org/10.14806/ej.17.1.200.
- [444] Andrews, S. FastQC: A quality control tool for high throughput sequence data, 2010, http: //www.bioinformatics.babraham.ac.uk/projects/fastqc/.
- [445] Rotmistrovsky, K. and Agarwala, R. BMTagger: Best Match Tagger for removing human reads from metagenomics datasets, 2011, ftp://ftp.ncbi.nlm.nih.gov/pub/agarwala/ bmtagger/.
- [446] Bengtsson-Palme, J., Hartmann, M., Eriksson, K., Pal, C., Thorell, K., Larsson, D., and Nilsson, R. metaxa2: improved identification and taxonomic classification of small and large subunit rRNA in metagenomic data. *Molecular Ecology Resources*, 15(6):1403–1414, 2015, DOI: https://doi.org/10.1111/1755-0998.12399.
- [447] Nayfach, S. and Pollard, K. S. Average genome size estimation improves comparative metagenomics and sheds light on the functional ecology of the human microbiome. *Genome Biology*, 16(1):51, 2015, DOI: https://doi.org/10.1186/s13059-015-0611-7.
- [448] Wood, D. E., Lu, J., and Langmead, B. Improved metagenomic analysis with Kraken 2. Genome Biology, 20(1):257, 2019, DOI: https://doi.org/10.1186/s13059-019-1891-0.
- [449] Lu, J., Breitwieser, F. P., Thielen, P., and Salzberg, S. L. Bracken: estimating species abundance in metagenomics data. *PeerJ Computer Science*, 3(e104), 2017, DOI: https://doi.org/https://doi.org/10.7717/peerj-cs.104.
- [450] Li, D., Liu, C., Luo, R., Sadakane, K., and Lam, T. MEGAHIT: an ultra-fast single-node solution for large and complex metagenomics assembly via succinct de Bruijn graph. *Bioinformatics*, 31(10):1674–1676, 5 2015, DOI: https://doi.org/10.1093/bioinformatics/btv033.
- [451] Doster, E., Lakin, S., Dean, C., Wolfe, C., Young, J., Boucher, C., Belk, K., Noyes, N., and Morley, P. MEGARes 2.0: a database for classification of antimicrobial drug, biocide and metal resistance determinants in metagenomic sequence data. *Nucleic Acids Research*, 48(D1):D561–D569, 2020, DOI: https://doi.org/10.1093/nar/gkz1010.
- [452] Li, H. and Durbin, R. Fast and accurate short read alignment with Burrows–Wheeler transform. *Bioinformatics*, 25(14):1754–1760, 7 2009, DOI: https://doi.org/10.1093/ bioinformatics/btp324.
- [453] Li, B., Yang, Y., Ma, L. P., Ju, F., Guo, F., Tiedje, J. M., and Zhang, T. Metagenomic and network analysis reveal wide distribution and co-occurrence of environmental antibiotic resistance genes. *ISME Journal*, 9(11):2490–2502, 2015, DOI: https://doi.org/10.1038/ismej. 2015.59.

- [454] DeSantis, T., Hugenholtz, P., Larsen, N., Rojas, M., Brodie, E., Keller, K., Huber, T., Dalevi, D., Hu, P., and Andersen, G. Greengenes, a chimera-checked 16S rRNA gene database and workbench compatible with ARB. *Applied and Environmental Microbiology*, 72(7):5069– 5072, 7 2006, DOI: https://doi.org/10.1128/AEM.03006-05.
- [455] von Meijenfeldt, F., Arkhipova, K., Cambuy, D., Coutinho, F., and Dutilh, B. Robust taxonomic classification of uncharted microbial sequences and bins with CAT and BAT. *Genome Biology*, 20(1):217, 2019, DOI: https://doi.org/10.1186/s13059-019-1817-x.
- [456] RStudio Team. RStudio: Integrated development for R, 2019, http://www.rstudio.com/.
- [457] Smith, C. J., Tribble, G. D., and Bayley, D. P. Genetic elements of *Bacteroides* species: A moving story. *Plasmid*, 40(1):12–29, 1998, DOI: https://doi.org/https://doi.org/10.1006/ plas.1998.1347.
- [458] Achard, A., Villers, C., Pichereau, V., and Leclercq, R. New *lnu(C)* gene conferring resistance to lincomycin by nucleotidylation in *Streptococcus agalactiae* UCN36. *Antimicrobial Agents and Chemotherapy*, 49(7):2716–2719, 7 2005, DOI: https://doi.org/10.1128/AAC.49. 7.2716-2719.2005.
- [459] Zheng, J., Wittouck, S., Salvetti, E., Franz, C. M. A. P., Harris, H. M. B., Mattarelli, P., O'Toole, P. W., Pot, B., Vandamme, P., Walter, J., Watanabe, K., Wuyts, S., Felis, G. E., Gänzle, M. G., and Lebeer, S. A taxonomic note on the genus *Lactobacillus*: Description of 23 novel genera, emended description of the genus *Lactobacillus* Beijerinck 1901, and union of *Lactobacillaceae* and *Leuconostocaceae*. *International Journal of Systematic and Evolutionary Microbiology*, 70(4):2782–2858, 4 2020, DOI: https://doi.org/10.1099/ijsem.0.004107.
- [460] Checcucci, A., Trevisi, P., Luise, D., Modesto, M., Blasioli, S., Braschi, I., and Mattarelli, P. Exploring the animal waste resistome: The spread of antimicrobial resistance genes through the use of livestock manure. *Frontiers in Microbiology*, 11:1416, 2020, DOI: https: //doi.org/10.3389/fmicb.2020.01416.
- [461] DairyNZ. A farmer's guide to managing farm dairy effluent. Report, Hamilton, New Zealand, 2015, https://www.dairynz.co.nz/media/2832537/farmers-guide-to-managing-fde.pdf.
- [462] Qian, X., Sun, W., Gu, J., Wang, X.-J., Sun, J.-J., Yin, Y.-N., and Duan, M.-L. Variable effects of oxytetracycline on antibiotic resistance gene abundance and the bacterial community during aerobic composting of cow manure. *Journal of Hazardous Materials*, 315:61–69, 2016, DOI: https://doi.org/https://doi.org/10.1016/j.jhazmat.2016.05.002.
- [463] Xie, W.-Y., Yang, X.-P., Li, Q., Wu, L.-H., Shen, Q.-R., and Zhao, F.-J. Changes in antibiotic concentrations and antibiotic resistome during commercial composting of animal manures. *Environmental Pollution*, 219:182–190, 2016, DOI: https://doi.org/https://doi.org/10.1016/ j.envpol.2016.10.044.
- [464] Sukhum, K. V., Vargas, R. C., Boolchandani, M., D'Souza, A. W., Patel, S., Kesaraju, A., Walljasper, G., Hegde, H., Ye, Z., Valenzuela, R. K., Gunderson, P., Bendixsen, C., Dantas, G., and Shukla, S. K. Manure microbial communities and resistance profiles reconfigure after transition to manure pits and differ from those in fertilized field soil. *mBio*, 12(3):00798–21, 10 2021, DOI: https://doi.org/10.1128/mBio.00798-21.

- [465] Pal, C., Bengtsson-Palme, J., Kristiansson, E., and Larsson, D. G. J. The structure and diversity of human, animal and environmental resistomes. *Microbiome*, 4(1):54, 2016, DOI: https://doi.org/10.1186/s40168-016-0199-5.
- [466] García, N., Gutiérrez, G., Lorenzo, M., García, J. E., Píriz, S., and Quesada, A. Genetic determinants for *cfxA* expression in *Bacteroides* strains isolated from human infections. *Journal of Antimicrobial Chemotherapy*, 62(5):942–947, 11 2008, DOI: https: //doi.org/10.1093/jac/dkn347.
- [467] Ferreira, L. Q., Avelar, K. E. S., Vieira, J. M. B. D., de Paula, G. R., Colombo, A. P. V., Domingues, R. M. C. P., and Ferreira, M. C. S. Association between the *cfxA* gene and transposon Tn4555 in *Bacteroides distasonis* strains and other *Bacteroides* species. *Current Microbiology*, 54(5):348–353, 2007, DOI: https://doi.org/10.1007/s00284-006-0411-0.
- [468] Giraud-Morin, C., Madinier, I., and Fosse, T. Sequence analysis of cfxA2-like β-lactamases in Prevotella species. Journal of Antimicrobial Chemotherapy, 51(5):1293–1296, 5 2003, DOI: https://doi.org/10.1093/jac/dkg221.
- [469] Boyd, E. and Barkay, T. The mercury resistance operon: from an origin in a geothermal environment to an efficient detoxification machine, 2012, https://www.frontiersin.org/article/ 10.3389/fmicb.2012.00349.
- [470] EFSA Panel on Biological Hazards (BIOHAZ)., Ricci, A., Allende, A., et al. Risk for the development of Antimicrobial Resistance (AMR) due to feeding of calves with milk containing residues of antibiotics. *EFSA Journal*, 15(1):e04665, 1 2017, DOI: https://doi.org/https: //doi.org/10.2903/j.efsa.2017.4665.
- [471] Tempini, P. N., Aly, S. S., Karle, B. M., and Pereira, R. V. Multidrug residues and antimicrobial resistance patterns in waste milk from dairy farms in Central California. *Journal of Dairy Science*, 101(9):8110–8122, 2018, DOI: https://doi.org/https://doi.org/10.3168/jds. 2018-14398.
- [472] Duse, A., Waller, K. P., Emanuelson, U., Unnerstad, H. E., Persson, Y., and Bengtsson, B. Risk factors for antimicrobial resistance in fecal *Escherichia coli* from preweaned dairy calves. *Journal of Dairy Science*, 98(1):500–516, 2015, DOI: https://doi.org/https://doi.org/10.3168/jds.2014-8432.
- [473] Thames, C., Pruden, A., James, R., Ray, P., and Knowlton, K. Excretion of antibiotic resistance genes by dairy calves fed milk replacers with varying doses of antibiotics. *Frontiers* in Microbiology, 3:139, 2012, DOI: https://doi.org/10.3389/fmicb.2012.00139.
- [474] Qian, X., Gunturu, S., Guo, J., Chai, B., Cole, J. R., Gu, J., and Tiedje, J. M. Metagenomic analysis reveals the shared and distinct features of the soil resistome across tundra, temperate prairie, and tropical ecosystems. *Microbiome*, 9(1):108, 2021, DOI: https://doi.org/10.1186/ s40168-021-01047-4.
- [475] Cytryn, E. The soil resistome: The anthropogenic, the native, and the unknown. Soil Biology and Biochemistry, 63:18–23, 2013, DOI: https://doi.org/https://doi.org/10.1016/j. soilbio.2013.03.017.

- [476] Fierer, N. Embracing the unknown: disentangling the complexities of the soil microbiome. *Nature Reviews Microbiology*, 15(10):579–590, 2017, DOI: https://doi.org/10.1038/nrmicro. 2017.87.
- [477] Chen, C., Pankow, C. A., Oh, M., Heath, L. S., Zhang, L., Du, P., Xia, K., and Pruden, A. Effect of antibiotic use and composting on antibiotic resistance gene abundance and resistome risks of soils receiving manure-derived amendments. *Environment International*, 128:233–243, 2019, DOI: https://doi.org/https://doi.org/10.1016/j.envint.2019.04.043.
- [478] Liu, J., Yu, F., Call, D. R., Mills, D. A., Zhang, A., and Zhao, Z. On-farm soil resistome is modified after treating dairy calves with the antibiotic florfenicol. *Science of the Total Environment*, 750, 1 2021, DOI: https://doi.org/10.1016/j.scitotenv.2020.141694.
- [479] Codjoe, F. S. and Donkor, E. S. Carbapenem resistance: A review. Medical Sciences (Basel, Switzerland), 6(1):1, 12 2017, DOI: https://doi.org/10.3390/medsci6010001.
- [480] Hiramatsu, K., Ito, T., Tsubakishita, S., Sasaki, T., Takeuchi, F., Morimoto, Y., Katayama, Y., Matsuo, M., Kuwahara-Arai, K., Hishinuma, T., and Baba, T. Genomic basis for methicillin resistance in *Staphylococcus aureus*. *Infection & Chemotherapy*, 45(2):117–136, 6 2013, DOI: https://doi.org/10.3947/ic.2013.45.2.117.
- [481] Sebastián, D., S., E. J., W., A. F., and Lifeng, Z. Identification and removal of potential contaminants in 16S rRNA gene sequence data sets from low-microbial-biomass samples: an example from mosquito tissues. *mSphere*, 6(3):00506–21, 1 2022, DOI: https://doi.org/10. 1128/mSphere.00506-21.
- [482] Pereira-Marques, J., Hout, A., Ferreira, R. M., Weber, M., Pinto-Ribeiro, I., van Doorn, L.-J., Knetsch, C. W., and Figueiredo, C. Impact of host DNA and sequencing depth on the taxonomic resolution of whole metagenome sequencing for microbiome analysis. *Frontiers in Microbiology*, 10:1277, 2019, DOI: https://doi.org/10.3389/fmicb.2019.01277.
- [483] Adler, A., Sturlesi, N., Fallach, N., Zilberman-Barzilai, D., Hussein, O., Blum, S. E., Klement, E., Schwaber, M. J., Carmeli, Y., and W., F. B. Prevalence, risk factors, and transmission dynamics of extended-spectrum-β-lactamase-producing Enterobacteriaceae: a national Survey of cattle farms in Israel in 2013. Journal of Clinical Microbiology, 53(11):3515–3521, 11 2015, DOI: https://doi.org/10.1128/JCM.01915-15.
- [484] Munk, P., Knudsen, B. E., Lukjancenko, O., Duarte, A. S. R., Van Gompel, L., Luiken, R. E. C., Smit, L. A. M., Schmitt, H., Garcia, A. D., Hansen, R. B., Petersen, T. N., Bossers, A., Ruppé, E., Lund, O., Hald, T., Pamp, S. J., Vigre, H., Heederik, D., Wagenaar, J. A., Mevius, D., and Aarestrup, F. M. Abundance and diversity of the faecal resistome in slaughter pigs and broilers in nine European countries. *Nature Microbiology*, 3(8):898–908, 8 2018, DOI: https://doi.org/10.1038/s41564-018-0192-9.
- [485] Lima, T., Domingues, S., and Da Silva, G. J. Manure as a potential hotspot for antibiotic resistance dissemination by horizontal gene transfer events. *Veterinary Sciences*, 7(3):110, 8 2020, DOI: https://doi.org/10.3390/vetsci7030110.
- [486] The Institute of Environmental Science and Research. Covid-19 wastewater testing results, https://www.esr.cri.nz/our-expertise/covid-19-response/wastewater-testing-results/.

- [487] Plassard, V., Gisbert, P., Granier, S. A., and Millemann, Y. Surveillance of extendedspectrum β-lactamase-, cephalosporinase- and carbapenemase-producing gram-negative bacteria in raw milk filters and healthy dairy cattle in three farms in Île-de-France, France. Frontiers in Veterinary Science, 8:41, 2021, DOI: https://doi.org/10.3389/fvets.2021.633598.
- [488] Oliver, D. M. and Page, T. Effects of seasonal meteorological variables on *E. coli* persistence in livestock faeces and implications for environmental and human health. *Scientific Reports*, 6(1):37101, 2016, DOI: https://doi.org/10.1038/srep37101.
- [489] Rothman, K. J., Gallacher, J. E. J., and Hatch, E. E. Why representativeness should be avoided. *International Journal of Epidemiology*, 42(4):1012–1014, 8 2013, DOI: https: //doi.org/10.1093/ije/dys223.
- [490] Durazzi, F., Sala, C., Castellani, G., Manfreda, G., Remondini, D., and De Cesare, A. Comparison between 16S rRNA and shotgun sequencing data for the taxonomic characterization of the gut microbiota. *Scientific Reports*, 11(1):3030, 2021, DOI: https://doi.org/10.1038/ s41598-021-82726-y.
- [491] Callahan, B. J., McMurdie, P. J., and Holmes, S. P. Exact sequence variants should replace operational taxonomic units in marker-gene data analysis. *The ISME Journal*, 11(12):2639– 2643, 2017, DOI: https://doi.org/10.1038/ismej.2017.119.
- [492] Alcock, B. P., Raphenya, A. R., Lau, T. T. Y., et al. CARD 2020: antibiotic resistome surveillance with the comprehensive antibiotic resistance database. *Nucleic Acids Research*, 48(D1):D517–D525, 1 2020, DOI: https://doi.org/10.1093/nar/gkz935.
- [493] Stewart, R. D., Auffret, M. D., Warr, A., Walker, A. W., Roehe, R., and Watson, M. Compendium of 4,941 rumen metagenome-assembled genomes for rumen microbiome biology and enzyme discovery. *Nature Biotechnology*, 37(8):953–961, 2019, DOI: https://doi.org/10. 1038/s41587-019-0202-3.
- [494] Foutz, C. A., Godden, S. M., Bender, J. B., Diez-Gonzalez, F., Akhtar, M., and Vatulin, A. Exposure to antimicrobials through the milk diet or systemic therapy is associated with a transient increase in antimicrobial resistance in fecal *Escherichia coli* of dairy calves. *Journal* of Dairy Science, 101(11):10126–10141, 2018, DOI: https://doi.org/https://doi.org/10.3168/ jds.2018-14598.
- [495] Stalder, T., Press, M. O., Sullivan, S., Liachko, I., and Top, E. M. Linking the resistome and plasmidome to the microbiome. *The ISME Journal*, 13(10):2437–2446, 2019, DOI: https://doi.org/10.1038/s41396-019-0446-4.
- [496] Veldman, K., van Tulden, P., Kant, A., Testerink, J., and Mevius, D. Characteristics of cefotaxime-resistant *Escherichia coli* from wild birds in the Netherlands. *Applied and En*vironmental Microbiology, 79(24):7556-7561, 12 2013, DOI: https://doi.org/10.1128/AEM. 01880-13.
- [497] Alcalá, L., Alonso, C. A., Simón, C., González-Esteban, C., Orós, J., Rezusta, A., Ortega, C., and Torres, C. Wild birds, frequent carriers of extended-spectrum β-lactamase (ESBL) producing *Escherichia coli* of CTX-M and SHV-12 types. *Microbial Ecology*, 72(4):861–869, 2016, DOI: https://doi.org/10.1007/s00248-015-0718-0.

- [498] Alonso, C. A., Alcalá, L., Simón, C., and Torres, C. Novel sequence types of extendedspectrum and acquired AmpC beta-lactamase producing *Escherichia coli* and *Escherichia* clade V isolated from wild mammals. *FEMS Microbiology Ecology*, 93(8):fix097, 8 2017, DOI: https://doi.org/10.1093/femsec/fix097.
- [499] Darwich, L., Vidal, A., Seminati, C., Albamonte, A., Casado, A., López, F., Molina-López, R. A., and Migura-Garcia, L. High prevalence and diversity of extended-spectrum βlactamase and emergence of OXA-48 producing *Enterobacterales* in wildlife in Catalonia. *PLOS One*, 14(8):e0210686–e0210686, 8 2019, DOI: https://doi.org/10.1371/journal.pone. 0210686.
- [500] Höfle, U., Jose Gonzalez-Lopez, J., Camacho, M. C., Solà-Ginés, M., Moreno-Mingorance, A., Manuel Hernández, J., De La Puente, J., Pineda-Pampliega, J., Aguirre, J. I., Torres-Medina, F., Ramis, A., Majó, N., Blas, J., and Migura-Garcia, L. Foraging at solid urban waste disposal sites as risk factor for cephalosporin and colistin resistant *Escherichia coli* carriage in white storks (*Ciconia ciconia*). *Frontiers in Microbiology*, 11:1397, 2020, DOI: https://doi.org/10.3389/fmicb.2020.01397.
- [501] World Health Organization, Food and Agriculture Organization, and World Organisation for Animal Health. WHO, FAO, and OIE unite in the fight against antimicrobial resistance, https://www.who.int/foodsafety/areas\_work/antimicrobial-resistance/ amr\_tripartite\_flyer.pdf.
- [502] Ministry of Health and Ministry for Primary Industries. New Zealand Antimicrobial Resistance Action Plan. Report, Ministry of Health, Wellington, New Zealand., 2017, https://www.mpi.govt.nz/dmsdocument/19391-NZ-Antimicrobial-Resistance-Action-Plan.

### Appendix A

# PCR reaction and gel electrophoresis conditions

PCR reaction	Primer set(s)	Reaction conditions <sup>a</sup>	Gel conditions <sup>b</sup>	Controls	Ref
ampC promoter region	$\mathrm{AmpC1^{71}/AmpC2^{120}}$	Initial denaturation at 94°C for 90 sec, then 30 cycles of 94°C for 30 sec, 57°C for 30 sec, and 72°C for 1 min, with a final ex- tension step at 72°C for 10 min.	2% agarose gel run at 90V for 40 min.	LT9031a (pos- itive <i>E. coli</i> ), FS192 (negative <i>Staphylococcus</i> <i>aureus</i> ) and non-template control.	[138]
pAmpC	MOXMF/MOXMR CITMF/CITMR DHAMF/DHAMR ACCMF/ACCMR FOXMF/FOXMR CMY-F/CMY-R	Initial denaturation at 94°C for 3 min, then 25 cycles of 94°C for 30 sec, 64°C for 30 sec, and 72°C for 1 min, with a final ex- tension step at 72°C for 7 min.	2% agarose gel run at 80V for 90 min.	$\begin{array}{c} \text{NZRM4402} \\ (\text{CMY-positive} \\ E. coli), \\ \text{NZRM4464} \\ (\text{FOX-5-} \\ \text{positive} K. \\ pneumoniae), \\ \text{NZRM4403} \\ (\text{DHA-1-positive} \\ E. coli), \\ \text{ATCC25922} \\ (\text{negative} E. \\ coli) \\ \text{and} \\ \text{non-template} \\ \text{control.} \end{array}$	[177]
CMY group	CMY-2-F/CMY-2-R	Initial denaturation at 94°C for 3 min, then 30 cycles of 94°C for 30 sec, 58°C for 30 sec, and 72°C for 1 min, with a final ex- tension step at 72°C for 10 min.	2% agarose gel run at 80V for 40 min.	NZRM4402 (CMY-positive <i>E. coli</i> ), ATCC25922 (negative <i>E. coli</i> ) and and non-template control.	[313]
CTX-M-1	CTX-1-SEQ- F/CTX-1-SEQ-R	Initial denaturation at 95°C for 5 min, then 30 cycles of 95°C for 30 sec, 60°C for 30 sec, and 72°C for 1 min, with a final extension step at 72°C for 1 min.	0.8% agarose gel run at 80V for 40 min.	DF0183g (CTX- M-15 positive <i>E.</i> <i>coli</i> ), DF0025d (CTX-M nega- tive <i>E. coli</i> ) and non-template control.	[313]

PCR reaction	Primer set(s)	Reaction conditions <sup>a</sup>	${f Gel}\ {f conditions^b}$	Controls	Ref
Clermont quadruplex PCR typing method	chuA.1b/chuA.2 yjaA.1b/yjaA.2b TspE4C2.1b/ TspE4C2.2b AceK.f/ArpA1.r	The PCR was under- taken in a Bio-Rad T100 Thermal Cycler (Bio-Rad, USA) using KAPA HiFi HotStart ReadyMix (KAPA BioSystems, Wilming- ton, USA) under the following conditions: initial denaturation at 94°C for 4 min, then 30 cycles of 98°C for 20 sec, 61°C for 20 sec, and 72°C for 5 min.	2% agarose gel run at 80V for 90 min.	AGR3560 (Clade III, IV or V positive <i>E.</i> <i>coli</i> ), AGR3575 (mixed <i>E. coli</i> DNA posi- tive for <i>arpA</i> , <i>chuA</i> , <i>yjaA</i> and TspE4C2 PCR products) and non-template control.	[76, 75]
Clermont typing method: Group C	trpAgpC.1/ tr- pAgpC.2	PCR was undertaken in a Bio-Rad T100 Ther- mal Cycler (Bio-Rad, USA) using KAPA HiFi HotStart ReadyMix (KAPA BioSystems, Wilmington, USA) under the following conditions: initial de- naturation at 94°C for 4 min, then 30 cycles of 98°C for 20 sec, 58°C for 20 sec, and 72°C for 5 min.	2% agarose gel run at 80V for 90 min.	AGR4288 ( <i>E.</i> coli phylogroup C positive control) and non-template control.	[318]
Clermont typing method: Group E	ArpAgpE.f/ ArpAgpE.r	PCR was undertaken in a Bio-Rad T100 Ther- mal Cycler (Bio-Rad, USA) using KAPA HiFi HotStart ReadyMix (KAPA BioSystems, Wilmington, USA) under the following conditions: initial de- naturation at 94°C for 4 min, then 30 cycles of 98°C for 20 sec, 58°C for 20 sec, and 72°C for 5 min.	2% agarose gel run at 80V for 90 min.	E. coli O157 (phylogroup E positive control) and non-template control.	[318]

<sup>a</sup>PCR was undertaken in a SensoQuest Lightcycler machine (SensoQuest, Germany) using HOT FIREPol<sup>®</sup> Blend Master Mix (10mM Mg) (Solis BioDyne, Tartu, Estonia) unless stated otherwise. <sup>b</sup>PCR products (2 µL) were mixed with 1 µL gel loading buffer (BlueJuice<sup>TM</sup>, ThermoFisher Scientific Inc., USA). Electrophoresis agarose gels were stained with RedSafe (Invitrogen, Auckland, New Zealand) and a 1kb+ ladder (Invitrogen, Auckland, New Zealand) (1.5 µL) was electrophoresed as a size standard. Gel images were captured and stored using a GelDoc XR+ (BioRad, New Zealand).

# Appendix B

# Bacterial strains isolated in this study

											_	_											
$\beta$ -lactam resistance genes <sup>c</sup>	ND	ND	ND	ND	ND	ND	ND	ND	bla <sub>EC-13</sub> , bla <sub>TEM-1</sub>	ND	ND	ND	$bla_{ m EC-13}$	ND	ND	ND	ND	$bla_{\rm EC}, \ bla_{\rm CMY-2}$	blaEC, blaCMY-2, blaCTX-M-1	ND	ND	ND	$bla_{\rm EC-8}, \ bla_{\rm OXA-1}$
WGS	No	No	No	No	No	No	No	No	$\mathbf{Y}_{\mathbf{es}}$	No	No	No	$\mathbf{Y}_{\mathbf{es}}$	No	No	No	No	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	No	No	No	$\mathbf{Y}_{\mathbf{es}}$
Phenotype <sup>b</sup>	cAmpC	cAmpC	cAmpC	cAmpC	cAmpC	cAmpC	cAmpC	cAmpC	cAmpC	cAmpC	cAmpC	cAmpC	cAmpC	cAmpC	cAmpC	cAmpC	pAmpC	pAmpC	pAmpC/ESBL	cAmpC	Unknown	Unknown	Unknown
Phylogroup	C	C	C	C	C	C	C	C	C	C	C	C	C	C	C	C	É	E	D	C	D	D	D
Isolation agar <sup>a</sup>	MCAZ	MCAZ	MCTX	MCTX	MCAZ	MCAZ	MCTX	MCTX	MCAZ	MCAZ	MCTX	MCTX	MCAZ	MCAZ	MCTX	MCTX	MCAZ	MCTX	MCTX	MCAZ	MCTX	MCTX	CHRO
Collection date	05/11/2018	05/11/2018	05/11/2018	05/11/2018	03/12/2018	03/12/2018	03/12/2018	03/12/2018	03/12/2018	03/12/2018	03/12/2018	03/12/2018	29/01/2019	29/01/2019	29/01/2019	29/01/2019	29/01/2019	29/01/2019	04/02/2019	04/02/2019	06/05/2019	06/05/2019	06/05/2019
$\operatorname{Farm}$													1	1				<del>, _</del>	4		1		
Source	FDE	FDE	FDE	FDE	FDE	FDE	FDE	FDE	Faeces	Faeces	Faeces	Faeces	FDE	FDE	FDE	FDE	Faeces	Faeces	Faeces	FDE	Faeces	Faeces	Faeces
Strain	$\rm DF0025c$	DF0025d	DF0025e	DF0025f	$\rm DF0030c$	DF0030d	DF0030e	DF0030f	DF0031.1c	DF0031.1d	DF0031.1e	DF0031.1f	DF0047c	DF0047d	DF0047e	DF0047f	DF0049.2c	DF0049.2e	DF0059.2e	$\rm DF0060c$	DF0102.4e	DF0102.4f	$\mathrm{DF0102.4g}$

$\beta$ -lactam resistance genes <sup>c</sup>	ND	ND	ND	bla <sub>EC-18</sub> , bla <sub>CMY-2</sub> , bla <sub>OXA-1</sub>	ND	ND	blaEC-18, blaCMY-2	ND	ND	ND	ND	ND	bla <sub>EC-18</sub> , bla <sub>CMY-2</sub>	ND	ND	ND	ND	ND	ND	$bla_{\rm EC}, \ bla_{\rm CMY-2}$	ND	ND	ND	ND	ND	$bla_{ m EC-18}, \ bla_{ m OMY-2}$	ND	blaEC-19, blaCTX-M-15	bla <sub>EC-18</sub> , bla <sub>CMY-2</sub>
WGS	No	$N_0$	$N_{O}$	$\mathbf{Yes}$	$N_{O}$	$N_0$	Yes	$N_0$	$N_{O}$	$N_0$	$N_{O}$	$N_0$	Yes	$N_0$	$N_0$	$N_0$	$N_0$	$N_0$	No	$\mathbf{Yes}$	No	$N_0$	$N_0$	$N_0$	$N_{O}$	$Y_{es}$	$N_{O}$	$\mathbf{Yes}$	Yes
$\mathbf{Phenotype}^{\mathbf{b}}$	Unknown	pAmpC	pAmpC	pAmpC	pAmpC	$_{ m cAmpC}$	pAmpC	pAmpC	pAmpC	pAmpC	pAmpC	pAmpC	pAmpC	pAmpC	pAmpC	pAmpC	pAmpC	pAmpC	pAmpC	pAmpC	pAmpC	pAmpC	pAmpC	ESBL	pAmpC	pAmpC	pAmpC	ESBL	pAmpC
Phylogroup	D	Α	Α	А	Α	B1	B1	B1	B1	B1	B1	B1	B1	B1	B1	B1	B1	B1	B1	Э	Э	ы	Э	Гц	B1	B1	B1	Ĺ	B1
Isolation agar <sup>a</sup>	CHRO	MCAZ	MCTX	CHRO	CHRO	MCAZ	MCAZ	MCAZ	MCTX	MCTX	MCAZ	MCTX	MCAZ	MCAZ	MCTX	MCTX	MCAZ	MCTX	MCTX	MCAZ	MCAZ	MCTX	MCTX	MCAZ	MCAZ	MCTX	MCTX	CHRO	MCAZ
Collection date	06/05/2019	07/10/2019	07/10/2019	07/10/2019	07/10/2019	07/10/2019	02/12/2019	02/12/2019	02/12/2019	02/12/2019	02/12/2019	02/12/2019	02/12/2019	02/12/2019	02/12/2019	02/12/2019	02/12/2019	02/12/2019	02/12/2019	02/12/2019	02/12/2019	02/12/2019	02/12/2019	02/12/2019	02/12/2019	02/12/2019	02/12/2019	02/12/2019	02/12/2019
Farm	1	4	4	4	4	1	4	4	4	4	4	4	4	4	4	4	4	4	4	4	4	4	4	4	4	4	4	4	4
Source	Faeces	Faeces	Faeces	Faeces	Faeces	FDE	Faeces	Faeces	Faeces	Faeces	Faeces	Faeces	Faeces	Faeces	Faeces	Faeces	Faeces	Faeces	Faeces	Faeces	Faeces	Faeces	Faeces	FDE	FDE	FDE	FDE	FDE	FDE
Strain	DF0102.4h	DF0159.2d	DF0159.2e	$\mathrm{DF0159.2g}$	DF0159.2h	DF0166c	DF0181.1c	DF0181.1d	DF0181.1e	DF0181.1f	DF0181.1j	DF0181.1k	DF0181.3c	DF0181.3d	DF0181.3e	DF0181.3f	DF0181.3i	DF0181.3j	DF0181.3k	DF0181.4c	DF0181.4d	DF0181.4e	DF0181.4f	DF0183c	DF0183d	$\rm DF0183e$	DF0183f	DF0183g	DF0183i

<sup>a</sup> MCAZ, MacConkey (MC) agar + 1 µg/ml ceftazidime; MCTX, MC agar + 1 µg/ml cefotaxime; CHRO, CHROMagar<sup>TM</sup> ESBL.

<sup>b</sup> cAmpC, Putative Ampc hyperproducers; pAmpC, Plasmid-mediated AmpC phenotype (CMY-2); ESBL, ESBL phenotype. <sup>c</sup> ND, Not done ( $\beta$ -lactam resistance genes identified only from isolates that were sequenced in this study).

## Appendix C

# E. coli AST zone sizes (mm)

Strain	CTX <sup>a</sup>	CPD	FOX	TET	STR	CIP
DF0025c	24.6	7.2	12.9	6.6	6.6	33.5
DF0025d	24.0	6.6	13.2	6.3	6.6	31.4
DF0025e	24.7	7.0	13.5	6.6	6.6	32.2
DF0025f	23.8	7.2	13.5	6.6	6.6	32.1
DF0030c	26.1	7.2	13.5	6.9	6.6	32.2
DF0030d	23.7	7.3	13	6.6	6.6	31.2
DF0030e	24.7	7.0	12.3	6.6	6.6	31.8
DF0030f	24.2	6.6	12.1	6.5	6.6	34.7
DF0031.1c	24.9	7.4	13.3	6.6	6.6	30.8
DF0031.1d	24.5	6.8	12.7	6.6	6.6	30.9
DF0031.1e	25.9	6.66	13.4	6.4	6.6	33.7
DF0031.1f	25.6	6.2	13.5	6.5	6.6	32.3
DF0047c	24.8	6.6	13.6	6.3	6.6	30.1
DF0047d	26.4	6.6	13.9	6.4	6.6	32.8
DF0047e	26.1	7.3	13.2	6.6	6.6	32.0
DF0047f	23.7	6.3	12.4	6.4	6.6	31.5
DF0049.2c	16.7	6.6	6.9	23.6	17.0	32.1
DF0049.2e	15.9	6.6	7.3	23.2	16.5	31.9
DF0059.2e	9.9	6.6	6.6	6.6	6.6	32.5
DF0060c	25.6	6.6	10.8	6.5	6.6	32.6
DF0102.4e	17.9	12.4	16.0	22.7	6.6	31.7
DF0102.4f	19.1	12.8	16.8	22.4	6.5	33.6
DF0102.4g	18.4	12.1	16.3	22.7	6.6	33.7
DF0102.4h	17.9	12.1	16.1	21.6	6.3	32.3
DF0159.2d	15.7	6.6	7.7	23.1	7.8	29.3
DF0159.2e	15.6	6.6	7.1	25.0	8.2	30.4
DF0159.2g	16.0	6.6	8.4	23.0	8.6	32.6
DF0159.2h	13.8	6.6	6.9	21.4	7.6	29.4
DF0166c	23.9	6.9	12.8	24.0	18.1	36.0
DF0181.1c	18.4	6.6	11.5	6.6	6.4	33.3
DF0181.1d	16.3	6.6	7.7	21.8	17.4	32.7
DF0181.1e	19.5	6.6	12.4	6.6	7.1	33.5
DF0181.1f	18.9	6.6	12.2	6.6	7.2	35.2
DF0181.1j	17.1	6.6	11.5	6.6	6.9	36.1
DF0181.1k	17.1	6.6	11.6	6.6	6.5	34.4
DF0181.3c	17.2	6.6	11.3	6.6	6.7	35.8
DF0181.3d	17.8	6.6	12.3	6.6	7.7	35.8
DF0181.3e	16.3	6.6	10.4	6.6	7.1	34.9
DF0181.3f	18.9	6.6	11.6	6.6	7.6	35.6
DF0181.3i	16.8	6.6	10.6	6.6	6.7	34.7
DF0181.3j	17.6	6.6	10.6	6.6	7.5	35.1
DF0181.3k	16.7	6.6	11.2	6.6	7.1	34.0
DF0181.4c	19.0	6.8	10.2	25.1	17.7	34.0
DF0181.4d	16.1	6.6	9.7	22.2	17.3	32.8
DF0181.4e	18.3	6.6	9.0	23.9	17.0	32.3

Strain	CTX <sup>a</sup>	CPD	FOX	TET	STR	CIP
DF0181.4f	16.4	6.6	9.2	23.0	16.3	32.4
DF0183c	7.4	6.6	17.9	22.5	16.3	29.2
DF0183d	17.8	6.6	11.2	6.6	8.2	36.5
DF0183e	17.3	6.6	11.9	6.6	7.0	36.0
DF0183f	19.6	6.6	12.4	6.6	7.5	34.5
DF0183g	8.1	6.6	18.0	22.3	16.3	30.3
DF0183i	16.2	6.6	6.7	21.4	16.6	30.5

<sup>a</sup> CTX, cefotaxime (30µg); CPD, cefpodoxime (10µg); FOX, cefoxitin (30µg); TET, tetracycline (30µg); STR, streptomycin (10µg); CIP, ciprofloxacin (5µg). Zone sizes in mm.

### Appendix D

# AmpC-producing $E. \ coli$ AST zone sizes for selected antibiotics

Isolate metadata and antimicrobial susceptibility testing zone size (mm) for cefotaxime, cefpodoxime and cefoxitin.

Isolate	$\mathrm{CTX}^{\mathrm{a}}$	CPD	FOX	AmpC	Phylogroup	Phenotype
DF0025c	24.6	7.2	12.9	Chrom	С	Intermediate
DF0030c	26.1	7.2	13.5	Chrom	С	Intermediate
DF0031.1c	24.9	7.4	13.3	Chrom	С	Intermediate
DF0047c	24.8	6.6	13.6	Chrom	С	Intermediate
DF0049.2c	16.7	6.6	6.9	CMY-2	Е	Resistant
DF0059.2e	9.9	6.6	6.2	CMY-2	D	Resistant
DF0060c	25.6	6.6	10.8	Chrom	С	Intermediate
DF0159.2d	15.7	6.6	7.7	CMY-2	А	Resistant
DF0166c	23.9	6.9	12.8	Chrom	B1	Intermediate
DF0181.1c	18.4	6.6	11.5	CMY-2	B1	Resistant
DF0181.1d	16.3	6.6	7.7	CMY-2	B1	Resistant
DF0181.3c	17.2	6.6	11.3	CMY-2	B1	Resistant
DF0181.4c	19.0	6.8	10.2	CMY-2	Е	Resistant
DF0183d	17.8	6.6	11.2	CMY-2	B1	Resistant
DF0183i	16.2	6.6	6.7	CMY-2	B1	Resistant

<sup>a</sup> CTX, cefotaxime (30µg); FOX, cefoxitin (30µg); CPD, cefpodoxime (10µg).

### Appendix E

# Antimicrobial resistance genes identified in $E. \ coli$ genome sequences

Gene	AMR gene family	Drug class	Resistance mechanism
aac(3)- $IId$	AAC(3)	Aminoglycoside	Antibiotic inactivation
aadA1	ANT(3")	Aminoglycoside	Antibiotic inactivation
aadA5	ANT(3")	Aminoglycoside	Antibiotic inactivation
aph(3')-Ia	APH(3')	Aminoglycoside	Antibiotic inactivation
aph(3")-Ib	APH(3")	Aminoglycoside	Antibiotic inactivation
aph(6)-Id	APH(6)	Aminoglycoside	Antibiotic inactivation
catA1	Chloramphenicol acetyl-	Phenicol	Antibiotic inactivation
	transferase $(CAT)$		
dfrA17	Trimethoprim resistant dihy-	Diaminopyrimidine	Antibiotic target replacement
	drofolate reductase dfr		
dfrA5	Trimethoprim resistant dihy-	Diaminopyrimidine	Antibiotic target replacement
	drofolate reductase dfr		
mph(A)	Macrolide phosphotrans-	Macrolide	Antibiotic inactivation
	ferase (MPH)		
sul1	Sulfonamide resistant sul	Sulfonamide	Antibiotic target replacement
sul 2	Sulfonamide resistant sul	Sulfonamide	Antibiotic target replacement
tet(A)	Major facilitator superfamily	Tetracycline	Antibiotic efflux
	(MFS) antibiotic efflux pump		
tet(B)	Major facilitator superfamily	Tetracycline	Antibiotic efflux
	(MFS) antibiotic efflux pump		

# Appendix F

# Complete circular plasmid pDF0059.2e\_3





Complete circular plasmid pDF0059.2e\_3 assembled from Nanopore and Illumina data showing cluster of orthologous groups on the outer circle and physical linkages of ARGs and mobile elements on the inner circle.





Appendix G

# Appendix H

# Details of the SNP annotations (n=7) identified in the core genome of five IncI1 plasmids

	-00031		-00047		-00077		_00078				_00078		$_{-00095}$		$_{-00095}$		
Locus tag	IPPFNOJB		IPPFNOJB		IPPFNOJB		IPPFNOJB				IPPFNOJB		IPPFNOJB		IPPFNOJB		
Effect	Synonymous variant	c.513T>C p.Thr171Thr	Missense variant c.76G>C	p.Gly26Arg	Start lost c.2_5 del TGCC	ins GGCA p.MetProl	Missense variant c.537_546	del AGACTGGAAA ins	CGACGGTAAT	p.TrpLys181GlyAsn	Missense variant c.531G>T	p.Lys177Asn	Missense variant c.1152G>T	p.Gln384His	Missense variant c.595T>G	p.Ser199Ala	
$\mathbf{Product}$	Hypothetical	protein	Methyltransferase		DNA primase		Conjugal	transfer protein			Conjugal	transfer protein	Pilus assembly	protein PilO	Pilus assembly	protein PilO	DF0181.1c 1: 183e.
183e	Ĺ		G		IJ		H				C		C		А		81.1c. n
181.1c	T		G		IJ		H				C		C		A		.2g 1: 1
159.2g	C		C		IJ		H				C		C		C		pDF0159
59.2e	T		C		H		A				А		C		А		159.2g.
$49.2e^{a}$	C		C		IJ		Ð				C		А		C		F0059.2e 1:
$\operatorname{Ref}$	Ð		G		IJ		Ð				C		C		Α		9.2e. pD
Position	22701		34544		66791		66891				66906		82769		83326		049.2e 1: 26
Reference	DF0183e		DF0183e		DF0183e		DF0183e				DF0183e		DF0183e		DF0183e		<sup>a</sup> 49.2e. pDF0.

pDF0183e\_1.



# Gene synteny comparison of five IncI1 plasmids



Gene synteny comparison of five IncI1 plasmids carrying the bla<sub>CMY-2</sub> gene.



# Surrounding genetic region of the Shufflon A coding sequence



Surrounding genetic regions of the Shufflon A coding sequence in the five E. coli IncI1 plasmids. The cluster of orthologous groups are colour coded as indicated in the figure key.

## Appendix K

# Antimicrobial use on Dairy 1

Amount (mg) of antimicrobial products used on Dairy 1 between October 2018 and December 2019.

Antimicrobial product	Amount (mg)	Administration route <sup>a</sup>	Active ingredi- ent	Class
Betamox LA	6,000	Parenteral other	Amoxycillin	Aminopenicillins
Bivatop <sup>®</sup> 200	20,000	LCT par	Oxytetracycline	Tetracyclines
Bomacure	20,000	Parenteral other	Cephapirin	First-generation
				cephalosporins
Metri-Clean <sup>TM</sup> or	7,000	Parenteral other	Cephapirin	First-generation
Metricure <sup>®</sup>				cephalosporins
Penethaject	966,570	Parenteral other	Penethamate	Penicillins
Vibrostrep <sup>TM</sup>	80,000	Parenteral other	Streptomycin	Aminoglycosides
Orbenin DC	12,000	DCT	Cloxacillin	Penicillins
Intracillin <sup>®</sup> 1000	336,000	Parenteral other	Procaine peni-	Penicillins
Milking Cow			cillin G	
Excede LA	14,000	LCT par	Ceftiofur	Third-generation
				cephalosporins
Intracillin <sup>®</sup> 300	495,000	Other	Procaine peni-	Penicillins
			cillin G	
Marbocyl 10%	1,200	Other	Marbofloxacin	Quinolones
Phoenix Pharma-	22,500	Other	Procaine peni-	Penicillins
cillin 300			cillin G	
Forcyl	40,320	Other	Marbofloxacin	Quinolones
Total	2,020,590			

<sup>a</sup>DCT, dry cow therapy; LCT imam, lactating cow therapy intramammary; LCT par, lactating cow therapy parenteral.

Amount	(mg)	of	antimicrobia	l per	class	used	on	Dairy	1
--------	------	----	--------------	-------	-------	------	----	-------	---

Class	Amount (mg)	$PCU^{a}$	% of total
Aminoglycoside	80,000	0.68	3.96
Aminopenicillin	6,000	0.05	0.30
First-generation cephalosporin	27,000	0.23	1.34
Third-generation cephalosporin	14,000	0.12	0.69
Penicillin	1,832,070	15.50	90.67
Quinolone	41,520	0.35	2.05
Tetracycline	20,000	0.17	0.99
Total	2,020,590	17.09	100.00

<sup>a</sup>PCU, Population correction unit.

Date	Amount (mg)	PCU <sup>a</sup>
October 2018	248,782	2.10
November 2018	93,494	0.79
December 2018	58,830	0.50
January 2019	78,496	0.66
February 2019	97,492	0.82
March 2019	127,996	1.08
April 2019	30,500	0.26
May 2019	21,698	0.18
June 2019	48,996	0.41
July 2019	50,898	0.43
August 2019	376,648	3.19
September 2019	362,268	3.06
October 2019	175,656	1.49
November 2019	183,840	1.55
December 2019	64,996	0.55
Total	2,020,590	17.09

### Antimicrobial use per month on Dairy 1

 $^{\rm a}{\rm PCU},$  Population correction unit.
### Appendix L

### Antimicrobial use on Dairy 4

Amount (mg) of antimicrobial products used on Dairy 4 between October 2018 and December 2019.

Antimicrobial	Amount	Administration	Antimicrobial	Class
product	(mg)	$\operatorname{route}^{\mathbf{a}}$	ingredient	
Betamox LA	6,000	Parenteral other	Amoxycillin	Aminopenicillins
Bivatop <sup>®</sup> 200	10,000	LCT par	Oxytetracycline	Tetracyclines
Bomacure	750,000	Parenteral other	Cephapirin	First-generation cephalosporins
Mastalone <sup>®</sup>	49.19	Parenteral other	Oxytetracycline, oleandomycin, neomycin	Multiple classes
Penethaject	43,329	Parenteral other	Penethamate	Penicillins
Tylo 200	5,000	Parenteral other	Tylosin	Macrolides
Vibrostrep <sup>TM</sup>	50,000	Parenteral other	Streptomycin	Aminoglycosides
Orbenin DC	84,000	DCT	Cloxacillin	Penicillins
Intracillin <sup>®</sup> 1000 Milking Cow	1,000	Parenteral other	Procaine peni- cillin G	Penicillins
Penclox 1200 <sup>TM</sup>	72,000	Parenteral other	Penicillin G and cloxacillin	Penicillins
Excede LA	10,600	LCT par	Ceftiofur	Third-generation cephalosporins
Intracillin <sup>®</sup> 300	121,800	Other	Procaine peni- cillin G	Penicillins
Marbocyl 10%	8,300	Other	Marbofloxacin	Quinolones
Phoenix Pharma-	255,000	Other	Procaine peni-	Penicillins
cillin 300			cillin G	
Total	$1,\!417,\!078.19$			

<sup>a</sup>DCT, dry cow the rapy; LCT imam, lactating cow the rapy intramammary; LCT par, lactating cow the rapy parenteral.

Class	Amount (mg)	PCU <sup>a</sup>	% of total
Aminoglycoside	50,000	0.19	3.53
Aminopenicillin	6,000	0.02	0.42
First-generation cephalosporin	750,000	2.83	52.93
Third-generation cephalosporin	10,600	0.04	0.75
Penicillin	577,129	2.18	40.73
Quinolone	8,300	0.03	0.59
Tetracycline	10,000	0.04	0.71
Multiple classes	49.19	0.00	0.00
Macrolide	5,000	0.02	0.35
Total	1,417,078.19	5.36	100.00

### Amount (mg) of antimicrobial per class used on Dairy 4

### Antimicrobial use per month on Dairy 4

Date	Amount (mg)	PCU <sup>a</sup>
November 2018	57,399	0.22
December 2018	2,400	0.01
January 2019	44,100	0.17
February 2019	13,800	0.05
March 2019	11,200	0.04
April 2019	42,000	0.16
May 2019	32,000	0.12
June 2019	57,000	0.22
July 2019	149,800	0.57
August 2019	66,200	0.25
September 2019	804,447.19	3.04
October 2019	56,000	0.21
November 2019	58,632	0.22
December 2019	22,100	0.08
Total	1,417,078.19	5.36

<sup>a</sup>PCU, Population correction unit.

Ν	
ppendix	
A	

# Antimicrobial products used on Dairy 1 and Dairy 4

fication <sup>a</sup> Administration route <sup>b</sup>	acycline), Parenteral other tycin), cin)	LCT par	Parenteral other	Parenteral other	Other			Other					Parenteral other	Parenteral other		Parenteral other	
NZVA classi	Green (oxytetı Red (oleandon Yellow (neomy	$\operatorname{Green}$	Green	Green	Green			Green					Yellow	Yellow		Yellow	
Class	Multiple classes	Tetracyclines	Penicillins	Penicillins	Penicillins			Penicillins					Aminopenicillins	First Generation	Cephalosporins	First-generation	Cephalosporins
Antimicrobial ingredient	Oxytetracycline, Oleandomycin, Neomycin	Oxytetracycline	Penethamate	Procaine penicillin G	Procaine penicillin G			Procaine penicillin G					Amoxycillin	Cephapirin		Cephapirin	
Condition	Mastitis - clinical	Sick - undiagnosed, theileria	Mastitis – clinical or subclinical	Mastitis – clinical or subclinical	Calving disorder - unspecified, white line disease, uterine in-	fection, between claw/footrot,	severe ınjury, retaıned mem- branes, sick - undiagnosed	Between claw/footrot, abomas-	sal disorder - unspecified, pneu-	monia or respiratory disease,	other lameness, calving trouble,	retained membranes	Between claw/footrot	Metri-check (treatment of en-	dometritis)	Metri-check (treatment of en-	dometritis)
Product	Mastalone	$\operatorname{Bivatop}^{\textcircled{B}}$ 200	Penethaject	Intracillin <sup>®</sup> 1000 Milking Cow	Intracillin <sup>®</sup> 300			Phoenix	Pharmacillin 300				Betamox LA	Bomacure		Metri-Clean <sup>TM</sup> or	$Metricure^{(B)}$

Product	Condition	Antimicrobial ingredient	Class	NZVA classification <sup>a</sup>	Administration
					route
Orbenin DC	Mastitis and reduces risk of new	Cloxacillin	Penicillins	Yellow	DCT
	infections during dry period				
$Penclox^{TM}$ 1200	Mastitis – clinical or subclini-	Penicillin G and cloxacillin	Penicillins	$\operatorname{Yellow}$	Parenteral other
	cal, teat injury				
Vibrostrep	Wooden tongue, sick - undiag-	Streptomycin	Aminogly cosides	Yellow	Parenteral other
	nosed				
Excede LA	Pneumonia or respiratory ill-	Ceftiofur	Third-generation	$\operatorname{Red}$	LCT par
	ness, calving disorder		Cephalosporins		
Marbocyl 10%	Pneumonia or respiratory ill-	Marbofloxacin	Quinolones	Red	Other
	ness, mastitis - clinical				
Forcyl	Mastitis - clinical, pneumonia	Marbofloxacin	Quinolones	$\operatorname{Red}$	Other
	or respiratory illness, sick - un-				
	diagnosed				
Tylo $200$	Mastitis - clinical	Tylosin	Macrolides	Red	Parenteral other

# <sup>a</sup>NZVA, New Zealand Veterinary Association. <sup>b</sup>LCT imam, lactating cow therapy intramammary; DCT, dry cow therapy; LCT par, lactating cow therapy parenteral.

### Appendix N

# Antimicrobial use per route of administration

Amount (mg per active ingredient per population correction unit (% total PCU)) of antimicrobials per route of administration on Dairy 1 and Dairy 4 between October 2018 and December 2019.

Route <sup>a</sup>	Dairy 1	Dairy 4
Dry cow therapy	$0.10 \ (0.59)$	0.32(5.93)
LCT parenteral	0.29 (1.70)	0.08 (1.49)
Parenteral other	11.97 (70.00)	3.51 (65.36)
Other	4.73 (27.67)	1.46 (27.18)

<sup>a</sup>LCT, lactating cow the rapy.

### Appendix O Faecal sample collection questionnaire

Faecal sampling from dairy cows receiving system	hic antimicrobial treatment: Rose Collis
Date:	
Farm name:Dairy 1Dairy 4 (please circle)	
Sick cow number: Contro	ol cow number:
Form completed by:	_ Antibiotics administered by:
Illness and treatment:	2. Treatment prescribed: (please circle)
<b>1. Diagnosis:</b> (please circle)	Intracillin Bivatop 200
Calving trouble Between claw/footrot	Phoenix Pharmacillin 300 Vibrostrep
Calving paralysis Retained membranes	Other (please specify)
Sick – undiagnosed Laminitis	3. Dosage prescribed: (please specify)
Other (please specify)	Dosage:
	Length of treatment:
4. What was this diagnosis based on: (please circle)	5. Antimicrobial method of administration: (please
Clinical symptoms Positive culture and sensitivity test	circle): Injectable
Other (please specify)	Other (please specify):
Additional information:	
6. Have any other products, besides those listed in Q2, be 12-months? Please specify:	een used to prevent or treat disease in this cow in the last
7. When was this cow last treated with antibiotics? Pleas	e list the disease and treatment prescribed (antimicrobial
and dose):	
Faecal sample #3: SICK COW Cow #:	Faecal sample #3: CONTROL Cow #:
Date: Sample collected by:	Date: Sample collected by:
Time faecal sample collected:	<sup>X</sup> Time faecal sample collected:
Additional comments:	Additional comments:
Faecal sample #2: SICK COW         Cow #:	Faecal sample #2:         CONTROL         Cow #:
Date: Sample collected by:	Date: Sample collected by:
Time faecal sample collected:	Time faecal sample collected:
Additional comments:	Additional comments:
Eaecal sample #1: SICK COW Cow #	Eaecal sample #1: CONTROL Com #:
Date: Sample collected by:	A Date: Sample collected by:
Time faecal sample collected:	Time faecal sample collected:
Additional comments:	Additional comments:
	X

Faecal sample collection from dairy cows receiving systemic antimicrobial treatment

Appendix P

Sample metadata of faecal samples collected for 16S rRNA V3-V4 amplicon sequencing

Collection	late	$\theta / 08 / 2019$	9/08/2019	9/08/2019	11/08/2019	11/08/2019	(11/08/2019)	11/08/2019	11/08/2019	11/08/2019	13/08/2019	13/08/2019	13/08/2019	16/08/2019	16/08/2019	17/08/2019	17/08/2019	17/08/2019	18/08/2019	18/08/2019	19/08/2019	19/08/2019	19/08/2019	26/08/2019	26/08/2019	1/09/2019	1/09/2019	1/09/2019	1/09/2019	7/09/2019	7/09/2019
s	•	t G	t (	0] [0	t I	t	t I	t		0	t	t	0	0	t	t	t	0	t	0	t	0	t I	t		is		is			1
Illnes		Footrc	Footrc	Contre	Footrc	Footrc	Footrc	Footrc	Contre	Contre	Footrc	Footrc	Contre	Contre	Footrc	Footrc	Footrc	Contre	Footrc	Contre	Footrc	Contre	Footrc	Footrc	Contre	Mastit	Contre	Mastit	Contre	LDA	Contre
Class		$\operatorname{Beta-lactam}$	Beta-lactam	NA	Beta-lactam	Beta-lactam	Beta-lactam	Beta-lactam	NA	NA	Beta-lactam	Beta-lactam	NA	NA	Beta-lactam	Beta-lactam	Beta-lactam	NA	Beta-lactam	NA	Beta-lactam	NA	Beta-lactam	Beta-lactam	NA	Beta-lactam	NA	Beta-lactam	NA	Beta-lactam	NA
Antimicrobial		Procaine penicillin G	Procaine penicillin G	NA	Procaine penicillin G	Procaine penicillin G	Procaine penicillin G	Procaine penicillin G	NA	NA	Procaine penicillin G	Procaine penicillin G	NA	NA	Procaine penicillin G	Procaine penicillin G	Procaine penicillin G	NA	Procaine penicillin G	NA	Procaine penicillin G	NA	Procaine penicillin G	Procaine penicillin G	NA	Penethamate hydrio- dide	NA	Penethamate hydrio- dide	NA	Procaine penicillin G	NA
Treatment		Intracillin	Intracillin	NA	Intracillin	Intracillin	Intracillin	Intracillin	NA	NA	Intracillin	Intracillin	NA	NA	Intracillin	Intracillin	Intracillin	NA	Intracillin	NA	Intracillin	NA	Intracillin	Intracillin	NA	Penethaject	NA	Penethaject	NA	Pharmacillin 300	NA
ple Farm	r	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 1	Dairy 1	Dairy 1	Dairy 1	Dairy 4	Dairy 4
Animal Sam	orde	1 $S1$	2 $S1$	1C/2C S1	2 S2	1 S2	3 S1	4 $S1$	3C/4C S1	1C/2C S2	3 S2	4 S2	3C/4C S2	5C $S1$	5 S1	2 S3	1 S3	1C/2C S3	5 S2	5C S2	4 S3	3C/4C S3	3 S3	5 S3	5C S3	6 S1	6C S1	6 S2	6C S2	7 S1	7C S1
Type		1020 Treated	<sup>1</sup> 021 Treated	1022 Control	1023 Treated	1024 Treated	1025 Treated	1026 Treated	1027 Control	1028 Control	1029 Treated	1030 Treated	1031 Control	<sup>1</sup> 032 Control	1033 Treated	1034 Treated	1035 Treated	1036 Control	1037 Treated	1038 Control	1039 Treated	1040 Control	1041 Treated	1042 Treated	1043 Control	1046 Treated	1047 Control	1048 Treated	1049 Control	1050 Treated	1051 Control
Lab ID		processed_DG	processed_DG	processed_DG	processed_DG	processed_DG	processed_DG	processed_DG	processed_DG	processed_DG	processed_DG	processed_DG	processed_DG	processed_DG	processed_DG	processed_DG	processed_DG	processed_DG	processed_DG	processed_DG	processed_DG	processed_DG	processed_DG	processed_DG	processed_DG	processed_DG	processed_DG	processed_DG	processed_DG	processed_DG	processed_DG

Lab ID Type	Animal	Sam	ole Farm	Treatment	Antimicrobial	Class	Illness	Collection
		order						date
processed_DG052 Treat	ed 7	S2	Dairy 4	Pharmacillin 300	Procaine penicillin G	$\operatorname{Beta-lactam}$	LDA	8/09/2019
processed_DG053 Conti	ol 7C	S2	Dairy 4	NA	NA	NA	Control	8/09/2019
processed_DG054 Treat	ed 8	$\mathbf{S1}$	Dairy 4	Intracillin 300	Procaine penicillin G	Beta-lactam	Footrot	8/09/2019
processed_DG055 Treat	ed 9	$\mathbf{S1}$	Dairy 4	Intracillin 300	Procaine penicillin G	Beta-lactam	Footrot	8/09/2019
processed_DG056 Conti	ol 8C/9C	$\mathbf{S1}$	Dairy 4	NA	NA	NA	Control	8/09/2019
processed_DG057 Treat	ed 9	S2	Dairy 4	Intracillin 300	Procaine penicillin G	Beta-lactam	Footrot	9/09/2019
processed_DG058 Treat	ed 8	S2	Dairy 4	Intracillin 300	Procaine penicillin G	Beta-lactam	Footrot	9/09/2019
processed_DG059 Conti	ol 8C/9C	S2	Dairy 4	NA	NA	NA	Control	9/09/2019
processed_DG060 Treat	ed 6	S3	Dairy 1	Penethaject	Penethamate hydrio- dide	$\operatorname{Beta-lactam}$	Mastitis	13/09/2019
processed_DG061 Conti	ol 6C	S3	Dairy 1	NA	NA	NA	Control	13/09/2019
processed_DG062 Treat	ed 9	S3	Dairy 4	Intracillin 300	Procaine penicillin G	Beta-lactam	Footrot	14/09/2019
processed_DG063 Treat	ed 8	S3	Dairy 4	Intracillin 300	Procaine penicillin G	Beta-lactam	Footrot	14/09/2019
processed_DG064 Conti	ol 8C/9C	$\mathbf{S3}$	Dairy 4	NA	NA	NA	Control	14/09/2019
processed_DG065 Treat	ed 7	S3	Dairy 4	Pharmacillin 300	Procaine penicillin G	Beta-lactam	LDA	16/09/2019
processed_DG066 Conti	ol 7C	S3	Dairy 4	NA	NA	NA	Control	16/09/2019
processed_DG067 Treat	ed 11	$\mathbf{S1}$	Dairy 4	Excede LA	Ceftiofur	Beta-lactam	Caesarean	17/09/2019
processed_DG068 Treat	ed 10	$\mathbf{S1}$	Dairy 4	Intracillin 300	Procaine penicillin G	Beta-lactam	LDA	17/09/2019
processed_DG069 Conti	ol 10C/11C	S1	Dairy 4	NA	NA	NA	Control	17/09/2019
processed_DG070 Treat	ed 12	$\mathbf{S1}$	Dairy 1	Excede LA	Ceftiofur	Beta-lactam	Caesarean	18/09/2019
processed_DG071 Conti	ol 12C	S1	Dairy 1	NA	NA	NA	Control	18/09/2019
processed_DG072 Treat	ed 10	S2	Dairy 4	Intracillin 300	Procaine penicillin G	Beta-lactam	LDA	18/09/2019
processed_DG073 Treat	ed 11	S2	Dairy 4	Excede LA	Ceftiofur	Beta-lactam	$\mathbf{Caesarean}$	18/09/2019
processed_DG074 Conti	ol 10C/11C	S2	Dairy 4	NA	NA	NA	Control	18/09/2019
processed_DG075 Treat	ed 13	$\mathbf{S1}$	Dairy 4	Excede LA	Ceftiofur	Beta-lactam	Caesarean	19/09/2019
processed_DG076 Conti	ol 13C	$\mathbf{S1}$	Dairy 4	NA	NA	NA	Control	19/09/2019
processed_DG077 Treat	ed 13	S2	Dairy 4	Excede LA	Ceftiofur	Beta-lactam	Caesarean	20/09/2019
processed_DG078 Conti	ol 13C	S2	Dairy 4	NA	NA	NA	Control	20/09/2019
processed_DG079 Treat	ed 12	S2	Dairy 1	Excede LA	Ceftiofur	Beta-lactam	Caesarean	20/09/2019
processed_DG080 Conti	ol 12C	S2	Dairy 1	NA	NA	NA	Control	20/09/2019
processed_DG082 Treat	ed 12	S3	Dairy 1	Excede LA	Ceftiofur	Beta-lactam	Caesarean	22/09/2019
processed_DG083 Conti	ol 12C	$\mathbf{S3}$	Dairy 1	NA	NA	NA	Control	22/09/2019

Type	Anima	al Samp	ole Farm	Treatment	Antimicrobial	Class	Illness	Collection
		order						date
34 Treated	13	S3	Dairy 4	Excede LA	Ceftiofur	Beta-lactam	Caesarean	26/09/2019
35 Control	13C	S3	Dairy 4	NA	NA	NA	Control	26/09/2019
86 Treated	10	S3	Dairy 4	Intracillin	Procaine penicillin G	Beta-lactam	LDA	26/09/2019
87 Treated	11	S3	Dairy 4	Excede LA	Ceftiofur	Beta-lactam	Caesarean	26/09/2019
88 Control	10C/11	IC S3	Dairy 4	NA	NA	NA	Control	26/09/2019
89 Treated	14	$\mathbf{S1}$	Dairy 4	Marbocyl	Marbofloxacin/Penetha	Quinolone/Beta-	Mastitis	29/09/2019
				$10\%/{ m Penetha-}$	hydriodide	lactam		
				ject				
090 Control	. 14C	$\mathbf{S1}$	Dairy 4	NA	NA	NA	Control	29/09/2019
091 Treated	14	S2	Dairy 4	Marbocyl	Marbofloxacin/Penetha	Quinolone/Beta-	Mastitis	30/09/2019
				$10\%/\mathrm{Penetha-}$	hydriodide	lactam		
				ject				
3092 Control	. 14C	S2	Dairy 4	NA	NA	NA	Control	30/09/2019
1093 Treated	14	S3	Dairy 4	Marbocyl	Marbofloxacin/Peneths	Quinolone/Beta-	Mastitis	9/10/2019
				$10\%/{ m Penetha-}$	hydriodide	lactam		
				ject				
2094 Control	. 14C	S3	Dairy 4	NA	NA	NA	Control	9/10/2019
3095 Treated	15	$\mathbf{S1}$	Dairy 4	$\operatorname{Pharmacillin}$	Procaine penicillin G	Beta-lactam	Footrot	11/10/2019
3096 Control	15C	$\mathbf{S1}$	Dairy 4	NA	NA	NA	Control	11/10/2019
3097 Treated	15	S2	Dairy 4	Pharmacillin	Procaine penicillin G	Beta-lactam	Footrot	12/10/2019
1098 Control	15C	S2	Dairy 4	NA	NA	NA	Control	12/10/2019
1099 Treated	15	S3	Dairy 4	Pharmacillin	Procaine penicillin G	Beta-lactam	Footrot	17/10/2019
100 Control	15C	S3	Dairy 4	NA	NA	NA	Control	17/10/2019
3101 Treated	16	$\mathbf{S1}$	Dairy 4	Intracillin	Procaine penicillin G	Beta-lactam	Swollen vulva	21/10/2019
3102 Control	16C	$\mathbf{S1}$	Dairy 4	NA	NA	NA	Control	21/10/2019
3103 Treated	16	S2	Dairy 4	Intracillin	Procaine penicillin G	Beta-lactam	Swollen vulva	23/10/2019
3104 Control	16C	S2	Dairy 4	NA	NA	NA	Control	23/10/2019
3105 Treated	17	$\mathbf{S1}$	Dairy 4	Intracillin	Procaine penicillin G	Beta-lactam	Footrot	23/10/2019
106 Control	17C	$\mathbf{S1}$	Dairy 4	NA	NA	NA	Control	23/10/2019
107 Treated	17	S2	Dairy 4	Intracillin	Procaine penicillin G	Beta-lactam	Footrot	23/10/2019
3108 Control	17C	S2	Dairy 4	NA	NA	NA	Control	23/10/2019
G113 Treated	17	S3	Dairy 4	Intracillin	Procaine penicillin G	Beta-lactam	Footrot	30/10/2019

ID Type	Anime	al Samp ordor	le Farm	Treatment	Antimicrobial	Class	Illness	Collection
d DC114 Control	17C	Ci de	Dairy 4	NA	NA	NA	Control	30/10/9019
d_DG115_Treated	16	e S	Dairy 4	Intracillin	Procaine nenicillin C	Reta-lactam	Swollen viilva	30/10/2019 30/10/9019
d DG116 Control	16C	S3	Dairy 4	NA	NA	NA	Control	$\frac{30}{10}2019$
d_DG117 Control	18C/19	)( S1	Dairy 4	NA	NA	NA	Control	22/11/2019
d_DG118 Treated	18	$\mathbf{S1}$	Dairy 4	Penethaject	Penethamate hydrio- dide	Beta-lactam	Mastitis	22/11/2019
d_DG119 Treated	20	$\mathbf{S1}$	Dairy 4	Intracillin	Procaine penicillin G	Beta-lactam	Footrot	22/11/2019
d_DG120 Treated	19	$\mathbf{S1}$	Dairy 4	Penethaject	Penethamate hydrio- dide	Beta-lactam	Mastitis	22/11/2019
ad_DG121 Control	18C/19	)( S2	Dairy 4	NA	NA	NA	Control	23/11/2019
ad_DG122 Treated	20	S2	Dairy 4	Intracillin	Procaine penicillin G	Beta-lactam	Footrot	23/11/2019
ed_DG123 Treated	18	S2	Dairy 4	Penethaject	Penethamate hydrio- dide	Beta-lactam	Mastitis	23/11/2019
ed_DG124 Treated	19	S2	Dairy 4	Penethaject	Penethamate hydrio- dide	Beta-lactam	Mastitis	23/11/2019
$ed_DG125$ Treated	21	$\mathbf{S1}$	Dairy 4	Intracillin	300 Procaine peni- cillin G	Beta-lactam	Footrot	27/11/2019
ad_DG126 Control	21C	$\mathbf{S1}$	Dairy 4	NA	NA	NA	Control	27/11/2019
ed_DG127 Treated	21	S2	Dairy 4	Intracillin	300 Procaine peni- cillin G	Beta-lactam	Footrot	28/11/2019
ad_DG128 Control	21C	S2	Dairy 4	NA	NA	NA	Control	28/11/2019
$^{ m ed}$ DG129 Treated	21	S3	Dairy 4	Intracillin	300 Procaine peni- cillin G	Beta-lactam	Footrot	2/12/2019
ad_DG130 Control	21C	S3	Dairy 4	NA	NA	NA	Control	2/12/2019
ed_DG131 Treated	18	S3	Dairy 4	Penethaject	Penethamate hydrio- dide	Beta-lactam	Mastitis	2/12/2019
ad_DG132 Control	18C/19	)( S3	Dairy 4	NA	NA	NA	Control	2/12/2019
ed_DG133 Treated	20	S3	Dairy 4	Intracillin	300 Procaine peni- cillin G	Beta-lactam	Footrot	2/12/2019
ed_DG134 Treated	19	S3	Dairy 4	Penethaject	Penethamate hydrio- dide	Beta-lactam	Mastitis	2/12/2019
d_DG135 Control	22C	$\mathbf{S1}$	Dairy 4	NA	NA	NA	Control	20/07/20

Collection date	20/07/20	21/07/20	21/07/20	4/08/20	4/08/20	5/08/20	6/08/20	6/08/20	12/08/20	12/08/20	12/08/20	12/08/20		13/08/20	13/08/20	14/08/20	14/08/20	14/08/20		14/08/20	14/08/20	14/08/20	15/08/20	15/08/20	20/08/20		20/08/20	20/08/20	20/08/20
Illness	Penetration on leg	Control	Penetration on leg	Penetration on leg	Control	Footrot	Footrot	Control	Control	Footrot	Control	Retained mem-	Drane	Control	Footrot	Footrot	Control	Retained mem-	$\mathbf{brane}$	Control	Footrot	Control	Footrot	Control	Retained mem-	brane	Control	Footrot	Control
Class	Beta-lactam	NA	Beta-lactam	Beta-lactam	NA	Beta-lactam	Beta-lactam	NA	NA	Beta-lactam	NA	Beta-lactam		NA	$\operatorname{Beta-lactam}$	Beta-lactam	NA	Beta-lactam		NA	Beta-lactam	NA	Beta-lactam	NA	$\operatorname{Beta-lactam}$		NA	Beta-lactam	NA
Antimicrobial	Procaine penicillin G	NA	Procaine penicillin G	Procaine penicillin G	NA	Procaine penicillin G	Procaine penicillin G	NA	NA	Procaine penicillin G	NA	Procaine penicillin G		NA	Procaine penicillin G	Procaine penicillin G	NA	Procaine penicillin G		NA	Procaine penicillin G	NA	Procaine penicillin G	NA	Procaine penicillin G		NA	Procaine penicillin G	NA
Treatment	Intracillin	NA	Intracillin	Intracillin	NA	Intracillin	Intracillin	NA	NA	Intracillin	NA	Intracillin		NA	Intracillin	Intracillin	NA	Intracillin		NA	Intracillin	NA	Intracillin	NA	Intracillin		NA	Intracillin	NA
ple Farm r	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4		Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4		Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4	Dairy 4		Dairy 4	Dairy 4	Dairy 4
Animal Sam orde	22 S1	22C S2	22 S2	22 S3	22C S3	23 S1	23 S2	23C S1	23C S3	23 S3	24C S1	24 S1		25C S1	25 S1	25 S2	25C S2	24 S2		24C S2	26 S1	26C S1	26 S2	26C S2	24 S3		24C S3	26 S3	$26C  ext{S3}$
Type	d_DG136 Treated	d_DG137 Control	d_DG138 Treated	d_DG143 Treated	d_DG144 Control	d_DG145 Treated	d_DG146 Treated	d_DG147 Control	d_DG148 Control	d_DG149 Treated	d_DG150 Control	d_DG151 Treated		d_DG152 Control	d_DG153 Treated	d_DG154 Treated	d_DG155 Control	d_DG156 Treated		d_DG157 Control	d_DG158 Treated	d_DG159 Control	d_DG160 Treated	d_DG161 Control	d_DG162 Treated		d_DG163 Control	d_DG164 Treated	d_DG165 Control
Lab ID	processed	processed	processed	processe	processed	processed	processed	processed	processed	processed	processed	processed		processed	processed	processed	processed	processed		processed	processed	processed	processed	processed	processed		processed	processed	processed

Lab ID Type	Anima	al Sampl	le Farm	Treatment	Antimicrobial	Class	Illness	Collection
		order						date
processed_DG167 Treated	25	S3	Dairy 4	Intracillin	Procaine penicillin G	Beta-lactam	Footrot	20/08/20
processed_DG168 Control	25C	S3	Dairy 4	NA	NA	NA	Control	21/08/20
processed_DG169 Treated	27	$\mathbf{S1}$	Dairy 4	Intracillin	Procaine penicillin G	Beta-lactam	Footrot	4/09/20
processed_DG170 Control	27C	$\mathbf{S1}$	Dairy 4	NA	NA	NA	Control	4/09/20
processed_DG171 Treated	27	S2	Dairy 4	Intracillin	Procaine penicillin G	Beta-lactam	Footrot	5/09/20
processed_DG172 Control	27C	S2	Dairy 4	NA	NA	NA	Control	5/09/20
processed_DG173 Treated	28	$\mathbf{S1}$	Dairy 4	Intracillin	Procaine penicillin G	Beta-lactam	Footrot	7/09/20
processed_DG174 Control	28C	$\mathbf{S1}$	Dairy 4	NA	NA	NA	Control	7/09/20
processed_DG175 Treated	28	S2	Dairy 4	Intracillin	Procaine penicillin G	Beta-lactam	Footrot	8/09/20
processed_DG176 Control	28C	S2	Dairy 4	NA	NA	NA	Control	8/09/20
processed_DG177 Treated	27	S3	Dairy 4	Intracillin	Procaine penicillin G	Beta-lactam	Footrot	11/09/20
processed_DG178 Control	27C	S3	Dairy 4	NA	NA	NA	Control	11/09/20
processed_DG179 Control	28C	S3	Dairy 4	NA	NA	NA	Control	15/09/20
processed_DG180 Treated	28	S3	Dairy 4	Intracillin	Procaine penicillin G	Beta-lactam	Footrot	16/09/20
processed_DG181 Treated	29	$\mathbf{S1}$	Dairy 4	Intracillin	Procaine penicillin G	$\operatorname{Beta-lactam}$	Footrot	21/09/20
processed_DG182 Control	29C	$\mathbf{S1}$	Dairy 4	NA	NA	NA	Control	21/09/20
processed_DG183 Treated	29	S2	Dairy 4	Intracillin	Procaine penicillin G	Beta-lactam	Footrot	22/09/20
processed_DG184 Control	29C	S2	Dairy 4	NA	NA	NA	Control	22/09/20
processed_DG185 Treated	30	$\mathbf{S1}$	Dairy 4	Intracillin	Procaine penicillin G	Beta-lactam	Footrot	23/09/20
processed_DG186 Control	30C	$\mathbf{S1}$	Dairy 4	NA	NA	NA	Control	23/09/20
processed_DG187 Treated	30	S2	Dairy 4	Intracillin	Procaine penicillin G	Beta-lactam	Footrot	24/09/20
processed_DG188 Control	30C	S2	Dairy 4	NA	NA	NA	Control	24/09/20
processed_DG189 Treated	30	S3	Dairy 4	Intracillin	Procaine penicillin G	Beta-lactam	Footrot	29/09/20
processed_DG190 Control	30C	S3	Dairy 4	NA	NA	NA	Control	29/09/20
processed_DG191 Treated	29	S3	Dairy 4	Intracillin	Procaine penicillin G	Beta-lactam	Footrot	29/09/20
processed_DG192 Control	29C	S3	Dairy 4	NA	NA	NA	Control	29/09/20

### Appendix Q

## Case study of 23 cows treated with procaine penicillin G



Shannon (A) and Chao1 (B)  $\alpha$  diversity of faecal samples from cows treated with procaine penicillin G (n=23).



Appendix R

Ceftiofur case study: *Enterobacteriaceae* abundance

Counts an cows (n=3	d relat ).	ive abu	indance of ASVs	belonging to <i>1</i>	Enterobacteriace	<i>ae</i> identified in t	the faecal sampl	es from ceftiof	cur treated (n=3	) and control
			Counts				Relative abur	ıdance		
			ASV30	ASV5332	ASV10841	ASV13004	ASV30	ASV5332	ASV10841	ASV13004
Type	$\operatorname{Grp}^{a}$	$Ord^b$	Escherichia-	Salmonella	Escherichia-	Escherichia-	Escherichia-	Salmonella	Escherichia-	Escherichia-
			Shigella		Shigella	Shigella	Shigella		Shigella	Shigella
Treated	11	$\mathbf{S1}$	13	0	0	0	0.000272794	0	0	0
Treated	11	S2	0	0	വ	0	0	0	0.00010498	0
Treated	11	S3	23	0	0	0	0.000482555	0	0	0
Control	11C	$\mathbf{S1}$	12	0	0	0	0.00025181	0	0	0
Control	11C	S2	32	0	0	0	0.000671324	0	0	0
Control	11C	S3	0	0	0	0	0	0	0	0
Treated	12	$\mathbf{S1}$	11	0	0	0	0.000231068	0	0	0
Treated	12	S2	0	0	0	0	0	0	0	0
Treated	12	S3	14	0	0	0	0.000293914	0	0	0
Control	12C	$\mathbf{S1}$	19	0	0	0	0.000398732	0	0	0
Control	12C	S2	0	0	0	0	0	0	0	0
Control	12C	S3	3	0	0	0	6.29736E-05	0	0	0
Treated	13	$\mathbf{S1}$	44	43	0	0	0.000923535	0.000902546	0	0
Treated	13	S2	9	0	0	0	0.000188857	0	0	0
Treated	13	S3	0	0	0	2	0	0	0	4.19956E-05
Control	13C	S1	Q	0	0	0	0.000104875	0	0	0
Control	13C	S2	21	0	0	0	0.000440815	0	0	0
Control	13C	S3	×	0	0	0	0.000167817	0	0	0
<sup>a</sup> Group	$^{\rm b}$ Sa	ample o	rder.							

### Appendix S

# Virulence factors from *E. coli* genomes

Isolate	Virulence factors
DG079c	astA, cia, f17A, f17G, iss, terC, traT
DG079h	astA, cia, f17A, f17G, gad, iss, terC, traT
DG082f	astA, cia, f17A, f17G, iss, terC, traT

### Appendix T

### Taxonomic classification at the phylum level of processed reads from faeces, FDE, soil and milk



Taxonomic classification at the phylum level of processed reads from faeces.



Taxonomic classification at the phylum level of processed reads from farm dairy effluent (FDE)



Taxonomic classification at the phylum level of processed reads from soil.



Taxonomic classification at the phylum level of processed reads from milk.

### Appendix U

### Heat map showing normalised values of antimicrobial drug, biocide and metal resistance determinants at the gene level



Heat map showing normalised values of antimicrobial drug, biocide and heavy metal resistance determinants at the gene level identified in faeces samples on Dairy 1 and Dairy 4 from October 2018 and December 2019. ARG abundance was expressed as copy of ARG per copy of 16S rRNA gene and individual gene variants were combined by group. The class of ARGs is indicated by colour in the figure legend.



Heat map showing normalised values of antimicrobial drug, biocide and metal resistance determinants at the gene level identified in farm dairy effluent samples on Dairy 1 and Dairy 4 from October 2018 and December 2019. ARG abundance was expressed as copy of ARG per copy of 16S rRNA gene and individual gene variants were combined by group. The class of ARGs is indicated by colour in the figure legend.



Heat map showing normalised values of antimicrobial drug, biocide and metal resistance determinants at the gene level identified in milk samples on Dairy 1 and Dairy 4 from October 2018 and December 2019. ARG abundance was expressed as copy of ARG per copy of 16S rRNA gene and individual gene variants were combined by group. The class of ARGs is indicated by colour in the figure legend. The waste milk sample (DF0167) was excluded from this analysis.



Heat map showing normalised values of antimicrobial drug, biocide and metal resistance determinants at the gene level identified in soil samples on Dairy 1 and Dairy 4 from October 2018 and December 2019. ARG abundance was expressed as copy of ARG per copy of 16S rRNA gene and individual gene variants were combined by group. The class of ARGs is indicated by colour in the figure legend.

Appendix V

The predicted taxonomic ranks of contigs harbouring acquired antimicrobial resistance genes















### Appendix W

### Supplementary feed on Dairy 1 and Dairy 4 during the study period

Collection date	Dairy 1	Dairy 4 <sup>a</sup>
October 2018	No extra feed	Dairy pellets, baleage, maize
		silage
November 2018	No extra feed	Maize silage, baleage, DDG,
		dairy pellets
December 2018	No extra feed	Maize silage, peas
January 2019	Crop (turnips)	Maize silage, dairy pellets
February 2019	Baleage, crop (turnips)	Grass silage
March 2019	Grass silage	Grass and maize silage
April 2019	Grass and maize silage,	Grass and maize silage,
	DDG, tapioca, lucerne, hay	straw, soya hull, baleage
May 2019	Maize silage, DDG, tapioca,	Maize silage
	lucerne, hay	
June 2019	Maize silage, kale	Straw and baleage
July 2019	Maize silage, kale	Maize silage
August 2019	Maize silage	DDG, soya meal, lucerne,
		maize silage
September 2019	Maize silage	Hay, DDG, lucerne (ad libi-
		$tum$ , baleage ( $ad \ libitum$ )
October 2019	No extra feed	Maize silage, soya, DDG,
		molasses
November 2019	No extra feed	Maize silage
December 2019	No extra feed	Maize silage

<sub>a</sub> DDG, Distillers dried grains.

Appendix X

### DRC16 form



### STATEMENT OF CONTRIBUTION DOCTORATE WITH PUBLICATIONS/MANUSCRIPTS

We, the candidate and the candidate's Primary Supervisor, certify that all co-authors have consented to their work being included in the thesis and they have accepted the candidate's contribution as indicated below in the *Statement of Originality*.

Name of candidate:	Rose Moyra Collis						
Name/title of Primary Supervisor:	Prof. Patrick Biggs						
Name of Research Output and full reference	e:						
s, P. J., Midwinter, A. C., French, N. P., Toombs-Ruane, L., & Coekson, A. L. (2019). Extended-spectrum bela-	actamase-producing Enterobacteriaceae in dairy farm environments: A	Now Zealand perspective, Foodborne Pathogens and Disease, 16(1).					
In which Chapter is the Manuscript /Publish	ned work:	Chapter 2 (Lit. review)					
Please indicate:							
The percentage of the manuscript/Published Work that was contributed by the candidate:     80							
and	and						
<ul> <li>Describe the contribution that the candidate has made to the Manuscript/Published Work:</li> </ul>							
This paper was the work of the candidate who researched the literature and wrote the review paper with the guidance and editing by the co-authors.							
For manuscripts intended for publication please indicate target journal:							
Candidate's Signature:							
Date:	28/01/2022						
Primary Supervisor's Signature:	Prof Patrick Biggs	Digitally signed by Prof Patrick Biggs DN: cn=Prof Patrick Biggs, c=NZ, o=Massey University, ou=School of Vaterinary Science, email=p-biggs@university.com Date: 2022.01.31 10:25-X7 +1300'					
Date:	31-Jan-22						

(This form should appear at the end of each thesis chapter/section/appendix submitted as a manuscript/ publication or collected as an appendix at the end of the thesis)

GRS Version 4– January 2019

DRC 16